



School of Health Professions Theses Faculty of Health Theses

2012

Empathy: Its Significance and Monitoring in the Dietetic Consultation for Chronic Disease Management

Tracey Parkin

Let us know how access to this document benefits you

General rights

All content in PEARL is protected by copyright law. Author manuscripts are made available in accordance with publisher policies. Please cite only the published version using the details provided on the item record or document. In the absence of an open licence (e.g. Creative Commons), permissions for further reuse of content should be sought from the publisher or author. **Take down policy**

If you believe that this document breaches copyright please contact the library providing details, and we will remove access to the work immediately and investigate your claim.

Follow this and additional works at: https://pearl.plymouth.ac.uk/hp-theses

Recommended Citation

Parkin, T. (2012) *Empathy: Its Significance and Monitoring in the Dietetic Consultation for Chronic Disease Management.* Thesis. University of Plymouth. Retrieved from https://pearl.plymouth.ac.uk/hp-theses/9 This Thesis is brought to you for free and open access by the Faculty of Health Theses at PEARL. It has been accepted for inclusion in School of Health Professions Theses by an authorized administrator of PEARL. For more information, please contact openresearch@plymouth.ac.uk.



PEARL

PHD

Empathy: Its Significance and Monitoring in the Dietetic Consultation for Chronic Disease Management

Parkin, Tracey

Award date: 2012

Awarding institution: University of Plymouth

Link to publication in PEARL

All content in PEARL is protected by copyright law.

The author assigns certain rights to the University of Plymouth including the right to make the thesis accessible and discoverable via the British Library's Electronic Thesis Online Service (EThOS) and the University research repository (PEARL), and to undertake activities to migrate, preserve and maintain the medium, format and integrity of the deposited file for future discovery and use.

Copyright and Moral rights arising from original work in this thesis and (where relevant), any accompanying data, rests with the Author unless stated otherwise*.

Re-use of the work is allowed under fair dealing exceptions outlined in the Copyright, Designs and Patents Act 1988 (amended), and the terms of the copyright licence assigned to the thesis by the Author.

In practice, and unless the copyright licence assigned by the author allows for more permissive use, this means,

That any content or accompanying data cannot be extensively quoted, reproduced or changed without the written permission of the author / rights holder

That the work in whole or part may not be sold commercially in any format or medium without the written permission of the author / rights holder

* Any third-party copyright material in this thesis remains the property of the original owner. Such third-party copyright work included in the thesis will be clearly marked and attributed, and the original licence under which it was released will be specified. This material is not covered by the licence or terms assigned to the wider thesis and must be used in accordance with the original licence; or separate permission must be sought from the copyright holder.

Download date: 28. Oct. 2024

This copy of the thesis has been supplied on condition that anyone who consults it is understood to recognise that its copyright rests with its author and that no quotation from the thesis and no information derived from it may be published without the author's prior consent.

EMPATHY: ITS SIGNIFICANCE AND MONITORING IN THE DIETETIC CONSULTATION FOR CHRONIC DISEASE MANAGEMENT

by

TRACEY PARKIN

A thesis submitted to the University of Plymouth in partial fulfilment for the degree of

DOCTOR OF PHILOSOPHY

School of Health Professions

Faculty of Health

April 2011

ABSTRACT

Tracey Parkin

EMPATHY: ITS SIGNIFICANCE AND MONITORING IN THE DIETETIC CONSULTATION FOR CHRONIC DISEASE MANAGEMENT

Communication between the patient and healthcare professional is pivotal in enabling effective self-care management to occur (Street et al., 2009) which in turn leads to improved health outcomes (Kravitz et al., 1993). However, there is little published data exploring the most effective tools for evaluating whether these communication skills occur and what particular attributes enhance the process (Goodchild, Skinner & Parkin, 2005; Heisler et al., 2003; Parkin & Skinner, 2003). This study aims to identify patient-centred communication skills occurring in consultations and to explore their link with a tool-recording agreement on reported decisions made.

Quantitative analysis was conducted on 86 dietetic consultations across four outpatient diabetes services. Audio recordings from 20 of these were qualitatively analysed.

Greater agreement on reported decisions correlated significantly with level of empathy demonstrated (τ = .283, p = .0005). In consultations featuring agreement, dietitians expressed more empathy (p = .02), used more active listening skills, asked significantly more exploratory questions on self-care practices (U = 18.5, p = .007), provided significantly more supportive/collaborative information-giving exchanges (U = 11, p = .003) and were more likely to set an agreed agenda at the start of consultations.

In contrast, consultations featuring disagreement had low levels of empathy, fewer active-listening skills and exploratory questions, significantly greater numbers of *persuasive* information-giving exchanges (U = 17, p = .007) and more recommending exchanges. Generally, agendas were not set.

In conclusion, recording patient/healthcare professional agreement on reported decisions made during a consultation is a simple tool that can indicate the presence of patient-centred communication skills. Active-listening skills allow expressions of empathy that facilitate patient involvement and interactive dialogue. The measure of agreement should be used frequently as a marker of effective dietetic consultations and to provide further data on the relationship between patient-centred communication and implementation of behaviour change for improved health outcomes.

CONTENTS

LIST OF TABLES	15
LIST OF FIGURES	17
ACKNOWLEDGEMENTS	19
AUTHOR'S DECLARATION	21
CHAPTER ONE: SETTING THE SCENE	25
1.0 SIGNIFICANCE OF STUDY	25
1.01 Aims and research questions	25
1.02 Structure of thesis	26
1.1 INTRODUCTION	28
1.2 TREATMENT OF DIABETES	32
1.2.1 Self-care Management	34
1.3. POSSIBLE RATIONALE FOR POOR SELF-CARE MANAGEMENT	35
1.3.1 Lifestyle / Dietary compliance	35
1.3.2 Understanding	
1.3.3 Emotional factors	40
1.3.4 Drug compliance	42
1.3.5 The Structure of Healthcare Delivery	44
1.4 THE FACTORS DEMONSTRATING IMPROVED SELF-CARE MANAGEMENT	46
1.4.1 Role of the Dietitian	51
1.5 SUMMARY	52
CHAPTER TWO: LITERATURE REVIEW	55
CHAPTER TWO: LITERATURE REVIEW	55 55
CHAPTER TWO: LITERATURE REVIEW 2.0 Background 2.1 Aim	55 55 57
CHAPTER TWO: LITERATURE REVIEW 2.0 BACKGROUND 2.1 AIM 2.2 SEARCH METHOD	55 55 57 57
CHAPTER TWO: LITERATURE REVIEW	55 57 57 60
CHAPTER TWO: LITERATURE REVIEW	55 57 57 60 60
CHAPTER TWO: LITERATURE REVIEW	55 57 57 60 60 61
CHAPTER TWO: LITERATURE REVIEW	55 57 60 60 61 62
CHAPTER TWO: LITERATURE REVIEW	55 57 60 60 61 62 68
CHAPTER TWO: LITERATURE REVIEW	55 57 60 60 61 62 68 69
CHAPTER TWO: LITERATURE REVIEW	55 57 60 60 61 62 68 69 71
CHAPTER TWO: LITERATURE REVIEW 2.0 BACKGROUND 2.1 AIM 2.2 SEARCH METHOD 2.2 SEARCH METHOD 2.2.1 Criteria for inclusion 2.2.2 Selection of papers 2.2.3 Search Strategy 2.3 RESULTS 2.3.1 Methodological quality 2.3.2 Recall as an outcome measure 2.3.3 Empathy and agreement on recall. 2.3.4 Patient autonomy	55 57 60 60 61 62 68 68 69 71 73
CHAPTER TWO: LITERATURE REVIEW 2.0 BACKGROUND 2.1 AIM 2.2 SEARCH METHOD 2.2.1 Criteria for inclusion 2.2.2 Selection of papers 2.2.3 Search Strategy 2.3 RESULTS 2.3.1 Methodological quality 2.3.2 Recall as an outcome measure 2.3.3 Empathy and agreement on recall 2.3.4 Patient autonomy 2.3.5 Self-efficacy, self-care management and agreement	55 57 60 60 61 62 68 69 71 73 73
CHAPTER TWO: LITERATURE REVIEW 2.0 BACKGROUND 2.1 AIM 2.2 SEARCH METHOD 2.2.1 Criteria for inclusion 2.2.2 Selection of papers 2.2.3 Search Strategy 2.3 RESULTS 2.3.1 Methodological quality 2.3.2 Recall as an outcome measure 2.3.3 Empathy and agreement on recall 2.3.4 Patient autonomy 2.3.5 Self-efficacy, self-care management and agreement 2.3.6 Communication, decision-making and self-care management	55 57 60 60 61 62 68 68 69 71 73 73 74
CHAPTER TWO: LITERATURE REVIEW 2.0 BACKGROUND 2.1 AIM 2.2 SEARCH METHOD 2.2.1 Criteria for inclusion 2.2.2 Selection of papers 2.2.3 Search Strategy 2.3 RESULTS 2.3.1 Methodological quality 2.3.2 Recall as an outcome measure 2.3.3 Empathy and agreement on recall 2.3.4 Patient autonomy 2.3.5 Self-efficacy, self-care management and agreement 2.3.6 Communication, decision-making and self-care management 2.3.7 Recall and adherence.	55 57 57 60 60 61 62 62 63 71 73 73 74 75
CHAPTER TWO: LITERATURE REVIEW 2.0 BACKGROUND 2.1 AIM 2.2 SEARCH METHOD 2.2.1 Criteria for inclusion 2.2.2 Selection of papers 2.2.3 Search Strategy 2.3 RESULTS 2.3.1 Methodological quality 2.3.2 Recall as an outcome measure 2.3.3 Empathy and agreement on recall 2.3.4 Patient autonomy 2.3.5 Self-efficacy, self-care management and agreement 2.3.7 Recall and adherence 2.4 DISCUSSION.	55 57 60 60 61 62 68 69 71 73 73 74 75 76
CHAPTER TWO: LITERATURE REVIEW 2.0 BACKGROUND 2.1 AIM 2.2 SEARCH METHOD 2.2.1 Criteria for inclusion 2.2.2 Selection of papers 2.2.3 Search Strategy 2.3 RESULTS 2.3.1 Methodological quality 2.3.2 Recall as an outcome measure 2.3.3 Empathy and agreement on recall 2.3.4 Patient autonomy 2.3.5 Self-efficacy, self-care management and agreement 2.3.6 Communication, decision-making and self-care management 2.3.7 Recall and adherence 2.4 DISCUSSION 2.5 CONCLUSION	55 57 57 60 60 61 62 68 69 71 73 73 74 75 76 83
CHAPTER TWO: LITERATURE REVIEW	55 57 57 60 60 61 62 68 63 73 73 73 74 75 76 83 85
CHAPTER TWO: LITERATURE REVIEW 2.0 BACKGROUND 2.1 AIM 2.2 SEARCH METHOD 2.2.1 Criteria for inclusion 2.2.2 Selection of papers 2.3 Search Strategy 2.3 RESULTS 2.3.1 Methodological quality 2.3.2 Recall as an outcome measure 2.3.3 Empathy and agreement on recall 2.3.4 Patient autonomy 2.3.5 Self-efficacy, self-care management and agreement 2.3.7 Recall and adherence 2.4 DISCUSSION 2.5 CONCLUSION CHAPTER THREE: PATIENT-CENTRED COMMUNICATION 3.0 PATIENT-CENTRED CARE	55 57 57 60 61 62 61 62 63 71 73 73 73 74 75 76 83 85

3.2 PATIENT-CENTRED COMMUNICATION	
3.3 Емратну	94
3.3.1 Measurement of empathy	97
3.3.2 Importance of Empathy	
3.3.3 Empathy and emotions	
3.4 ACTIVE-LISTENING SKILLS	107
3.4.1 Information	
3.4.2 Lack of involvement	113
3.4.3 Facilitation	115
3.5 CONCLUSIONS	116
3.6 SUMMARY	118
CHAPTER 4: PATIENT SATISFACTION, PATIENT AUTONO	MY AND
PATIENT SELF-EFFICACY AND THEIR INFLUENCE ON EM	PATHY AND
AGREEMENT ON RECALL	121
	121
4 1 PATIENT SATISFACTION	122
4.2 PATIENT AUTONOMY	
4.3 SEI F-EFFICACY	
4.4 RELATIONSHIP BETWEEN EMPATHY. PATIENT SATISFACTION. PA	TIENT AUTONOMY
AND PATIENT SELF-EFFICACY WITH AGREEMENT ON REPORTED DEC	SISIONS MADE . 129
4.5 CONCLUSION	
DIETETIC CONSULTATIONS	
DIETETIC CONSULTATIONS	
DIETETIC CONSULTATIONS	133
DIETETIC CONSULTATIONS	133
DIETETIC CONSULTATIONS. 5.0 INTRODUCTION. 5.1 SELECTION OF A TOOL. 5.2 CODING FOR EMPATHY 5.3 L Empathic Communication Coding System (ECCS)	133
5.0 INTRODUCTION	133
 5.0 INTRODUCTION. 5.1 SELECTION OF A TOOL. 5.2 CODING FOR EMPATHY 5.2.1 Empathic Communication Coding System (ECCS) 5.2.2 Analysis using Empathy Coding System. 	133
 5.0 INTRODUCTION. 5.1 SELECTION OF A TOOL. 5.2 CODING FOR EMPATHY 5.2.1 Empathic Communication Coding System (ECCS) 5.2.2 Analysis using Empathy Coding System 5.3 CODING FOR PATIENT-CENTRED COMMUNICATION SKILLS. 5.3 1 Roter Interactive Analysis System (RIAS) 	133
 5.0 INTRODUCTION. 5.1 SELECTION OF A TOOL. 5.2 CODING FOR EMPATHY 5.2.1 Empathic Communication Coding System (ECCS) 5.2.2 Analysis using Empathy Coding System	
 5.0 INTRODUCTION	133
 5.0 INTRODUCTION. 5.1 SELECTION OF A TOOL. 5.2 CODING FOR EMPATHY 5.2.1 Empathic Communication Coding System (ECCS) 5.2.2 Analysis using Empathy Coding System	133 133 134 135 135 135 137 140 141 ctor VR-MICS/D 143 145
 DIETETIC CONSULTATIONS. 5.0 INTRODUCTION. 5.1 SELECTION OF A TOOL. 5.2 CODING FOR EMPATHY	133
 5.0 INTRODUCTION. 5.1 SELECTION OF A TOOL. 5.2 CODING FOR EMPATHY 5.2.1 Empathic Communication Coding System (ECCS) 5.2.2 Analysis using Empathy Coding System	133
 DIETETIC CONSULTATIONS. 5.0 INTRODUCTION. 5.1 SELECTION OF A TOOL. 5.2 CODING FOR EMPATHY	133
 DIETETIC CONSULTATIONS. 5.0 INTRODUCTION. 5.1 SELECTION OF A TOOL. 5.2 CODING FOR EMPATHY 5.2.1 Empathic Communication Coding System (ECCS) 5.2.2 Analysis using Empathy Coding System 5.3 CODING FOR PATIENT-CENTRED COMMUNICATION SKILLS. 5.3.1 Roter Interactive Analysis System (RIAS) 5.3.2 Verona Medical Interview Classification System – Do 5.4 COMPARISON OF THE RIAS/VERONA AND REFLECTIVE-PRACTI SYSTEMS 5.4.1 Analysis Using RIAS Coding System 5.4.2 Analysis Using RIAS Coding System 	133
 DIETETIC CONSULTATIONS. 5.0 INTRODUCTION. 5.1 SELECTION OF A TOOL. 5.2 CODING FOR EMPATHY	133
 DIETETIC CONSULTATIONS. 5.0 INTRODUCTION. 5.1 SELECTION OF A TOOL	133
 DIETETIC CONSULTATIONS. 5.0 INTRODUCTION. 5.1 SELECTION OF A TOOL. 5.2 CODING FOR EMPATHY	133
 DIETETIC CONSULTATIONS. 5.0 INTRODUCTION. 5.1 SELECTION OF A TOOL. 5.2 CODING FOR EMPATHY	133
 DIETETIC CONSULTATIONS. 5.0 INTRODUCTION. 5.1 SELECTION OF A TOOL. 5.2 CODING FOR EMPATHY	133

CHAPTER SIX: RATIONALE FOR USING MIXED METHODOLOGY	159
6.0 Overview of Mixed-Methods Research	159
6.01 Issues in Mixed-Methods Research	160
6.02 Rationale for Using Mixed-Methods Research	160
6.1 DESIGNING A MIXED-METHODOLOGY STUDY	161
6.1.1 Theorising	161
6.1.2 Weighting	162
6.1.3 Mixing	162
6.1.4 Timing	163
6. 2 CONSIDERATION OF THE FOUR TENETS FOR DESIGNING THE MIXED	
METHODOLOGY	163
6.2.1 Theorising	164
6.2.2 Weighting	165
6.2.3 Mixing	165
6.2.4 Timing	167
6.3 Study design	167
6.4 OBSERVATIONAL METHOD FOR DATA COLLECTION	168
6.5 SUMMARY	170
CHAPTER SEVEN: METHODOLOGY FOR PHASE I (QUANTITATIVE)	173
7.0 Phase I: Research Questions and Hypotheses	173
7.1 RECRUITMENT	174
7.1.1 Dietitians	174
7.1.2 Patients	174
7.2 PARTICIPANTS	176
7.2.1 Dietitians	176
7.2.2 Patients	177
7.3 Метнод	178
7.4 Measures	180
7.4.1 Questionnaires	180
7.4.2 Ordering of Questionnaires in Patient Booklet	183
7.5 POWER CALCULATION	184
7.6 ANALYSIS OF QUESTIONNAIRES	185
7.6.1 Quantitative Analysis of Questionnaires	185
7.6.2 Transforming the Qualitative Open Question into a Quantitative	
Measure	185
7.6.2.1 Decisions Reported on Questionnaires	185
7.7 ANALYSIS OF AUDIO RECORDINGS	189
7.7.1 Extraction of Decisions and Generation of Quantitative Data Set	189
7.7.2 Coding for Empathy Using the Empathic Communication Coding	!
System (ECCS)	191
7.7.3 Screening for Interference in the Interpretation of Empathy	193
7.8 ETHICS	193
(.9 SUMMARY	194
CHAPTER EIGHT: RESULTS FOR PHASE I	195

8.0 Statistical Analysis	196
8.1 Results	196
8.1.2 Decision Topics	
8.1.3 Level of agreement on reported decisions made in the co	nsultation
8.2 Емратну	201
8.2.1 Impact of Empathy on Consultation Outcomes	202
8.2.2 Does Greater Empathy Result in Greater Agreement?	203
8.2.3 The effect of decision recapping / action planning at the e	nd of the
consultation	203
8.2.4 Impact of the Various Components of Empathy on Agree	ment204
8. 2.5 Can the relationship between empathy and agreement b	e explained
by differences within the consultations?	205
8.2.6 Empathy and patient satisfaction	211
8.3 MEDIATING VARIABLES (PATIENT AUTONOMY AND SELF-EFFICACY).	211
8.4 MODERATING VARIABLES (DEMOGRAPHIC DATA AND LENGTH OF CO	NSULTATION)
8.4.1 Presence of complications and duration of diabetes on ag	greement 212
8.4.2 Length of consultation	
8.4 CONCLUSION	216
CHAPTER NINE: INTRODUCTION TO PHASE II	219
9.0 QUALITATIVE ANALYSIS	219
9.1 THEMATIC ANALYSIS	221
9.1.1 Approach taken to thematic analysis	222
9.1.2 Sequential map	223
9.2 SUMMARY	223
CHAPTER TEN: QUALITATIVE METHOD PHASE II	225
	225
10.1 IDENTIFICATION OF QUALITATIVE DATA EMBEDDED IN PHASE 1	
10.2 Statistical analysis of sample	
10.3 METHOD FOR THEMATIC ANALYSIS	
10.3.1 Familiarisation with data	231
10.3.2 Coding for communication skills	231
10.3.3 Identification of information-giving themes	232
10.3.4 Reviewing identified themes	236
10.3.5 Defining and naming themes	239
10.4 COMPARISON OF DATA FROM CONSULTATIONS WITH AGREEMENT	ON REPORTED
DECISIONS MADE AND THOSE WITH DISAGREEMENT	240
10.5 SEQUENCING OF INFORMATION-GIVING EXCHANGES AND EMPATHI	С
OPPORTUNITIES IN CONSULTATIONS	240
10.4 SUMMARY	241
CHAPTER ELEVEN: RESULTS FROM PHASE II QUALITATIVE	ANALYSIS
	243

11.0 EXPLORATION OF KEY COMMUNICATION SKILLS	244
11.1 IS THERE A DIFFERENCE IN THE USE OF ACTIVE LISTENING SKILLS IN	
CONSULTATIONS WITH AGREEMENT ON REPORTED DECISIONS MADE AND THOSE W	/ITH
DISAGREEMENT?	244
11.1.2 Is there a difference in the use of exploratory questions in	
consultations with agreement on reported decisions made and those with	า
disagreement?	246
11.1.3 Is there a difference in agenda setting in consultations with	
agreement on reported decisions made and those with disagreement?2	247
11.2 EXPLORATION OF THE FOUR IDENTIFIED INFORMATION-GIVING THEMES IN THE	Ξ
CONSULTATIONS	248
11.2.1 Persuasive information-giving exchanges	249
11.2.2 Supportive/collaborative information-giving exchanges	253
11.2.3 Confirming/permitting information-giving exchanges	258
11.2.4 Recommending/teaching/instruction information-giving exchanges	3
	261
11.3 IS THERE A DIFFERENCE IN THE TYPE OF INFORMATION-GIVING EXCHANGES	
USED IN CONSULTATIONS WITH AGREEMENT ON REPORTED DECISIONS MADE AND	
THOSE WITH DISAGREEMENT?	264
11.4 WHAT IS THE RELATIONSHIP BETWEEN INFORMATION-GIVING EXCHANGES AN	D
EMPATHY IN THE CONSULTATIONS REVIEWED?	265
11.4.1 Pattern of supportive/collaborative information-giving exchanges.2	267
11.5 EXPLORING THE COMMUNICATION SKILLS THAT PRECEDE EMPATHIC	
STATEMENTS IN CONSULTATIONS	268
11.6 CONCLUSION	269
CHAPTER TWELVE: DISCUSSION	273
12.1. EVEL OF ACREEMENT ON REPORTED DECISIONS MADE IN THE CONSULTATIO	
12.1 LEVEL OF AGREEMENT ON REPORTED DECISIONS MADE IN THE CONSULTATIO	אוי 270
	213
12.2 Level OF PATIENT ENGAGEMENT AND CONSULTING TIME	207
12.5 THE FOCUS OF THE CONSULTATION AND ITS RELEVANCE TO IMMUEDIATE NEED	, 200
	202
12.5 PRESENCE OF EMPATHY IN THE CONSULTATION	202
12.6 RELATIONSHIP BETWEEN PATIENT-CENTRED COMMUNICATION SKILLS AND	200
	296
12 7 DOES PATIENT AUTONOMY OR SELE-EFEICACY AFFECT SUCCESSEU	200
CONSULTATION OUTCOMES?	302
12.8 DOES TRAINING INFLUENCE THE LEVELS OF EMPATHY SEEN AND THE	J02
APPLICATION OF PATIENT-CENTRED COMMUNICATION SKILLS NEEDED TO SUPPORT	-
THIS?	303
12.91 IMITATIONS AND FUTURE RESEARCH DIRECTIONS	304
12.10 CONCLUSIONS	307
	201
CHAPIER THIRTEEN: IMPLICATIONS FOR PRACTICE AND FUTURE	
RESEARCH	311

APPENDIX	317
1.1 RECRUITMENT, PARTICIPANTS AND METHOD FOR STUDY	
EXPLORING TOOLS SUITABLE FOR ANALYSING DIETETIC	
CONSULTATIONS	319
1.2 CODING FOR EMPATHY USING THE EMPATHIC COMMUNICATION	ON
CODING SYSTEM (ECCS)	323
1.3 SUMMARY OF CODES OBTAINED USING ECCS	326
1.4 OVERVIEW OF CODING CATEGORIES FOR RIAS	327
1.5 SUMMARY OF CODES USING RIAS	328
1.6 SUMMARY OF CODES USING VERONA	329
1.7 SUMMARY OF CODES OBTAINED USING THE REFLECTIVE	
PRACTICE CODING SYSTEM	330
2.1 PATIENT LETTER AND INFORMATION SHEET	331
2.2 PATIENT CONSENT FORM	333
2.3 DIETITIAN CONSENT FORM	334
2.4 DIETITIAN QUESTIONNAIRE BOOKLET	335
2.5 PATIENT QUESTIONNAIRE BOOKLET	338
2.6 INITIAL CODING FRAME FOR IDENTIFYING DECISIONS	344
2.7 ILLUSTRATION OF THE TEN SUB-CATEGORIES FOR FOOD	346
2.8 AMENDED EMPATHY CODING LEVELS FOR ECCS	347
3.1 CODING MANUAL FOR QUALITATIVE ANALYSIS	349
GLOSSARY	369
REFERENCES	375
BIBLIOGRAPHY	419

List of Tables

Table 1.1 Summary of key areas for self-care management in diabetes
Table 2.2: Search terms used to identify the measurement of consultation skills /
behaviours against outcome of recall / goal or decisions
Table 2.3: Study design, participants, behaviours, measures, results, and implications
of findings of papers reviewed
Table 3.4: Operational definition of patient-centred communication adapted from
(Epstein et al, 2005)
Table 3.5: Dimensions of empathy highlighted by Stepien & Baerstein (2006)
Table 5.6: Identification of empathic opportunities (ECCS)
Table 5.7: Coding system for empathic responses made to patient empathic
opportunities (ECCS)
Table 5.8: Empathic opportunities and dietetic response to these for the 12 dietetic
consultations analysed
Table 5.9: The responses of dietitians to empathic opportunities
Table 5.10: Revised coding levels for empathy adapted from Empathic Communication
Coding System (ECCS)
Table 5.11: Coding categories for RIAS
Table 5.12: Coding categories VERONA Doctor VR-MICS/D
Table 5.13: Reflective Practice Coding System
Table 5.14: Summary of scores obtained with reflective-practice coding tool 152
Table 6.15: Reasons for mixed-methods research (adapted from Johnson &
Onwuegbuzie, 2004)
Table 7 16: Number of training courses in behaviour change and communication skills
attended by dietitians in this study since qualifying
Table 7 17: Demographic data for patients with complete data set included in the study
compared to patients who did not participate
Table 7 18: Latin square used for ordering of questionnaires in patient booklet 183
Table 7.19: Final coding themes used for identified decisions
Table 7.20: Ten sub-categories identified for food theme
Table 7.20: Ferroula for calculating agreement on reported decisions made in the
consultation
Table 7.22: Coding categories for complete agreement on decisions made and partial
acreement on decisions made
Table 7.23: Coding system for empathic responses made to patient empathic
statements (ECCS)
Table 8.24: Summary of decision topics reported on patient and distitian questionnaires
and decision topics taken from audio recordings
Table 8 25: Food sub-categories ranked in order of occurrence
Table 0.25. Food Sub-categories failed in order of occurrence
Table 8.26: Patient and dictitian agreement on decisions reported by both parties 199
Table 8.27: Mean number of reported decisions per consultation
Table 8.28: Distillar responses to patients empathic opportunities
I able 8.29: Iviean level of response by distitians to empathic statements made by
Zoble 0.20: Mass and modion visiting for any other and
able 8.30: Iviean and median values for empathy and agreement on reported
decisions made in consultations with and without action planning/decision recapping

Table 8.31: Influence of individual dietitians on level of empathic response to statements of challenge and impact on level of agreement on reported decisions mat	ade
Table 8.32: Frequency of ECCS codes used for responses by dietitians to patient	200
Table 8.33: Mean scores (SEM) for agreement on reported decisions, empathy, level empathy for different empathic statements and number of decisions recorded for individual dietitians.	.209 el of
Table 8.34: Mean number of decisions and complications	.213
Table 9.35: Phases of thematic analysis (adapted from Braun & Clarke, 2006)	222
Table10.36: Summary of recordings selected for analysis from participating dietitian where there was complete agreement and complete disagreement on reported	S
decisions made	. 227
Table10.37: Demographic data, appointment type, consultation length and empathy scores for consultations selected for qualitative analysis	.229
Table 10.38: Key features of the four identified themes for coding information-giving	ļ
exchanges adapted from Kiuru et al (2004)	.238
Table11.39: Frequency of occurrence of active listening skills used in consultations	.245
Table 11.40: Frequency of patient-specific questions used in consultations	.247
Table 11.41: Number of consultations that set an agenda with the patient	.248
Table 11.42: Summary of occurrence of information-giving exchange themes in	
consultations with agreement on reported decisions made and those with disagreen	nent
	.264
Table 11.43: Summary of the pattern of occurrence of supportive/collaborative	
information-giving exchanges in consultations	.268
Table11.44: Frequency of behaviours preceding empathic statements of emotion (S	E),
progress (SP), and challenge (SC) in consultations	.269
Table A. 45 Codes for empathic opportunities	.323
Table A.46 Coding levels for empathic responses ECCS	.325
Table A.48 Overview of coding categories for RIAS	.327
Table A. 49 Summary of codes obtained using RIAS	. 328
Table A.50 Summary of codes obtained using Verona	. 329
Table A.51 Summary of codes obtained using reflective practice coding system	.330
Table A.52 Initial codes for decisions	.345
Table A.53 Illustration of the ten subcategories for food	.346
Table A.54 Amended coding levels for the ECCS	.348
Table A.55 Additional codes used for dietitian talk	.357
Table A.56 Additional codes used for patient talk	.357
Table A.57 Key features of the four identified themes for information-giving exchange	jes
adapted from (Kiuru et al, 2004)	.359

List of Figures

Figure 1.1: The diabetes care team and the person with diabetes	48
Figure 1.2 Factors that may influence the outcome of the consultation	50
Figure 3.4: Dimensions of natient-centredness (Bensing, 2000)	88
Figure 4.5: Causal relationship between empathy and agreement on reported decis	ions
made in the consultation	130
Figure 6.6: Sequential Explanatory Design adapted from Creswell (2009)	166
Figure 7.7: Flow diagram illustrating number of patients contacted and the final num	nber
of patients consenting to take part in the study	175
Figure 8.8: Agreement on reported decisions for individual dietitians	207
Figure 8.9: Mean level of empathy for individual dietitians	208
Figure 8.10: Mean length of consultation for new and follow-up appointments and le	evel
of agreement on reported decisions	214
Figure 9.11: Model illustrating the interaction between empathy, communication skill	lls
and information exchange (adapted from Neumann et al, 2009)	220
Figure 10.12: Initial coding map used to identify information-giving exchanges within	า
the consultations	234
Figure 11.13: Examples of <i>persuasive</i> information-giving exchanges	252
Figure 11.14: Examples of supportive/collaborative information-giving exchanges	256
Figure 11.15. Examples of supportive/collaborative information-giving exchanges	257
Figure 11.17: Examples of commanding/toaching/instructing information-giving	209
exchanges	262
Figure 11 18: Thematic Map of information-giving exchanges and empathic stateme	nts
for consultations	266
Figure A.19 Example of <i>persuasive</i> information-giving exchange	361
Figure A.20 Example of <i>confirming</i> information-giving exchange	362
Figure A.21 Example of <i>confirming</i> information-giving exchange	363
Figure A.22 Example of recommending information-giving exchange	364
Figure A.23 Example of supportive/collaborative information-giving exchange	366

ACKNOWLEDGEMENTS

I would like to thank all my supervisors for their support and guidance whilst conducting this research.

Dr T C Skinner for believing that I could get started

Professor Frasier Reid for agreeing to take on this challenge

Dr Paul Farrand for his consistent and valuable guidance and support over the last four years

Dr Ian Dennis for his statistical advice and guidance

Professor Anne de Looy for supporting me through the final stages and the many insightful comments that have helped develop my thinking and supported the writing of this thesis

I would also like to thank those dietitians who agreed to take part in the study, as without their cooperation this research would not have been possible.

Finally, I would like to say a big thank you to my family for their patience, support, and understanding throughout this process.

In particular, I wish to thank my husband Murray without whom this thesis would never have been completed.

AUTHOR'S DECLARATION

At no time during the registration for the degree of Doctor of Philosophy has the author been registered for any other University award without prior agreement of the Graduate Committee.

The fees for this PhD were provided by the School of Health Professions.

Presentations and Conferences Attended are listed below:

ABSTRACT and POSTERS:

Diabetes Annual Professional Conference March 2006

Parkin T (2006) "What behaviours support patient-centred diabetic consultations?" (p 363) *Diabetic Medicine* 23 (Suppl 2) p120

ABSTRACTS and ORAL PRESENTATIONS:

Doctoral Research Conference University of Plymouth 25th June 2009

Parkin T, de Looy A, Dennis I, Farrand P "Exploring Patient and Professional

Agreement on Decisions made in the Consultation"

ORAL PRESENTATIONS:

3rd DIETS Conference, Lisbon, Portugal, September 2009

What do patients expect from us? Ensuring patient-centred care

Dietetic Service Clinical Governance Conference, Glasgow, March 2010

What is a patient-centred dietetic consultation?

Dietetic Department meeting, Barnstable, April 2010

What do patients expect from us? Ensuring patient-centred care

6th Cyprus Dietetic and Nutrition Conference (CyDNA), Nicosia, Cyprus

September 2010

de Looy A, & Parkin T "Are dietitians really competent to support weight loss strategies?"

South West Dietitians Research and Clinical Dietetics Study Day,

Plymouth, June 2010

Effective patient consultation skills - where is the evidence?

Word count of main body of thesis: 66178

Signed

Date 20th April 2011

CHAPTER ONE: SETTING THE SCENE

1.0 Significance of study

Professionals are increasingly expected to demonstrate the impact of their care on patient outcomes (NHS, 2011). The management of long term chronic conditions poses challenges in this area however, as many of the markers that are currently used to measure successful health outcomes can be influenced by a variety of factors other than the professionals input. This study is therefore significant in that it identifies an outcome measure that relates directly to the dietitians input in the consultation. It provides evidence to support the use of specific patient-centred communication skills/behaviours for chronic disease management and provides a mechanism to explain how these skills/behaviours interact to impact on the outcome measure of agreement on reported decisions made. Furthermore the outcome measure of level of agreement on reported decisions made is shown to be a reliable indicator of the presence/absence of these patient-centred communication skills/behaviours in the consultation.

1.01 Aims and research questions

The overall aim of this study is to identify specific patient-centred communication skills/behaviours occurring in consultations for chronic disease management and to explore their link with a tool recording agreement on reported decisions made.

Empathy, a crucial component of patient-centred communication, is explored alongside other key skills of active listening and information exchange. Mixed methodology design is employed to allow both quantitative and qualitative

research questions to be addressed. The quantitative analysis addresses the following three questions:

- What is the relationship between empathy and agreement on reported decisions made in the consultation?
- 2. To what extent do the various components of empathy (statements of challenge, statements of progress and statements of emotion) impact on agreement on reported decisions made in the consultation?
- 3. What is the relationship between patient satisfaction and empathy?

The qualitative analysis addresses the following four questions:

- 1. What are the components of information giving in the consultation?
- 2. How do the components of information giving relate to empathy?
- 3. How does information giving relate to agreement on reported decisions made in the consultation?
- 4. How do the components of information giving differ between consultations with agreement on reported decisions made and those with disagreement?

1.02 Structure of thesis

The thesis is structured as follows; Chapter one explores the complex nature of self-care management highlighting issues that may impact on an individual's ability to self-care, using diabetes as an example of a chronic disease. Lifestyle, particularly diet, is a major component of self-care management in diabetes (NICE, 2003; NICE, 2008a; NICE, 2009) and as a consequence dietitians have a significant role to play in the management of this condition. This management is delivered predominately through individual patient consultations and relies heavily on the effective communication skills of the dietitian. Chapter two explores the limited available data on the measurement of communication skills/behaviours against the outcome measure of agreement on reported decisions made. Chapter three provides a broader overview of the literature in this area by exploring patient-centred communication skills reported to impact on outcome measures other than agreement on decisions made. The literature reviewed identifies empathy, information exchange and active listening as important patient-centred communication skills, these skills are needed to support self-care management and therefore their potential impact on patient/professional agreement on decisions made in the consultation is explored. Chapter four reviews the role of patient self-efficacy, patient autonomy and patient satisfaction in patient-centred care. The hypothesised links to empathy and agreement on reported decisions made are outlined, and this provides the rationale for the measurement of these concepts alongside empathy, active listening skills and information exchange. The pilot study described in chapter five aims to clarify suitable methods for the direct measurement of empathy, active listening skills and information exchange in the dietetic consultation. It highlights difficulties with current tools for measuring active listening skills and information exchange, and provides the rationale for a mixed methods study to address questions raised. The design of a two phase mixed methods study to answer questions is outlined in chapter six. The quantitative phase of the study design (Phase I) is detailed in chapter seven, the results of this are presented in chapter eight. The qualitative study design (Phase II) is covered in chapter nine, qualitative methods used are detailed in chapter ten and the qualitative results presented in chapter eleven. The discussion in chapter twelve incorporates the results from both the quantitative

and qualitative analysis as per mixed methodology. The implications of the findings from this study for dietetic practice and other healthcare professionals working with chronic disease are summarised in the final chapter.

1.1 Introduction

The World Health Organisation (WHO) describes chronic diseases as those that persist over a long period of time, that are progressive in nature and that may result in complete or partial disability or even lead to death (WHO, 2009). The Department of Health considers chronic diseases to be conditions that cannot be cured, but that can be managed through medication and/or therapy (Department of Health, 2011). Chronic diseases include a range of conditions: the more common being heart disease, stroke, cancer, diabetes, respiratory disease, obesity, arthritis, chronic obstructive pulmonary disease (COPD), and mental health issues (WHO, 2009).

Chronic disease or long-term conditions are reported to be the most common cause of mortality in the world, representing 60% of all deaths (WHO, 2009). Currently there are around 15 million people in England with at least one long-term condition. Numbers are expected to increase, in particular those people with two or more conditions (Department of Health, 2011). This increase will have a huge impact on resources and care delivery. The specific aspects of the medical management will vary. However, there are generic components to the management of these conditions that will revolve around the individual's ability to learn to live with the condition and manage it on a day-to-day basis (Barlow et al, 2002; Department of Health, 2001; Lorig et al, 2001). This day-to-day management by the individual is referred to as self-care management and will

involve training and education from professionals and others in the specifics of the disease management process through medication and/or therapy (Clement, 1995; Department of Health, 2010a).

Diabetes Mellitus is one of the most common chronic diseases and, as such, provides a good model for exploring the issues concerned with self-care management. It currently affects approximately 2.2 million people in England (Department of Health, 2010c) and the incidence of diabetes is growing at a rapid rate. There are two major classifications of diabetes: type 1 and type 2. The majority (85-90%) of people diagnosed with diabetes will have type 2 (Department of Health, 2007). In some parts of the country, type 2 diabetes is increasing at a rate of 15% per year due to the ageing population and an increase in obesity (Department of Health, 2007). Diabetes UK (2009) estimates that by 2025, over four million people will have diabetes.

Diabetes has a major impact on the health of the individual: increasing the risk of morbidity and mortality and reducing the average life expectancy by 7-15 years (Department of Health, 2007; Roper et al, 2001; Dawson et al, 2008). Standard 3 of the National Service Framework for Diabetes (NSF for Diabetes) states that

"people with diabetes are empowered to enhance their personal control over the day-to-day management of their diabetes in a way that enables them to experience the best possible quality of life" (Department of Health, 2003; Department of Health, 2010c).

In common with other long-term conditions, the task of managing the condition lies with the individual. People with diabetes may spend on average three hours per year with a professional discussing their diabetes management and the remaining 8,752 hours caring for themselves (Department of Health, 2007). Self-care is therefore an important component of the management of diabetes, as all behaviours are actioned by the patient. Improvements in self-care depend on the patient's ability to successfully engage in a range of self-care behaviours (Aiken, Bingham & Piette, 2005; Glasgow et al, 1997; Goodall & Halford, 1991; Harrington, Noble & Newman, 2004; Hartz et al, 2006). Information is required to support these self-care behaviours; the ability of professionals to communicate this information to patients forms an essential component of effective self-care management.

The aim of self-care management is to prevent deterioration in health resulting from the chronic condition and to preserve function and quality of life. In diabetes, this will focus on reducing complications associated with macrovascular and microvascular disease. Macrovascular disease (large vessel disease) such as heart disease and stroke (Bell, 1994; Dawson et al, 2008) greatly increases in people with diabetes. The predicted risk of heart disease varies from 17% to 22.5%, depending on the risk score used (Kirk et al, 2007). In general, people with diabetes show a threefold increase in risk for all cardiovascular diseases (Bertoni et al, 2007; Stamler et al, 1993). Microvascular complications (small blood vessel disease) are a common consequence of having diabetes and occur in the kidneys (nephropathy), eyes (retinopathy), nerves (neuropathy), and limbs (peripheral vascular disease). Consequently, people with diabetes are more likely to suffer with heart disease and 80% will die of cardiovascular disease (Department of Health, 2007). Stroke, blindness, kidney failure, gangrene, and amputations are also common long-term problems associated with having diabetes. Treatment of these

specific complications currently accounts for 5% of the NHS budget (Department of Health, 2007), with up to 10% of hospital budgets used to care for people with diabetes (NICE, 2008a). However, economic models predict that the cost of managing the long-term complications in type 2 diabetes will rise to 40-50% by 2030 (Bagust et al, 2002). Diabetes therefore has a huge impact on current and future NHS resources and the general health economy, as well as the overall impact on the individuals themselves.

Many of the complications that are seen in diabetes can be avoided with earlier diagnosis and better preventative care (ADVANCE Collaborative Group et al, 2008; DCCT, 1993; Gillies et al, 2007; Nathan et al, 2005; Norris et al, 2005; Pan et al, 1997; Tuomilehto et al, 2002; UK Prospective Diabetes Study (UKPDS) Group, 1998; Uusitupa et al, 2000; Yamaoka & Tango, 2005).

Studies have demonstrated that intensive lifestyle intervention at the prediabetic state (impaired fasting glucose tolerance IGT) can significantly reduce the progression to diabetes by 58% (Gillies et al, 2007; Tuomilehto et al, 2002; Uusitupa et al, 2000; Yamaoka & Tango, 2005). Interventions that target both diet and lifestyle appear to be the most effective for reducing the incidence of type 2 diabetes (Li et al, 2008; Orozco et al, 2008). However, early diagnosis does present some challenges. When patients are diagnosed earlier, glucose levels may not be high enough to cause symptoms, thus patients are presented with a diagnosis at a time when they feel fit and well.

There are a range of issues that individuals will need to address in terms of dayto-day coping and self-care management skills when living with diabetes. These are highlighted below.

1.2 Treatment of diabetes

The main aims of treating diabetes are to alleviate those symptoms that will help people feel better; to regulate blood glucose levels in order to prevent high or low extremes (hyperglycaemia or hypoglycaemia); to reduce the risk of (macrovascular) cardiovascular disease and diabetic microvascular complications (retinopathy/ neuropathy/nephropathy); to maintain quality of life (NICE, 2008a; NICE, 2009; Williams & Pickup, 2004). These aims are achieved by a combination of drug treatment and lifestyle (diet/activity) management to control and modify parameters known to influence outcomes (Table 1.1).

Controlling blood glucose levels by:

- Eating carbohydrate in the right amount, at the right time and considering the type of carbohydrate eaten
- Checking blood glucose levels to monitor the balance between carbohydrate foods eaten and insulin and or oral hypoglycaemic medication
- Taking the correct amount of medication (tablets or insulin) at the right time of the day

Controlling blood pressure (BP) by:

- Eating specific foods in the right amount and following principles of the "Dietary Approach to Stopping Hypertension" (DASH) (Sacks et al., 2001) including reduction in salt intake
- Taking medication to help lower blood pressure (often 2-3 types of medication are required to lower BP)

Controlling cholesterol levels by:

- Targeting specific foods in the diet and/or by
- Taking cholesterol-lowering medication

Reducing or maintaining weight by:

- Increasing activity
- Eating smaller portions
- Taking medication where prescribed

Reducing cardiovascular risk by:

- Managing blood lipids
- Maintaining or reducing body weight
- Stopping smoking
- Increasing activity

Table 1.1 Summary of key areas for self-care management in diabetes

Other factors associated with reducing the risk of a cardiovascular event, such as smoking cessation, increasing physical activity and eating a healthy diet, are also targeted (Fagard, 2009; Hua & Tuomilehtoa, 2007; Whiteley et al, 2005). In summary, people with diabetes are required to engage in a set of complex behaviours in order to manage their condition and reduce the risk of long-term complications (Table 1.1).
1.2.1 Self-care Management

The literature consistently highlights that suboptimal control of blood pressure, lipids, and glucose is commonly experienced in diabetes; and that this is associated with poor long-term outcomes (DCCT, 1993; UK Prospective Diabetes Study (UKPDS) Group, 1998). Although there are some physiological factors that contribute to poor control (such as insulin resistance and other comorbidities) a major contributor is the low level of patient self-care (Benner et al, 2002; Donnan, MacDonald & Morris, 2002; Morris et al, 1997). Improvements in self-care depend on the patient's ability to successfully engage in a range of behaviours that will ultimately result in successful management of the key areas identified in Table 1.1 (Aiken, Bingham & Piette, 2005; Glasgow et al, 1997; Goodall & Halford, 1991; Harrington, Noble & Newman, 2004; Hartz et al, 2006).

The support provided by healthcare professionals (particularly dietitians) in supporting lifestyle and dietary change is fundamental to fostering self-care management in several key areas. These are: understanding food choice and lifestyle change, supporting the emotional changes, and understanding drug regimens more fully. The delivery of patient training and education in these areas may be provided through group education sessions (DAFNE Study Group, 2002; Davies et al, 2008; Deakin et al, 2006) and / or more commonly through one to one consultations (Duke, Colagiuri & Colagiuri, 2009). Training and education in these areas requires the professional to utilise a range of communication and teaching strategies in order to engage patients successfully in the self-care management of their condition (Anderson et al, 1991; Roach et al, 1992; Skinner et al, 2008; Stetson et al, 1992). However, this support can

only be effective if the structure of the healthcare system is also facilitatory. Each of these components can be explored for its significance in supporting self-care management by the patient.

Self-care management in diabetes can therefore be summarised as: effective lifestyle change and effective medication use. Evidence suggests that more effective self-care management leads to better metabolic control and improved functioning (DAFNE Study Group, 2002; Davies et al, 2008; Lorig, Ritter & Jacquez, 2005; Norris et al, 2002).

1.3. Possible rationale for poor self-care management

1.3.1 Lifestyle / Dietary compliance

There are problems documented with adherence to lifestyle management, which is a major component of diabetes care (Dietrich, 1996; Donnan, MacDonald & Morris, 2002; Foreyt & Poston, 1999; Huang, Gorawara-Bhat & Chin, 2005; Mc Nabb, 1997; Morris et al, 1997; Toobert, Hampson & Glasgow, 2000). Patient-reported adherence to lifestyle management is consistently poor, eg a rate of 33% for adherence to activity recommendations compared to 47% for foot-care (Toobert, Hampson & Glasgow, 2000) and 50% for dietary recommendations (Ary et al, 1986). In patients over 65 years of age, the figure for dietary compliance drops to 25%, with no further change in activity compliance (Huang, Gorawara-Bhat & Chin, 2005). Another small study (N= 75) reported daily dietary compliance rates of 26.7% (Travis, 1997). A large cross-sectional study (N = 5104): Data from the Diabetes Attitudes Wishes and Needs (DAWN), shows self-reported diet-adherence rates of 37% and 39% in people with type 2 and type 1 diabetes respectively (Peyrot et al, 2005).

However, the real levels of self-care may be much lower than this, as selfreported data regarding exercise and dietary behaviours are known to have a bias (Adams et al, 1999; Kristal et al, 1989; Krutz, 1990; Lichman et al, 1992) and are therefore unreliable (Mc Nabb, 1997).

The difficulty with dietary compliance may well reflect the multiple layers of potential dietary change and the many barriers in people's lives that may inhibit these changes from taking place (Sherman et al, 2000). Food is more than just fuel for the body and plays many roles in our day-to-day lives. The type and amount of food eaten may be determined by social, cultural, or religious factors, peer group pressure, food status and emotional drivers (Sherman et al. 2000: Vijan et al., 2005). The communication skills of the healthcare professional are therefore important when counselling patients on this complex area of dietary behaviour change (Gable, 2007; Pearson & Rapoport, 2007). Focus groups suggest that cost and smaller portion sizes were common barriers to dietary adherence in patients with diabetes (Vijan et al, 2005). The lack of desire to plan for meals may also affect appropriateness of food choices (Kavookjian et al, 2005), as may the lack of ideas for cooking (Brekke et al, 2004). Situational factors also play a role, eg eating out, a lack of appropriate food choices when eating out (Brekke et al, 2004), inappropriate food offers from others and lack of time to plan and prepare food (Ary et al, 1986; Maldonato et al, 1995; Tang et al, 2008). Interestingly, a survey of 446 people with type 2 diabetes showed that a modified diet of sugar and fat reduction was seen as more of a burden to participants than taking oral hypoglycaemics. This finding was related to age, with younger patients reporting this as more of an issue than older patients (Vijan et al, 2005). A strict diet aimed at weight loss that modified sugar, fat and

calorie intake was rated as burdensome as taking twice-daily insulin. Strict timings of meal times were also rated as burdensome (Vijan et al, 2005). Selfreported adherence was much higher for both, oral hypoglycaemics and insulin than moderate dietary changes (Vijan et al, 2005). It may be that individuals see these dietary changes as restrictive (Gentili et al, 2001) and impacting on their perceived quality of life (Watkins et al, 2000). Habit also plays a role in food choice and behaviour (Griffin et al, 1999; Lally, Chipperfield & Wardle, 2008). Dietary change and behaviour take time and require self-discipline (Kiuru et al, 2004) and people may forget and slip back into old routines (Brekke et al, 2004).

Lack of coping skills and the use of food to deal with emotions may also contribute to difficulties in adapting dietary intake (Rosenstock, 1985; Travis, 1997). Many factors interlink to affect the complex food choices that are made on a daily basis and account for the fact that dietary changes are generally the hardest for people to make when they have diabetes (Brown et al, 1998; Harrington, Noble & Newman, 2004; Peyrot & Rubin, 1994; Rubin, Peyrot & Saudek, 1991; Sullivan & Joseph, 1998). In addition, where conflict exists in balancing health and wellbeing, liberties are often taken with diet in order to minimise impact on wellbeing (Maclean, 1991). Lack of adherence may therefore be unintentional as life takes over (Rustveld et al, 2009). Numerous decisions may need to be taken when small changes in the diet are advised, such as cost considerations (Golin, DiMatteo & Gelberg, 1996; Jeon et al, 2009), availability, taste and impact on the rest of the family to name but a few. In addition, the positive benefits from making dietary changes must continually outweigh any barriers to change in order for these changes to be maintained

(Nestle et al, 1998). Uncertainty over the consequences of dietary changes and long-term benefits may further limit change from taking place (AbuSabha & Ahterberg, 1997). Not all of the food choices that people make are conscious decisions and there may be up to 200 daily food choices made that are unconscious (Wansink & Sobal, 2007). When multiple dietary changes are required, the level of complexity regarding food choices increases, resulting in a larger number of potential barriers that may inhibit or prevent these dietary changes from being maintained in the long term. Good communication skills are therefore needed to help the professional engage with the patient. Also needed is an understanding of the range of issues that may influence patient behaviour and food choices (Epstein, Alper & Quill, 2004).

Diabetes is a multi-faceted disease and different aspects of management may take precedence at different times (Glasgow & Toobert, 1988). Adherence to one part of the diet or diabetes regimen may be unrelated to adherence to another part of their self-care management (Goodall & Halford, 1991; Kasila et al, 2003; Maldonato et al, 1995; Webb et al, 1984; Wing et al, 1985). Diabetes management is therefore not a unitary concept, but a complex package of care. Patients will be engaged in different stages of change for differing elements of the diabetes management (Kasila et al, 2003) requiring the juggling of multiple unrelated behaviours on a daily basis (Asimakopoulou, Hampson & Morrish, 2002). The complexities of diabetes management may in part account for the poor levels of compliance and concordance seen in the literature (Kravitz et al, 1993; Sherman et al, 2000).

1.3.2 Understanding

There may be a more general lack of understanding of the importance of the need to take medication effectively (Rubin, 2005) or change lifestyle, and the importance attached to these changes may be low (Levy, 2009). Patients may not appreciate the seriousness of their condition (Arundel et al, 2003; Clark & Hampson, 2003; McCaul, Glasgow & Schafer, 1987) or misunderstand both the condition and its management (Holmstrom & Rosengvist, 2005; Kravitz et al, 1993; Lerman, 2005). Additionally, the information provided may not be clearly perceived by the patient as beneficial and therefore is unlikely to aid patient understanding (Street, 1992a). One study guestioned 200 patients with type 2 diabetes found that 64% thought that diabetes was a chronic disease and only 76% of patients with longstanding diabetes agreed that it was a chronic condition (Skinner et al, 2003c). This lack of understanding will ultimately be reflected in the lack of application to lifestyle change in the individual patient (Anderson et al, 2003; Skinner, 2004). Change only occurs in a specific healthrelated behaviour if the patient believes that continuing the current behaviour will put them more at risk and that change will decrease this risk (Harvey & Lawson, 2009; Travis, 1997). As with other personal belief models, if the patient believes that diabetes is not serious then they will be unlikely to change their behaviour in order to reduce the risks (Gentili et al, 2001). These studies suggest that healthcare professionals need to be communicating more clearly and effectively with patients in order to increase patient understanding (Heisler et al. 2002). Good communication skills will also improve professional understanding of the patient's perspective, allowing misconceptions to be addressed (Holmstrom & Rosenqvist, 2005).

Issues regarding health literacy and the ability of patients to grasp and understand information being delivered are therefore important (Powell, Hill & Clancy, 2007), lack of understanding results in patients being unable to interpret and use provided information to improve self-care management practices (Ishikawa et al, 2009). Good professional communication skills are essential as these can support appropriate delivery of accessible information to patients (Holli, Calabrese & O'Sullivan Maillet, 2003).

1.3.3 Emotional factors

Emotional factors as well as psychosocial factors also play a role in self-care management. Depression has been reported to be more prevalent in people with diabetes, and studies suggest having diabetes is a risk factor for developing depression (Eren, Erdi & Sahin, 2008). A recent large population study (N=37,291) found 20% of people with diabetes were depressed, compared to 4% of the non-diabetic controls. Interestingly, those individuals who reported symptoms of depression and anxiety at baseline had an increased risk of developing type 2 diabetes at 10-year follow up. This finding was independent of other known risk factors such as socioeconomic, lifestyle and markers of metabolic syndrome (Engum, 2007). Depression greatly reduces people's ability to engage in self-care management. In diabetes, this has been related to poorer glycaemic control and a greater risk of associated micro and macro vascular complications (Glasgow, Toobert & Gillette, 2001; Lustman et al, 2000; The DAWN, 2002). In a 5-year follow-up study, depression increased the hazard risk of microvascular complications by 1.36 and macrovascular complications by 1.24 (Lin et al, 2010). Depressed patients also have lower adherence rates to drug therapy (Cramer, 2004) and in one meta-analysis, non-

adherence was reported as 3 times greater in depressed patients compared to non-depressed patients (Lustman et al, 2000). Exercise, diet, and medication are particularly difficult areas to maintain when depressed (Lin et al, 2004), with high levels of depressive symptoms being reported in patients with poor adherence to diet (Ciechanowski, Katon & Russo, 2000). This effect appears to be more marked in men then in women (NauAikens & Pacholski, 2007). It has been suggested that this effect is mediated through higher perceptions of barriers and reduction in self-efficacy (Chao et al, 2005; Sacco et al, 2007). The concept of "burnout" has been used to explain this unwillingness to change, ie previous lack of success, constant failure and general fatigue with trying to bring about change, especially when efforts are not rewarded (Polonsky, 1996). There are also the day-to-day stresses and hassles of living with diabetes and trying to make things fit (Albright et al, 2001; Anderson, 1995). In one study, emotions accounted for 41.3% of reasons for failure to adhere to a dietary regimen. This effect was significant (p < .05) for younger patients (< 65 years of age) and for females (Travis, 1997).

In addition, there is a higher incidence of eating disorders in the diabetic population that predisposes this group to higher morbidity and mortality rates (Herpertz et al, 2001; Steel et al, 1989; Takii et al, 2008). In type 1 diabetes, this will contribute to poor control, as insulin is omitted in order to prevent weight gain, resulting in elevated blood glucose levels (Goebel-Fabbri, 2009; Takii et al, 2008). Binge eating, restraint and body dissatisfaction frequently occur in individuals with type 2 diabetes, resulting in low self-esteem and further contributing to poor outcomes (Yannakoulia, 2006). Type 2 patients are more likely to be overweight and will have a greater body mass index (BMI) and this

is a predisposing factor for developing type 2 diabetes (Sacco et al, 2007). A greater BMI has also been found to increase depression in diabetes (Sacco et al., 2007). Binge eating is reported as preceding the diagnosis of type 2 diabetes (Herpertz et al, 2001). This could be one of the causes of obesity, leading to type 2 diabetes and will contribute to poor self-care practices.

The method of treatment may add to the emotional distress and burden felt by individuals, with those patients requiring insulin reporting greater levels of diabetes-related emotional distress (Delahanty et al. 2007). Changing behaviours relating to exercise and weight control may have a far more dramatic effect on health outcomes, as these changes may mediate the relationship between depression and outcomes (Wing, Phelan & Tate, 2002). Emotions therefore play an important role with regard to the patient's ability to cope, eg emotional wellbeing has been seen to contribute significantly to effective self-care management (The DAWN, 2002). These studies suggest that addressing emotions raised in the consultation is important in supporting the patient's ability to cope. This requires good communication skills in order to listen to and respond to the emotions raised appropriately (Gable, 2007).

1.3.4 Drug compliance

Drug compliance also plays a part in the self-care management of diabetes. Data on medication adherence highlights difficulties with single drug use, with only 33% of patients on monotherapy taking their medication for more than 80% of their days on therapy (Morris et al, 1997). This suggests that more complicated medication regimens involving multiple drug use, may achieve lower levels of compliance. Other studies and reviews looking at drug

compliance have highlighted decreasing adherence as medication prescription use increases (Claxton et al, 2001; Donnan et al, 2002; Odegard & Gray, 2008). Maintaining adequate medication use over a period of time is also problematic, with studies showing adherence rates falling to 16 - 80% over a 6-24 month period (Cramer, 2004). Medication adherence was also found to be poor in patients who are socially deprived and who have both macrovascular and microvascular complications and in those people with a long duration of diabetes (Donnan et al, 2002). People with a long duration of diabetes are more likely to have complications and other co-morbidities (Maddigan et al, 2005) resulting in multiple medication use. This complexity may add to the daily struggle to manage diabetes (Albright et al, 2001; Shah et al, 2003; Stewart & Liolitsa, 1999) contributing to poor adherence to their medication. This is similar to findings from a large study of 7227 patients with chronic kidney disease, where adherence to medication was found to decline with severity of condition (Schmitt et al, 2010). 33% of patients with long-term severe chronic kidney disease had poor medication adherence.

Interventions for improving effective long-term use of self-medicated drugs are complex and are reported as being generally not effective (Haynes et al, 2005). It has been suggested that improving outcomes may be made by taking time to explain to patients how best to use their medication (Odegard & Capoccia, 2007), as well as explaining its function, and developing strategies to support this (Bezie et al, 2006). A review by Haynes et al (2000) suggests that this may be too simplistic a view but that it could form part of an overall care package that includes dedicated time spent in discussion with patients, reviewing drug use, function and strategies to support effective drug uptake. A more recent

review by Haynes et al (2005) also noted that telling patients about adverse effects of their medication did not affect their use of the medication. Forgetting to take medication and running out of medication are commonly cited reasons for non-adherence to medication therapy (Hill-Briggs et al, 2005; Odegard & Gray, 2008). Habit can therefore be seen to play a role in medication adherence, and it has been suggested that patients should be taught at the start of drug therapy to "form habits integrated within their cognitive procedures" to help assist adherence to the drug treatment (Reach, 2005, p.419). Data from the Diabetes Control and Complications Trial (DCCT) suggests that targeting tight control at diagnosis significantly reduces the risk of cardiovascular events by 15% and decreases mortality by 14% (Gaede et al., 2008). Additional studies predict reductions in microvascular complications when drug compliance is improved (Yu, Yu & Nichol, 2010). Focusing on drug compliance may therefore be one way by which health outcomes in diabetes could be improved (Kogut et al, 2004). Theory-based interventions to address medication adherence in people with Type 2 diabetes are currently under development (Farmer et al, 2008). A recent intervention into medication adherence in chronic disease in adolescents and children suggests that a behavioural component is required alongside the education to enhance medication adherence (Dean et al, 2010). The delivery of a behavioural component in patient education will require good communication skills from the professional to be effective (NICE, 2007).

1.3.5 The Structure of Healthcare Delivery

The structure of current healthcare delivery systems, including patients returning for management review, access to the services of health providers

and access to healthcare information may also contribute to poor self-care (Eccles et al, 2007; Renders et al, 2000). Eccles et al (2007) have suggested that restructuring the delivery of care may to lead to more effective outcomes as it better deals with the needs of the patients (Hiscock et al, 2001). Continuity of care and/or having a personal source of care, have positive effects on adherence (Rosenstock, 1985). Healthcare teams may have different impact on individuals within consultations and within teams and across different organisational structures (Francis et al. 2008) leading to further variability in patient outcomes for self-care management (NHS, 2011). Although professionals can be a source of influence in promoting positive dietary changes they may not always be seen as helpful (Goldstein et al, 2004).

As well as a variety of healthcare delivery systems, patients are also exposed to multiple sources of advice, not just from professionals, but also from well - meaning friends and relatives (Chapman et al, 2005). Lack of support and family issues have been cited as common barriers to following dietary recommendations in people with type 2 diabetes (Vijan et al, 2005). Younger patients and those with a shorter duration of type 2 diabetes who work long hours and live in rural communities are reported as having lower dietary compliance (Beisecker & Beisecker, 1990; Shah et al, 2003). This may reflect lack of patient engagement, as a shorter duration of diabetes will mean less chance of complications. Individuals are likely to feel fit and well and other life issues may take priority (Shah et al, 2003). Advice on diabetes can be obtained from a variety of media sources (Huang et al, 2005) including the internet. This advice may be conflicting and this will add to the confusion over which self-care behaviours should be undertaken. Reasons for poor self-care are therefore

varied but may relate to one or more of the following: lack of education, poor understanding, conflicting information, inaccurate information and advice, inappropriate drug treatment and dietary changes.

The life of patients outside their diabetes will greatly influence their self-care management strategies. Social support from families remains an important issue, with strong family support being a consistent predictor of adherence (Beverly & Wray, 2010; Glasgow & Toobert, 1988; Kelsey et al, 1996; Searle et al., 2007; Vijan et al., 2005. This is especially true for diet (Albright et al, 2001; Stephens et al, 2010; Tang et al, 2008) and particularly when support provided is positive rather than coercive (Stephens et al., 2010). Married men achieve better dietary adherence compared to those who are single (August & Sorkin, 2010). Social control using persuasion resulted in better dietary control among married patients (August & Sorkin, 2010). Other dietary features shown to benefit from social support are adherence to a healthy eating plan, spacing out carbohydrate over the day (Tang et al, 2008) and adherence to lipid lowering diets in hypercholesterolemic men (Bovbjerg et al, 1995).

1.4 The factors demonstrating improved self-care management

As previously quoted, Standard 3 of the National Service Framework for Diabetes (NSF for Diabetes) states that "*people with diabetes are empowered to enhance their personal control over the day-to-day management of their diabetes in a way that enables them to experience the best possible quality of life*" (Department of Health, 2003; Department of Health, 2010c). The healthcare professional becomes a critical component in empowering the patient to good and sustained management.

A major part of improving self-care management is the acquisition of the necessary knowledge and skills needed to enable effective day-to-day self-care management to occur (Anderson et al, 1991; Department of Health, 2003; King et al, 2002). In practice, these skills and knowledge are delivered by the diabetes care team, who aim to work in a multi-professional way, providing consistency of care and consistent messages with regard to diabetes management (Department of Health, 2003; Department of Health, 2007; NHS, 2011). Contact with members of the diabetes care team will depend on the stage of the diabetes, the presence of complications and whether the patient is managed primarily in the community setting or in the acute setting. Figure 1.1 illustrates some of the members of the diabetes care team who may be involved.





The work of the team is only as effective as each of its component parts and the attitude of the practitioner at the point of diagnosis is also crucial, as this conveys the sense of importance and seriousness of the disease and may influence subsequent compliance levels (Dietrich, 1996). Patient perceptions of professionals at diagnosis, in terms of reassurance and provision of a clear action plans, has a long-lasting impact resulting in less diabetes distress and better outcomes 1-5 years after diagnosis (Polonsky et al., 2010). Beliefs regarding treatment effectiveness have been shown to be predictive of dietary intake (Hampson et al, 1995). The assessment of personal belief models and individual strategies in patients that are consistent with their perspectives are therefore important factors to be considered (Hampson, Glasgow & Foster,

1995; Vijan et al, 2005). Time also needs to be spent raising awareness and exploring misconceptions (Sherman et al, 2000), which might have arisen due to different explanatory models used by patients and professionals (Loewe & Freeman, 2000). These studies highlight the importance of communication skills within the consultation to assess the patient's perceptions, beliefs, needs, and areas of misconception. Worry, guilt, uncertain acceptance of diagnosis as well as unclear treatment goals are further factors that may contribute to patient distress (Delahanty et al, 2007). This further highlights the need for good communication skills to identify and address patient concerns appropriately. Failure to address these issues due to the poor communication skills of professionals involved in care delivery may increase levels of poor compliance (Freeman & Loewe, 2000). Empowering patients to take more responsibility for the management of their condition (Poskiparta et al, 2001) may help to improve outcomes, as demonstrated by those patients who belong to patient organisations and support groups (Selvais et al, 2008; Simmons, 1992). Lack of problem-solving skills may also contribute to the inability of patients to implement required changes (Bonnet et al, 1998). This is particularly true for those self-care areas that demand complex behavioural lifestyle adjustments, such as exercise and diet (Glasgow & Osteen, 1992; Heisler et al, 2007).

The age and gender of the physician may also influence the outcome of the consultation and should be considered when reviewing data (Hall et al, 1994; Scmid Mast et al, 2007; Siminoff et al, 2006). Figure 1.2 summarises key factors that may affect consultation outcome.



Figure 1.2 Factors that may influence the outcome of the consultation

Earlier diagnosis may mean patients are presenting fit and well, with no obvious symptoms of diabetes, whilst medicating for the newly diagnosed condition of diabetes. Furthermore, the drugs taken to manage the long-term effects of diabetes may cause side effects, resulting in individuals feeling unwell (American Diabetes Association Position Statement, 2002). Consequently individuals may perform a cost-benefit analysis and conclude that the future benefits of continuing with the treatment are outweighed by current factors such as feeling unwell (Kolbe, 2002). This is a difficult issue to deal with, as earlier diagnosis and intensive management have been shown to lower overall risk associated with diabetes (Ambady & Chamukuttan, 2008; Gaede et al, 2008). This raises the question of how to engage people with diabetes effectively while acknowledging these complex sets of behaviours (Anderson et al, 1991). The

aim is to enable effective self-care management and for these behaviours to continue over time to improve health outcomes.

1.4.1 Role of the Dietitian

The dietitian has a major role to play in the diabetes care team, as dietary management features in both the short term (to alleviate symptoms) and the long-term (to reduce the risk of complications) (Choudhary, 2004; Diabetes and Nutrition Study Group of the European Association for The Study of Diabetes, 2000). Diet may require additional adaptation depending on the age of the patient, the presence of complications, and other co-morbidities (NICE, 2008b; NICE, 2009). Poor manipulation of diet may add to the burden of greater drug use in order to manage the condition (Franz 2004). The effectiveness of drugs could be improved by the appropriate manipulation of the diet (Delahanty & Halford, 1993; Fleming, 1996; Franz 2004; Garg, 1998; Gillies et al, 2007; Harsha, 1999; Jenkins et al, 2005; Mulrow et al, 2000; Thomas & Elliott, 2009). As highlighted, dietary change is complex; there are many possible factors that can influence food choice and behaviour and these will differ from person to person. The ability to communicate effectively with patients is an essential skill needed by dietitians in order to manage this complex area of behaviour change (British Dietetic Association, 2008b; Davidson et al, 1987; Horacek, Salomón & Nelsen, 2007; Pearson & Rapoport, 2007; Whitehead et al, 2009).

Dietitians also play a key role in the ongoing support and training of the rest of the diabetes care team to ensure that consistent and accurate messages around food are delivered by all. There is therefore a need to know that

dietitians are practising effectively (Cant & Aroni, 2008; Jarret & Payne, 1995) in order to instil effective practice in others (Ross, Bower & Ibbald, 1994).

1.5 Summary

Diabetes mellitus is a complex chronic disease and as such is a good model for the study of long-term conditions and interventions that will improve self-care management. Poor self-care management has huge implications both to the individual with diabetes and to the larger health economy (Bagust et al, 2002). There is evidence to show that the risk of complications and therefore cost implications and impact to individuals can be reduced with improved control of the diabetes (DCCT, 1993; UK Prospective Diabetes Study (UKPDS) Group, 1998). However, good diabetes management requires the patient to engage in a complex range of self-care management behaviours, which will be rooted in their individual life situations (Moser et al, 2008). The evidence in this chapter suggests that patients struggle with self-care management for a variety of reasons, the main factors being complexity, permanency, and the requirement for substantial behavioural change from the individual (Moser et al, 2008). The different aspects of the self-care management taking priority over others at different times during the course of the patient's life with diabetes (Kolbe, 2002), adds further complexity.

This raises the question of how professionals should engage with people with diabetes to enable empowerment and enhanced personal control to occur in order to meet standard 3 of the National Service Framework (Department of Health, 2003).

Although systems play a role in effective diabetes care, a major contributing factor to the patient's self-care management will be the interaction that occurs between the patient and the professionals involved in the delivery of care. The patient consultation is the main vehicle through which patients are educated and supported in managing their diabetes. In order to improve self-care management there is a need to review the consultation process to ensure that professionals are able to work effectively with patients (Department of Health, 2003; Department of Health, 2010c) to improve long-term health outcomes (Aiken, Bingham & Piette, 2005).

The communication process that occurs in the consultation is pivotal to this success. Professionals have an important role to play, not just in providing the necessary technical skills and information required for management, but also in understanding the issues from the patient's perspective. Developing this understanding through good communication will allow information provision to be tailored to meet the individual's needs (Epstein, Alper & Quill, 2004; Thorne, Paterson & Russell, 2003), thereby supporting self-care management that is achievable for that patient. Good communication skills are more likely to result in improved self-care resulting in improvement in long-term health outcomes (Epstein et al, 2005; Heisler et al, 2007; Stewart, 1995).

CHAPTER TWO: LITERATURE REVIEW

This chapter describes the literature review undertaken to identify behaviours and skills used in consultations that have been highlighted as supporting patient-centred care. In particular, it reviews those skills and behaviours that have been linked to improvements in outcomes. The method used for the literature review is outlined and a summary of the key findings is provided.

2.0 Background

The primary function of the professional's consultation is to provide the patient with education and information to help develop their knowledge and understanding and provide them with skills for effective self-care management (Anderson et al, 1991; King et al, 2002). This should help the patient to implement lifestyle changes and improve adherence to medication/treatment that will enhance or improve their health and reduce the risk of long-term complications associated with diabetes (Ellis et al, 2004). Education of the patient involves much more than just offering information (Matteucci & Giampietro, 2003), as information alone does not change behaviour. This is a finding that is consistent across healthcare professional groups (Arborelius, 1996; Clement, 1995; Coates & Boore, 1996; Contento, Balch & Bronner, 1995; Cullen, Baranowski & Smith, 2001; Daneman et al, 1985; Day, 2000; Dunn, 1988; Glasgow, Marrero & Fisher, 1999; Goodall & Halford, 1991; McClusky & Lovarini, 2005; Norris, Engelgau & Narayan, 2001; Schnoll & Zimmerman, 2001). Provision of information should aim to improve patient's knowledge and understanding, encourage informed decision-making (Anderson et al, 1991; Kavookjian et al, 2005). This requires an individualised approach to setting goals, gathering information, and giving advice (Chapman et al, 2005). This

process should also allow consideration of the psychological, social, and emotional impact of having a chronic disease such as diabetes (Funnell et al, 2005; Rapoport, 1998). The way that healthcare professionals communicate with patients is acknowledged as having an important role to play in the transfer of information to assist patient understanding and this requires further clarification (Griffin et al, 1999). This is particularly pertinent for dietitians, whose primary tool for engaging patients in behaviour change is through the communication process that occurs in the consultation (Pearson & Rapoport, 2007). A study in 2005 of people with diabetes and their interactions with providers in a range of primary and specialist clinic settings demonstrated that patient self-care management and patient-provider communication were independent (Aiken, Bingham & Piette, 2005). This highlights the further need for research into effective communication between patients and professionals to improve health outcomes.

In a normal clinical setting, at the end of the consultation, all professionals record in the patient's notes the decisions that have been made regarding care management, (BDA, 2008; Department of Health, 2008). The majority of these decisions will inevitably require the patient to implement new behaviours or change their current behaviours (as outlined in Chapter One). The ultimate aim of these decisions is to improve self-care management practices, with a view to enabling better long-term health outcomes for that individual (Kaplan, Greenfield & Ware, 1989). If patients cannot recall decisions made in the consultation then they cannot implement them and improvements in health outcomes are unlikely to occur. Understanding the processes that lead to clear and accurate recall of decisions made in the consultation is therefore a good

starting point for addressing the issue of appropriate communication skills in consultations (Shaw et al, 2005). The ability to recall the decisions made at the end of the consultation is an important first step for patients in enabling them to implement long-term improvements in outcomes. However, the consultation is a two-way communication process involving the patient and the professional. A more accurate measure of communication effectiveness would therefore be the level of patient and professional agreement on reported decisions made in the consultation.

2.1 Aim

The aim of this review is to identify behaviours and/or skills used in consultations that are linked to the outcome measure of recall.

2.2 Search method

A literature search was conducted to identify published papers through the following search engines: Medline, Embase, CINHAL, AHMED, and PsycInfo. The three known papers in this area were not retrieved in the initial search when including the terms *dietitian/dietician/dietetics*. The difficulty in retrieving papers initially may reflect the limited research in this area. As a consequence, a backward search was conducted meshing the key words on the three known papers in this area. This search was then further expanded and refined to capture other papers relevant to the key search terms (see Table 2.2). The concepts used in the search are equally legitimate domains for other professional groups and it was reasoned that it would remain valid for the questions involving professional behaviours, without the need for the inclusion of the terms *dietitian/dietetics*.

As the earliest article retrieved on dietetic-patient interaction was from 1987, the search was limited to articles published in English between 1985 and 2011. The completion date for the initial search was December 2005. This was the start date for the exploratory study detailed in Chapter Five. The search was re-run from December 2005 to March 2011; no further papers were retrieved.

The search strategy involved retrieval of relevant abstracts and scanning according to selection criteria, before full text copies of papers were obtained and reviewed as outlined in Figure 2.3.



Figure 2.3: Search strategy used for literature review

2.2.1 Criteria for inclusion

The results included articles that mentioned specific behaviours and/or skills that had been measured in consultations in relation to self-care outcomes, specifically: goals, decisions made, and recall.

2.2.2 Selection of papers

To determine which papers were to be selected for review, the titles of the abstracts and the abstracts were reviewed. The paper was included if the information given suggested that a behaviour or skill of the professional had been measured against an outcome measure of goals, recall or decisions and self-care. If there was any doubt regarding these criteria, from the information given in the title and abstract, or if the abstract was absent or not clear enough to draw conclusions, the full article was retrieved for further consideration. Studies were eliminated if a clear behaviour or skill had not been measured, or if the measured skill had not been assessed against the outcome measures listed.

2.2.3 Search Strategy

The databases were searched using the key words shown in Table 2.2. The Medline search below clarifies the search process used:

	AND	AND	AND				
Treatment outcomes	Motivation	Patient education	Self-care				
Agreement Decisions Patient compliance Patient adherence Patient concordance	Personal autonomy Self-efficacy Goals	Professional patient interaction Recall Communication skills Communication behavio* Empathy Professional patient relations Attitude of health personnel Consultation skills Patient-centred* care					
		Patient focused care					
*Donates various suffices searched including American and English spellings, eg							
benaviour/benaviours, behavior/behaviours							

Table 2.2: Search terms used to identify the measurement of consultation skills / behaviours against outcome of recall / goal or decisions.

Medline and Embase had similar search strategies applied, whereas the search

terms used for PsycInfo, CINAHL and AMED were amended slightly to reflect

the databases being searched.

2.3 Results

Due to the limited number of papers initially retrieved a further search was conducted on some of the key themes identified in these papers, specifically empathy, empowerment, motivation, goal setting, adherence strategies, behaviour change, and cognitive behaviour therapy. These key terms were then searched independently against *dietitian/dietetics/dietician* to see if this enabled retrieval of any further papers in the field of dietetics. A total of 1,336 references were retrieved and none of these met the inclusion criteria. One additional paper was identified from a conference abstract and the author was contacted and a copy of the findings was provided.

A total of seven papers met the inclusion criteria, six were empirical studies (three observational, and three cross-sectional) and one was a qualitative study.

Papers are grouped in a logical way according to the behaviour/skill or outcome measure assessed. Table 2.3 shows a summary of the study design, participant characteristics, identified behaviours, outcome measures, results, and implications reported by authors.

Author	Study design	Participants	Identified behaviours	Measures	Results related to behaviours and outcomes	Implications
Coughlan (2003)	Observational study 1 centre, (secondary care)	N=30 patients with diabetes Gender male=50% N=2 male consultants	Empathy	Empathic communication coding system (ECCS) RIAS global affect ratings Levinson's empathic clues and missed opportunities Patient autonomy (HCC)	Empathy correlated to agreement on decisions made ($r = .486$, $p =$.016) using the ECCS. The male physicians in this study were found to be significantly more empathic to female patients than male patients t(16.16) = -2.35, p = .032). The more empathic the consultant the shorter the consultation ($r =41$, $p = .026$) using Levinson's coding	More empathic the consultants the greater the agreement on recall and the shorter the consultation.
Goodchild et al (2005)	Observational study 1 centre, (secondary care)	N= 40 patients with diabetes Mean age = 55.7years Gender male = 57.5% Type 1 diabetes 40% N = 3 female dietitians (3-12 years working in diabetes)	Empathy	Patient and professional recall agreement on goals/ decisions Patient satisfaction (MISS) Patient autonomy (HCCQ) and Empathic opportunities presented in the consultation and providers response to these (ECCS)	The more empathic the professionals' response to emotional opportunities, the more satisfied patients were with their consultations ($r = .41$, df = 15, $p = .05$). There was a non-significant trend that the more empathic opportunities that arise during a consultation, the higher the agreement between patient and dietitian on what was discussed ($r = 0.28$, $p = .07$). Patients reported more autonomy support when they created more empathic opportunities during their consultation. ($r = .29$, $p = .07$).	Professionals response to empathic opportunities may be a useful component to dietetic consultations

Heisler et al (2002)	Cross sectional study 1 VA centre (survey)	N= 2,000 surveyed (66% completed survey of the total sample this equates to N= 1,314) Mean age 67 years Gender male 98%	Decision- making Clear information on decisions treatment options Desire to participate in decision- making needs assessing	Provider participation decision-making style (PDM style) Providers communication (PCOM) Patient self- management Patient understanding of diabetes self-care	Provider communication significant predictor of self- management β 0.18; p < .001 Understanding was strongly and independently associated with self-management β 0.25; p < .001	Provider styles enhance self- management through increased patient understanding or self confidence
Heisler et al (2003)	Cross sectional study In 2 settings	 N = 127 patients Gender male 81% Mean age 64 years 100% type 2 97% were on oral medication 67% had diabetes for 4 or more years N = 50 providers 56% males Mean 12 years in practice professionals (endocrinologists and primary physicians) 	Shared decision- making	Recall agreement on treatment goals and treatment strategies Self-efficacy Patient centeredness of providers (HCCQ) Patients belief in treatment efficacy	When there is agreement on goals there is higher patient self- efficacy and assessments of diabetes self-management. Shared responsibility in treatment decisions The stronger the belief in efficacy of treatment the greater agreement on gaols $p = .013$ Patients more educated more likely to agree with providers goals $p = .015$ Sharing responsibility in treatment decisions resulted in greater agreement on treatment plans $p = .026$ Patients who shared providers top treatment strategy had higher	Need to focus on strategies that increase agreement on goals and goal strategies as these improve self- efficacy and self- management. Greater concordance on goals may be mechanism by which patient provider communication and contribute to improved patient outcome

					self-efficacy $p = .005$	
					Patients who agreed with	
					providers on more treatment	
					goals rate diabetes self-	
					management more positively	
					<i>p</i> = .004	
Kravitz et	Cross sectional	N = 1751 patients	None	Self- reported data on	Proportion recalling	Patients failed to
al (1993)	study	Moon ogo -61		recall of 15 possible	recommendations to take	recall important
	Data collected	wear age =01		recommendations	medications (≥ 90%) exceeded	medical advice did
	from physicians	years		provided by physician	the fraction recalling	not always adhere
	in four	Gender male =45%			recommendations to follow a	to advice recalled
	specialities			plan	and perform various self-care	Even when patients
	(internal	13% diabetes		Patient questionnaire	activities (22% to 84%)	do recall
	medicine, family	50% hypertensive		administered 3-4		information many
	practice,	5% in in past year		months later asked	Adherence to recommendations	patients do not
	endocrinology,	beart failure		extent to which they	was correlated to reduced serum	incorporate
	cardiology)	neartrailare		had performed the 15	glucose $r =33$ and	treatment
	In three systems	Of these patients		behaviours during the	glycohaemoglobin ($r =25$) in	recommendations
	of care health			past 4 weeks and	reduced diastelia blood processor	into their daily lives
	maintenance	(16% had diabetes			in hypertensive nationts	Self-reported
	organisations,	and hypertension)		disease activity	(r = -15)	adherence
	large	3% had diabetes		diocado donvity	(/ = 110)	measures are at
	multidisciplinary	and heart disease				least weakly
	groups and solo					associated with
	practices)	6% had				clinical measures
	In three cities	hypertension and				and the
		neart disease				associations tend
		4% had diabetes.				direction one would
		hypertension and				
		heart disease				erheor
						Research needed
						to help patients
						understand recall
						and carry out

						recommendations
Parkin & Skinner (2003)	Observational study 1 centre (secondary care)	N= 141 patients with diabetes Mean age = 55.6 years Gender female 54% Type 1 =31% Professionals all female N= 2 dietitians N= 7 diabetes specialist nurses (1-17 years diabetes practice)	Perception of autonomy and autonomous motivation	Patient and dietitian recall agreement on a) goals b) decisions Patient satisfaction (MISS) Patient autonomy (HCCQ) Autonomous motivation (TSRQ)	Patient and professional's perceptions of consultation don't correlate. Patient and professionals disagreed on the issues discussed 19.6% of the time, on the decisions made 20.7% of the time and goals set 44.3% of the time. Autonomy support in the consultation was associated with greater autonomous motivation for self- care ($r = .31$; $p < .001$) More controlled motivation was associated with less agreement on issues discussed and goals set ($r = .20$; $r = .24$; $p < .05$). Multiple regression suggests more autonomy support; older age predicted more autonomous motivation in patients, accounting for 29% of variance seen	Increasing behaviours that support patient autonomy may be one way in which agreement on recall can be improved.

Huang et al (2005)	Qualitative study Exploratory Semi- structured interviews using grounded theory approach	N = 28 patients all type 2 diabetes mean age 74 years gender female 57% 27 out of the 28 had complications	Self- reported health goals Factors that shape health goals Self-care practices and how these relate to self- care behaviours	 Key themes explored 1.Patients healthcare goals 2.Diabetes mellitus care goals 3.Expereinces with diabetes mellitus 4. Priorities for self-care strategies 	Patients describe healthcare goals in global and functional terms rather than biomedical Patients couched goals in aspects of daily routine life rather than the specifics of diabetes mellitus care Patients predominant healthcare goal was to maintain independence in activities of daily living	Enhance communication by exploring patient's specific circumstances; reframing diabetes treatment goals in patients own language Crucial step for developing individualised care
			care behaviours influencing these	4. Priorities for self-care strategies5.Daily self-care tasks	independence in activities of daily living Greatest variability in self-care was around diet and exercise.	Crucial step for developing individualised care plans

Table 2.3: Study design, participants, behaviours, measures, results, and implications of findings of papers reviewed

2.3.1 Methodological quality

In order to be significant, study findings need to be generalised to the larger population. Huang et al (2005) selected a group of patients who volunteered to talk to interviewers. This group comprised older patients (>65 years), with type 2 diabetes, and were predominately African American. The ethnic identity of this group may influence the applicability of findings to the wider population (Huang, Gorawara-Bhat & Chin, 2005). Three of the studies focused on patients in one diabetes centre. However, these studies did not explore whether the participating patients were typical of the usual patient population attending these clinics (Coughlan, 2005: Goodchild, Skinner & Parkin, 2005: Parkin & Skinner, 2003). In addition, patients attending secondary care for diabetes management may have greater disease severity and therefore present with differing expectations and complexity of needs that may impact on measured outcomes (Maddigan, Feeny & Johnson, 2005). One of these studies reported only patient gender; additional demographic data was not provided (Coughlan, 2005). Heisler et al (2003) used two large settings and it would be supposed that this would be more typical of the larger population. The population studied in Heisler et al (2002) were predominately older male patients and therefore did not represent the population as a whole. Kravitz et al (1993) presented a good methodological study, recruiting a large number of patients (N = 20,223) from a wide range of settings and therefore the sample of patients used was representative of the population group as a whole. Data in this study is presented on participating and non-participating clinicians and patients.

In terms of the professional groups involved in the studies, one study looked at dietitians (Goodchild, Skinner & Parkin, 2005) and one looked at nurses and dietitians (Parkin & Skinner, 2003). Data was not provided on the number of consultations completed by nurses and dietitians in the later study, so comparisons between groups cannot be drawn. In addition, each study collected data from a single site. The practice of individuals working in the same centre may be influenced by exposure to the working practice of other professionals and to delivery of care within that setting (Francis et al, 2008). This may have skewed results, and therefore the findings may not reflect the practice of nurses/dietitians as a whole. This may also be true for the small data set from the two consultants working in the same site in the Coughlan (2005) study. However, data collected on consultations for physicians/clinicians was much larger in the Heisler et al (2002) and Kravitz et al (1993) studies and covered multiple settings. Therefore, findings are more likely to accurately reflect practice in that group as a whole.

2.3.2 Recall as an outcome measure

Recall as an outcome measure was used in five of the identified studies, although different methods were used to measure recall. Immediate auditory recall was used in three of the studies (Coughlan, 2005; Goodchild, Skinner & Parkin, 2005; Parkin & Skinner, 2003) whereby patients and professionals where asked to independently report all decisions or goals made in the consultation. The other two studies used prompt lists to assist recall (Heisler et al, 2003; Kravitz et al, 1993).
A total of 181 nurse/dietitian consultations were assessed using immediate auditory recall. In the Coughlan (2005) study patients were asked to recall decisions made at the end of the consultation, and consultant decisions were extracted from the letters that were dictated to the patient's GP at the end of the consultation. The agreement rate on recalled decisions for patients seeing dietitians and nurses (N= 141) was 47.1% (Parkin & Skinner, 2003). Although Goodchild et al (2005) measured recall on issues discussed in 40 dietetic consultations; the exact level of recall was not reported in their study. The rate of recall for patients and consultants (N=30) was reported as 64.08%, however this appears to be over-inflated due to the method used to calculate recall agreement (Coughlan, 2005). When recall agreement was recalculated using the method in Parkin & Skinner (2003), an agreement rate of 50% was recorded, which is similar to the Parkin & Skinner agreement rate.

Heisler et al (2003) used prompt lists to assist recall. Patients and providers were contacted by post and asked to rank their top three goals from a list of eleven possible treatment outcomes. They were then asked to rate their top three strategies from a list of nine treatment strategies. These lists were posted to patients within three months of seeing the primary care physician, and emailed to the latter on the day that the patients attended the clinic. Overall agreement on the top three treatment outcomes was low (κ from 0.00 to 0.24) and in addition, priorities listed by professionals were found to differ from those of the patients. Overall agreement on treatment strategies was also low ($\kappa < 0.4$). Once again there were differences between patients and professionals in recalled treatments. Professionals and patients achieved a 40% agreement on one treatment goal and a 36% agreement on two treatment goals. The low

agreement rates achieved may reflect the delay in collecting data, as recall is known to deteriorate over time (Campos & Alonso-Quecuty, 2006; Pesudovs, Luscombe & Coster, 2006). It may also reflect the differing agendas of the patients and professionals in this study, indicating potential problems with the communication that occurred between patients and physicians in the consultation.

Kravitz et al (1993) also used a prompt list with 15 specific items. The items were read out to patients during a telephone interview to identify whether the physician had recommended any of the 15 specific behaviours as part of their treatment plan. Physicians were not asked directly about recommendations that they had made, but were asked after the visit whether they had counselled the patient about compliance with therapy for at least 3 minutes. Patients, whose physicians reported doing so were significantly (p < .05) more likely to recall having received 12 of the 15 recommendations. Recall was greatest for recommendations to take prescribed medications and lower for recalled advice on diet, exercise, and performance of various self-care activities.

2.3.3 Empathy and agreement on recall

Godchild et al (2005) measured empathy against recall agreement in 40 dietetic consultations, involving three dietitians, using the Empathic Communication Coding System (ECCS) (Bylund & Makoul, 2002). This provides a direct measure of empathy from direct statements of emotion, progress, or challenge made by patients during consultations. Although not significant, a trend was noted between increasing number of empathic opportunities and the level of agreement between dietitians and patients on reported issues discussed (r =

.28, p = .07). In addition, the more empathic the response by the professional to emotional opportunities offered, the more satisfied the patient (r = .41, df =15, p = .05). The satisfaction measure used focused on the patient-provider relationship (Kinnersley et al., 1996), suggesting that the communication that occurred between the patient and the dietitian was a significant factor in this study. Greater autonomy support was associated with a greater number of empathic opportunities occurring in the consultation (r = -.29, p = .07).

Coughlan (2005) measured empathy in 30 consultations involving two consultant physicians and patients with diabetes. This study uses three different tools to measure empathy: the ECCS, the Roter Interactive Analysis System (RIAS) of global affective ratings (which provides a measure of the emotional context of the dialogue by rating the overall affective impression of speakers) and Levinson's coding of empathic clues and missed opportunities. (Levinson codes patient-initiated clues, which are defined as hints, direct statements of aspects of the patient's life, expressions of feelings that offer opportunity for empathy, physician-initiated clues and open-ended questions that invite possible discussion of emotions). Empathy was found to correlate with agreement on decisions made (r = .486, p = .016) using the ECCS. The male physicians in this study were found to be significantly more empathic to female patients than male patients (t(16.16) = -2.35, p = .032). The more empathic the consultant the shorter the consultation (r = -.41, p = .026) using Levinson's coding system. The coding systems were found to correlate positively to each other, suggesting that they were all measuring a similar construct of empathy. Although communication skills were not measured directly in this study, the measurement of empathy using the ECCS codes

professional responses to the patient's empathic opportunities. As communication skills are needed to demonstrate empathy (Egan, 2002), the levels of coding for empathy achieved will be reflecting different communication skills used by the professionals in the consultation, thus highlighting the importance of good communication skills to ensure that empathy is expressed appropriately.

2.3.4 Patient autonomy

Parkin & Skinner (2003) found greater autonomy support was associated with greater autonomous motivation for self-care (r = .31; p < .001), with more controlled motivation being associated with less agreement on issues discussed and goals set (r = .2; r = .24; p < .05). Patient autonomy was correlated with satisfaction in this study. Patients were found to report more autonomy support when they created more empathic opportunities during their consultation (r = .0.29, p = .07).

2.3.5 Self-efficacy, self-care management and agreement

Heisler et al (2003) found that patients who shared their professional's top treatment strategy had higher self-efficacy. Patients who agreed on more treatment goals also rated their diabetes self-management more positively. Sharing of responsibility when making treatment decisions resulted in greater agreement on treatment strategies and goals. Patients who were more educated and had a greater belief in the efficacy of their treatment were more likely to agree on treatment goals and strategies.

2.3.6 Communication, decision-making and self-care management

Heisler et al (2002) surveyed 2,000 patients with a postal questionnaire that achieved a 66% completed response rate. The provider participatory decisionstyle questionnaire used four questions that examined "how often the professional offered patient choices in medical care; Discussed the pros and cons of each choice with patient; Got patients to state which choice or option they would prefer; Took patient preferences into account when making treatment decisions" (p 245). They found that higher ratings of provider participatory decision-making and patient provider communication were associated with higher self-management assessments p < .01. Patient understanding was strongly associated with self-management (β :0.25; p < .001). Provider participatory decision-making was found to be a significant predictor for observing a good diet (p = .003).

The study by Huang et al (2005) used semi-structured interviews with 28 patients with type 2 diabetes aged 65 and older. Patients in this study defined their goals in reference to aspects of daily routine life rather than specifics of diabetes management. Seventy one per cent of patients reported maintaining independence in daily activities as a goal. Remaining alive and healthy was the next most frequently cited goal, whereas adhering to medications, controlling weight and managing other risk factors were much lower on the list of self-reported goals. The greatest variability in patient self-care practices were in diet and exercise with 29% of patients reporting that they had not altered their diet in any way since diagnosis. This study highlights the need to review the type of language used when working with patients, ie the use of language that moves away from biomedical terms where the dietitian takes time to explore the

circumstances of patients so that goals and decisions can be framed in a way that patients can understand and relate to.

2.3.7 Recall and adherence

In this study Kravitz et al (1993) measured patient recall of physician recommendations made in the consultation, and then their adherence to these recalled recommendations. Four months after patients were asked to recall recommendations made in the consultation, they were questioned about their adherence. Adherence to diet and medication was found to be higher among patients who were able to recall these recommendations being made. Sixty nine per cent of the diabetic patients (n = 601) reported following a diabetic diet, 52% reported following a low-fat diet and 74% of hypertensive patients (n =1291) reported following a low-salt diet. Interestingly 41% of patients in this study reported following specific self-care behaviours while not recalling a specific recommendation to do so. Self-reported adherence was measured against clinical indexes: for type1 diabetes this was correlated with lower blood glucose levels (r = -.33) and lower glycohaemoglobin level (r = -.25). In type 2 diabetes higher self-reported adherence was correlated with fewer hyperglycaemic symptoms (polyuria and polydipsia), lower blood glucose levels, and lower body mass index. Self-reported adherence in hypertensive patients was correlated to lower diastolic blood pressure and lower BMI, and selfreported measures in heart patients correlated with less dyspnoea (r = -0.17). In general, self-reported adherence rates in patients with diabetes, hypertension, and heart disease were all found to correlate in those patients who were less obese (r = -.28). Self-reported adherence was found to be associated with improvements in clinical measures.

2.4 Discussion

The final outcome of the patient consultation is the recording of decisions or treatment recommendations that have been made with regard to care management. If patients cannot recall decisions made in the consultation, then they will be unable to implement them. The communication process that occurs in the consultation is therefore essential for engaging patients in order to allow effective dialogue to occur (Poskiparta et al, 2001; Rosal et al, 2001; Vickery & Hodges, 1986). The establishment of a good rapport with patients facilitates the process of understanding and information exchange (Brown et al, 1998; Gable, 2007; Roter, 2000) and would be expected to lead to decisions that are relevant to the individual, thereby increasing the chances of recall.

This review aimed to establish whether specific communication skills such as empathy, active listening, and communication behaviours have been linked to the outcome measure of recall. Although there are studies that have examined patient recall of information (Anderson et al, 1979; Flocke & Stange, 2004; Madan & Tichansky, 2005; Mauffrey et al, 2008; Michie et al, 1997; Page et al, 1981; Skinner et al, 2003b; Ward & Sanson-Fisher, 1996) these have failed to explore the exact presence or absence of communication skills in the consultation that may account for the level of agreement or disagreement obtained.

Although the study by Kravitz et al (1993) did not measure specific communication behaviours, it was included in this review as it provides compelling evidence to support the use of recall as an intermediate measure of long-term outcomes. The findings suggest that recall indicates client

engagement and commitment to decisions made in the consultation. This leads to adherence, and self-reported adherence relates to improvements in clinical measures. Although this was reported as a weak association, this may reflect the complex interaction of the multiple behaviours impacting on self-care management and their role in improving outcomes. In addition, there is evidence to suggest that self-reported data is likely to have a bias, as people tend to over-report (Adams et al, 1999; Kiviniemi & Rothman, 2006; Kristal et al, 1989; Schrimer et al, 2005). Therefore, the weaker associations may also be a result of self-reporting of adherence. Despite the challenges associated with measuring adherence, when patients recalled recommendations, their adherence particularly to diet and medications was higher, suggesting that recall may be a useful intermediate measure of outcomes. This association between adherence and recall was seen in three separate chronic conditions (diabetes, heart disease and hypertension) and this suggests that recall could be used as an intermediate measure of adherence for patients with a variety of chronic conditions. Furthermore, the communication skills used by the professionals and the level of empathic engagement of the patient in the consultation are all factors that are likely to contribute to patient recall. Recall may therefore be a useful indicator of the communication skills used in the consultation.

If recall is an important intermediate measure of outcomes, then there is a need to consider how recall data is reliably collected, quantified, and measured. The studies reviewed used different measures to assess different types of recall, eg auditory (Coughlan, 2005; Goodchild, Skinner & Parkin, 2005; Parkin & Skinner, 2003) or using visual prompts (Heisler et al, 2003; Kravitz et al, 1993). The use

of prompt lists (written material) is reported to increase recall, rather than relying on auditory memory alone (Campos & Alonso-Quecuty, 2006; Jansen et al, 2008b; Mauffrey et al, 2008). Age differences are known to impact on auditory recall (Pesudovs, Luscombe & Coster, 2006) and cognitive decline is accelerated in older patients and those people with diabetes (Jansen et al. 2008a; Stadtlander & Coyne, 1990). This population group may therefore have greater difficulty recalling decisions made, and this will impact on the level of agreement recorded (Asimakopoulou, Hampson & Morrish, 2002). This may account for the low recall levels seen in studies where patients are greater than 65 years of age (Barnard et al, 2006; Jansen et al, 2008b; Pesudovs, Luscombe & Coster, 2006; Rost & Roter, 1987). However, when younger patients (< 65 years of age) use prompt lists to aid recall, a 46% agreement rate on recall was achieved (Madan & Tichansky, 2005). This is comparable to the 47.1% obtained using auditory recall (Parkin & Skinner, 2003). This suggests that the method of collecting recall data may be an important consideration when working with older patients (Stadtlander & Coyne, 1990). The studies assessing recall in this review all had patients with a mean age of less than 65 years and it was unlikely that age would be a factor influencing the recall data reviewed.

The lower rate of recall achieved in Heisler et al (2003), when compared to Parkin & Skinner (2003) is likely to be a function of the delay in collecting recall information, as recall is known to decline over time (Campos & Alonso-Quecuty, 2006). Equally, poor communication skills may have resulted in the lack of clarity on goals recalled. Professionals may have chosen goals from the list that they felt they should have made, and patients may be choosing goals that

are broad and so may not reflect their perception of the decisions made (D' Eramo-Melkus & Demas, 1989). The prompt list used by Kravitz et al (1993) used very broad statements such as, "Follow a diabetic diet" (p 1872), but what does this actually mean? Variations in perception of what this statement means may impact on the measures obtained (D' Eramo-Melkus & Demas, 1989; Simmons et al. 2007; Street, 1992a). Patients often did not recall elements of advice provided and equally did not always adhere to advice that was recalled (Kravitz et al., 1993; Theunissen et al., 2003). In addition, lack of specific detail on recalled decisions makes it difficult for patients and professionals to assess the impact of changes made. Lack of clarity may result in patients interpreting decisions differently from the intended meaning, resulting in changes to biomedical measures that differ from those expected (Kravitz et al, 1993). Lack of clarity may be a reasonable contributing factor to the lack of adherence seen in this study. This also suggests that, to acquire meaningful data on decisions made, auditory recall may provide more meaningful data as it allows patients to express their perception of decisions made rather than using a pre-constructed list.

The issues raised over clarity of recall, as well as ability to recall decisions, suggests that there is a need to help patients understand (Heisler et al, 2002) and recall recommendations. If patients can recall and understand recommendations and decisions made this may increase the chances of their successful implementation (Kravitz et al, 1993). Having raised the importance of effective communication, clear guidance on how to achieve this improvement is lacking (Heisler et al, 2003; Kravitz et al. 1993). In addition, Heisler (2002) noted that reported understanding of self-care behaviours was likely to be a

result of professional communication that enhanced self-management through increased patient understanding. This relationship still held even when controlling for socio-demographics and health variables, such as age and health status. Patient perceptions of goal importance will vary (Kuijer & De Ridder, 2003) and may explain the variations seen in the reviewed studies (Heisler et al. 2003; Kravitz et al, 1993; Parkin & Skinner, 2003). There is a need to understand this difference in perception in order to set meaningful and relevant goals for individuals (Bradley et al, 1999; De Fine Olivarius et al, 2001; Huang et al, 2005). Those goals that support the ability of patients to live independently may be more successful than goals that focus on preventing complications alone (Huang et al, 2005), suggesting the need to focus on the agenda of the patient in the consultation. Focusing on the language used to frame goals may have an impact on this communication process (Thorne et al, 2003). Reflecting on behaviours and responses to information is seen to be important, as it requires the professional to consider the patient as a person rather than as a condition (Anderson et al, 2003; Bradley et al, 1999; Gentili et al, 2001; Goodchild et al, 2005; Huang et al, 2005; Kravitz et al, 1993; Krichbaum et al, 2003). Bogardus et al (2004) showed that patient agreement with treatment plans increased patient adherence. Therefore actively involving patients in treatment decisions through facilitated discussions (Naik et al, 2008; Poskiparta, et al, 1998) may be a useful strategy to increase agreement on recall and ultimately improve health outcomes.

Older patients with complex needs prefer professionals to make decisions for them (Huang et al, 2005; Little et al, 2001b), highlighting further the need for the professional to be able to listen and understand the patient's need for

information and involvement, as some patients may not wish to be actively involved in decision-making (Wetzels et al, 2007). This need to understand the individual's situation and preference for involvement requires the professional to be able to use a range of active listening skills, including reflection, summarising and exploratory questions, as these help to demonstrate listening and understanding (Egan, 2002; Gable, 2007). The ability to demonstrate understanding of a patient's situation, emotions and thoughts is a process referred to as empathy (Egan, 2002), in other words, the ability to identify with another person's situation and perspective by 'putting yourself in their shoes' (Hemmerdinger et al, 2007). This ability to understand the patient appears to be crucial to developing effective dialogue with the patient (Bellet & Maloney, 1991; Egan, 2002; Gable, 2007; Reynolds & Scott, 1999). However, this review identified only 2 studies that had attempted to link empathy with the outcome measure of agreement on recall. One study identified a trend between empathic opportunities and agreement (Goodchild et al, 2005), whereas the other was able to correlate empathy with agreement (Coughlan, 2005). However, as previously mentioned, both of these studies were small and exploratory in nature and further larger studies across a range of sites is needed to substantiate the importance of the link between empathy and recall (Coughlan, 2005; Goodchild et al, 2005). In other studies, it has been suggested that patient demographics may impact on measures of empathy and recall (Golin et al, 1996; Heisler et al, 2002; Kravitz et al, 1993; McDonald-Miszczak et al, 2005; Rost et al, 1990) and should therefore be included in any researching in this area.

Involving patients in shared decision-making (Heisler et al. 2002) and providing them with choices through increasing autonomy (Parkin & Skinner, 2003) may be another method for improving self-care and this has been linked to recall agreement (Parkin & Skinner, 2003). Patient autonomy requires the use of good communication skills that enable the professional to acknowledge the perspectives of patients, support the initiatives of patients, offer choice about treatment options, provide relevant information, minimize pressure and control (Williams et al, 1998). The Coughlan (2005) study did not find a significant correlation between empathy and autonomy support. This study suggests the lack of correlation may be contributed to the presence of a 'consultant effect'. whereby patients scored consultants highly as a result of their status, expertise and their perceived power and authority over treatment (Coughlan, 2005, p.43). In contrast, patients in dietetic consultations created more empathic opportunities when they reported more autonomy support from the dietitian (Goodchild et al, 2005). This appears to suggest that there may be differences in the way that dietitians and consultants communicate with patients. In addition Parkin & Skinner (2003) found that when nurses and dietitians were less controlling, patients reported greater levels of autonomy support. This was linked to more agreement, particularly for older patients. Equally greater control in the consultation over decisions made may reflect higher levels of patient selfefficacy (Heisler et al, 2003). Improving self-efficacy may improve self-care management ability (Krichbaum et al, 2003). This issue of self-efficacy is particularly important in dietetics, as it has been shown to have strong associations with dietary adherence (Senécal et al, 2000; Williams et al, 1998).

2.5 Conclusion

The limited data obtained from the review suggests that further work is needed to establish the links between communication skills in the consultation and the outcome measure of agreement on recall of decisions made.

The empirical studies reviewed were cross-sectional and therefore hard conclusions cannot be drawn about directional effects of behaviours in relation to adherence and outcomes. It has been suggested that patient adherence may be mediated by recall (Bartlett et al, 1984; McCann & Weinman, 1996) and the evidence from Kravitz et al (1993) appears to support this finding. Memory of advice is also important (McCann & Weinman, 1996) and agreeing with recommendations through improved collaboration (Heisler et al, 2003; Stewart, 1995) can promote adherence and goal attainment (Bogardus et al.., 2004). Greater agreement on recall of decisions made in the consultation may therefore be a marker for success of the consultation, leading to improved patient outcomes (Heisler et al, 2003).

Agreement on recall may therefore be a useful intermediate measure of longterm outcomes in chronic long-term conditions. However, the literature reviewed highlights significant gaps in understanding of the specific communication skills and process needed to achieve agreement on recall (Bradley et al, 1999; Watson & McKinstry, 2009). The Watson & McKinstry review of interventions to improve recall of medical advice provides general guidance on behaviours, such as the importance of repetition of decisions and the positioning of important information within the consultation. However, this

still fails to clarify the specific communication skill set needed in the consultation to improve recall.

Although some skills such as empathy have been measured, and tentative links with agreement on recall made, further larger studies are required to substantiate the relationship of empathy with agreement on recall. In addition, the demonstration of empathy requires the use of active listening skills, which may in turn influence a patient's perception of autonomy support and selfefficacy leading to agreement on recall. The skills of empathy and active listening therefore require additional exploration to clarify their role and importance in achieving agreement on recalled decisions made in the consultation.

The questions that this review raises are as follows:

Does greater empathy result in higher levels of patient/professional agreement on reported decisions made in the consultation as suggested?

What are the specific communication skills needed to support empathy in the consultation?

If decisions are made in the consultation, then at some stage during the consultation, information exchange would have occurred. So how does information exchange relate to empathy and communication skills in the consultation and what is their impact on agreement on reported decisions made?

CHAPTER THREE: PATIENT-CENTRED COMMUNICATION

The literature reviewed in Chapter 2 has highlighted a significant gap in our understanding of the link between specific communication skills/behaviours in the consultation with the outcome measure of agreement on recall. However, tentative links between empathy and agreement on reported decisions were made. This chapter provides further supporting evidence for the need to explore the following:-

- The presence of empathy in the consultation and its importance in patient-centred communication.
- The literature supporting the use of patient-centred communication for effective self-care management in long-term conditions.
- The communication skills that may be needed to support empathy and information-giving in the consultation
- The potential impact of these communication skills on patient/professional agreement on decisions made in the consultation.

3.0 Patient-centred care

The approach required to manage chronic long-term conditions, such as diabetes, differs from that used for medical acute care management (Anderson, 1995; Anderson & Funnell, 2005b; Department of Health, 2001; Glasgow et al, 2001). In the past, professionals would have seen themselves as being in charge of, and responsible for, the treatment of patients (Anderson, 1995). The professional would have made decisions on how the condition was to be managed (Byrne & Long, 1976; Fieldhouse, 1979) without regard to the impact of the condition or these changes on the individual (Cohen et al, 1994; Weston et al, 1989). Professionals would have seen themselves as 'experts' with

patients as 'passive' recipients of information provided (Glasgow et al. 2001). Patients who failed to conform would have been labelled as non-compliant (Glasgow et al, 2001) and strategies aimed at increasing their level of compliance would have been implemented (Anderson & Funnell, 2000). These would have contributed to increases in patient distress levels (Hiscock et al. 2001) and thus impacted on outcomes. Metabolic control and patient compliance/adherence would have been the primary measures of the educational approaches used (Brown, 1990; Glasgow & Osteen, 1992; Padgett et al, 1988; Wolpert & Anderson, 2001). However, there are problems with using this approach for a chronic disease (Glasgow & Anderson, 1999). Acute conditions respond well to expert-led approaches, as changes that are made during an acute period of illness tend to be short term, as the illness abates and patients revert back to their usual habits. However, with chronic diseases the changes needed to manage the condition are not short term, they require longterm commitment and involve major changes in the way that the patients conduct their day-to-day lives (Anderson & Funnell, 2005b; Glasgow & Anderson, 1999; Moser et al, 2008; Wolpert & Anderson, 2001). They also involve changes to the way that individuals perceive themselves and their sense of health and well-being (Hiscock et al, 2001). This lack of consideration of the needs of the individual and the complexity of the changes required may in part account for the poor levels of compliance documented in the literature (Ary et al, 1986; Donnan et al, 2002; Glasgow & Toobert, 1988; Glasgow et al, 2001; McNabb, 1997; Thorne & Paterson, 2001; Toobert et al, 2000) and highlighted in Chapter One.

Understanding this basic difference between acute medical care and chronic care is essential and it means that chronic disease management carries a different emphasis to acute disease management (Anderson, 1995). Focusing on the behaviours in the consultation that help engage people actively in their day-to-day care may be more productive than focusing on information-giving alone. In particular, when these behaviours are underpinned by theory (Abraham & Michie, 2008; Cullen et al, 2001; Ley, 1982), this may lead to improved diabetes outcomes (Murphy & Kinmouth, 1995). Improvement requires a subtle change in the emphasis of the consultation: focusing on the individual and their health problems rather than solely on the management of their disease (Anderson & Funnell, 2005b; Greenfield & Kaplan, 1988; Kinnersley et al, 1999; Lewin et al, 2001; Peel et al, 2005).

The patient therefore becomes a key player in the successful implementation of the care package and as a result their needs are considered alongside the medical management of the condition (Anderson & Funnell, 2005b; Anderson et al, 1982; Rubin et al, 2002; Wilson, 2002). This need to understand and work more closely with patients to enable effective self-care management is reflected in the literature. Patient education moves from professional-led medically driven concepts of care to a chronic disease model that focuses on patient-centred care (Anderson, 1995; Brown, 1990; Deakin et al, 2006; Norris, Engelgau & Narayan, 2001; Padgett et al, 1988; Skinner et al, 2003a; Stewart et al, 2000). A key example of a patient-centred collaborative approach is the model of patient empowerment that has been used in diabetes education (Anderson, 1995; Anderson et al, 1995; Anderson et al, 1991; Anderson et al, 1982; Funnell et al, 2005). This approach encourages professionals to focus on

the patient as an expert within the consultation (Anderson, 1995; Lorig et al,

2001; Peel et al, 2005; Smith et al, 1999) thus challenging the professionals'

view of themselves as expert providers (Funnell & Anderson, 2004). The

shifting emphasis means that the role of the professional is changing from that

of an expert provider to that of an expert teacher facilitator and guide

(Anderson, 1995; Smith et al, 1999). This approach recognises that

"Knowing about an illness is not the same as knowing about a person's life" and that, by default patients are the primary decision makers in control of the daily self-management of their diabetes" (Funnell & Anderson, 2004, p 124).

Patient-centred care brings with it therefore some challenges in terms of how

best to balance evidence based medicine and patient-centred medicine

(Anderson, 1995; Anderson & Funnell, 2005b; Bensing, 2000; Smith et al,

1999). Bensing (2000) uses the following model to explore the dimensions of

patient-centredness (Figure 3.4).



Figure 3.4: Dimensions of patient-centredness (Bensing, 2000)¹

This suggests that patient-centredness involves listening to the patient in order to understand their needs and values (de Haes, 2006) and where they wish to

¹ Reprinted from Bensing, J. (2000) 'Bridging the gap. The separate worlds of evidence-based medicine and patient-centred medicine', *Patient Education & Counseling, 39 pp.* 17-25 with permission from Elsevier)

be on the axes shown in Figure 3.4 (Bensing, 2000). The professional's use of communication skills is therefore important. The listening and questioning skills employed will help in identifying the level of control that the patient wishes to have in the consultation, and the information that the patient requires from the professional to address their needs (Bensing, 2000; Cant & Aroni, 2008; Fallowfield & Jenkins, 1999; Hambly et al, 2009; Lang et al, 2000; Levinson & Roter, 1995).

The consultation therefore becomes a learner-centred activity with the professional facilitating the process of patient learning (Anderson et al, 1991; Coles, 1990; Lorenz et al, 1996; Maldonato et al, 1995). The growing awareness of the need to facilitate patient learning in order to enable effective self-care management is underpinned in various national guidance documents on patient education (NICE, 2003) and care for people with diabetes (Diabetes UK, 2009). These include the National Service Framework (NSF) for diabetes and the National Institute of Clinical Excellence (NICE) guidance documents on patient care and group education (Department of Health, 2001; Department of Health, 2003; Department of Health, 2007; Department of Health and Diabetes UK, 2005; NICE, 2003; NICE, 2008a). Changes within the health service also mirror this shift in culture, in particular the 'Creating a patient-led NHS: Delivering the NHS improvement plan' and the new white paper 'Equity and excellence: Liberating the NHS'. Both continue this theme and place great emphasis on patient-led services that increase choice, personalised care and empowerment of people to improve their health (Department of Health, 2005; Department of Health, 2010b; Department of Health, 2011).

3.1 How equipped are dietitians to deliver patient-centred care?

The dietetic profession recognises the importance of communication to engage effectively with patients and this is reflected in the pre-registration education and training curriculum. Here, psychology, sociology, and communication training form an integral part of the curriculum for student dietitians, helping to develop the skills needed for working with others effectively (British Dietetic Association, 2008b). The concept of patient-centred care is reflected in the dietitians' professional code of conduct, which states that dietitians should:

"respect the decisions of service users concerning their own health and independence, even when such decisions conflict with professional opinion... give sufficient information and time to enable them to make informed decisions about their health and social care...present information in a way that service users can understand" (British Dietetic Association, 2008a, p.8).

However, these are broad concepts and specific guidance on how to achieve these aims is not provided, leaving their application open to individual interpretation. This issue of interpretation has been raised by others who found that whilst Canadian dietitians were familiar with the concept of 'client-centred' nutritional counselling, they required further clarification to remove ambiguity from their interpretation of this term (MacLellan & Berenbaum, 2003). Whilst most dietitians would agree that patient-centred care is essential for their practice (Brown et al, 1998; Whitehead et al, 2009), the inability to describe this process hinders critical evaluation of the effectiveness of the dietetic consultation.

Professional issues concerning time in the consultation (Whitehead et al., 2009) and resources, as well as differing goals of patients and professionals (Heisler et al, 2003; Julien et al, 2009; Kravitz et al., 1993; Kuijer & De Ridder, 2003;

Levy, 2009: Parkin & Skinner, 2003: Vijan et al, 2005) have been cited as additional challenges to implementing a patient-centred approach and may contribute to the perceived difficulties of implementation amongst dietitians (Maclellan, 2006). This may account for the difficulties reported by Rosal (2001) in identifying evidence of integration of skills needed to support behaviour change in American dietitians. Canadian dietitians also report the need for further training in behaviour modification skills (Barr et al, 2004). In the UK, a report on dietetic consulting skills highlighted that dietitians generally recognise the need for additional skills to support behavioural approaches for delivering patient-centred care, but in practice feel that their skill set to support this approach may not be adequate (Rapoport, 1998; Rapoport & Perry, 2000; Stetson et al. 1992). Although there is a move to consider psychological and counselling skills in much greater depth in dietetic practice (Hunt, 1995; Isselmann et al, 1993; Pearson & Rapoport, 2007) particularly for consideration in student training (Curry & Jaffe, 1998; Davidson et al, 1987; Horacek et al, 2007; Huyck, 1986; Lawrence, 2003; Long et al, 1983; Pearson & Rapoport, 2007), clear definitions and research in this area are lacking.

Dietitians are not alone in struggling to conceptualise and therefore utilise the concept of patient-centred care for self-care management. Doctors and nurses have also been reported to struggle with the concept (Brown, 1990; Golin et al, 1996; Griffin et al, 1999; Norris et al, 2001; Norris et al, 2002; Padgett et al, 1988; Stewart, 1995). Additionally, the numerous definitions of 'patient-centred' care create problems when reviewing research in this area, resulting in limited and mixed evidence on the effects of interventions on patient behaviours and health status (Lewin et al, 2001). Patient-centredness has been

described as a set of professional attitudes, a set of knowledge (Mead & Bower, 2000a) and consultation behaviours (Stewart, 1984; Stewart, 1995). Add to this the variety of measures used to assess patient-centredness and this may explain why the evidence on the effectiveness of patient-centred care is inconsistent (Lewin et al, 2001; Mead & Bower, 2000a; Whitehead et al, 2009). There is some evidence to suggest that patient-centred approaches are more effective in enhancing patient communication and thereby improving subsequent patient health (Anderson et al, 1995; Golin et al, 1996; Griffin et al, 1999; Stewart, 1995). For example, interventions designed to increase patient participation and autonomy (the degree to which a person feels that they are in control) have resulted in improvements in self-care behaviour and glycosylated haemoglobin (HbA1c) (Anderson et al., 1991; Davies et al, 2008; Greenfield & Kaplan, 1988; Harrington et al, 2004). However, work is still required to strengthen the associations seen between patient-centred care and effective communication as demonstrated through improved health outcomes (Bensing, 2000; de Haes & Bensing, 2009; Epstein et al, 2005).

3.2 Patient-centred Communication

Further exploration and definition of the concept of patient-centredness is required to provide clear direction and understanding of process and outcomes (Epstein et al, 2005; Lewin et al, 2001; Mead & Bower, 2000b). Epstein et al (2005) suggest the need to distinguish between patient-centredness, patient centred care, and patient-centred communication, as the overlapping use of these terms in the literature distorts findings of what exactly is being examined. The authors clarify patient-centredness as a moral philosophy with three clear values: Considering patients needs, wants, perspectives and individual

experiences; Offering patients opportunities to provide input into and participate in their care; Enhancing partnership and understanding in the patient-physician relationship. These three broad values encompass the five dimensions of patient-centredness as described in Mead & Bower's (2000) review: Understanding a patient's illness in general within a broader bio-psychosocial framework; understating an individual's experience of an illness; sharing power and responsibility; the therapeutic alliance; the influence of the personal qualities of the doctor.

Epstein et al (2005) go on to state that "patient-centred communication focuses on the relationship between the patient the health care professional and family members that promote patient-centredness" (p 1517). The operational definition of patient-centred communication, which this paper provides, focuses on four key aspects as outlined in Table 3.4.

- **1.** Eliciting and understanding the patient's perspective-concerns, ideas, expectations, needs, feelings and functioning.
- **2.** Understanding the patient within their unique psychosocial context
- **3.** Reaching a shared understanding of the problem and its treatment with the patients that is concordant with the patient's values
- **4.** Helping patients to share power and responsibility by involving them in choices to the degree that they wish

Table 3.4: Operational definition of patient-centred communication adapted from (Epstein et al, 2005)²

These definitions provided further direction and clarity when studying patient-

centred communication and outcomes (Epstein et al, 2005). The key essence

² Reprinted from Epstein et al. (2005) 'Measuring patient-centred communication in patientphysician consultations: Theoretical and practical issues' *Social science & Medicine*, 61 pp.1516-1528 with permission from Elsevier

of patient-centred communication is said to derive from Carl Rogers work in the late 1950s and his development of the concept of "client-centred counselling". His guiding principles, when working with patients, were respect, genuineness, and empathy (Rogers, 1957). These principles are still considered to be essential for effective patient-centred communication and provide the scaffold for effective relationships between patients and professionals (Anderson, et al, 1982). Epstein et al (2005) define a similar set of principles for patient-centred communication as being attentiveness, critical curiosity, informed flexibility, and presence. Understanding how empathy impacts on the consultation process and its subsequent impact on intermediate outcomes is essential (Street et al, 2009).

3.3 Empathy

Empathy is a central construct in patient-centred communication (Reynolds & Scott, 1999). Carl Rogers first defined the concept in the 1950s (Rogers, 1957) and later he defines empathy as a "*process*" and goes on to say that..."*being empathic is a complex, demanding and strong yet also subtle and gentle way of being*" (Rogers, 1980, p143). Others have defined empathy as the ability to identify with another person's situation, by being able to "*put yourself in their shoes*" and understand their perspective of the situation (Hemmerdinger et al, 2007). Empathy requires the professional to be non-judgemental in their ability to understand another's feelings and in their ability to communicate that understanding back to the individual (Wiseman, 1996). Empathy is a separate construct from sympathy: empathy is concerned with sharing understanding whereas sympathy is about sharing feelings (Egan, 2002; Hemmerdinger et alk, 2007; Norfolk et al, 2007). Empathy is concerned with the ability to

demonstrate an understanding of another's situation and communicate this back to the individual where helpful (Egan, 2002; Norfolk et al, 2007). Egan (2002) and Rogers (1980) describe empathy as an active process of listening, which involves the professional engaging with patient's feelings and perceptions of the world that they are experiencing, and then reflecting their words and experience back to provide meaning. The experience of being listened to by someone who demonstrates understanding is itself a process that helps individuals to reflect (Ong et al, 2000; Rogers, 1980) and is said to be an intrinsically therapeutic process (Egan, 2002). For a chronic disease such as diabetes, this acknowledgement and understanding may help to reduce the burden of living with the condition (Jansen et al, 2010). Empathy is therefore about understanding an individual within and through the context of their lives (Rogers, 1980). This allows the discrepancies in a patient's viewpoint and reality to be explored more easily (Egan, 2002).

Empathy may occur at a non-verbal level, but there are times when it must be verbalised to demonstrate that understanding and awareness of what someone is going through (Egan, 2002). This process is referred to as behavioural empathy (Hemmerdinger et al, 2007), and requires the use of the patient's own words to reflect back what has been heard (Gable, 2007; Lussier & Richard, 2007; Rogers, 1980). It is important that communication of this understanding back to the patient is conducted in a language that reflects both content and feeling accurately (Egan, 2002; Reynolds & Scott, 1999) and shows the same level of intensity (Egan, 2002; Gable, 2007).

Being non-judgemental is important when reflecting back a patient's thoughts, feelings, and ideas. These non-interpretative reflections demonstrate that the professional is listening and trying to understand, for example: "*that must have been hard*"... "It sounds' like it was difficult..." (Egan, 2002; Gable, 2007). Basic empathic listening has been described as a crucial skill for effective listening in the consultation (Lussier & Richard, 2007; Norfolk et al, 2007) and as such, empathy should be measured to assess its presence in (and therefore impact on) the consultation and outcomes. Rogers (1980) describes early work by therapists such as Truax (1967) and their attempts to measure empathy, such as developing a coding system for psychotherapists looking at empathy which stated the lowest level as:

"bored, disinterested, actively offering advice with no communication of awareness of patients current feelings"

Whereas the highest level of empathy coding stated that:

"voice tone reflects seriousness and depth of empathic grasp accurately interprets patient's present acknowledged feelings voices meanings in patient's experience, sensitive and accurate" (cited in Rogers, 1980, p144).

The coding system contained eight levels to convey empathy, demonstrating the complexity of attempting to study and measure this concept. This suggests that a range of skills may be needed to effectively demonstrate empathy when communicating. This adds further weight to Egan's statement that empathy is an *"interpersonal communication skill"* (Egan, 2002, p 48).

Other researchers have highlighted the importance of empathy in establishing a therapeutic dialogue and its relationship to effective communication skills (Neumann et al, 2009; Norfolk et al, 2007). The latter have developed a conceptual framework to explain how empathy underpins the consultation dialogue and suggest that empathic listening skills enable negotiation and

establishment of treatment plans, and adherence to these, alongside better patient satisfaction (Nofolk et al, 2007). The model they propose suggests that empathy may play a role in the way that information is delivered and the subsequent flow of the consultation. Particular importance can be attached to the patient/provider rapport. The quality of the patient/provider relationship may affect outcomes (Mead & Bower, 2000a). Adherence to diet has also been reported to relate to perceptions of the patient/provider relationship (Epstein et al, 2005; Maddigan et al, 2005) and may be crucial to the successful working relationship between patients and professionals (Barlow et al, 2002; Ellis et al, 2004; Norfolk et al, 2007).

A review of the literature suggests that developing empathy and identifying and handling emotional problems may be useful skills for the professional in the delivery of effective patient-centred communication (Lewin et al, 2001; Mercer et al, 2008).

3.3.1 Measurement of empathy

Although empathy is an accepted construct in dietetic education and training (Isselmann et al,1993; Pearson & Rapoport, 2007) very little research has been conducted to explore either the presence of empathy in dietetic consultations (Goodchild et al, 2005) or how best to educate student dietitians to improve their empathic skills in practice (Spraggins et al, 1990). In contrast, medical schools have invested a lot of energy into trying to measure and assess empathy in medical student training (Bellet & Maloney, 1991; Hemmerdinger et al, 2007; Spiro, 1992). Nurses have also started to address issues around definitions of empathy to assist in identification and measurement that will

improve patient outcomes (Brunero et al, 2010; Hope-Stone & Mills, 2001; Yu & Kirk, 2008).

Difficulties in measuring empathy have been highlighted, particularly the most important dimension of empathy and exploring the best methods for validating this (Stepien & Baernstein, 2006). The medical literature provides a number of dimensions of empathy that include emotive, moral, cognitive, and behavioural dimensions, as summarised in Table 3.5 (Stepien & Baernstein, 2006).

Emotive	being the ability to imagine patient's emotions
Moral	being the physician's internal motivation to emphasise
Cognitive	being the intellectual ability to identify and understand patient's emotions and perspectives
Behavioural	being the ability to convey understanding of those emotions and perspectives back to the patient

Table 3.5: Dimensions of empathy highlighted by Stepien & Baerstein (2006)³

In the nursing literature, empathy has been summarised by Wiseman (1996) as seeing the world as others see it, being non-judgemental, being able to understand another's feelings and being able to communicate that understanding. In addition, Kunyk & Olson (2000) have described empathy as a professional state, a communication process, as caring and as a special relationship. The parameters of empathy within these definitions have been found to vary, with some researchers describing empathy as a communication process whereas others have defined empathy as preceding a broader

³ Reprinted from Stepien, K. A. & Baernstein, A. (2006), 'Educating for Empathy', *J Gen Inter Med*, 21 pp. 524-530 with permission from Elsevier

communication model (Kunyk & Olson, 2000). Reynolds & Scott (1999), in their exploration of empathy in nursing, suggest that the definition of empathy should reflect what nurses should be doing during their interactions with patients. They state that the definition of empathy for nursing should be an

"accurate perception of the client's world and an ability to communicate this understanding to the client, in order that the client is aware of the nurse's perception of the client's position" (p 367).

This behavioural approach to the definition of empathy resonates with the small number of studies in the dietetic literature, with previous surveys of dietitians highlighting this behavioural dimension as one that they recognise as important (Cant & Aroni, 2008; Maclellan, 2006). The varying definitions of empathy have also led to a number of different measures to assess this concept (Hemmerdinger et al. 2007; Pedersen, 2009; Yu & Kirk, 2008). Typically these have been conducted in one of three ways: either through self-assessment guestionnaires which focus on the individual's assessment of their level of empathy; through the use of questionnaires that explore individual's perception of empathy from care givers that they have encountered and through the use of standardised assessment tools that are used to measure the observed empathy demonstrated in clinical encounters. These standardised assessment tools may include indirect measures of empathy (Levinson et al, 2000; Suchman et al, 1997) or direct measures of empathy (Bylund & Makoul, 2002; Hemmerdinger et al, 2007). These measures may be applied directly whilst observing consultations or when viewing video recordings of consultations. However, the range of measures used prevents direct comparisons between studies conducted so far (Butow et al, 2002; Del Piccolo et al, 2008; Hulsman, 2009; Jansen et al, 2010; Levinson et al, 2000; Pedersen, 2009; Zimmermann et al, 2007).

Hojat et al (2002) used a questionnaire to measure professional perception of empathy across a range of psychiatrists and physicians. They showed significant differences in professional perception of empathy, with psychiatrists scoring higher than physicians. They also suggested possible gender effects on empathy, with females being more empathic than males. Direct measures of empathy confirm these findings and demonstrate significant differences in empathy between male and female physicians (Bertakis, 2009; Bylund & Makoul, 2002; Roter et al, 2002; Scmid Mast et al, 2007). This suggests that the professional's gender should be considered when reviewing data on empathy (Bylund & Makoul, 2002; Nicolai & Demmel, 2007).

During the consultation, the professional needs to identify significant clues to the patient's thoughts and feelings (verbal and/or non-verbal) and then make sense of these clues (Hampson et al, 1995; Zimmermann et al, 2007) in order to demonstrate empathy. As patients seldom verbalise emotions directly (Butow et al, 2002) but through indirect cues (Suchman et al, 1997), the challenge for the professional is to identify moments when an empathic response is required. It could be argued that the subtleties of empathy are difficult to assess for a number of reasons, including those of definition, but also the reliance on good communication skills that are needed to respond to both verbal and non-verbal clues from patients (Cant & Aroni, 2008).

3.3.2 Importance of Empathy

Patients are reported as wanting practitioners who are empathic as well as technically proficient (Spiro, 1992). The perception of physician empathy has also been noted to reduce time and expense in medical consultations (Bellet & Maloney, 1991; Levinson et al, 2000). Patient perception of physician empathy

has been shown to significantly influence patient satisfaction and compliance (Kim, Kaplowitz & Johnston, 2004). The engagement of patients through empathic listening may encourage them to contribute more to setting and developing their own goals resulting in a clear understanding on the course of action that the patient is planning to take (Ammerman et al, 2002; Bensing, 2000; Michie et al, 2003). Responding empathically to patient cues may help to reduce misperceptions of decisions taken and goals set in the consultation impacting on outcomes (Butow et al, 2002; Levinson et al, 2000; Salmon et al, 2004). Effective empathic communication may therefore result in selfmanagement education that better meets the patient's needs, and as such lead to greater recall of information and decisions made (Norfolk et al, 2007). Perception of empathy in medical consultations has been shown to impact on consultation outcomes (Bertakis, 2009; Comstock et al, 1982; Larson & Yao, 2005; Levinson, et al, 2000; Ong et al, 2000). Whether empathy impacts on dietetic outcomes has yet to be fully established. Empathy has been linked to agreement on recall in one small study (Coughlan, 2005) and linked to patient enablement in another (Mercer et al, 2008). Patient enablement is a measure of the patient's perception of their ability to cope with life and their illness, to understand their illness, to keep healthy and to be confident in their ability to help themselves. Patient enablement may be an early indicator of potential health improvement (Mercer et al, 2008).

Empathy has also been linked to compliance to recommendations and improved patient satisfaction in medical consultations (Epstein et al, 2007; Kim et al, 2004; Stewart, 1984; Zachariae et al., 2003). Greater expressed empathy may result in patients feeling understood (Hall et al, 2002; Jackson et

al, 2001; Street et al, 2005) thereby increasing levels of patient satisfaction (Golin et al, 1996; Hall et al., 2002; Kim et al, 2004; Mora et al, 2008; Neumann et al, 2007). Having the opportunity to express difficulties and frustrations with care is a therapeutic exercise, particularly if these challenges are dealt with in a non- judgemental manner (Anderson & Zimmerman, 1993; Arborelius & Österberg, 1995). The greater satisfaction reported could be a result of the acknowledgement and exploration of issues raised, confirmation of the difficulties that patients were experiencing, and an illustration of the impact of good empathic listening skills on consultation outcomes (Bertakis, 2009). Studies with physicians suggest that being more empathic and responding appropriately to patient cues can increase patient satisfaction (Levinson & Roter, 1993). A large study involving 1520 patients was able to detect a small effect (r = 0.28) of patient perceived empathy with patient satisfaction (Fuertes et al, 2008). In addition, direct measures of empathy have been linked to satisfaction in a small number of dietetic consultations (Goodchild et al, 2005).

Bylund & Makoul (2002) have raised concerns that the questionnaires used to measure patient perception of empathy may in fact be measuring a much broader concept of empathy than those used to assess empathy directly through observation. Patient perception questionnaires assessing empathy may be influenced by additional factors such as the characteristics of the patients completing the questionnaires (Pedersen, 2009). These factors contribute to the potential difficulties encountered when trying to compare data from studies that utilise different measures (Pedersen, 2009). Careful consideration therefore needs to be taken when selecting methods for measuring empathy.

Hope-Stone & Mills (2001) cite lack of time, the environment, and communication difficulties as barriers to implementation of empathy in consultations. Evidence regarding these factors is conflicting. Some studies suggest that greater use of empathy will lengthen the consultation time (Marvel et al, 1999) others suggest that greater use of empathy will shorten the consultation time (Coughlan et al, 2006; Levinson et al, 2000), whereas one study suggests no effect on consultation length (Goodchild et al, 2005).

Lack of time may not therefore be the sole reason for the lack of empathy, it may relate to other communication difficulties. If professionals cannot articulate their empathic understanding to patients then the patient has no reason to believe they want to listen or understand (Egan, 2002). Good communication skills are therefore needed to demonstrate empathy (Egan, 2002; Ong et al, 2000).

3.3.3 Empathy and emotions

Discussing patient's fears and the emotional impact of their condition on their daily habits can help them to cope with their emotions (Ary et al, 1986; Weston et al, 1989). Facilitating this discussion through effective listening and use of empathy is therefore an essential part of the patient-centred communication process (Epstein et al, 2007). However, studies show that doctors and nurses are often unaware of patients needs and concerns (Barry et al, 2000; Street et al, 1993) with professionals missing empathic opportunities in up to 21% of primary care physician consultations and up to 38% of the time in surgical cases (Epstein et al, 2007; Levinson et al, 2000). In another smaller study, physicians missed 70% of the empathic opportunities presented in 16

consultations (Easter & Beach, 2004). Missing empathic opportunities has been shown to result in issues being raised a second or even third time during the consultation and may contribute to the longer consultations (Levinson et al, 2000). Emotional cues are reported as being the most missed (Butow et al, 2002; Zimmermann et al, 2007) and as occurring in the lowest frequency in medical consultations. In one study, surgeons and oncologists were found to respond empathically to only 10% of emotional opportunities raised, preferring to shift the conversation from an emotional to a biomedical context (Morse et al, 2008). This may reflect the difficulty that professionals have in responding to emotional issues (Eide et al, 2004; Epstein et al, 2007; Guimond et al, 2003; Salmon et al, 2004).

Emotional cues are said to elicit more distress in physicians than informational cues, and are often cited as a reason why these cues may be ignored more frequently (Barry et al, 2000; Butow et al, 2002; Kim et al, 2004; Zimmermann et al, 2007). Taped consultations for oncologists and surgeons revealed a lack of expression of empathy 37.5% of the time (Fallowfield et al, 2002). It is not clear whether this lack of expression of empathy holds true for dietitians. The one study that has looked at empathy found that dietitians missed only 4 empathic opportunities in 40 consultations (Goodchild et al, 2005). However, the small sample size in this study makes it difficult to draw strong conclusions about the representativeness of these findings for the dietetic profession as a whole. Morse et al (2008) also showed that surgeons tended to respond to factual terms rather than the abstract or negative concerns that patients may raise. This suggests problems with identifying empathic opportunities when emotions are not directly expressed (Suchman et al, 1997). Poor communication skills of

the professionals involved, and the lack of ability to judge patient emotions (Street et al, 2009) may therefore account for the high levels of missed cues in medical consultations (Butow et al, 2002; Easter & Beach, 2004; Street et al, 2009).

Whilst professionals may wish to shield themselves from patient emotions, these emotions are potentially a valuable therapeutic resource (Jarret & Payne, 1995; Williams et al, 1998). Developing skills to enable effective use of empathy in consultations would benefit practitioners and their patients (Eide et al, 2004; Ford et al, 2000; Hope-Stone & Mills, 2001). The failure to respond appropriately to patient emotions and pick up on cues may impact on the patient's ability to listen and respond appropriately to information given in the consultation (Hulsman, 2009; Neumann et al, 2007).

The low frequency of statements of emotion in consultations may be explained by the fact that emotions are hard for patients to express, and they may be embarrassed and/or frightened to fully verbalise their fears (Del Piccolo et al, 2008; Jarret & Payne, 1995; Little et al, 2001a). Qualitative interviews with patients following GP consultations highlight frequent concerns that are seldom raised directly or spontaneously in the consultation (Barry et al, 2000). Professionals therefore need to prompt patients by asking about their feelings or concerns in order for these to be expressed (Barry et al, 2000; Butow et al, 2002; Jansen et al, 2010; Zimmermann et al, 2007). Once emotional issues have been raised, it is important for professionals to listen attentively to and validate any feelings expressed through appropriate empathic responses (Barry et al, 2000; Butow et al, 2002; Epstein et al, 2007; Suchman et al, 1997; Zachariae et al, 2003). Lack of listening and the resultant lack of demonstration
of empathy could create anxieties whereby patients feel that they are not being listened to and responded to adequately (Eide et al, 2004; Epstein et al, 2007; Poskiparta et al, 2001). Stress is known to narrow the attention span (Zachariae et al, 2003), stressed patients may therefore experience a reduced ability to process information and to pick up on cues and respond appropriately to information provided (Hulsman, 2009; Neumann et al, 2007). This factor may result in fewer decisions being recalled by the patient (Jansen et al., 2010). Furthermore Morse et al (2008) demonstrated that in surgical consultations empathic opportunities occurred at a higher frequency towards the end of the consultation. If patients are voicing unmet agendas towards the end of their session, then this leaves little time to address appropriately any issues raised and is likely to impact negatively on patient recall.

The use of empathy in the consultation allows the professional to demonstrate understanding of the uniqueness of the patient's situation, but requires good communication skills to support this process (Norfolk et al, 2007). Neumann et al (2009) have proposed a therapeutic model with a hypothetical link between empathic communication and suggested links to outcomes. This model suggests that the social-emotional aspects of communication may directly influence the informative, participative, and educative part of the dialogue. This provides further evidence for the need to focus on patient/professional communication during the consultation in an attempt to understand how empathy impacts on these communication processes and ultimately on consultation outcomes.

When exploring the role of empathy in the consultation it is important to remember that it may or may not have an emotional component. Empathy also encompasses an understanding of the patient's perceptions and opinions (Bylund & Makoul, 2002; Mead & Bower, 2000b; Norfolk et al, 2007). Clarification of the empathic components that relate to recall is also required, as this may provide guidance for future developments in improving the communication skills of professionals (Kim et al, 2004).

3.4 Active-listening Skills

Employing active-listening skills can help professionals to develop an understanding and appreciation of the individual's viewpoint (Simpson et al, 1991) and this is acknowledged in the Toronto Consensus Statement (1991), which has been endorsed by an international panel of experts on patient– physician communication. The skills required for active listening are the same core skills used in counselling (Gable, 2007; Shillitoe, 1994). They consist of reflection, summarising, paraphrasing and appropriate exploratory questioning and are essential for developing an effective dialogue and for demonstrating empathy (Peel et al, 2005; Poskiparta et al, 2001; Rosal et al, 2001; Vickery & Hodges, 1986; Zimmermann et al, 2007).

Good use of these active-listening skills has been found to promote patient participation through encouraging patients to take part and express opinions and their concerns (Barry et al, 2000; Beck et al, 2002; Beisecker & Beisecker, 1990; Eide et al, 2004; Eldh, 2006; Epstein et al, 2007; Kaplan et al, 1989; Little et al, 2001a; Maguire et al,1996; Michie, Miles & Weinman, 2003; Peyrot & Rubin, 1994; Poskiparta et al, 2001; Roter et al, 2008; Stetson et al, 1992;

Street, 1991; Street et al, 1993; Street et al, 2005; Weston et al, 1989; Williams et al, 1998; Zandbelt et al, 2007). Understanding patient's perceptions of their condition and how it can be managed (Bertakis et al, 1991; Miller & Rollnick, 2002; Rollnick et al, 1999) helps the professional in determining possible drivers for change (DiMatteo et al., 1980), as well as considering patient preferences (Epstein et al, 2004; Poskiparta et al, 2001; Stetson et al, 1992; Thorne et al, 2003). Summarising and paraphrasing are a small but important part of the active-listening skills employed in the consultation (Guimond et al, 2003) and are essential in helping to clarify patient and professional understanding of the issues being discussed.

Asking reflective questions helps the professional to further explore the understanding of the patient's perspective, ideas, and current self-care practices (Bertakis et al, 1991; Levinson & Roter, 1993; Poskiparta et al, 2000; Poskiparta et al, 2001). This process should enable the professional to provide information and explain ideas in a way that patients can understand and respond to, and is reported to translate into better outcomes (Ahlén et al, 2007). If patients are not asked they do not generally volunteer information (Barry et al, 2000; Greenfield & Kaplan, 1988). Questions are therefore an important part of the consultation process (Bensing, 2000) and used well they help to build rapport (Bertakis et al,1991; Mead & Bower, 2000a; Poskiparta et al, 1998). Questions can be an excellent way to encourage patients to think and reflect, and can encourage patient involvement and develop problem-solving skills (Guimond et al, 2003; King et al, 2002; Kolb, 1984; Lang et al, 2000; Maguire et al, 1996; Toobert & Glasgow, 1991; Weston et al, 1989; Willems et al, 2005).

Patients feel involved when professionals attend to their experiences of living with the condition and attempt to understand their problems and priorities associated with their long-term, self-care management (D' Eramo-Melkus & Demas, 1989; Entwistle et al, 2008; Kravitz et al, 1994). In addition, this exploration of the patient's understanding allows personal models of diabetes to be examined in order to dispel any myths and misconceptions that patients may have (Gentili et al, 2001; Sherman et al, 2000) in order to provide reliable information to enable informed choice to occur (Anderson et al., 1995; Glasgow et al, 1997; Goodall & Halford, 1991). Lang et al, (2000) suggest three approaches to engaging patients in the consultation; firstly, through direct facilitation by using questions such as *"How do you feel about the amount you eat, how would you describe it?"*; secondly, through an indirect approach to facilitation using questions such as *"Ok, can you tell me more about that?"* or *"what else?"* and thirdly by engaging the individuals through questioning based on active listening to clues *"It sounds as if things have been difficult."*

3.4.1 Information

An understanding of how information is given during the consultation is essential to the process of patient understanding and engagement, as information provision on its own does not change behaviour (Arborelius, 1996; Cullen et al, 2001; Goodall & Halford, 1991; McClusky & Lovarini, 2005; Norris et al, 2001). How and why information is given, and the individual's response to and use of the information provided are important factors to consider when reviewing information exchange. Patients are reported as wanting understandable information (Beisecker & Beisecker, 1990; Cox et al, 2006) and it is recognised that delivery of too much complex information could leave

patients feeling overwhelmed (Entwistle et al, 2008). There is a need therefore to use active-listening techniques to reflect on information and its use (Rollnick et al, 1999). Checking with individuals that the information provided is understood by them will help to inform rather than overwhelm (Entwistle et al, 2008; Epstein et al, 2004).

Exploring how issues raised by individuals in the consultation impact on their lives will enable better provision of information to meet the individual's needs (Anderson, 1995; Bertakis et al, 1991; D' Eramo-Melkus & Demas, 1989; Epstein et al, 2004; Thorne et al, 2003; Wolpert & Anderson, 2001). It requires that the professional demonstrate an understanding of the patient's view of their condition and its treatment before information is delivered (Del Piccolo et al., 2005; D' Eramo-Melkus & Demas, 1989; Edwards et al, 2003; Epstein et al, 2004; Thorne et al, 2004; Wikblad, 1991). Exploratory guestions such as "How do you feel about the amount you eat?" can result in greater sharing of ideas, concerns, and expectations on the part of the patient (Lang et al, 2000). Open questions should be encouraged as they facilitate patient participation, which is known to lead to improvements in outcomes (Bertakis et al, 1991; Harrington et al, 2004; Kaplan et al, 1989; Miller et al, 2002; Roter, 2000; Street et al, 1993; Willams, 1995; Willems et al, 2005). Patients report valuing opportunities to discuss their own ideas in the consultation and better outcomes have been observed as a result (Kaplan et al, 1989; Kravitz et al, 1994; Street et al, 1993; Wikblad, 1991; Williams et al, 1998). Active discussion allows the professional to match the content of the professional intervention to the patient's personal goal structure (Gable, 2007; Gebhardt, 2006) thereby improving outcomes and consultation effectiveness. Equally, active discussion can help to identify misconceptions, as patient understanding may differ considerably from the

professional (Chapelle et al,1997; Lang et al, 2000). Understanding the language that patients use to describe their diabetes provides further insight into the drivers for their behaviour (Freeman & Loewe, 2000). It is important therefore, to explore how individuals are currently using and interpreting the information that they possess (Eldh, 2006; Epstein et al, 2004). Misinterpretations may account for lack of understanding, application, or retention of the provided information (Chapelle et al, 1997), further contributing to poor outcomes.

Patients may depend on past experiences when making decisions (Sherman et al., 2000). Using questioning to explore what went well and the barriers to achievement will help to build on past success and avoid potential pitfalls occurring again (Rosal et al, 2001; Senécal et al, 2000). Building on existing ideas to problem solve collaboratively with patients should help to guide patients in their focused reflections (Moran et al, 2008; Poskiparta et al, 2001; Skinner, 2004). This will result in more active participation, as patients are likely to seek information to clarify treatment goals (Heisler et al, 2002; Poskiparta et al, 2001; Street et al, 1993). In addition, building on patient success helps to increase their self-efficacy (Bandura, 1977), which is acknowledged as an important component of successful behaviour change (Egan, 2002; Gable, 2007; Rollnick et al, 1999). Exploratory questions will facilitate this process and may be a fourth domain of the patient-centred model of care (Little et al, 2001b). Facilitation of patient understanding may enhance patient trust (Moran et al, 2008), motivation, and sense of personal control - leading to change (Feste & Anderson, 1995; Keating et al, 2004; Najavits & Weiss, 1994; Thorn, 2000).

Within the consultation setting, the amount and type of information provision will vary (Rosal et al, 2001), as general advice will not suit all patients (Thorne, Paterson & Russell, 2003). Increasing the interactivity of the patient/professional exchange may enhance patient-centred care (Roter et al., 2008). When patients are able to express their views greater adherence has been reported (Rost, 1989), with patient participation increasing when information provided with explanations is based on individual needs (Eldh, 2006) and offered with support (Street et al, 1993). Engaging the patient in this way requires good communication skills (Little et al, 2001b), as effectively reflecting the needs of individuals will necessitate less talk from the professional and more from the patient (Davies et al, 2008; Roter et al, 2008). The Mead et al (2008) postal survey of patients showed that positive evaluation of the communication skills of GPs was strongly associated with patient enablement.

The literature clearly states that telling people what to do doesn't work (Arborelius, 1996; McClusky & Lovarini, 2005; Norris et al, 2001). To avoid this didactic approach, information should be presented in a neutral manner (Rollnick, 1996). There is some disagreement over whether information-giving is a patient-centred behaviour (Mead & Bower, 2000b). Some studies make distinctions between the provision of biomedical information and psychosocial information (Mead & Bower, 2000a; Roter et al, 1987; Street, 1992a). However, it is clear that patients cannot make choices or decisions without adequate information (Ogden et al, 2002). The patient's self-care skills and ability to understand and work with their condition is far more important than the possession of knowledge on its own (Goodall & Halford, 1991). It is important therefore, that any information provided is framed in a way that makes sense to

the patient (Huang et al, 2005; Michie et al, 2003; Thorne et al, 2003), is relevant to them (Anderson, 1995; Entwistle et al, 2008) focusing on their concerns (Donnelly & Anderson, 1990; Ockene et al, 1995; Shultz et al, 2001) and needs (Anderson, 1995; Gentili et al, 2001; Kravitz et al, 1993; Levy, 2009; Vijan et al, 2005; Watson & McKinstry, 2009; Wolpert & Anderson, 2001). This process of information sharing suggests a more collaborative approach to working with patients that involve them in the decision-making process.

The negotiation of decisions allows exploration and understanding of the patient's perspective, thereby ensuring that information is understood and used appropriately (Bertakis et al, 1991; Miller et al, 2002; Rollnick et al, 1999; Roter, 2000; Street et al, 2009; Street et al, 1993; Williams, 1995). This allows the professional to check the patient's level of understanding and self-care practice, and identify potential gaps in knowledge and misunderstandings before providing information that is personalised and tailored to the individual (Barry et al, 2000; Lang et al, 2000; Rosal et al, 2001; Thorne et al, 2003). This procedure thereby avoids the use of redundant information (Rollnick, 1996). Rollnick (1999) suggests that this exploratory approach allows motivation to change to be elicited from the patient, rather than imposed by the professional, further increasing the chances of success (Miller & Rollnick, 2002; Rollnick, 1996; Satia et al, 2001; Spahn et al, 2010).

3.4.2 Lack of involvement

Not all patients will want to be involved in their decision-making, and some may prefer a more directed approach to care (McKinstry, 2000). In a study of American high school graduates, Swenson et al (2004) found that older patients

and those with less formal education were more likely to prefer a professionalled consulting style. This preference may be partly explained by the complex needs associated with older patients, and may be related to the expectations of certain social groups (Little et al, 2001b). Asking reflective questions during information-giving sequences will enable the professional to make a judgement about how involved the patient wishes to be and will therefore help to facilitate the appropriate information-giving behaviour (Bertakis, Roter & Putnam, 1991; Levinson & Roter, 1993; Poskiparta et al, 2000; Poskiparta et al, 2001). Furthermore, some patients may find it hard to engage in the self-care process and will therefore adopt a passive role (McKinstry, 2000). Although patients who are socially deprived and those with a lower level of education may adopt a more passive role (Greenfield & Kaplan, 1988), given the opportunity they can be encouraged to participate to a much greater extent in the consultation (Mercer et al, 2008). Patients are less motivated towards self-care when scolded (Wikblad, 1991) and when their cues, questions, or offered information are ignored (Anderson et al, 1991; McWilliam et al, 1994). Information exchanges that are controlling are potentially disempowering (Anderson, 1995; Beck, Daughtridge & Sloane, 2002; Hall et al, 1994), as few patients are motivated by pressure from others (Anderson et al, 1991; Bertakis et al, 1991; Carter et al, 1982; Satia et al, 2001). This type of approach may increase resistance to change (Hall et al, 1994; Miller & Rollnick, 2002) and may even result in patients doing the opposite of that proposed by the professional (Miller & Rollnick, 2002). Understanding patient resistance or lack of desire to change is important. Adopting empathic communication demonstrates respect through non-judgemental responses, and is more likely to lead to appropriate information delivery for the patient and an improved chance of long-term and

sustained change. Directive information to coerce behaviour change may result in decisions that do not meet the perceived needs of the patient.

3.4.3 Facilitation

Patient-centred communication therefore promotes interaction and collaboration, enabling the development of appropriate care plans (Ammerman et al, 2002; Michie et al, 2003; Poskiparta et al, 2001) and strategies for achieving dietary goals (Rosal et al, 2001; Street et al, 1993; Trudeau & Dubé, 1995). The respect developed through facilitative communication has been shown to yield strong results in terms of patient satisfaction and self-reported adherence in medical consultations (Fuertes et al, 2008; Hall et al, 2002; Theunissen et al, 2003).

Collaborative relationships are said to support adaptive learning in patients to help develop problem-solving skills (Health Development Agency, 2003), whereas problem-solving (Lorenz et al, 1996; Spahn et al, 2010) and decisionmaking abilities have been indicated as independent predictors of self-care practises (Clark & Hampson, 2001; Lorig et al, 2005; Toobert & Glasgow, 1991).

Collaborative working with patients is therefore proposed as a method to achieve better control in consultations (Aiken et al, 2005; Flocke & Stange, 2004; Greenfield & Kaplan, 1988; Spahn et al, 2010; Stewart, 2001; Street, 1992b; Trudeau & Dubé, 1995) and is a key feature of patient-centred counselling models (Anderson, 1995; Egan, 2002; Rosal et al, 2001). Collaborative working requires professionals to be less domineering and

coercive and more patient-centred in their approach (Moran et al, 2008; Stewart, 2001; Street et al, 2005; Williams, 1995). Provision of advice without pressure or control supports intrinsic patient autonomy (Feste & Anderson, 1995; Rubin et al, 2002; Shultz et al, 2001; Stewart, 2001; Street et al, 2005; Williams, 1995). It has also been shown to improve glycaemic control (Williams et al, 1998). These greater facilitation skills are particularly important for complex areas of behavioural change such as diet (Heisler et al, 2007; Pearson & Rapoport, 2007) where they have been shown to improve dietary implementation (Trudeau & Dubé, 1995).

3.5 Conclusions

This chapter has provided an overview of the essential skills needed for effective patient-centred communication. Empathy is a crucial component in patient-centred communication. The ability to acknowledge and respond to patient empathic cues and accurately reflect back understanding requires good communication skills, particularly active-listening skills (Neumann et al, 2009; Norfolk et al, 2007). This suggests that in order to explore empathy appropriately and to provide meaningful data, information needs to be collected not only on the presence of potential empathic opportunities in the consultation, but also on the communication skills used to demonstrate empathy in response to these cues (Neumann et al, 2009; Norfolk et al, 2007). Lack of active listening skills may limit patient expressions of empathy and adversely impact on levels of recall.

These identified communication skills are essential for effective self-care management strategies that utilise a range of behavioural approaches (NICE,

2007). A number of behavioural approaches have been applied in dietetic and diabetic practice to support patient-centred care (Anderson et al, 1995; Bridle et al, 2005; Clark & Hampson, 2001; Deakin et al, 2006; Epstein et al, 2005; Feste & Anderson, 1995; Georgiadis et al, 2006; Jones et al, 2003; Povey et al, 1999; Rubak et al, 2005; Salmela et al, 2009; Schnoll & Zimmerman, 2001; Spahn *et al*, 2010; Williams et al, 1998; Williams et al, 2004). However, effective use of these patient-centred communication skills will dictate whether these models are applied effectively in practice (Brug et al, 2007; Michie et al, 2003). The lack of strong evidence to support the effectiveness of the application of models to patient-centred care in practice may reflect the lack of appropriate communication skills and behaviours of the professionals who use them (Hampson et al, 1995; Salmela et al, 2009). Furthermore, the NICE (2007) guidelines on behaviour change suggest that behaviour-change training should focus on key competencies and skills rather than particular behaviour-change theories and models.

These reports add to the need for research that reviews the specific patientcentred communication skills used in consultations and their impact on outcomes, before mapping identified skills to the relevant behaviour-change theories as suggested by Abrahams & Michie (2008).

The studies reviewed in this chapter are of mixed quality and include both quantitative and qualitative research. The majority are conducted on physician consultations and there is limited data available exploring dietetic consultations (Cant & Aroni, 2008; Whitehead et al, 2009). While these studies appear to suggest that skills such as empathy, active listening, information exchange and

patient involvement are important for patient-centred communication, care should be taken when extrapolating data from medical to dietetic consultations. Professional and patient expectations have been shown to differ and this may impact on the communication process that occurs within the different consultations (Kravitz et al, 1993; Zanbelt et al, 2004). Female physicians demonstrate greater empathy and more supportive communication than their male colleagues (Roter et al, 2002) and as dietetics is predominately a female profession gender may be an additional factor influencing communication in dietetic consultations. There is variation in the definitions of patient centred communication and in the range of indirect and direct measures employed for assessing the communication skills/behaviours in studies reviewed. This adds to the lack of clarity on the specific skills being assessed and prevents direct comparison between studies (Epstein et al, 2005; Stewart, 1995). This chapter highlights the complex nature of patient-centred communication and the need therefore to explore the interrelationships of a number of clearly defined skills/behaviours to enable their interaction effects to be examined more fully. Linking identified skills and behaviours to a measurable outcome has the potential to provide feedback on skill use as well as providing a direct measure of the effectiveness of the dietitians' interaction on consultation outcomes.

3.6 Summary

This chapter has highlighted the need for a patient-centred approach when working with individuals with a chronic long-term condition. This approach requires the professional to engage with patients using a range of patientcentred communication skills. This skill set encompasses empathy, active-

listening skills (reflections, paraphrasing, summarising, exploratory questions, supportive and positive statements), and information exchange.

The overlapping nature of these patient-centred communication skills means that they cannot be studied in isolation, as meaning is derived from their combined use. In order to address the question of empathy and its effect on agreement on reported decisions made, active listening skills and information exchanges will need to be explored alongside empathy. This will enable the identification of their impact on each other and on patient/professional agreement on the reported decisions that are made.

CHAPTER 4: PATIENT SATISFACTION, PATIENT AUTONOMY AND PATIENT SELF-EFFICACY AND THEIR INFLUENCE ON EMPATHY AND AGREEMENT ON RECALL

This chapter provides information on the concepts of patient autonomy, selfefficacy, and patient satisfaction, and the rationale for including these measures in the study to explore patient-centred communication and its impact on agreement on reported decisions made.

4.0 Introduction

The previous chapter has highlighted the importance of patient-centred communication skills to engage in effective dialogue with the patients, to improve self-care management, leading to better health outcomes. In particular the importance of empathy and its impact on agreement on recall have been raised. However, the communication process that occurs in the consultation is complex and other factors may play a role in influencing behaviours and outcomes (Newman et al, 2004). To provide a clearer picture of this process, additional concepts that have been measured against empathy and recall need to be included in the design of the study. The concepts of patient satisfaction, patient autonomy, and patient self-efficacy have been measured and correlated with empathy and agreement on recall (Goodchild et al, 2005; Hurley & Shea, 1992; Parkin & Skinner, 2003; Zachariae et al, 2003). These concepts therefore need to be considered when designing a study to explore empathy and agreement on reported decisions made. The rationale for inclusion of these measures in the study design is provided below.

4.1 Patient Satisfaction

Patient satisfaction has been used widely to measure quality of services and delivery from professionals (Health Services Research Group, 1992), with studies focusing on assessment of satisfaction in relation to parameters measured in the consultation or resultant outcomes. Assessment of medical consultations has shown that when doctors have failed to provide a positive patient-centred approach, patients have reported less satisfaction with the consultation (Krupt et al, 2000; Roter, 2000). In contrast, greater patient satisfaction resulted when consultations included: sharing of understanding, greater involvement of the patient (including information-giving) and flexible discussion moving between relevant topics (Crow et al. 2002; Fossum & Arborelius, 2004). Encouraging patients to raise issues and discuss symptoms therefore not only increases patient satisfaction, it also has a positive impact on patient perception of communication and outcomes (Little et al, 2001a; Williams et al, 1998). In their qualitative analysis of 44 GP consultations, Moran, et al (2008) provided further insight into the dialogue process, suggesting that providing patient reassurance alone is not enough to increase patient satisfaction, but rather it is the partnership-building component of the interaction that increases patient satisfaction. This highlights the importance of the patient/professional interaction in the consultation and the engagement of patients in the consultation process. Although one study found that activating patients through engagement decreases satisfaction, as patients were more frustrated with the service as a result, other studies have not been able to replicate this (Greenfield & Kaplan, 1988; McCann & Weinman, 1996).

Rost & Roter (1987) found that patient adherence might be mediated by patient satisfaction. Other studies suggest that recall of advice as well as memory is influenced by patient satisfaction (Bartlett et al, 1984; McCann & Weinman, 1996). In their study of 840 renal patients, Coyne et al (1995) suggested that the extent of the dietary change recommended for the individual might be a factor to consider when assessing patient satisfaction. They demonstrated that the greater the impact of the dietary change the lower the level of patient satisfaction, participants whose prescribed protein intake more closely matched their usual intake were more satisfied and had better dietary adherence.

Content analysis involving people with diabetes and their perceptions of physician consultations has identified further factors that support satisfaction in the consultation. These included patient autonomy, empathy, feeling worthy, being attended to (Williams et al, 1998), reinforcement (Vickery & Hodges, 1986), feeling safe and confident, and being in agreement about goals (Hornsten et al, 2005). In the latter paper, the authors suggested that satisfaction and patient-centred care have many features in common; whereas Zachariae et al (2003) found that greater empathy in the consultation was associated with greater patient satisfaction and with higher frequency of patientcentred behaviours. However, these two studies used questionnaires, rather than direct measures of observed skills used by physicians and nurses. As perception of care has been shown to vary from practice the measures of perceptions of skills in these studies may not provide a reliable indicator of the actual skills used in practice (Latter et al, 2007; Parkin & Skinner, 2003; Pill et al, 1999). A study to quantify GP behaviours in the consultation with patient satisfaction further highlights the complexity of this type of assessment.

Rowland-Morin & Carroll (1990) quantified GP behaviours of silences, language reciprocity and reflective use of interruptions with patient satisfaction, and were able to contribute 27% of the variance in satisfaction to these communication behaviours. This study highlights further the need for good communication skills to support patient satisfaction.

Doctors have been shown to be poor at predicting patient satisfaction in the consultation (Hall et al, 1999; Mc Kinstry et al, 2006; Merkel, 1984; Zachariae et al, 2003). A small study of nurses and dietitians obtained a similar finding, where patient satisfaction and professional satisfaction as an outcome measure for the consultation were not found to correlate (Parkin & Skinner, 2003). This suggests that data collection that focuses solely on patient or professional perception may not provide a clear picture of the process that has occurred. Zachariae et al (2003) were able to show that more empathic physicians assessed patient satisfaction more accurately. To assess patient satisfaction accurately, prior knowledge of the empathic skills of the professionals involved is therefore needed before using the perception measure of satisfaction. This may not be a feasible option in a busy clinic setting, as it is adding an additional layer to the assessment of consultation skills and their effectiveness. However, it does highlight that patient satisfaction measures alone may not be adequate to assess the effectiveness of consultations due to the overlapping nature of the behaviours associated with satisfaction and patient-centred care (Sixma et al, 1998). One systematic review also suggested that age and health status might affect satisfaction, with satisfaction being reported as lower in sicker patients. The authors also suggested that improving patient satisfaction may be one way

to improve patient health, with the patient/professional relationship as the main route to achieving this (Crow et al, 2002).

Patient satisfaction data therefore needs to be collected in conjunction with other data in order for the findings to provide any useful meaning (Health Services Research Group, 1992). The studies reviewed appear to suggest that an improved relationship between the patient and the provider, as demonstrated through good rapport, may be one way to improve patient satisfaction. This will require good active listening skills on the part of the professional, as these are needed to demonstrate empathy and understanding (Egan, 2002). Greater empathy in the consultation might therefore be the mechanism by which greater levels of patient satisfaction occur.

4.2 Patient autonomy

Patient autonomy is the patient's perceived ability to make an informed choice. Perceived autonomy support is the measure of the patient's perception of support in making their choice, which is not coerced, but informed by the heath care professional (Deci & Ryan, 1985). This concept is derived from Self-Determination Theory, which states that behaviour is autonomously motivated when willingly chosen and initiated by the individual (Deci & Ryan, 1985). Self-Determination Theory suggests that autonomous support is predictive of autonomous motivation.

The causal role of autonomy support in developing autonomous motivation and perceived competence has been demonstrated in smoking cessation studies, resulting in greater abstinence (Williams et al, 2006). Applying this model to

self-care management has resulted in significant improvements in glycaemic control up to 12 months after the intervention and weight loss (Williams et al, 1998). Greater perceived autonomy support resulted in greater autonomous motivation to manage diabetes, leading to greater self-care and improved metabolic control (Arborelius, 1996; Golin et al, 1996; Greenfield & Kaplan, 1988; Kaplan et al, 1989; Mead & Bower, 2000a; Skinner & Cradock, 2000; Street et al, 1993; Williams et al, 2000; Williams et al, 2004).

Computer-assisted interventions applying Self -Determination Theory have also resulted in changes in lipids, diabetes distress, and depressive symptoms (Williams et al, 2007). This theory illustrates the importance of perceived autonomy support in changing perceived competence, leading to changes in health outcomes. Training professionals to increase their support of patient autonomy may be an important route towards improving diabetes outcomes (Parkin & Skinner, 2003; Street et al, 1993; Williams et al, 2005). In addition, a small study of 40 dietitian consultations found that patients reported more autonomy support when they created more empathic opportunities during their consultation (Goodchild et al, 2005). This study further highlights the possible role of empathy and communication skills in promoting patient autonomy in the consultation.

A small study of 77 patients showed that when patients reported higher autonomous motivation they also reported higher maintenance of diet (P < 0.0001) (Shigaki et al, 2010). Longitudinal studies have shown that greater autonomous motivation was related to compliance of dietary self-care activities up to six months later (Williams et al, 2004). In their study of 365 patients with

type 2 diabetes, Julien et al (2009) demonstrated the ability of autonomous motivation to predict dietary compliance ($\beta = 0.22$) up to one year after the consultation. This study also suggested that active planning strategies might be another path by which people with diabetes can develop their autonomous motivation. Active planning is the process whereby patients have the ability to make informed choices about dietary behaviours with the encouragement of professionals who help them to identify, initiate and solve problems in order to implement their chosen solution (Williams et al, 2004).

Studies suggest that supporting patients through active listening and empathic understanding may help support patients sense of autonomy leading to agreement on recall.

4.3 Self-efficacy

Self-efficacy is a person's belief in their ability to succeed in a particular situation (Bandura, 1977). A patient's perceived level of self-efficacy has been demonstrated to be a powerful influencer of adherence in diabetes care (Glasgow & Osteen, 1992; McCaul et al, 1987; Nakahara et al, 2006; Rubin et al, 1993). Self-efficacy is an indicator that patients have the confidence to overcome barriers associated with self- care practices (Kingery & Glasgow, 1989; McCaul et al, 1987). In their study of 256 Japanese people with type 2 diabetes, Nakahara et al (2006) found that adherence to diet correlated significantly with diabetes self-efficacy (r = 0.56, P < 0.01) and was found to be positively correlated with future glycaemic control. However, these patients

were not typical of a UK population⁴, as average BMI for this group was 22 and there was a low presentation of diabetic complications, despite the long duration of diabetes. The study of 638 members of the Quebec Diabetes Association by Senécal et al (2000) more typically reflects the UK population. The authors found that self-efficacy was significantly associated with self-reported adherence to diet ($\beta = 0.54$). Self-efficacy has been shown to account for 56% of the variation in studies looking at dietary changes and their effect on cholesterol levels (van Beurden et al, 1991), suggesting that self-efficacy is an important measure for assessing dietary change. Self-efficacy has also been shown to be powerful on specific behavioural tasks (Maibach & Murphy, 1995), such as modelling strategies to avoid overeating in stressful situations (Hinton & Olson, 2001), and on the use of goal setting and self-monitoring to increase dietary behaviour change (Schnoll & Zimmerman, 2001). Lack of promotion of self-efficacy in goal setting may contribute to poor goal agreement and recall (Cullen et al., 2001). Encouraging patients to set their own goals is therefore a vital step in increasing self-efficacy (Michie et al, 2003). Social support has also been shown to increase self-efficacy (Sousa et al, 2004). The daily difficulties of living with diabetes with limited social support may decrease self-efficacy (Nakahara et al, 2006). Self-efficacy and outcome measures are therefore important indicators for self-care and should be targeted for change by interventions from the professional (Miller et al, 2002). In young people with type 1 diabetes, self-efficacy has been shown to be a predictor of self-care behaviours (Johnston-Brooks et al, 2002), whereas other studies suggest that improving self-efficacy may lead to improved self-care management ability

 $^{^4}$ 80% of people with type 2 diabetes in the UK are overweight at diagnosis presenting with a BMI > 25 (Astrup & Finer, 2000)

(Krichbaum et al, 2003). Patients who exhibit greater control over the decisions made in the consultation have greater agreement on the decisions made and higher levels of self-efficacy (Heisler et al, 2003). Self-efficacy is therefore an important factor to consider in dietetic consultations, as this may lead to improvements in self-care management (Hurley & Shea, 1992; Knight et al, 2006). Although self-efficacy is a necessary condition for behaviour change, one study suggests that perceived self-efficacy alone may not be enough to explain behaviour or motivate action (AbuSabha & Ahterberg, 1997). This concept therefore needs to be explored alongside other measures to assess its role in improving patient recall.

4.4 Relationship between empathy, patient satisfaction, patient autonomy and patient self-efficacy with agreement on reported decisions made

The hypothesis here is that greater levels of empathy in the consultation will result in greater agreement on reported decisions made in the consultation. Greater empathy will result in greater levels of patient's satisfaction, but the effect of the dietitian's empathy may be mediated differently by the patient's perception of autonomy support and patient self-efficacy.

The casual model in Figure 4.5 explains the postulated relationship between empathy and agreement on reported decisions made in the consultation. It illustrates the cause and effect sequence between the independent variable of empathy and the identified mediating variables of self- efficacy and patient autonomy (Baranowski et al, 2003).



Figure 4.5: Causal relationship between empathy and agreement on reported decisions made in the consultation

The presence of professional empathy helps to build a good therapeutic relationship through active listening (Norfolk et al, 2007). When patients feel understood and listened to, this may help to increase their perception of autonomy support provided by the professional (arrow 1). As patient autonomy increases, levels of patient and professional agreement on reported decisions may increase (arrow 2) (Parkin & Skinner, 2003). Active listening, as evidenced through empathy skills, may help to develop self-efficacy in patients (arrow 3). Increased self-efficacy results in increased self-reported adherence to dietary self-care activities (Senécal et al, 2000). This may result from the increased agreement on dietary self-care decisions made in the consultation (arrow 4).

This relationship needs to be explored through statistical analysis. Moderating factors such as age, gender, and duration of diabetes and presence of complications need to be considered when analysing data as these factors may impact on outcomes (Baron & Kenny, 1986; Kenny, 2008).

4.5 Conclusion

The complex inter-relationships between empathy, self-efficacy, patient autonomy and agreement on reported decisions made have yet to be fully explored, as does the understanding of how these factors relate to informationgiving and the subsequent impact on agreement on reported decisions made. The complex nature of dietary behaviour-change and relationship to outcome measures requires a number of predictors to be addressed to try and explain and expand on this area (AbuSabha & Ahterberg, 1997; Epstein et al, 2005; Newman et al, 2004).

The model proposed in Figure 4.5 provides some indicative measures of communication behaviours tested through concepts of self-efficacy and patient autonomy. In addition, patient satisfaction will be measured to assess its impact on empathy. Validated measures are available for assessing patient satisfaction, patient self-efficacy, and patient autonomy. The measurement of the presence of specific communication skills of empathy, active listening, and information giving in the consultation will be addressed in the next chapter.

CHAPTER FIVE: EXPLORING TOOLS SUITABLE FOR ANALYSING DIETETIC CONSULTATIONS

The patient-centred communication skills identified previously in Chapter Three are: empathy, active listening skills, shared decision-making, and information exchange. This chapter describes the exploratory study set up to identify suitable tools for analysing patient-centred communication skills in dietetic consultations. It includes tool selection and application thereof, and the resultant findings in terms of tool-suitability for analysing patient-centred communication behaviours.

5.0 Introduction

The majority of studies exploring communication skills in medical consultations have been conducted using a variety of interactive coding tools, which explore in detail the dialogue between doctor/physician and patients (Cegala, 1997; Dent et al, 2005; Ford et al, 2000; Ram et al, 1999; Roter, 2003. There is limited reported use of these coding tools for exploring consultations with other professionals (Eide et al, 2011; Goodchild et al, 2005; Roter, 2003). This raises the question of how effective these tools will be for identifying patient-centred communication skills in dietetic consultations, not only in terms of content, but also in the light of differing patient/professional expectations (Anderson, 1981; Hampson & Mckay, 1996; Kravitz et al, 1993; Levy, 2009; Schrimer et al, 2005; Street et al, 1993; Zandbelt et al, 2004). The emotional, social and psychological impact of dietary changes will result in different issues being addressed, which in turn may require a different set of skills and or behaviours to be utilised by dietitians and other professionals in order to improve outcomes (Anderson, 1981). There is some evidence to suggest that patients are more

expressive when interacting with nurses than physicians (Campbell et al, 1990). Whether this is true for dietitian/patient interaction has yet to be established. The question remains of how best to assess the presence of patient-centred behaviours used by dietitians in dietetic consultations.

5.1 Selection of a Tool

A pilot study was designed to explore the use of a number of coding tools and identify those tools suitable for exploring patient-centred communication skills in dietetic consultations. Patient-centred communication skills have been identified as empathy and active listening skills (which include reflections, paraphrasing, summarising, exploratory questions, positive statements and information-giving). Fifteen patients with diabetes consented to have the dietitian make an audio-recoding of their consultations. Only 12 of these recordings were of a suitable quality to be transcribed. The following results describe the process of coding the 12 transcribed tapes with the interactive coding tools highlighting issues that arose with the use of these tools (See Appendix 1.1 for details of study recruitment, participants, and design).

Tools based on verbal behaviour coding scales are considered to be more reliable than broadly-defined behaviours for communication analysis, as they are less subjective in observer interpretation (Mead & Bower, 2000a). Consequently, verbal behaviour coding tools were selected for exploratory analysis.

The first section of this chapter explores the measurement of empathy using the Empathic Communication Coding System (ECCS) (Bylund & Makoul, 2005).

The remaining section explores the measurement of patient-centred communication skills (active listening) using the Roter Interactive Analysis System (RIAS) (Roter, 2005), the Verona medical interview Classification system – Doctor VR-MICS/D (Del Piccolo et al, 2005) and the Reflective Practice Coding System (Anderson et al, 1995). This includes: reflecting, summarising, paraphrasing, exploratory questions, and information exchange.

5.2 Coding for Empathy

5. 2.1 Empathic Communication Coding System (ECCS)

The empathic communication coding system (ECCS) was chosen, as this tool focuses on behavioural empathy by exploring the professional's verbal response to direct empathic opportunities raised by the patient (Bylund & Makoul, 2002). This measure has previously been used to explore dietetic consultations (Goodchild et al, 2005). Although the ECCS appeared to have potential for measuring empathy in dietetic consultations, a number of problems were identified with the coding system and changes were recommended. The original ECCS coding system used (*ibid*) has since been amended by Bylund & Makoul (2005), whereby a further coding level for empathic responses by professionals has been added which is similar to the one proposed by Goodchild et al (2005). A sample of 100 consultations showed the inter-coder reliability of this additional coding level to be acceptable (Cohen's kappa = 0.74).

The exploratory analysis seeks to assess the usefulness of the 2005 version of the ECCS for coding dietetic consultations. The additional coding level should provide greater differentiation of skills used to demonstrate behavioural

empathy. In addition, the ECCS codes from video recordings this exploratory

analysis will assess the feasibility of using the coding system on audio

recordings.

The ECCS coding involves listening to the recordings while coding to allow for

tonal qualities of voice to aide interpretation and codes empathy in two stages.

The first stage involves coding the empathic opportunities presented by the

patient (Table 5.6).

Statement of Emotion: Emotion is defined as "an affective statement of consciousness in which joy, sorrow, fear or the like is experienced."

Statement of Progress: the patient states or describes a positive development in physical condition that has improved quality of life, a positive development in the psychosocial aspect of the patient's life, or a recent, very positive, life-changing event.

Statement of Challenge: the patient states or describes a negative effect a physical or psychosocial problem is having on the patients' quality of life, or a recent devastating, life-changing event.

Table 5.6: Identification of empathic opportunities (ECCS) adapted from Bylund & Makoul, 2002⁵

The second stage involves coding the dietitian's response to the empathic

opportunities presented. These are scored on a scale of 0 - 6 (Table 5.7). Full

coding tables are presented in Appendix I.2.

⁵ Reprinted from Bylund, C. L. & Makoul, G. (2002) 'Empathic communication and gender in the physician-patient encounter', *Patient Education & Counseling*, 48(3), pp. 207-216 with permission from Elsevier

Level 6: Statement of shared feeling or experience
Level 5: Confirmation
Level 4: Acknowledgement with pursuit
Level 3: Acknowledgement
Level 2: Implicit recognition of patient perspective
Level 1: Perfunctory recognition of patient perspective
Level 0: Denial of patient perspective

Table 5.7: Coding system for empathic responses made to patient empathic opportunities (ECCS) adapted from Bylund & Makoul 2005⁶

If the dietitian's response to an empathic statement fell into more than one of the coding levels of the ECCS, the highest level was taken as the coded response. Codes between levels 3 and 6 illustrate that the dietitian had conveyed explicit recognition of the patient's perspective and demonstrated some form of empathy (Bylund & Makoul, 2002). All codes allocated were entered onto an Excel spreadsheet. The mean level of empathy for each consultation was computed by taking the sum of the empathic responses and dividing by the total number of empathic opportunities made in the consultation (Bylund & Makoul, 2002). The same method was used to compute the mean level of empathy for each empathic component (statement of emotion, statement of progress and statement of challenge).

5.2.2 Analysis using Empathy Coding System

Dietitians in this sample did not miss any of the empathic opportunities raised by the patient. The mean number of empathic opportunities per consultation was 16 (range 3-29), with a mean level of empathy of 2.99 (SD 0.52). A

⁶ Reprinted from Bylund, C. & Makoul, G. (2005) 'Examining empathy in medical encounters an observational study using the empathic communication coding system' *Health Communication*, 18(2), pp123-140 with permission from Elsevier

summary of the number of empathic opportunities provided and the dietitian's

Empathic statements made by patient	Number of empathic statements made	Mean empathic response to empathic statements (SD)
Statement of Emotion (SE)	28	2.56 (1.67)
Statement of Progress (SP)	60	2.68 (1.36)
Statement of Challenge (SC)	106	2.73 (1.08)
Total empathic		
opportunities presented	194	
(EOP)		

response to these is provided in Table 5.8.

Table 5.8: Empathic opportunities and dietetic response to these for the 12 dietetic consultations analysed

This data shows that patients are presenting with more statements of challenge in the consultation, than statements of progress or statements of emotion. The empathic response to statements of challenge is also greater than the empathic response to statements of progress and to statements of emotion.

Table 5.9 provides more detailed information on the different coding levels applied to the empathic responses made by dietitians to the patient empathic

opportunities presented in the consultations.

Level	Empathy description	<i>n</i> (N=194)	Percentage
6	Shared feeling and experience	0	0
5	Confirmation	25	13%
4	Acknowledgement with pursuit	29	15%
3	Acknowledgement	119	61%
2	Implicit recognition	19	10%
1	Perfunctory responses	0	0
0	Denial	2	1%

Table 5.9: The responses of dietitians to empathic opportunities

A full summary of the codes obtained using the ECCS can be found in Appendix 1.3.

Dietitians did not share feelings or experiences with patients; therefore codes were not allocated to the empathic level 6 "statement of shared feeling and experience." It is questionable whether this code fits into these coding criteria as provision of own experiences and feelings may override or belittle the patient's experience (Gable, 2007). Previous studies with dietitians have also noted the lack of use of this code in consultation analysis (Goodchild et al, 2005). Codes were also not allocated to level 1 "perfunctory recognition of patient perspective" as coding was from audio recordings rather than video and so this coding was unable to pick up on non-verbal behaviour. Previous studies using the ECCS have noted minimal responses to this coding level (Goodchild et al, 2005) with a reported response of 1.6% (Bylund & Makoul, 2002) to 4.6% (Bylund & Makoul, 2005). Given its limited use and the fact that level 1 does not have a positive impact on the measure of empathy, this coding level will be removed for the analysis of the audio recordings. *Implicit recognition* will then move from a level 2 to a level 1 position. *Minimal encouragers* are found in the ECCS coding under acknowledgement level 3 (Bylund & Makoul, 2002; Bylund & Makoul, 2005), and these will be moved down to a level 2 response. Goodchild et al (2005) have suggested the splitting of *minimal encouragers* from level 3. Coding for *minimal encouragers* at level 2 takes into consideration that acknowledgment through empathy is more than a "yes," "no" or "mmm" response. Behavioural empathy is about demonstrating listening through verbal responses, which illustrate the listener's attempts at understanding what has been said, eg "It sounds as if things have been tough over the last few weeks"

or "when you say it was hard to manage, could you explain that a bit more". Thus, acknowledgement with pursuit and confirmation demonstrate through reflection and questioning that one is hearing what the patient has said and is trying to understand (Gable, 2007; Goodchild et al, 2005; Shillitoe, 1994). The coded responses for empathy that reflect positive engagement should be the ones that are coded at levels 3-6, as suggested by Bylund & Makoul (2005). The revised coding level definitions are outlined in Table 5.10 below.

Level	Empathic description of category
6	Shared feeling or experience
5	Confirmation (let me check I heard you correctly and I understand)
4	Acknowledgement with pursuit (I hear you - help me understand)
3	Acknowledgement reflective listening (I hear you) active
2	Heard non-active minimal encouragers (mm, yes.)
1	Implicit recognition (hearing but not listening)
0	Denial not listening

Table 5.10: Revised coding levels for empathy adapted from EmpathicCommunication Coding System (ECCS) Bylund & Makoul 2005

5.3 Coding for patient-centred communication skills

The RIAS, Verona, and reflective-practice coding tool were selected for exploratory analysis of patient-centred communication skills. The RIAS and the Verona are comprehensive tools that enable coding of all patient and physician dialogue in the consultation and are therefore expected to capture all aspects of patient-centred communication skills highlighted. The reflective-practice coding tool focuses on specific verbal responses of the professional that illustrate patient-centred working as shown by: reflections on patient's feelings or goals, exploration of issues raised by patients and problem-solving alongside nonjudgemental responses. These tools may provide a detailed picture of the patient-centred communication skills used by dietitians in consultations.

5.3.1 Roter Interactive Analysis System (RIAS)

The RIAS was chosen as this has been used widely in doctor/patient communication, and therefore is considered to be a reliable tool for looking at content analysis of the consultation (Roter, 2005; Roter, 2009; Roter et al, 1997). It looks at both patient and doctor dialogue in detail. The coding frame employed allows exploration of empathy, information -giving, and listening skills.

A manual copy of the coding analysis system by Debra Roter (Roter, 2005) was used for analysis. Although a CD-ROM for analysis does exist, it was not felt to be necessary for the purpose of this exploratory study. The RIAS codes both patient and professional dialogue and focuses heavily on task-focused categories, of which there are 29 that ask questions and give information and 14 that are socio-emotional. A section of the coding categories is shown in Table 5.11 (Full coding categories are provided in Appendix 1.4).
Socio-emotional Exchange	Task-Focused Exchange	
Personal remarks, social	Transition words	
conversation	Gives orientation, instructions	
Laughs, tells jokes	Paraphrase/Checks for understanding	
Shows approval – direct	Bid for repetition	
Gives compliment – general	Asks for understanding	
Shows agreement or understanding	Asks for opinion	
Back-channel responses	Asks Questions (closed-ended) – medical condition	
Empathy	Asks Questions (closed-ended) –	
Shows concern or worry	Therapeutic regimen	
Reassures, encourages or shows	Asks Questions (closed-ended) – lifestyle	

Table 5.11: Coding categories for RIAS (Roter, 2005)

Consultations were coded directly from the audio recordings, allowing tone of

voice to help in defining the meaning of the words and utterances. For

example,

"the word OK, could be coded as agreement, asking for understanding, indicating transition or coded as a back-channelling response" (Roter, 2005).

The transcript was also marked for the number of utterances that would require coding. A gap of more than one second was coded as a separate utterance with this system. All codes allocated were entered onto an Excel spreadsheet. Any issues relating to the use of RIAS for coding the dietetic consultations were noted for discussion later. There are no reports in the literature of the RIAS being used to code dietetic consultations.

5.3.2 Verona Medical Interview Classification System – Doctor VR-MICS/D

The Verona Medical Interview Classification System has been developed to look at interviewing skills of doctors and in particular to measure responses to emotional issues that patients raise. It looks in detail at both patient and doctor responses, focusing on the emotional content of speech (Del Piccolo *et al*, 2005). This system has two separate coding systems: one for coding patient speech and one for coding doctor speech. The inter-rater and intra-rater reliability of the English translation of the Verona has shown to be good for both patient and doctor coding systems, eg a doctor agreement of coding categories of 88-91% and a Cohen kappa value of 0.86-0.91 (*ibid*). For this analysis, the doctor coding system was chosen to enable the dietitian's speech to be explored in detail. There are no reports in the literature of the VR-MICS/D being used to review dietetic consultations.

The identification and classification of verbal events were coded directly onto the transcripts after the transcripts had been divided into units of speech. A unit of speech was defined as an utterance with a start and an ending. The utterance could end because the speaker chose to stop of his or her own volition or because of an interruption. However, when the utterance changed content or process category, this too counted as a unit of speech (Del Piccolo et al, 2005). According to the Verona classification, a unit of speech may be as short as one word or as long as several sentences (Del Piccolo et al, 2005). Units of speech were coded and assigned to one of 22 mutually exclusive categories. These were then assigned under the following six headings: patient facilitating, patient supporting, patient education, negative talk, information gathering, and patient involving (see Table 5.12). The recordings were played

while the coding of the transcripts took place, as tone of voice might have added additional meaning to the words uttered. Categories classified as patient-centred are printed in italics.

Patient facilitating	Patient supporting
Conversation	Agreement
Welcome, taking leave, small talk, jokes	Reassurance
Facilitations	Appraisal
Neutral Affirmative reflecting	Respect, empathy, legitimating, participation
Transitions	
Orienting expressions	
Orienting explanations	
Patient education	Negative talk
Giving information	Disapproval
Illness management	
Psychological/psychosocial	
Giving instruction	
Illness management	
Psychological/psychosocial	
Brief answers	
Questions	1
Information gathering	Patient involving
Closed ended and open ended	Asking for repetition
directive Illness management	Asking for understanding
Psychological/psychosocial	Check
Non-directive	Clarifying, summarising, checking
	(interrogative reflecting)
Questions/completing patient agenda	Asks for opinion
	Illness management
	Psychosocial topics

Table 5.12: Coding categories VERONA Doctor VR-MICS/D⁷

⁷ Permission to use VR-MICS/D granted by Professor L Del Piccolo

All codes allocated were entered onto an Excel spreadsheet. Any issues relating to the use of the VERONA Doctor VR-MICS/D for coding the dietetic consultations were noted for discussion later.

5.3.3 Reflective-Practice Coding Tool

The reflective-practice coding tool was chosen as this focuses heavily on patient-centred care by looking at responses to patient's dialogue that support patient-centred working (Anderson et al, 1995). It has been used in professional training sessions on empowerment to provide participants with a marker of how, as a result of training, their responses have changed to become more empowering and less didactic and directive in the consultation (Anderson et al, 1995; Anderson et al, 1991). The empowerment model explores patient involvement, ownership, and sense of control (Anderson, 1995; Funnell & Anderson, 2004; Funnell et al, 2005). Essentially, this coding tool reflects on how empowering a consultation is, by coding verbal responses to patient dialogue that allow reflection, exploration and problem solving. A greater number of positive responses reflect greater patient-centred working (Anderson & Funnell, 2005a).

The reflective-practice coding system codes responses to patient's verbal expressions as outlined in Table 5.13. The recordings were listened to whilst coding responses. Examples of responses obtained from transcripts are in the right-hand column.

	Codes	Responses
+ 2	Focusing on feelings or goals	"So was that worrying you then?" "So how do you feel you're coping at the moment then?"
+1	Problem exploration	Ok, so what affect does it have on your diabetes? You mentioned twice that it's not good for your diabetes." "So what were your thoughts on that?"
0	Miscellaneous	
- 1	Solving problems for the patient	"Well you could still put it in a book"
- 2	Judging the patient	<i>"Well at least its healthy and not many calories in it."</i>

Table 5.13: Reflective Practice Coding System (Anderson & Funnell 2005)⁸

The scores for each professional response to statements made by the patient were added together and then divided by the total number of responses per consultation. All codes allocated were entered onto an Excel spreadsheet. Any issues relating to the use of the reflective-practice coding tool for coding the dietetic consultations were noted for discussion later.

5.4 Comparison of the RIAS/Verona and Reflective-practice Coding Systems

The following results describe the process of coding the 12 transcribed recordings with the interactive coding tools, highlighting any issues that arose with the use of these coding tools.

⁸ Copyright 2000, The Art of Empowerment. Modified with permission from the *American Diabetes Association*

5.4.1 Analysis Using RIAS Coding System

The RIAS system codes diet as a lifestyle issue, but as diet forms a major part of the treatment for diabetes; it could be argued that this should fall into the therapy category. The coding system states that when the doctor specifically mentions diet in relation to the patient's medical condition, ie losing weight, then it is coded under the therapy category. As diet is a major part of the ongoing and immediate treatment of diabetes, the decision was made to code diet as follows. When discussing therapeutic effects of food, ie those foods with a low glycaemic index (that slow down the release of blood sugar) or foods to treat hypoglycaemia, a therapy code was assigned. When particular foods or meals were being discussed in terms of personal preferences, a lifestyle code was assigned.

The coding criteria for counselling-type statements (reflections, paraphrasing, summarising) and empathy statements were very narrow, for example the following reflective question. *"When you say let yourself go a bit?"* demonstrates active listening. The dietitian is attempting to understand and clarify the patient's perspective by reflecting back the patient's words and phrasing them as a question. This response, however, was coded as a closed question, as per the manual. Patient-centred communication techniques such as these were not coded, resulting therefore in a loss of richness in the exchange.

The coding criteria for questions provided guidance on the use of closed and open questions and whether the questions were focused on lifestyle, therapy, or medical aspects of care. However, despite this level of detail, the subtleties of

the phrasing of the questions were not captured with the coding criteria, as illustrated above with reflective questioning. Overall the socio-emotional categories were either too broad or not clear enough to give full interpretation: an issue that has also been noted by others (Eide et al, 2004; Sandvik et al, 2002). Nearly half of the socio-emotional exchanges (49%) in dietetic consultations were positive. This category included dietitian approvals, agreement, or reassurance of something that the patient had said. Approximately a third (29%) of the task-focused exchanges were informationgiving. A full summary of codes obtained using the RIAS can be found in Appendix 1.5. Where data has been coded for empathy and information-giving it has lacked depth and direction. Although this tool was able to indicate the amount of dialogue that was concerned with information giving, there is no clear indication of how and why information exchanges occurred within the dietetic consultation. In addition some of the active listening skills were not being captured and coded adequately, resulting in a loss of data on effective patientcentred communication skills.

The multiple questions used in consultations are also slightly leading as they are giving the patient a number of options to choose from, which again may constrain or alter the flow of the conversation. Discrepancies in coding questions have also been highlighted by others (Sandvik et al, 2002). Patient responses of "*mms…*" were coded as agree, as there is no specific code for this type of response. It is also unclear from listening to the recordings whether the patient is agreeing with the dietitian or whether these are automatic responses. Silences and pauses are also not coded, despite the fact that these are known

to play a major role in counselling to elicit responses (Anderson et al, 1991; Eide et al, 2004; Gable, 2007; Sandvik et al, 2002).

5.4.2 Analysis Using Verona Doctor VR-MICS/D Coding

As with the RIAS coding system, the Verona system also posed questions on how to code diet-related questions. Some questions required yes and no answers so were coded under closed-ended illness management and therapy. Other diet-related questions were coded under open-ended directive illness management, and psychosocial. This was especially true for food questions, which allowed for more than a yes or no response and where the patient also provided information around lifestyle. Multiple questions were coded twice, as per the coding schedule. This raises similar issues to the RIAS coding as to which question is answered first and how this subsequently impacts on the flow of the consultation. Statements were made during the consultations that were not questions or information, or direct answers to patients' questions. However, there is no discrete category code for these statements; so a best-fit category was chosen. Leading questions also lacked a coding category.

There was some ambiguity on how to code responses that confirmed that someone was working or applying information in the right way. Would that be an empathy code or an agreement code, as illustrated below?

"Ahh right OK. So it sounds as if some of the things on here have been ongoing, but actually from what you're saying are slightly improving and moving forward."

It was difficult to code information giving, as the coding system only looks at content to place information into a category. It does not allow coding that differentiates between information givers (the patient or the professional) and whether the information giving follows the flow of the conversation. Interruptions were also difficult to code effectively, although they were coded for in terms of units of speech, but this does not allow any interpretation of the effect of the interruption on the subsequent direction of the consultation.

When the dietitian ignored statements that patients made and moved on to another topic, this was coded as a transition statement. However, this isn't really a transition statement as the patient has been ignored. Failure to listen to what the patient wants to say could alter the flow of the conversation and alter the final outcome (Anderson et al, 1991; McWilliam et al, 1994). Multiple repetitions could not be coded as such. Instead, they were identified and coded as individual utterances, losing the richness of this data in the final analysis. The Verona did provide clear data on specific questions, such as asking for patient understanding, asking for patient opinion, and checking patient understanding. It provided an overview of the communication that occurred in the dietetic consultations as follows. The largest percentage of codes for the Verona fell in the category of patient education 592/1744 (34%), with patient facilitating 470/1744 (27%) and information gathering 370/1744 (21%) following behind. As with the RIAS, this tool was able to indicate that a substantial amount of the dialogue was concerned with information exchange. A full summary of the codes for the analysis using Verona Doctor VR-MICS/D can be found in Appendix 1.6.

5.4.3 Analysis Using Reflective-Practice Coding

Responses were only coded for issues that the patient had raised in the conversation. *Minimal encouragers* such as *"mmmmh"*, and *"ahhh"* were not coded. Although it could be argued that *minimal encouragers* helped keep patients talking, they were not providing any assistance in reflecting on where the patient was, or how the patient had got to a certain point, as per the coding system. All food questions relating to collection of information around a 24-hour diet history were coded as 0. It could be argued that some of the responses were reflective, but the main purpose was obtaining information on what the patient was eating. Although reflection was employed, it was used to check that dietary information was correct, rather than how the patient thought or felt about their diet. If the patient had described their diet in terms of how they wanted it to be different or how they felt about it, and this was reflected back to the patient, a score would have been allocated as per the coding system. If a discussion occurred around potential solutions this was given a score of 0, but if the dietitian was telling the patient what to do then a score of –1 was given.

Large chunks of the conversation were coded as 0 and therefore ignored. This may be irrelevant, as these areas did not engage the patient in self-reflection and problem solving. This tool was very good at identifying facilitative responses from the dietitian in terms of problem exploration and exploration of patient's feelings and goals. It therefore provides clear guidance and coding on supportive interactive dialogue in consultations. However, a constructive patient-centred dialogue would be expected to contain more elements than just self-reflection and problem solving. This un-coded dialogue may impact on conversation flow and information exchange. Overall the reflective statements

of dietitians were positive. Dietitians scored twice as many positive reflective statements (212) compared to negative reflective statements (97) in the consultations (See summary in Table 5.14; full details in Appendix 1.7).

	Total	Mean per consultation
Total responses coded	1844	154 (SD 41.28)
Positive responses	212	17.67(SD 10.87)
Negative responses	97	8.08 (SD 6.84)

Table 5.14: Summary of scores obtained with reflective-practice coding tool

5.5 Discussion

The RIAS, Verona and reflective-practice coding tool all failed to adequately identify or code multiple questions, information-giving, and interruptions; features that occurred consistently in the dietetic consultations explored. Information giving is an important aspect of a patient-centred consultation and a major part of a dietetic consultation. Interruptions and multiple questions may be factors that impact on information giving within the consultation. These factors may alter the consultation outcomes and therefore warrant further investigation.

Patient-centred communication skills such as active listening are essential for an empathic dialogue to occur and are likely to impact on conversation flow (Norfolk et al, 2007). However patient-centred communication behaviours such as reflections, paraphrasing and reflective exploratory questions and positive statements were difficult to code with the RIAS and the Verona coding systems. This is an important omission, as positive reflections may influence patient's thoughts and feelings and so affect the way that the consultation proceeds, and the way that information is delivered and received (Holmström et al, 2003; Sandvik et al, 2002). Understanding the impact that patient-centred communication behaviours have on information flow and the subsequent impact on consultation outcomes is essential (Pedersen, 2009). This understanding will allow professionals to improve their interactions with patients, resulting in more productive outcomes.

5.5.1 Information-giving

This study highlighted discrepancies over coding of information-giving and information exchanges using the RIAS. This is an issue that has been identified by others (Mead & Bower, 2000a), who have suggested that information exchange and information-giving by physicians should be treated as patient-centred communication; as information around disease, treatment, side-effects and symptoms are all needed to inform choices. The RIAS still fails to address the question of who raises the information in the consultation, the level of patient engagement, and the subsequent impact of these factors on conversation flow and outcomes.

The Verona also identified a large proportion of the consultation as being focused around information exchange. However, like the RIAS, the Verona is unable to provide a clear understanding of the process of information exchange that occurred in the dietetic consultation. They both fail to explain whether the information was provided unsolicited by the dietitian. In addition, questions on how the information was delivered (as part of a collaborative exchange or a set of instructions) and how the patient received the information remain

unanswered. These factors may impact on subsequent conversation flow. Information itself is crucial for patient evaluation of care (Hall & Dornan, 1988). When delivered appropriately, information can empower patients to make informed decision about their care (Anderson et al, 1991; Epstein et al, 2004; Loewe & Freeman, 2000; Thorne et al, 2003). How information is given and received may serve to encourage personal reflection enhancing self-care practices and improving outcomes (Marilynn et al, 1995).

The reflective-practice coding tool was found to be a useful measure in this respect as the positive measures coded illustrated engagement in reflection and self-discovery. These responses encouraged patients to develop their own solutions to issues raised (Anderson et al, 1995; Anderson et al, 1991; Vickery & Hodges, 1986) and are said to improve outcomes (Clark & Hampson, 2001; Rollnick et al, 1999). However, the reflective practice tool failed to identify who asked for the information, so raising the question: was it a professional-led information-giving exchange or a patient-led exchange? Questions concerning the intent of the advice provided and subsequent impact on conversation flow also remain unanswered. In addition there were large chunks of the dialogue that were coded as 0. This dialogue may provide additional insight into the process of information exchange within the consultation, particularly around food and dietary issues, which were not coded specifically with this tool.

Patients are reported to prefer the presentation of information to be made frequently as a continuous process, rather than as a single chunk of information (Wikblad, 1991). This suggests that to achieve more meaningful data on information-giving exchanges that occur in the consultation, there is a need to

review the flow of information-exchange that occurs. The patient and provider contributions to these information-giving exchanges will provide a more meaningful interpretation of the process of information-giving that occurs in dietetic consultations and its subsequent impact on consultation outcomes (Goss et al, 2004; Sandvik et al, 2002). The lack of detail around information giving in the consultation has been identified by others who also suggest a sequential approach is taken to analysis (Eide et al, 2004; Sandvik et al, 2002). A sequential approach will provide further insight into the effect of informationgiving sequences within the consultation and their subsequent impact on outcomes.

5.5.2 Interruptions

Interruptions occurred in all 12 of the analysed recordings, however, none of the coding tools explored coded interruptions. Studies have noted that interruptions may act as indicators of patient wishes to contribute information, resulting in new information being delivered 70% of the time (Realini et al, 1995). Activating patients by engaging them in the consultation has been shown to improve outcomes (Harrington et al, 2004). Interruptions may be one method used by patients to gain control and pursue their own agenda (Sandvik et al, 2002). Alternatively, professionals may use interruptions to take back control of the conversation. Patients report being more satisfied when they interrupt the doctor but less satisfied when the doctor interrupts them (Rost, 1989). The presence of interruptions and the potential impact this has on information flow and outcomes further consideration.

5.5.3 Empathy

The RIAS and Verona had narrow criteria for coding empathy, making it difficult to assign speech to this category. Between 1 and 3 empathic opportunities per consultation were identified with these tools. Sandvik et al (2002) have also highlighted the narrow definitions for empathy when coding with the RIAS. The RIAS codes statements that paraphrase, interpret name or recognise the emotional state of the person (Roter, 2005). The Verona codes expressions that interrupt emotion expressed by the patient (Del Piccolo et al, 2005). Both of these tools focus on coding emotional content. However, the concept of empathy also encompasses an understanding of the patient's perceptions and opinions (Bylund & Makoul, 2002; Mead & Bower, 2000b; Norfolk et al, 2007).

In contrast, the empathy-communication coding system (ECCS) was able to detect between 3 and 29 empathy statements per consultation, highlighting the suitability of the choice of this tool for coding empathy. The ECCS records different empathic responses to the different empathic statements made by patients, allowing the interaction between patient and dietitian to be analysed in more depth. The use of the ECCS with minor amendments made to the coding levels should provide a detailed picture of behavioural empathy in dietetic consultations. In addition, the varied responses obtained to empathic opportunities may shape the way that information is exchanged within the consultation and warrants further investigation.

5.6 Summary

Exploratory analysis aimed to clarify suitability of the ECCS for measuring empathy from audio recordings and to identify a suitable tool to use alongside

this for measuring other patient-centred communication skills in the dietetic consultation. This analysis has confirmed that the ECCS would be a suitable tool for exploring empathy in the dietetic consultation. It provides a more detailed exploration of empathy than the RIAS and Verona. In addition, information is provided on how professionals responded to the empathic issues raised. A minor change to the coding levels has been suggested and will be considered when the ECCS is used for the main data analysis in order to explore empathy in the dietetic consultations.

Although the RIAS, VERONA and reflective-practice coding tool provided some information on patient-centred communication skills, they did not adequately identify and code patient-centred communication skills such as positive reflections, probing questions, paraphrasing, reflective exploratory questions and positive supportive statements. This lack of identification of specific patient-centred skills could result in the richness of the patient-centred exchange being lost. In addition, these tools were unable to provide adequate detail on information giving in the consultation, in particular, how the information was provided: was it requested by patients or provided unsolicited by the dietitian? In exploring the use of a different interactive coding tool, Zandbelt et al, (2005) have raised this issue of lack of detail on information giving. This suggests that a different approach to measuring and analysing information giving within the consultation is required in order to explore how informationgiving impacts on conversation flow and outcomes.

The process of empathic listening and patient-centred communication is likely to impact on both the flow of information and the way that information is delivered

in the consultation (Norfolk et al, 2007). Far from answering questions, this exploratory analysis has raised further important questions that need to be considered, as follows: How do patient-centred communication skills impact on the information-giving process in the consultation? What effect does empathy have on information giving within the consultation?

Further, clarification is required on the relationship between empathy, information-giving and active-listening skills, and their subsequent impact on agreement on reported decisions made in the consultation. The design and implementation of a study to address these questions appropriately are detailed in the following chapter.

CHAPTER SIX: RATIONALE FOR USING MIXED METHODOLOGY

This chapter explores the issues that need to be considered when designing and setting up a study using mixed methodology. It illustrates how the study to explore empathy, information giving, and agreement on recall in dietetic consultations was designed, explaining the reasons for the two separate phases of data collection and analysis.

6.0 Overview of Mixed-Methods Research

Mixed-methods research combines qualitative and quantitative methodological approaches in the same study. It seeks to consider the central research question before deciding which qualitative or quantitative method is best suited to examine this (Creswell, 2009; Teddlie & Tashakkori, 2003). It is, however, more than just collecting and analysing quantitative and qualitative data in the same study. Philosophical considerations as to how and when the data will be collected, analysed and mixed are also required (Teddlie & Tashakkori, 2003). Both qualitative and quantitative approaches are used in tandem so that the overall strengths of the study are greater than when the approaches are used independently (Creswell, 2009; Creswell & Plano Clark, 2007). Specifically, combining approaches allows a more expanded and deeper understanding of complex issues than can be obtained from the use of qualitative or quantitative research alone (Creswell, 2009). Furthermore, it allows the strengths of both qualitative and quantitative approaches to be utilised in addressing the research question (Johnson & Onwuegbuzie, 2004). Combining both approaches is often referred to as the 'third wave' or 'third research movement' as it looks beyond the paradigm wars of qualitative versus quantitative research (Creswell

& Plano Clark, 2007; Johnson & Onwuegbuzie, 2004) and focuses on their combined strengths.

6.01 Issues in Mixed-Methods Research

Mixed methods research presents the researcher with a number of challenges. Firstly, those adopting this approach need to be familiar with both quantitative and qualitative data analysis (Creswell, 2009), as care is needed in the interpretation of the data and of the transferability of any findings to the wider community (Tashakkori & Teddlie, 1998).

Secondly, the timing of the data collection and interpretation of data is influenced by decisions regarding the weighting of the qualitative and quantitative aspects of the research, as well as the intended mixing that will be applied to the data (Creswell, 2009). Failure to adequately consider the weighting and intended mixing of data can result in a study that is not true mixed methodology (Creswell & Plano Clark, 2007).

6.02 Rationale for Using Mixed-Methods Research

Quantitative studies are used when identifying factors that influence outcomes, to gain insight into the best predictors of these outcomes or the usefulness of an intervention (Creswell, 2009). Conversely where a concept or phenomenon requires further exploration (as there is little published research in that area) then qualitative approaches are employed (Creswell, 2009). A mixed methods design is therefore appropriate when either qualitative or quantitative approaches in isolation are inadequate in addressing the research question (Creswell, 2009). However mixed-methods researchers need to clearly establish their rationale for mixing the qualitative and quantitative methods in their study. Five main reasons for the use of mixed methods research have been advanced (Johnson & Onwuegbuzie, 2004) and outlined in Table 6.15.

1. Triangulation	Data sets from qualitative and quantitative approaches seek convergence to provide corroboration and understanding
2. Complementary	Clarification of the results from one method is used with results from another
3. Initiation	Data is used to discover paradoxes or contradictions and so used to reframe research questions
4. Development	Using the findings from one method to inform another method
5. Expansion	Broadening the range of research and understanding by using different measures for different components of the issue

Table 6.15: Reasons for mixed-methods research (adapted from Johnson &Onwuegbuzie, 2004)

6.1 Designing a Mixed-Methodology Study

Once the rationale for the mixed methodology has been established, there is a further need to consider the type of quantitative and qualitative approaches to be taken. The design of the mixed methodology should then be considered by attending to four key design tenets (Creswell, 2009).

6.1.1 Theorising

Theorising is an important aspect of mixed methods research, as theory provides an overarching "'lens' that shapes the types of questions asked, the types of participants enrolled in the study, the data-collection methods used and the implications of findings from the study" (Creswell, 2009, p 208). Theory may

provide a framework that guides the entire study design. Theory may also include a "broad explanation of behaviour and attitudes complete with variables, constructs and hypotheses" (*ibid* p 61). Starting with a theoretical model may be one way of developing themes within qualitative research; hence the theorising provides a framework for both qualitative and quantitative research to occur (Creswell, 1998; Creswell, 2009).

6.1.2 Weighting

This refers to the priority given to qualitative and quantitative methods in the study and establishes the context for the treatment of the data (Creswell, 2009). These methods can have equal importance in a study, or one method may take precedence over the other (Creswell, 2009). The questions that are being examined influences the ordering of the chosen methods. If the predominant research questions are about concepts or phenomena, then qualitative methods may take priority, whereas quantitative methods are used to clarify aspects of the findings obtained (Creswell, 2009). Alternatively, quantitative methods may be given priority where clarification is needed as to which factors influence outcomes, whereas qualitative methods are used to explore possible reasons for observed outcomes (Creswell, 2009).

6.1.3 Mixing

Mixing relates to how and when the data from the quantitative and qualitative methods are brought together in the study (Creswell, 2009). It is an important consideration, as unequal priority attached to methods may result in unequal evidence within the same study, with a subsequent unintended impact on interpretation of results (Tashakkori & Teddlie, 1998). This issue can be

addressed by taking care with the design, conduct, and interpretation of data from each method (*ibid*). Creswell (2009) describes how the mixing of the quantitative and qualitative research can occur at any of one of three points: data collection, data analysis or data interpretation. Creswell describes the way that the data is mixed as either: *connected, integrating* or *embedding* and these different approaches to mixing are described as follows. '*Connected*' refers to the analysis of data and results from one study in order to identify participants for the follow-up study. '*Integrating*' refers to the merging of the qualitative and quantitative data sets whilst '*embedding*' refers to the use of secondary data to support findings from the primary data set (Creswell, 2009).

6.1.4 Timing

Data collection for mixed methods research can occur sequentially or concurrently. If data is collected sequentially it is collected in discrete phases so that one method follows the other. The ordering of the phases is dependent on the aims of the study. If quantitative and qualitative data are collected concurrently then data for both methods is collected at the same time. On a practical level, the concurrent method reduces the time needed for data collection (Creswell, 2009).

6. 2 Consideration of the four tenets for designing the mixed methodology In the set-up of this two-phase study, the rationale for the use of mixed methodology and the design tenets employed are explored below.

6.2.1 Theorising

This study is observational in design, in that data is collected through audio recording patient and professional consultations in the real world, ie the usual clinic setting. This allows the usual consulting behaviour of the dietitians to be assessed. The study has a defined outcome measure of agreement on reported decisions made in the consultation, which is supported by the theoretical framework outlined in Chapter Three. The relationship between the presence of empathy in the consultation and agreement on reported decisions can be tested with validated quantitative measures. However, there is a need to explore this process further, to better understand the effect that empathy has on information-giving and the subsequent effect that it has on agreement on reported decisions made in the dietetic consultation (Connor et al. 2009: O'Brien & Petrie, 1996; Pedersen, 2009; Watson & McKinstry, 2009). Although there is theoretical support for this relationship, the empirical evidence to support the arguments and the detailed understanding of the relationship between empathy, conversation flow and outcomes is lacking. Qualitative analysis is therefore needed to explore differences in information-giving behaviours that occur in consultations (Teddlie & Tashakkori, 2003). To allow detailed exploration of the interaction between empathy, communication skills, and the subsequent flow of information giving that occurs in the consultation (Pedersen, 2009; Schrimer et al, 2005). The outcome marker of agreement on reported decisions made will allow further differentiation by exploring how information-giving behaviours and empathy vary between consultations with agreement and those with disagreement on decisions made. This will provide a unique insight into the consultation process and its effect on the outcome measure of agreement on decisions made. This study uses the results from the

qualitative analysis to explore and expand on findings from the data collected by observation of the consultation (the quantitative data set). It therefore complies with the development rationale listed in Table 1 (Johnson & Onwuegbuzie, 2004).

6.2.2 Weighting

This study's primary focus is to explore empathy in the consultation and its impact on patient and professional agreement on reported decisions made. The relationship that empathy may have with self-efficacy, patient autonomy and patient satisfaction and the subsequent effect this may have on agreement on reported decisions made in the consultation are also examined. To test these relationships, a quantitative approach is required and this forms Phase I of the study. Consequently, quantitative methods of data collection are given precedence over qualitative methods in this study.

The qualitative method adds a secondary dimension by helping to explore further the process by which empathy impacts on agreement on reported decisions made in the consultation and forms Phase II of the study. This process is said to be analogous to hierarchical analysis in quantitative research, which attempts to explain predictors of observations obtained from quantitative analysis (Tashakkori & Teddlie, 1998).

6.2.3 Mixing

The embedded nature of the qualitative data means that quantitative analysis must occur first in order to identify data samples suitable for qualitative analysis. A sequential approach to analysis of data was therefore adopted (Figure 6.6).

The data from the qualitative analysis is secondary to the quantitative data and was used to build on and expand on this data (Creswell, 2009; Johnson & Onwuegbuzie, 2004). The two data sets are brought together in the discussion of the results, providing a detailed explanation of the research questions. Figure 6.6 illustrates the sequential explanatory design employed for this mixed method study, using the mixed methods notation described by Creswell (2009).

Figure has been removed due to Copyright restrictions

Figure 6.6: Sequential Explanatory Design adapted from Creswell (2009)

In addition, this study also incorporates a within-stage mixed-model design (Johnson & Onwuegbuzie, 2004). This arises as a result of an open question presented on the quantitative questionnaire, which explores decisions made in the consultation. The decisions were converted into numerical codes that were then subjected to statistical analysis (Johnson & Onwuegbuzie, 2004). This data will be presented alongside the quantitative data set in Chapter Eight.

6.2.4 Timing

This study employed a concurrent embedded strategy. Qualitative and quantitative data were collected simultaneously, with the qualitative data embedded in the quantitative data set (Creswell, 2009; Johnson & Onwuegbuzie, 2004).

6.3 Study design

This mixed methods study was conducted in two phases (Figure 6.6). Phase I is a quantitative study that aims to identify the presence of empathy in the consultation and establish its relationship to agreement on reported decisions made. In addition, patient satisfaction, autonomy, and self-efficacy were measured with validated questionnaires providing quantitative measures. These measures may have a mediating effect on empathy and the statistical relationship between these measures was explored.

Following this statistical analysis, Phase II of the study takes a qualitative approach to explore the presence of information-giving exchanges and patientcentred communication skills within consultations, to better understand their relationship to empathy. Consultations with complete agreement and complete disagreement on reported decisions made were selected from the Phase I data, to provide contrasting data sets for the qualitative analysis. This allowed the presence or absence of patient-centred communication skills and informationgiving exchanges to be explored when they occurred in consultations that featured contrasting agreements on reported decisions made.

6.4 Observational method for data collection

This study uses audio recordings in order to collect information on current practice and behaviours in the consultation. Making audio and video recordings of consultations is a well-established method for exploring communication between professionals and patients in the usual clinic setting (Glasgow et al, 2006). This method helps to develop an understanding of the structure and patterns of patient and clinician communication, and is more objective then self-reported data (Cant & Aroni, 2008; Lu & Dollahite, 2010). It does not rely on retrospective accounts from the individuals (which could be misleading or inaccurate due to imprecise recall) but allows the interaction dynamics of the consultation to be viewed objectively (Field & Hole, 2003; Lu & Dollahite, 2010).

Discrepancies in patient and professional perceptions of care and the perceived practice of professionals have been highlighted (Latter et al, 2007; Makoul et al, 1995; Parkin & Skinner, 2003; Pill et al, 1999; Watson & McKinstry, 2009). Training interventions that have used direct observation methods and videos have further highlighted discrepancies between professional's perception of care delivery and the reality of care that is delivered (Brug et al, 2007; Latter et al, 2007; Pill et al, 1999). These studies suggest that data on an individual's perception of their practice in the consultation is an unreliable method for reviewing current practice. The professional's perception of care delivery varies considerably from that of patients (Anderson et al, 1991; Makoul et al, 1995) and self-assessment of their skill level by professionals is often over-inflated (Kruger & Dunning, 1999) and does not correlate with patient perception of care delivery (Brug et al, 2007; Parkin & Skinner, 2003). These studies strengthen the need for observational data collection to provide insight into usual practice.

It has been suggested that professionals and patients may modify behaviours in the consultation as a result of being observed (Swift, 1996), with some behaviours being more sensitive to this than others (Coleman, 2000). It is further suggested that this 'reactivity' or observer effect may be increased when the event is being video recorded (Swift, 1996). Therefore, to minimise the observer effect and to remove a potential barrier to participation in this study, it was decided to use audio recordings of the consultations. In order to limit obtrusiveness, a small recording device was placed just behind the patient, out of their line of vision, thus increasing the likelihood that the patient would behave naturally throughout the consultation. A review of audio and visual recordings suggests that 97% of patients who had their consultations audio recorded did not feel that the recording device influenced their behaviour in the consultation or impacted on the level of care that they received from the professional (Themessl-Huber et al, 2008). Participants also reported forgetting about the presence of the recording device as the consultation proceeded, with some patients stating that they eventually forgot about the recording process completely.

Awareness of the recording device is likely to be accentuated at the start of data collection. The first recording from each dietitian was therefore discarded to allow for this effect. This procedure was successfully employed in a number of earlier studies exploring communication skills in nurses and dietitians (Goodchild et al, 2005; Parkin et al, 2006; Parkin & Skinner, 2003). Observational data collection has been shown to be feasible, acceptable, and practical (ThemessI-Huber et al, 2008). The disadvantage of observational methodology is that it is costly and time-consuming in terms of data collection

and processing and does not allow for the measurement of cause and effect (Creswell, 2009).

6.5 Summary

This chapter has described the rationale for the mixed methodology approach used for this study. It has provided an overview of the structure of the study, which aims to explore empathy and its effect on information giving using a sequential explanatory design. Phase I was undertaken to collect observational data from dietetic consultations and to analyse it to provide a quantitative data set. Phase II used a qualitative approach to better understand the findings in Phase I, by exploring a selection of consultations identified in Phase I as having contrasting agreement on reported decisions made. The methods and materials used for each of the phases will be explored in the following chapters. The combined discussion of the findings from Phase I and Phase II of this study will occur in Chapter Twelve. PHASE I

Quantitative Methods and Analysis

CHAPTER SEVEN: METHODOLOGY FOR PHASE I (Quantitative)

This chapter discusses the data collection process for Phase I of the study, including access to clinics, recruitment, ethical considerations, and methods used in the interpretation of the quantitative data collected.

Phase I focuses on the quantitative measurement of empathy, patient and professional autonomy, patient satisfaction, patient self-efficacy and agreement on reported decisions made in the consultation. The relationship between these factors has been explored in Chapter Four.

7.0 Phase I: Research Questions and Hypotheses

The data resulting from the analysis in Phase I attempts to answer the following research questions and test the hypotheses:

Research Question 1

What is the relationship between empathy and agreement on reported decisions made in the consultation?

Research Question 2

To what extent do the various components of empathy (statements of challenge, statements of progress and statements of emotion) impact on agreement on reported decisions made in the consultation?

Research Question 3

What is the relationship between patient satisfaction and empathy?

Hypothesis 1

Higher levels of empathy will result in greater agreement on reported decisions made in the consultation.

Hypothesis 2

Higher levels of empathy will result in greater levels of patient satisfaction.

7.1 Recruitment

7.1.1 Dietitians

Permission to access nine specialist dietitians was sought from the heads of service in five diabetes centres across the South West of England. Once permission was granted, visits were then arranged with the dietitians working in the diabetes centres to provide further information about the study.

Three dietitians from one Trust felt that the study would take up too much of their time and therefore declined to take part, and another two dietitians from a different Trust had numerous staffing issues that impacted on their work load preventing them from committing to the study. Four dietitians in the remaining three Trusts agreed to take part.

The number of dietitians agreeing to participate in the study represented 44% (4/9) of those approached and is similar to the median participation rate of 41% for GPs consenting to have consultations audio-recorded (Themessl-Huber et al, 2008).

7.1.2 Patients

Individuals with diabetes who were attending scheduled out-patient appointments with the diabetes dietitians who were participating in the study were eligible to take part. These routine out-patient clinics consisted of new and follow-up appointments for people with either type 1 or type 2 diabetes.

Patients who had scheduled out-patient appointments with the participating dietitians were contacted via letter. Information sheets explaining the study

were also provided (Appendix 2.1). Patients were asked if they would consider taking part when they next attended the clinic. Figure 7.7 illustrates the number of patients who were initially contacted with information on the study and the number of patients who consented to take part.

7.2.2.1 Inclusion and exclusion criteria

Patients were excluded from taking part if they were in any of the following categories: not fluent in English; under the age of 16; with learning difficulties, currently experiencing mental health issues; abusing drugs or alcohol.



Figure 7.7: Flow diagram illustrating number of patients contacted and the final number of patients consenting to take part in the study

The recruitment rate for this study was 92/110 (84%), which is similar to the recruitment rate of 83% reported in other studies using audio recordings of patient consultations (Themessl-Huber et al, 2008).

The average rate of non-attendance at clinics in this study was 47/157 (30%), which is comparable to figures obtained for dietetic obesity clinics103/313 (33%) in Hickson et al (2009) and dietetic diabetic clinics: 127/293 (43%) in Spikmans et al (2003). It is unlikely therefore that patient attendance at these clinics was adversely affected by the provision of written study information prior to the clinic visit.

Demographic data was collected on all patients scheduled to attend the dietetic clinic. Information on whether the appointment with the dietitian was a first visit or a follow-up visit was also recorded.

7.2 Participants

7.2.1 Dietitians

The following data was collected on participating dietitians: gender, number of years in practice and prior attendance on training courses covering communication and behaviour change.

The diabetes dietitians taking part in this study were all female and had specialised in diabetes for 10-11 years after qualifying. The number of training courses covering behaviour change and communication skills that participating dietitians had attended since qualifying is shown in Table 7.16.

	Years qualified	Number of training courses attended since qualifying
Dietitian 1	23	1
Dietitian 2	25	5
Dietitian 3	13	6
Dietitian 4	24	3

Table 7.16: Number of training courses in behaviour change and communication skills attended by dietitians in this study since qualifying

The earliest training course attended was a teaching skills course in 1985 and the most recent attendance on a course was in 2007. Courses included counselling skills, motivational interviewing, empowerment courses, and cognitive behavioural therapy for obesity and weight management. Data on training of dietitians in behaviour change skills is not held centrally, so it is difficult therefore to know whether the level of training presented in Table 7.16 is a typical reflection of the level of training in this area for the dietetic profession as whole.

7.2.2 Patients

A total of 92 patients were successfully recruited for the study. However, three recordings were discarded due to poor recording quality and a further three recordings were not included in the data analysis as patients failed to complete the questionnaires adequately. This left 86 patients with completed questionnaires and audio recordings available for analysis. Demographic data of the 86 patients who took part in the study was compared to the 67 who were not included, to ensure that the group that agreed to take part were representative of the population being studied (see Table 7.17).
	Patients	Patients not	p
	taking part	taking part	value
	(N = 86)	(N = 67)	
Female gender % (n)	35% (30)	34% (23)	.815
Median age (years)	54	49	.114 ^a
Range	21-85 years	19-89 years	
Type 2 diabetes % (<i>n</i>)	66% (57)	58% (39)	.184
Duration of diabetes (years) SEM	10.87 (1.31)	10.45 (1.40)	.414
Range	1month to 59 years	6 months to 50 years	
Complications present % (yes)	42% (36)	37% (25)	.782
Follow up appointment % (n)	34% (29)	46% (31)	.243

^a Due to significant Levene's test with independent t-test, this was recalculated using Mann-Whitney

Table 7.17: Demographic data for patients with complete data set included in the study, compared to patients who did not participate

Patients taking part in this study did not differ significantly from those not taking part across any of the characteristics considered.

7.3 Method

Once ethical and Trust approval had been granted, written consent was

obtained from the dietitians participating in the study (Appendix 2.2). In each

Trust, a process was established for distributing information sheets and

covering letters to patients one week prior to clinic attendance.

Data collection at clinics was co-ordinated with the dietitians in each Trust and this involved checking clinic dates and times as well as periods of annual leave to ensure that the researcher could attend the scheduled dietetic clinics. Data collection for the three Trusts took place over a 12-month period. Patients who were willing to take part in the study signed a consent form (Appendix 2.3) before entering the clinic room. The whole consultation was audio-recorded, but patients were reminded that they could withdraw their consent at any time during the consultation and the recording would be erased. The dietitian undertook the consultation following their usual procedures and at the end of the consultation the patient and dietitian were asked to complete a brief questionnaire.

The questionnaire booklet contained a set of validated questionnaires, all of which had the term *professional* changed to *dietitian* for the purpose of this study. The dietitian completed the questionnaire independently; in the clinic room after the patient had left the room. The patient completed the questionnaire booklet in an available side room before leaving the clinic. During this process, the researcher stayed with the patient to ensure that questionnaires were completed and to provide a reading service for those who had difficulty reading. This did not involve interpreting the meaning of the questions. The researcher administered all of the questionnaire booklets and gained the consent of all the patients, thus ensuring that a consistent approach was taken and thereby reducing the possibility of bias in the data collection caused by different researchers doing the work (Silverman, 2006).

Questionnaires were collected at the end of the clinic session and securely filed. All questionnaire booklets and audio recordings were dated and assigned ID codes to enable patient and dietitian questionnaire booklets and audio recordings to be matched.

7.4 Measures

7.4.1 Questionnaires

Questionnaire booklets were designed specifically for either the dietitian or the patient. The dietitian booklet (Appendix 2.4) contained an open question about decisions and the Health Care Climate questionnaire (HCC) (Williams et al, 1998). It also contained a section to record patient demographic data: age, gender, type of diabetes (Type 1 or Type 2), duration of diabetes, presence or absence of complications, whether the appointment with the dietitian was a first visit or a follow-up visit.

The patient booklet (Appendix 2.5) contained an open question about decisions, the Health Care Climate questionnaire (HCC), the Medical Interview Satisfaction Scale (MISS) (Kinnersley et al, 1996) and a self-efficacy questionnaire (SE) (Lorig et al, 2009).

Details of the open question and the validated questionnaires used in the questionnaire booklets are provided below:

7.4.1.1 Open Question

The first page of the questionnaire booklet for both patients and dietitians asked the same open question.

"Please could you write down any decisions that were made in the consultation today?"

This study design was influenced by the results from a previous study, which asked patients to list goals and decisions set in the consultation (Parkin & Skinner, 2003). This noted that either the listed goals were the same as the

decisions or that the goal section was left completely blank, while the decisions section was completed (Parkin & Skinner, 2003). Other studies have also recorded difficulties with patients interpreting the term goals (Bradley et al, 1999). Consequentially the word *decisions* was chosen for this open question, to help ensure that patients provided an answer to the question.

7.4.1.2 The Health Care Climate Questionnaire (HCC)

The Health Care Climate questionnaire (HCC:Williams et al, 1998) assesses the degree of autonomy support in the consultation as perceived by the patient. Autonomy support examines the extent to which providers are perceived by patients to elicit and acknowledge their perspectives, support their initiatives, offer choice about treatment options and provide relevant information while minimizing pressure and control. For example *'I feel that my dietitian has provided me with choices and options.'* ⁹ Respondents answer each of the 15 questions using a 7-item response option (strongly agree to strongly disagree). The HCC has been used in a number of studies (Williams et al, 1996; Williams et al, 2004) with alphas ranging from 0.92 to 0.96, demonstrating high internal consistency of the questionnaire.

Shortened versions of the HCC have been developed by selecting items that most represent autonomy support from the 15-item questionnaire. These range from a 4-item scale, which produced an alpha of 0.75 (Williams et al, 1999), to a 7-item HCC giving an alpha of 0.85 (Kasser & Ryan, 1999). A 6-item version of the HCC was selected for use in this study, with an alpha of 0.82 (Williams et al, 2004), which is freely available (University of Rochester, 2008). The

⁹ The wording in the questionnaire was changed from "doctor" to "dietitian" for this study

questionnaire is scored by summing the scores and dividing by the total number of questions in the HCC questionnaire. High average scores represent high levels of patient-perceived autonomy support.

7.4.1.3 The Medical Interview Satisfaction Questionnaire (MISS)

The Medical Interview Satisfaction Scale (MISS; Rose et al 2002; Wolf et al 1978) is a 29 item guestionnaire, using a 7 item response option (strongly disagree to strongly agree), which assesses the degree to which the individual was satisfied with the consultation. For example, 'I felt that the dietitian didn't take my problem seriously.⁷¹⁰ The 9-item subscale of the MISS, which focuses on the patient/provider relationship, was chosen for this study. This measures satisfaction with the patient/provider relationship, by focusing on perceived professional warmth and patients feelings of trust, confidence, and freedom to express themselves in the consultation setting. As such, it is a more refined marker of the process of patient satisfaction as it relates specifically to the patient provider relationship (Crow et al, 2002). This 9-item subscale of the MISS has a Cronbach alpha of .93 showing good reliability which is similar to the .96 Cronbach alpha for the complete 29-item MISS guestionnaire (Kinnersley et al, 1996). The questionnaire is scored by summing the scores and dividing by the total number of questions. High average scores represent high levels of patient satisfaction.

¹⁰ The wording in the questionnaire was changed from *"doctor"* "to "*dietitian* "the 9-item subscale of the MISS was obtained from Kinnersley et al, "A comparison of methods for measuring patient satisfaction with consultations in primary care" *Family Practice*, 1999, 41-51 by permission of Oxford University Press

7.4.1.4 Diabetes Self-efficacy Questionnaire (SE)

The diabetes self-efficacy questionnaire measures the patient's belief in their ability to manage their diabetes, and consisted of 8 items using a 10 point scale from not at all confident to totally confident, eg *"How confident do you feel that you can choose the appropriate foods to eat when you are hungry (for example snacks)?"* This scale has been developed by the Stanford Patient Education Research Centre and has an internal consistency reliability of 0.83 (Lorig et al, 2009; Stanford Patient Education Research Centre, 2009). The questionnaire is scored by summing the scores for each question and then dividing by the total number of questions. High average scores represent high levels of patient self-efficacy.

7.4.2 Ordering of Questionnaires in Patient Booklet

To reduce any potential bias that may arise from the order in which the questionnaires are presented in the patient booklet, a Latin square was employed (Field & Hole, 2003). This resulted in the three sets of questionnaires being ordered in six different ways (Table 7.18).

	Satisfaction (MISS)	Self–efficacy (SE)	Patient autonomy (HCC)
Questionnaire 1	1	2	3
Questionnaire 2	2	3	1
Questionnaire 3	3	1	2
Questionnaire 4	1	3	2
Questionnaire 5	3	2	1
Questionnaire 6	2	1	3

Table 7.18: Latin square used for ordering of questionnaires in patient booklet

Equal quantities of the six patient questionnaire booklets identified in Table 7.18 were taken to the three Trusts. The booklets were shuffled to ensure random selection of a questionnaire booklet for completion at the end of each clinic. This method helped to reduce bias from the questionnaire order and from the questionnaire layout. An example of a patient questionnaire booklet used in this study is provided in Appendix 2.5.

7.5 Power Calculation

There is little data on the relationship between empathy and agreement on reported decisions made in the literature. One small study of 30 medical consultations suggested a moderate correlation (r = 0.486) between empathy and agreement on reported decisions (Coughlan, 2005). The hypothesis for this study is directional in that higher levels of empathy are expected to result in greater agreement on reported decisions made in the consultation, suggesting the need for a one-tailed power calculation. Using the correlation coefficient from Coughlan and a one-tailed test, the sample size for 90% power was calculated as 37. This suggests that a sample size of 37 patients would be needed to detect a significant difference between agreement on reported decisions made and empathy in the consultation. However, data on drop-out rates from the Coughlan study were not reported and it is difficult to know therefore how many additional patients may be needed to ensure that an adequate sample size is obtained. Studies using a similar methodology have also failed to report drop-out rates, but do report variable recruitment rates of 55% to 80% (Parkin et al, 2006; Skinner et al, 2007). Given the range of uncertainty over drop-out rates, recruitment rates, quality of audio recordings and completed questionnaires, a larger sample size of between 80-100

participants was aimed for to ensure that an adequate sample size would be obtained.

7.6 Analysis of Questionnaires

The following section explores the analysis used for the questionnaires that were collected at the end of the consultation.

7.6.1 Quantitative Analysis of Questionnaires

The questionnaires used to measure patient autonomy, self-efficacy and satisfaction were scored and the scores for these measures were entered into SPSS v.17.0 for Windows (SPSS Inc., Chicago, IL, USA). Demographic data and information on consultation type (follow-up/new) for each consultation was also entered. Identification codes on the questionnaire booklets enabled patient and dietitian responses for each consultation to be matched prior to analysis using SPSS.

7.6.2 Transforming the Qualitative Open Question into a Quantitative Measure

The open question in the questionnaire booklet resulted in lists of decisions being reported by both dietitians and patients. The following section explains the process by which these decisions were converted into numerical codes to allow statistical analysis to be undertaken.

7.6.2.1 Decisions Reported on Questionnaires

To compare reported decisions made, each decision recorded on the

questionnaires was coded into a topic area using a coding frame (Swift, 1996).

This produced a list of 37 topic areas for reported decisions (Appendix 2.6).

Related topics were then grouped together under a themed heading. For example, decisions relating to sorting out tablets, insulin dose, injection sites, and insulin pumps were all placed under the heading 'Medication'. By grouping common themes together, the initial 37 topic headings were reduced to 13 key areas (Table 7.19)

	Final coding themes for decisions
1	Food
2	Weight management
3	Exercise/activity
4	Continue with diet no change
5	Administration
6	Food and drink monitoring diary
7	Education/support groups
8	Psychosocial
9	Alcohol
10	Medication
11	Hypoglycaemia treatment/management
12	Blood glucose management
13	Other problems

Table 7.19: Final coding themes used for identified decisions

Reported decisions relating to food were numerous and contributed to the bulk of the decisions made. To provide further insight into this large topic area, and to capture the richness of this data, the food theme was broken down into ten distinct sub-categories (see Table 7.20, full description in Appendix 2.7).

Food sub-category

1	Portion sizes
2	Fruit and vegetables
3	Fat
4	Carbohydrate counting/Glycaemic index (GI)
5	Snacks
6	Eating out
7	Sugar free drinks
8	Food labels
9	Salt
10	Drink water with meals

Table 7.20: Ten sub-categories identified for food theme

Initial coding identified 463 decisions from the patient and dietitian questionnaires. Codes were assigned to each and a second round of coding was conducted to check consistency of allocated codes to the reported decisions. This second round of coding identified the following discrepancies: Eleven decisions were coded that were subsequently felt to be statements of facts or descriptions of issues discussed rather than decisions made. For example, the following statement was originally coded as weight management "having weighed myself my weight had gone up about 3lbs since my last weigh in". However, this was judged to be a statement and not a decision. All statements of facts or descriptions of issues were removed from the list of decisions.

Thirteen additional decisions were coded, having been missed in the first round of coding. These decisions were reported in long sentences that contained one or more specific codes, eg "*I need to start reducing snacks or crisps in the evening and to eat a banana or apple instead.*" This would be coded twice in the food category as two specific decisions around food have been made: a reduction in snacks and an increase in fruit intake. These were coded according to the themes identified in Table 7.19 and Table 7.20 and added to the data set. The decision coding was then re-checked for a third time on the 465 decisions reported, to ensure consistent application of the themed headings to all reported decisions on the questionnaires. No further discrepancies were found.

The numerical codes for patient and dietitian-reported decisions were entered onto an Excel spreadsheet for each consultation. This allowed the total number of decisions reported under the themes to be collated. In addition, coding by numbers allowed direct comparison of reported decision topics for patients and dietitians. The total number of patient and dietitian-reported decisions for each consultation was entered into SPSS for analysis.

Agreement on reported decisions was calculated using the formula shown in Table 7.21

 $\frac{patient \ and \ dietitian \ agreement \ on \ reported \ decisions \ (PD)}{(patient \ reported \ decisions - \ PD) + \ PD + (dietitian \ reported \ decisions - \ PD)} \ X \ 100$

= % agreement on reported decisions

Table 7.21: Formula for calculating agreement on reported decisions made in the consultation

Table 7.22 shows that two categories were created to record decision agreement for patient and dietitian: complete agreement on decisions made and partial agreement on decisions made (Goodchild et al, 2005; Parkin et al, 2006; Parkin & Skinner, 2003).

Complete agreement	Patient-reported decision	Dietitian-reported decision
Topics and specific content of decisions are the same	Putting spreading fat on only 1 slice of bread Trying to avoid any fat on meats	Reduce spreading fat, spread one side only of bread Cut off visible fat on meat
Partial agreement	Patient-reported decision	Dietitian-reported decision
Topics the same but the specifics differed or there was overlap between topics reported by patient and dietitian	More exercise	½ hour walk 5 days a week
	Fewer snacks	I O reduce discuits

Table 7.22: Coding categories for complete agreement on decisions made and partial agreement on decisions made

This method was applied to all decisions obtained to provide additional

differentiation of agreement categories and also entered onto SPSS.

7.7 Analysis of Audio Recordings

7.7.1 Extraction of Decisions and Generation of Quantitative Data Set

Consultations were digitally recorded with an Olympus VN-2100PC model voice

recorder and downloaded onto a PC using Olympus Digital Wave Player v2.1.4

software. The researcher listened to the audio files and identified any decisions made during the consultations and then entered them onto a spreadsheet. Decisions needed to include an action that either the patient or dietitian was to undertake. *"Maybe we should think about increasing your fibre intake"* was not considered a decision, as it did not provide a clear action of how this was to be achieved. However, the following examples would have been recorded as decisions as clear actions were provided; *"make ½ lb of butter last 2 weeks", "reduce semi-skimmed milk form 1½ pints to ¾ pint a day"* or *"include a bedtime snack after gin and tonic."*

To ensure reliability, a second coder independently extracted decisions from 9 (10%) of the audio recordings. There were no discrepancies in extracted decisions between the researcher and the second coder and it was therefore assumed that the decisions extracted by the researcher from the remaining audio recordings were accurate.

The coding themes identified in Tables 7.19 and 7.20 were then applied to all extracted decisions from the audio recordings. The final codes allocated to the extracted decisions were entered onto an Excel spreadsheet to allow comparison between decision topics reported on questionnaires and decision topics recorded in the consultations. Data on number of decisions recorded per consultation were also entered onto SPSS.

7.7.2 Coding for Empathy Using the Empathic Communication Coding System (ECCS)

Following completion of the questionnaire analysis, the audio recordings were analysed for empathy. All recordings were coded for empathy using the Empathic Communication Coding System (ECCS) (Bylund & Makoul, 2005) incorporating the minor changes suggested in Chapter Five. This system provides a quantitative measure of behavioural empathy in the consultation. The ECCS codes the type of empathic statement made by patients and the dietitian's response to these statements. The first part of the coding measures the presence of three types of empathic statements (statement of emotion, statement of progress, statement of challenge) provided by the patient in the consultation (see Appendix 1.2). Each recording was examined to determine the timing of occurrence and type of empathic statement made by patients. This process was then repeated to allow coding of the dietitian's response to the empathic statements into one of six hierarchical categories, as illustrated in table 7.23. (Full ECCS coding definitions and examples are in Appendix 2.8.)

	Description of category
6	Statement of shared feeling or experience
5	Confirmation
4	Acknowledgement with pursuit
3	Acknowledgement
2	Minimal encouragers
1	Implicit recognition of patient perspective
0	Denial of patient perspective

 Table 7.23: Coding system for empathic responses made to patient empathic statements (ECCS)

7.7.2.1 Reliability of application of the Empathic Communication Coding System (ECCS)

Fifteen (17%) of the audio recordings were randomly selected to check the reliability of the application of the ECCS coding. The researcher coded the first 5 minutes of these audio recordings using the ECCS and then a second coder repeated the coding independently. Cohen's kappa was used to test for agreement between coders, as this method corrects for chance and reduces bias involved when looking at agreement and disagreement scores alone (Bakeman & Gottman, 1997). The time intervals of occurrence and non-occurrence of empathic statements over the first 5 minutes were compared for agreement. The time segments were broken into seconds (300 seconds) to aid correct identification of occurrence of empathic statements. This check gave a Cohen kappa of 0.75 for agreement of occurrence of empathic statements made in the consultations.

Once the coding for occurrence of empathic statements had been checked, Cohen's Kappa was used again to check for agreement on the codes assigned to the empathic statements by the two coders. This gave a Cohen kappa of 0.76 for agreement on assigned codes to empathic statements.

Following these reliability checks, the remaining audio recordings were coded with the amended version of the ECCS (Appendix2.8). Data was collected on the total number and type of empathic statements made in the consultations and entered onto SPSS. The mean level of empathy for each consultation was calculated by dividing the sum of the empathic responses by the total number of empathic opportunities made in the consultation (Bylund & Makoul, 2002). This method was also used to compute the mean level of empathy for each empathic

component (statement of emotion, statement of progress and statement of challenge) per consultation. The mean level of empathy for each empathic statement made, as well as the mean level of empathy for the consultation as a whole was also entered onto SPSS.

7.7.3 Screening for Interference in the Interpretation of Empathy

There was a concern that other factors may interfere with the interpretation of the correlation data between empathy and agreement on reported decisions made (Silverman, 2006). Immediacy of information is known to affect recall (Watson & McKinstry, 2009). Greater levels of agreement on reported decisions made may be influenced by the provision of action plans or repetition of decisions at the end of the consultation, rather than indicating the presence of empathy. To test for this effect, all consultations that recapped decisions and/or set action plans in the last 5 minutes of the consultation were identified. These consultations were compared to the remaining consultations where these behaviours did not occur. This process highlighted differences in the presence of empathy and the level of agreement on reported decisions made in the two data sets allowing interpretation of level of interference to be made. These findings are presented alongside the Phase I data on empathy in Chapter Eight.

7.8 Ethics

Ethical guidance for this study was obtained via the National Research Ethics Service (NRES) (2007) and granted by the Gloucester Ethics Committee. The research and development departments within the three Trusts provided final approval for the study to be undertaken within their hospitals. Ethical approval was required as the study was based within the National Health Service and

patient interactions with dietitians were to be explored. This research raised potential issues with regard to patient confidentiality and in order to comply with such issues all audio recordings and questionnaires were coded to ensure anonymity. Although dietitians collected data on patient gender, age, duration of diabetes and presence of complications the data collected did not contain any personal identifiable information such as names, addresses, or hospital numbers thereby ensuring anonymity and confidentiality. All digital audio recordings were downloaded onto a password-protected PC immediately after the consultation, conforming to the Data Protection Act (1998) applying to the storage of personal information.

7.9 Summary

This chapter has identified the methods used for recruiting dietitians and patients into the study and the methods of data collection employed. The measures for patient satisfaction, patient and professional autonomy and patient self-efficacy and empathy have been detailed. The retrieval of decisions from the questionnaires and audio recordings have also been described, including the processes put in place to check the reliability of the methods used for identifying and coding decisions made.

The method used for coding empathy from the consultation recordings has been described, as has the process used to check the reliability of the coding system employed. Additional checks were also carried out on the data collected, to screen for possible interference when exploring empathy and agreement on reported decisions made. The results obtained from the quantitative analysis of the data are presented in Chapter Eight.

CHAPTER EIGHT: RESULTS FOR PHASE I

Phase I of this study relates to the quantitative analysis of the data collected and explores the level of patient and dietitian agreement on reported decisions made in the consultation, noting the key topic areas in which decisions were made. Decisions extracted from the recorded consultations are compared to decision topics reported by patients and dietitians on questionnaires.

The relationship between agreement on reported decisions made and empathy, patient satisfaction, patient self-efficacy and patient autonomy is explored, along with the effect that the empathic statements of emotion, progress and challenge have on these factors. Patient and dietitian perception of autonomy support provided in the consultation is compared.

The effect of patient demographic data on agreement is also discussed. These include the impact on levels of recall of: age (Jansen et al, 2008a; McDonald-Miszczak, 2005), gender, and health status (Kiviniemi & Rothman, 2006) and duration of diabetes (Rost et al, 1990). Gender status is also reported to impact on empathy measures (Coughlan, 2005; Roter et al, 2002) as have the presence of complications, duration of diabetes and length of consultation (Heisler et al, 2002). The impact of these additional factors also needs to be assessed when undertaking the quantitative analysis of the data.

8.0 Statistical Analysis

The statistical tests used in this study are detailed in the appropriate sections of the following analysis. Data that reports mean values will include the standard error of the mean as per the American Psychological Association guidance (Field & Hole, 2003). All tests performed will report results calculated for two-tailed tests (unless stated otherwise) and the level of probability of significance used will be set at p < .05.

8.1 Results

The following section presents the descriptive statistics generated on reported and recorded decisions from consultations.

8.1.2 Decision Topics

Table 8.24 lists decision topics obtained from patient and dietitian questionnaires and audio recordings of consultations. The top three decision topics identified by patients and dietitians were the same: food, administration (eg appointments, letters etc) and exercise. A full description of decision topics is provided in Appendix 2.6.

Decision Topics	Patient- reported decisions	Dietitian- reported decisions	Decisions from audio recordings
Food	79	122	157
Administration (appointments etc)	25	39	43
Exercise	24	27	27
Weight	4	8	6
Continue / no change	18	19	18
Medication	11	12	14
Monitoring of food and drink intake	8	11	13
Education	9	13	10
Hypoglycaemia	6	4	9
Blood glucose monitoring	2	3	8
Psychosocial	3	6	5
Alcohol	4	4	6
Other	3	1	5
Total	196	269	321

Table 8.24: Summary of decision topics reported on patient and dietitian questionnaires and decision topics taken from audio recordings

Food decisions accounted for (157/321) 49% of the total decisions made in the consultations, and in order to provide further refinement to the analysis, this group was broken down into the sub-categories identified in Table 8.25.

Patient-reported decisions	Dietitian-reported decisions	Decisions from audio recordings
Portions	Portions	Portions
Fat/snacks	Fat/Fruit and vegetables	Fat
Eating out	Snacks	Fruit and Vegetables
Fruit and Vegetables	Eating out	Snacks
Carbohydrate/glycaemic index	Carbohydrate/glycaemic index	Eating out
Drink water with meals	Drink water with meals	Carbohydrate/glycaemic index
Salt	Salt/sugar free drinks	Drink water with meals
	Food labels	Sugar-free drinks
		Salt/food labels

Table 8.25: Food sub-categories ranked in order of occurrence

Food portion sizes were the most frequent food decision reported in the consultations, and Table 8.25 highlights differences in patient and dietitian reporting of food decisions made. To test for significance in the variation seen in reported and recorded food decisions, one-way ANOVA was conducted. The differences seen in fruit and vegetable decisions were found to be significant, with dietitians reporting significantly more fruit and vegetable decisions than were reported by patients or extracted from the recorded consultations, F(2, 251) = 5.190, p = .006.

8.1.3 Level of agreement on reported decisions made in the consultation

The 86 questionnaires analysed recorded 196 decisions by patients and 269 decisions by dietitans. Of these, 142 were reported by both parties (72% were patient-reported decisions, 53% were dietitian-reported decisions). Overall

agreement between patients and dietitians on reported decisions made was assessed to give an overall patient/dietitian agreement rate on reported decisions made of 44% (see Table 7.21, Chapter Seven for the full explanation).

Complete disagreement on all decisions reported by patients and dietitians occurred in 10 of the 86 consultations (12%). *'Complete disagreement'* meant that none of the topics or details of decisions reported by patient or dietitian on the questionnaires matched.

The 142 decisions reported by patients and dietitians were examined further to explore the level of agreement. The agreement code was broken into two categories; complete agreement (where the topic and detail reported by both parties was the same), and partial agreement (where the topics were the same but the specifics reported differed between parties). 31% of the reported decisions achieved partial agreement and results are presented in table 8.26

	Decisions reported by patients and dietitians	Percentage Agreement
Complete agreement on reported decisions	98/142	69%
Partial agreement on reported decisions	44/142	31%

Table 8.26: Patient and dietitian agreement on decisions reported by both parties

Decisions reported on questionnaires and on audio recordings were compared for each consultation. This analysis highlighted a number of decisions that were not made in the consultation: 37 (11.5%) for patients and 50 (15.5%) for dietitians.

Diifferences in the mean number of reported decisions made in the consultations were also analysed (Table 8.27), with dietitians reporting more decisions (3.13) being made in the consultations than patients (2.28). Wilcoxon signed rank test was used to explore this difference and found it to be significant (z = -4.42, p < .05).

	Mean number of decisions per
	consultation (SEM)
Decisions made per consultation (extracted from recordings)	3.73 (0.19), range 0-8
Patient-report of decisions	2.28 (1.57), range 0-5
Dietitian-report of decisions	3.13 (0.17), range 1-12
Patient and dietitian agreement on reported decisions	1.65 (0.11), range 0-5

Table 8.27: Mean number of reported decisions per consultation

The findings highlighted above indicate that patients and dietitians report few of the decisions made in the consultation, with patients reporting fewer decisions than dietitians. Patient and dietitian agreement on reported decisions accounts for less than half of the decisions made in the consultation. There are also differences in the reported decision topics, which may reflect the differing agendas of patients and dietitians. Patients and dietitians reported decisions that were not made in the consultation, and this may be further evidence of their differing agendas.

8.2 Empathy

The following section presents the descriptive statistics generated for the measure of empathy.

A total of 857 empathic opportunities occurred in the 86 consultations recorded. Dietitians missed only four of these empathic opportunities during delivery, and these were three statements of emotion and one statement of progress. The mean number of empathic opportunities provided by patients was 9.97 (SEM 0.57) with a range of 0-32 empathic statements per consultation. The level of empathic response to these by dietitians is illustrated in table 8.28.

Level	Empathy description	<i>n</i> (N=853)	Percentage
6	Shared feeling or experience	0	0
5	Confirmation	53	6.2%
4	Acknowledgement with pursuit	195	22.9%
3	Acknowledgement	235	27.5%
2	Minimal encouragers	252	29.5%
1	Implicit	96	11.3%
0	Denial	22	2.6%

Table 8.28: Dietitian responses to patients' empathic opportunities

The most frequent response to patients' empathic opportunities were *acknowledgement* and *minimal encouragers*. The response by dietitians to empathic opportunities raised by patients did not include *sharing feelings* or *experiences*; therefore no codes were allocated to level six.

The mean level of empathy was calculated for each empathic statement by taking the sum of empathic responses and dividing by the number of empathic

statements per consultation (Bylund & Makoul, 2002). The level of empathy was found to vary according to the type of empathic statement that patients made. The highest level of empathy was recorded for statements of challenge (2.84), with a mean level of empathy of 2.77, as shown in Table 8.29.

Empathic opportunities	Number of empathic statements made	Range of empathic statements made per consultation	Mean empathic response provided by dietitian
Statement of Emotion (SE)	147	0-12	1.33 (SEM 0.15)
Statement of Progress (SP)	249	0-10	2.33 (SEM 0.12)
Statement of Challenge (SC)	457	0-22	2.84 (SEM 0.10)
Total number of empathic opportunities presented (EOP)	853	0-32	2.77 (SEM 0.06)

Table 8.29: Mean level of response by dietitians to empathic statements made by patients

These findings demonstrate that patients present a varied number of empathic opportunities in the consultation. Dietitians mainly respond to these empathic opportunities with *minimal encouragers, acknowledgement,* and *acknowledgement with pursuit*. Dietitians are more empathic when responding to statements of challenge then statements of emotion.

8.2.1 Impact of Empathy on Consultation Outcomes

The following section provides the data on the inferential statistics used to

explore empathy. All correlations were conducted using Kendall's au coefficient,

independent t-test was performed on the continuous variables for the demographic data set. Independent t-test was adjusted for equal or unequal variances based on Levene's test. If Levene's test was significant then values were recalculated using Mann-Whitney *U*-test.

8.2.2 Does Greater Empathy Result in Greater Agreement?

Correlations were undertaken using a one-tailed test to explore the relationship between empathy and agreement on reported decisions made. There was a significant correlation between empathy and patient and dietitian agreement on reported decisions made in the consultation ($\tau = .283$, p = .0005). When dietitians are more empathic, greater levels of agreement on reported decisions occur as a result, and this finding supports the first hypothesis.

8.2.3 The effect of decision recapping / action planning at the end of the consultation

It was important to check that the correlation seen between empathy and agreement did not occur as a result of the dietitian using decision recapping or action planning in the consultation.

The last 5 minutes of the audio recordings were reviewed, and this resulted in the identification of 45 consultations where decision recapping and action planning occurred. These were compared to the remaining 41 consultations, where this process was absent. Independent t-test showed significant differences between the two groups for agreement on reported decisions made, t(84), = 2.74, *p* = .007. Consultations that used action planning or decision recapping at the end of the consultation achieved significantly greater levels of

agreement on reported decisions made. Also, the level of empathy demonstrated was significantly greater U = 530.500, N₁ = 1, N₂ = 45, p = .001, see Table 8.30.

	Mean agreement on reported decisions	Median value for empathy
Action plan/decision recap	1.93 (SEM .157)	2.92 (range 1.9)
No action plan/ decision recap	1.34 (SEM .147)	2.60 (range 3.5)

Table 8.30: Mean and median values for empathy and agreement on reported decisions made in consultations with and without action planning/decision recapping

These findings suggest that dietitians who are more empathic are also using decision recapping and/or action planning in their consultations, which is associated with greater levels of agreement on reported decisions made.

8.2.4 Impact of the Various Components of Empathy on Agreement

Correlations were conducted to explore the relationship between empathic components and agreement on reported decisions and number of decisions made.

High levels of empathy to statements of challenge were found to correlate significantly with agreement on reported decisions made ($\tau = .221$, p = .008). The number of patient decisions reported correlated with the dietitian's level of empathy ($\tau = .216$, p = .008), and in particular the dietitian's empathic response to statements of challenge ($\tau = .210$, p = .011).

The number of decisions reported by the dietitian was correlated to the dietitian's level of empathy (τ = .228, p = .005), and in particular to statements of emotion (τ = .231, p = .008) and statements of challenge (τ = .180, p = .028).

These findings suggest that responding to empathic statements resulted in dietitians reporting that more decisions were being made. When dietitians respond empathically to statements of challenge, patients and dietitians report a greater number of decisions being made and achieve greater agreement on reported decisions made in the consultation.

8. 2.5 Can the relationship between empathy and agreement be explained by differences within the consultations?

Due to the small number of dietitians used in this study, it was important to check that the correlations seen between empathy and agreement on reported decisions made did not occur as a result of a cluster effect from one individual - resulting in skewed data. Multiple regression analysis was therefore conducted to determine the differences within each of the dietitian's consultations for empathic responses and agreement on reported decisions. The first step was to create dummy variables for dietitian 1, 2 and 3, allowing the independent variable of dietitian to be entered as a dichotomous variable in the regression equation (Calder & Sapsford, 1996). Product variables (dietitian x empathy) were incorporated into the model to allow for the possibility of slope differences between the dietitians (Calder & Sapsford, 1996). A hierarchical method was used as follows: Block 1 dummy variables for dietitians were entered, followed by Block 2 - the mean empathic response to statements of challenge (as this had a significant correlation with agreement on reported decisions), followed by

Block 3 product variables (dietitian x empathy). The dependent variable used for this regression was patient and dietitian agreement on reported decisions made. A significant model emerged: F(4, 81) = 6.902, p < .0005. This model explains 21.7% of the variance (Adjusted $R^2 = .217$). Table 8.31 gives information for the predictor variables that were included in the final model.

Patient and dietitian agreement on reported decisions made in the			
consultation			
В	SE B	β	p value
-0.87	-0.31	-0.37	p = .006**
-0.12	0.30	-0.05	р = .691
0.04	0.31	0.02	<i>p</i> = .902
0.37	0.12	0.32	p = .002**
Adjusted R^2 for block 1 = 0.132Adjusted R^2 for block 2 = 0.217Adjusted R^2 for block 3 = 0.208		x 1 = 0.132 x 2 = 0.217 x 3 = 0.208	
	Patier repo B -0.87 -0.12 0.04 0.37	Patient and dia reported dec conBSE B-0.87-0.31-0.120.300.040.310.370.12Adjusted Adjusted Adjusted	Patient and dietitian agree reported decisions matrix consultation B SE B β -0.87 -0.31 -0.37 -0.12 0.30 -0.05 0.04 0.31 0.02 0.37 0.12 0.32 Adjusted R ² for block Adjusted R ² for b

Table 8.31: Influence of individual dietitians on level of empathic response to statements of challenge and impact on level of agreement on reported decisions made

The hierarchical model shows that Block 3, which contained the product variables (dietitian x empathy), did not lead to a significant improvement in R

square (R^2 = .208). This was not therefore included in the final model (see

Table 8.31), since the relationship between empathy and agreement on

reported decisions did not vary across dietitians. Within each dietetic

consultation where there was greater empathy, there was greater agreement on

reported decisions made.

8.2.5.1 Differences Between Dietitians

Further analysis was conducted using one-way ANOVA to explore the differences identified between dietitians on levels of agreement on reported decisions made. The difference was significant, F (3,82) = 5.310, p = .002, as was the difference between dietitians for the level of empathy, F (3,82) = 4.351, p = .007. Error bar charts illustrate the variability of the means for level of agreement (Figure 8.8) and level of empathy (Figure 8.9) for each dietitian.



Figure 8.8: Agreement on reported decisions for individual dietitians



Figure 8.9: Mean level of empathy for individual dietitians

These findings suggest significant differences in the practice of dietitians in consultations. Independent t-test confirmed that the differences between dietitian 1 and dietitian 3 were significant for agreement on reported decisions made (t = -3.386(df43), p =.002) and for differences in expressed empathy (t = -3.16 (df43), p = .003).

In comparison to dietitian 1, Dietitian 3 expressed significantly more empathy in her consultations and achieved significantly greater levels of agreement on reported decisions made.

The differences in practice were explored further to identify the different responses provided by each dietitian to empathic opportunities offered by patients in the consultation. The results are presented in Table 8.32.

Level	Empathy description	Dietitian	Dietitian	Dietitian	Dietitian
		1	2	3	4
6	Shared	0	0	0	0
5	Confirmation	16	10	13	12
4	Acknowledgement with pursuit	32	49	63	49
3	Acknowledgement	57	67	59	50
2	Minimal encouragers	74	65	58	61
1	Implicit recognition	40	28	7	20
0	Denial	10	6	2	5
	Total	229	225	202	197

Table 8.32: Frequency of ECCS codes used for responses by dietitians to patient empathic opportunities

Dietitian 3 used more *acknowledgement with pursuit* in response to empathic opportunities, whereas in contrast dietitian 1 used more *minimal encouragers*. Compared to the other dietitians, dietitian 1 used a greater number of *implicit recognition* to patients' empathic opportunities.

One-way ANOVA was conducted to examine these differences in empathic responses. There were significant differences in level 1 responses (*implicit recognition*), F(3,82)=5.651, p = .001, and level 4 responses (*acknowledgement with pursuit*), F(3,82) = 4.826, p = .004. Differences between dietitians for the remaining levels (0, 2, 3 and 5) were not significant: p > .05.

These findings highlight significant differences in the way that the dietitians in this study respond to empathic opportunities, suggesting differences in the way that dietitians practice. Differences were also found in the average number of decisions reported by the dietitians (Table 8.33). The mean scores for agreement on reported decisions, empathy, level of empathy for different empathic statements and number of decisions for each dietitian are presented below.

	Dietitian 1	Dietitian 2	Dietitian 3	Dietitian 4
	(<i>n</i> = 22)	(<i>n</i> = 26)	(<i>n</i> = 23)	(<i>n</i> = 15)
Number of decisions recorded in consultation	2.95 (0.42)	3.62 (0.31)	4.43 (0.35)	4.00 (0.39)
Patient-reported decisions	1.86 (0.24)	2.19 (0.22)	2.65 (0.21)	2.47 (0.24)
Dietitian-reported decisions	2.36 (0.29)	2.92 (0.18)	3.57 (0.23)	3.93 (0.64)
Agreement on reported decisions	1.00 (0.20)	1.65 (0.17)	2.09 (0.25)	1.93 (0.21)
Level of empathy	2.49 (0.16)	2.74 (0.09)	3.07 (0.09)	2.80 (0.10)
Empathic response to statements of emotion	1.48 (0.27)	1.77 (0.29)	0.65 (0.23)	1.40 (0.40)
Empathic response to statements of challenge	2.75 (0.22)	2.50 (0.96)	3.24 (0.14)	2.93 (0.17)

Table 8.33: Mean scores (SEM) for agreement on reported decisions, empathy, level of empathy for different empathic statements and number of decisions recorded for individual dietitians

Dietitian 1 set the fewest number of decisions per consultation (2.95 decisions) and had the lowest agreement on reported decisions made (31%). Dietitian 3 set the highest number of decisions per consultation (4.43 decisions) and had the highest level of agreement on reported decisions made (51%). This

difference in agreement on reported decisions (Figure 8.8) is therefore not a result of the number of decisions set in the consultation.

These findings suggest that higher levels of empathy in the consultation, particularly in response to statements of challenge, result in more decisions being made and recalled by patients. There are significant differences in the way that dietitians practice resulting in differences in expressed empathy, recall, and agreement on reported decisions made. These differences may reflect different communication skill sets used by the dietitians. Qualitative analysis will allow expansion of these results to provide insight into the communication processes that occur in consultations that may account for these differences.

8.2.6 Empathy and patient satisfaction

Correlations were conducted to explore the relationship between empathy and satisfaction using a 1-tailed test to test the hypothesis that higher levels of empathy would result in greater levels of patient satisfaction. In this study, empathy did not correlate to satisfaction ($\tau = -.032$, p = .340). However, a correlation was found between the total number of empathic statements of challenge made by the patient and patient satisfaction ($\tau = .246$, p = .023).

8.3 Mediating Variables (patient autonomy and self-efficacy)

The following section describes the correlations conducted to explore the relationship between patient perception of autonomy and self-efficacy with agreement on reported decisions made. The relationship between autonomy and self-efficacy with empathy was also explored.

Patient autonomy (τ = -.070, p = .417) and self-efficacy (τ = -.144, p = .081) did not correlate with patient and dietitian agreement on reported decisions made. In addition, patient autonomy (τ = .019, p = .806) and self-efficacy (τ = .012, p = .875) did not correlate with empathy.

In this study, patient autonomy and self-efficacy did not correlate with empathy or with agreement on reported decisions.

Correlations were also conducted to explore patient and dietitian perception of autonomy support. There was no correlation between dietitian autonomy and patient autonomy ($\tau = 0.014 \ p = .858$). Patient and dietitian perception of autonomy support in the consultation differed and this further highlights the importance of observational data to examine communication skills used in consultations.

8.4 Moderating variables (demographic data and length of consultation)

The following section explores the data on patient demographics and situational variables (length of consultation) as these may influence the levels of empathy and agreement on reported decisions made. Data is only presented for findings that were found to be significant (p < .05).

8.4.1 Presence of complications and duration of diabetes on agreement

Patients with complications reported significantly more decisions being made in the consultation t(84) = 2.199, p = .031. The final level of agreement on reported decisions made was also significantly greater, t(84) = 2.509, p = .014 (Table 8.34).

	Presence of	No complications	
	Complications <i>n</i> = 36	<i>n</i> = 50	
	Mean (SEM)	Mean (SEM)	
Patient-reported decisions	2.58 (0.19)	2.06 (0.15)	
Patient/dietitian agreement on reported decisions	1.97 (0.19)	1.42 (0.13)	

 Table 8.34: Mean number of decisions and complications

Duration of diabetes correlated significantly to levels of complete agreement on reported decisions made ($\tau = 0.251$, p = .003).

These findings suggest that the duration of diabetes and the presence of complications resulting from diabetes are additional factors that need to be considered when reviewing consultation outcomes. These factors significantly influenced the level of agreement on reported decisions made.

8.4.2 Length of consultation

Length of consultation was explored to ascertain what impact this had on the level of agreement on reported decisions made. Consultation length was found to correlate to agreement on reported decisions made ($\tau = .206$, p = .012). Consultation length also correlated to the number of decisions made in the consultation ($\tau = .312$, p < .0005) both for patient-reported decisions ($\tau = .182$, p = .026) and dietitian-reported decisions ($\tau = .245$, p = 0.02).

Consultation length is pre-determined by appointment type, as new patients are usually allocated longer consultation times than for follow-up appointments. In this study, new patient consultations were generally twice as long as those for follow up. Figure 8.10 illustrates the relationship between mean consultation
time for new/follow-up appointments and the level of agreement on reported

decisions made.



Figure 8.10: Mean length of consultation for new and follow-up appointments and level of agreement on reported decisions.

Figure 8.10 indicates that new and follow-up appointments require a minimum length of time before decisions are made and therefore before agreement on reported decisions can occur.

New patient appointments show an increasing number of decisions as consultation length increases. Although agreement on four reported decisions was achieved in less time (38.82 minutes) than that needed to achieve agreement on three reported decisions (42.90 minutes), this only occurred in two consultations and therefore cannot be considered as a normal occurrence. The same is true for agreement on five decisions found in only one consultation. Interestingly, in follow-up appointments, agreement on three reported decisions was achieved in less time (20.69 minutes) than that needed to achieve agreement on two reported decisions (26.45 minutes). Again this was seen in only a small number 5/29 (17%) of the follow-up appointments. A larger patient sample with a greater number of professionals would allow this pattern in agreement and consultation length to be explored more fully.

Length of consultation was correlated with the total number of statements of challenge ($\tau = .380, p < .0005$) as well as the total number of empathic opportunities provided by the patient ($\tau = .306, p < .0005$). There was no significant difference in the number of empathic opportunities presented in new appointments compared to follow-up appointments p > .05.

Longer consultations for both new and follow-up appointments were associated with more empathic statements from the patient, particularly statements of challenge. Longer consultations also resulted in greater agreement on reported decisions made, with duration of diabetes and/or the presence of complications significantly impacting on agreement. This suggests that patients with a longer duration of diabetes and complications are presenting more statements of challenge, resulting in longer consultations, which lead to greater agreement on reported decisions made.

8.4 Conclusion

The quantitative analysis resulted in findings that support the first hypothesis, which states that higher empathy is associated with greater levels of agreement on reported decisions made in the consultation.

The second hypothesis tested stated that higher empathy is associated with greater levels of patient satisfaction. Unfortunately, the data does not support this hypothesis.

Patient autonomy and self-efficacy did not correlate with empathy and agreement on reported decisions made in the consultation. However, the following variables: presence of complications, duration of diabetes and length of consultation were found to significantly impact on the level of agreement on reported decisions made.

The consultation length was also found to correlate with the number of empathic opportunities presented by the patient.

Significant differences were found between dietitians regarding the level of expressed empathy and the resulting level of agreement on reported decisions made, suggesting significant differences in the way that dietitians practise.

Full discussion of the data collected in this chapter will occur in Chapter Twelve.

PHASE II

Qualitative Methods and Analysis

CHAPTER NINE: INTRODUCTION TO PHASE II

Phase II is a qualitative exploration of a sub-sample of the embedded data collected in Phase I. Analysis of Phase I data enabled identification and selection of consultations where complete agreement and complete disagreement on reported decisions made between patients and dietitians occurred. This selection enabled an exploratory comparison of the communication skills and information-giving exchanges within the consultations and their subsequent impact on the level of agreement on reported decisions made. This chapter describes the qualitative method chosen for Phase II of this study.

9.0 Qualitative analysis

The qualitative analysis involves exploration of consultations to identify specific communication skills and behaviours that occurred in the consultations. Firstly, this identification allows comparison of these behaviours to be made between those consultations with agreement on reported decisions made and those with disagreement. To enable this comparison to occur, the data is presented as frequencies to illustrate occurrence within the consultations (Boeije, 2010; Boyatzis, 1998; Hancock, Windridge & Ockleford, 2007; Street et al, 1993). Secondly, although this data is presented quantitatively to allow comparison, it is also explored qualitatively to support the development of themes for the information-giving exchanges that occurred in the consultations (Joffe & Yardley, 2004).

Patient-centred communication skills are crucial for the expression of empathy (Egan, 2002; Neumann et al, 2009; Ong et al, 1995) and may influence the way in which that information is delivered in the consultation (Neumann et al, 2009). In addition, lack of active-listening skills may limit expressions of empathy, thereby impacting on patient recall as discussed in Chapter Three. Figure 9.11 illustrates a small part of the proposed model of interaction between empathy, communication skills, and information exchange, with outcomes as suggested by Neumann et al (2009). The links between both short-term and long-term outcomes are hypothesised links in this model.



Figure 9.11: Model illustrating the interaction between empathy, communication skills and information exchange (adapted from Neumann et al, 2009)⁸

¹¹ Reprinted from Neumann et al (2009) 'Analysing the "nature" and "specific effectiveness" of clinical empathy: A theoretical overview and contribution towards a theory-based research agenda', *Patient Education & Counseling*, 74 pp.339-346 with permission from Elsevier

9.1 Thematic analysis

A thematic approach will be employed to explore the information-giving exchanges that occurred in the consultations. This method was chosen as it provides a flexible approach to data analysis resulting in a summary of key features from the data set (Boyatzis, 1998; Braun & Clarke, 2006; Jupp, 2006). The flexible approach to analysis allows attention to be paid to the qualitative aspects of the material as it is analysed (Joffe & Yardley, 2004).

Theoretical thematic analysis uses a directed approach to data analysis, which means that theory is used to guide the selection of initial codes applied in analysis (Braun & Clarke, 2006; Corbin & Strauss, 2008; Creswell, 2009; Tuckett, 2005). For this study, the patient-centred communication skills identified in Chapter Three were coded to provide direction for the development of information-giving themes. The interpretative analysis of the coded data collected on communication skills and behaviours was used to guide the development of these information-giving themes (Braun & Clarke, 2006; Joffe & Yardley, 2004). Codes selected for initial theme development included: open and closed questions, professional requests for information from the patient, patient provision of unsolicited information, patient questions, statements and requests for information, patient concerns and queries.

In addition the dietitian's use of reflections, clarifications, paraphrasing and summarising were coded as well as supportive and confirming statements. The provisional coding scheme employed and the final coding scheme used to develop the themes for information-giving exchanges will be explored fully in Chapter Ten.

9.1.1 Approach taken to thematic analysis

A five-phase thematic approach was applied to develop the themes for

information-giving exchanges (Braun & Clarke, 2006), as outlined in Table 9.35.

Phase	
1.	Familiarisation with the data
2.	Generating initial codes
	(informed from initial pilot work and literature review)
3.	Searching for themes
4.	Reviewing themes
5.	Defining and naming themes

Table 9.35: Phases of thematic analysis (adapted from Braun & Clarke, 2006)

Application of these phases is not linear and involves movement backwards and forwards through the sequence of phases. This done in order to check reasoning, application, and relevance of the codes applied as analysis evolves and is further developed over time (Ryan & Bernard, 2003; Silverman, 2006). This movement backwards and forwards through the phases is called a 'recursive' process (Braun & Clarke, 2006).

The application of codes will enable the development of information-giving themes within the process of information-giving exchange that occurs in the consultations. The 'recursive' process allows the coder to check, search and recheck allocated codes and development of themes. Rigorous checking ensures consistency in the application of codes and themes in the data set.

The reliability of themes applied to the information-giving exchanges was assisted by the development of clear definitions, documented in a coding manual (Appendix 3.1). Validity of coding of themes was checked through a process of peer debriefing (Creswell, 2009). A second coder was used to review the qualitative study data and theme development and then to check the themes allocated to the information-giving exchanges.

9.1.2 Sequential map

To further understand the relationship between empathy and information-giving exchanges, a sequential map was employed (Chapter Eleven, Figure11.18). The map illustrates the flow of information-giving themes and empathic statements made in those consultations with agreement on reported decisions and those with disagreement. This provides additional insight into the possible mechanisms by which these variables interact with each other and their subsequent impact on the consultation outcome of agreement on reported decisions made.

9.2 Summary

The literature reviewed in Chapter Three suggests that the presence of active listening skills will support the professional's expressions of empathy because it demonstrates understanding of the patient's situation. Greater understanding of the patient's perspective through the use of active listening skills and empathy leads to information exchange that is more relevant and personal. The use of theoretical thematic analysis allows theory to inform the selection of codes chosen for analysis. The presence of active-listening skills and empathic opportunities in the consultations are used to guide the theme development (phase 3) for the information-giving exchanges, using the five-phase approach illustrated in Table 9.35. To reduce the risk of researcher bias in coding and interpretation of themes, data on agreement on reported decisions made was

kept separate during the coding process. Agreement coding was added back into the data set on completion of qualitative analysis, to allow comparison to occur.

The detailed methodology used for the qualitative analysis of Phase II applying this thematic approach is provided in the next chapter.

CHAPTER TEN: QUALITATIVE METHOD PHASE II

This chapter describes the qualitative analysis employed for Phase II of the study and explains the identification and selection of data from Phase I for this purpose. The application of thematic analysis to this data set as outlined in Chapter Nine is explored, detailing the steps taken to identify codes from the transcripts and the process of development of the information-giving themes.

The aim of the qualitative analysis is to identify how information is delivered in the consultation and to explore how this relates to empathy and agreement on reported decisions made in the consultation. The qualitative analysis attempts to answer the following questions:

Research Questions:

1. What are the components of information giving in the consultation?

2. How do the components of information giving relate to empathy?

3. How does information giving relate to agreement on reported decisions made in the consultation?

4. How do the components of information giving differ between consultations with agreement on reported decisions made and those with disagreement?

10.0 Data collection

The following section describes how the data for Phase II analysis was extracted from the Phase I data set.

10.1 Identification of qualitative data embedded in Phase 1

Ten of the 86 consultations analysed in Phase I were identified as having complete disagreement on reported decisions made between patient and dietitian. Complete disagreement is defined as being when the content and detail of all decisions reported by patient and dietitian for a consultation do not match (Goodchild, Skinner & Parkin, 2005; Parkin et al, 2006; Parkin & Skinner, 2003).

Forty-three consultations were identified in Phase I as having complete agreement on all reported decisions made. (The method for selection of decisions with complete agreement was outlined in Chapter Seven.) Ten of these consultations were randomly selected to provide a contrasting data set for analysis. Analysis of the two data sets enables comparison between information processes that occur in consultations and their subsequent impact on level of agreement on reported decisions made by patients and dietitians.

As a result, 20 audio recordings were selected from the Phase I data. The size of the sample chosen for qualitative analysis was determined in part by the limited number of audio recordings obtained that featured complete disagreement on reported decisions. However, 20 is considered to be an adequate sample size on which to undertake qualitative analysis (Green & Thorogood, 2009).

The 20 recordings selected for analysis included consultations from all four of the participating dietitians. The majority of recordings identified with disagreement on reported decisions came from dietitian 1, whereas recordings

with disagreement were not identified for dietitian 4, which suggests differences in the practice of the participating dietitians.

A summary of the number of recordings selected from participating dietitians is provided in Table 10.36, and this will be explored further in the discussion.

Dietitian	Number of recordings with complete agreement on reported decisions	Number of recordings with complete disagreement on reported decisions
1	2	7
2	3	2
3	4	1
4	1	0

Table 10.36: Summary of recordings selected for analysis from participatingdietitians where there was complete agreement and complete disagreement onreported decisions made

Selecting audio-recordings to represent the extremes of the consultation outcome (complete agreement versus complete disagreement) enables the relationship between the presence and /or absence of identified skills/behaviours and subsequent impact on agreement on reported decisions made to be examined. Clear evidence based guidance on the specific skills/behaviours needed to improve levels of agreement on reported decisions made can therefore be provided.

Approximately a third of patients 25/86 (29%) attended dietetic consultations with a partner and this had no significant effect on empathy or level of agreement on reported decisions made (p > .05). Consequently consultations which included a third party were not excluded from the qualitative analysis.

10.2. Statistical analysis of sample

Situational variables were compared between the two data sets and included: consultation length, whether new or follow-up appointments, gender, age, type of diabetes, duration of diabetes.

Statistical tests were conducted to check for differences in the measured variables between the data sets as follows:

A chi-square test was conducted on the nominal and ordinal data (gender, new/follow-up appointments, presence of complications, type of diabetes). An independent t-test was performed on the continuous variables (age, duration of diabetes, empathy and length of consultation). Independent t-test was adjusted for equal or unequal variances based on Levene's test. If Levene's test was significant, then values were recalculated using the Mann-Whitney *U*-test. All tests produced report results calculated for two-tailed tests and the level of probability of significance used was p < .05. The results of the statistical analysis are presented in Table 10.37.

	Recordings with complete agreement on reported decisions (<i>n</i> = 10)	Recordings with complete disagreement on reported decisions (<i>n</i> = 10)	p values
Appointment type (New)	9	6	.12
Gender (female)	5	2	.16
Mean age (years)	60	55.2	.34
Type of diabetes (type1)	7	4	.17
Duration of diabetes (years)	Median 25 (2months-48years)	Median 7 (1month-17 years)	**.01
Complications	6	3	.17
Length of Consultation (SEM)	32.24 (4.7)	21.31(2.64)	^a .058
Mean empathic response (SEM)	3.00 (0.12)	2.13 (0.31)	**.02
Statements of challenge	5.4	2.1	^a .055
Statements of emotion	2.1	1.2	.34
Statements of progress	1.8	2.8	.22

*significant at p < .05

^aalmost significant reflecting trend in main data set

Table 10.37: Demographic data, appointment type, consultation length and empathy scores for consultations selected for qualitative analysis

As can be seen from Table 10.37, consultations with agreement on reported

decisions had significantly higher levels of empathy, p = .02.

Trends were seen towards significance on the number of statements of

challenge and the length of the consultation. The failure to reach significance

with these measures suggests that the differences measured were not very

large and this is a consequence of the small sample size used for the qualitative

analysis (Brace et al, 2006). However, the trends towards significance that are

highlighted reflect the significant findings detailed in Chapter Eight for the full data set of consultations.

In addition, this analysis identified significant differences between the data sets on duration of diabetes p = .01. Patients showing disagreement on reported decisions made in consultations had a diabetes duration of 7 years (or less). Patients showing agreement on reported decisions had a diabetes duration of 25 years (or more). The median duration of diabetes varies considerably from the 10.87 years seen in the full data set, and these extreme differences in duration will therefore need to be considered when interpreting the findings from the qualitative data analysis.

10.3 Method for thematic analysis

The 20 audio recordings were transcribed and the thematic analysis was conducted using a five-phase approach as outlined in Chapter Nine (Table 9.35) (Braun & Clarke, 2006). The application of this approach is detailed below:-

10.3.1 Familiarisation with data

The recordings were listened to while coding to help clarify tonal qualities and the interpretation of identified themes (Creswell, 1998). Tonal qualities of patient's voices can provide additional information with regard to their emotional state, thus adding a different emphasis to the words and phrases used by the individual (Del Piccolo et al, 2005; Gable, 2007). It was essential, therefore, that recordings were listened to alongside the transcripts to check the transcripts for accuracy (Braun & Clarke, 2006). Notes were made in the margins of the hard copies of the transcripts highlighting interesting issues as they arose. A separate list of these issues was made to help direct later analysis.

Before coding began each transcript was numbered sequentially for units of speech, a unit of speech is defined as an uninterrupted episode of speech by either the dietitian or the patient, and may be one word long or may comprise of several sentences. It begins when a person starts speaking and ends when they have finished speaking (Del Piccolo et al, 2005). The numbering of these speech units enabled tracking and retrieval of identified features in the transcripts.

10.3.2 Coding for communication skills

Coding of distinct communication behaviours was conducted on all transcripts and was informed by the literature reviewed in Chapter Three and the process of initial listening to and familiarisation with the recordings. The transcripts were read and assigned preliminary communication codes. This process involved movement backwards and forwards during the coding process to ensure that

codes were allocated consistently on all transcripts. In order to check code assignment and to amend codes where necessary, the audio recordings were listened to again whilst simultaneously following the transcripts.

The following communication behaviours were coded; active listening skills, (reflection, clarification, summarising and paraphrasing, confirming statements and supportive statements). The dietitian's questions were coded if they specifically asked about patient's perceptions, thoughts, ideas, opinions, and feelings or explored the patient's self-care management practices. All of these skills have been identified in Chapter Three as important for developing effective dialogue with patients (Gable, 2007; Peel et al, 2005; Poskiparta et al, 2001; Rosal et al, 2001; Vickery & Hodges, 1986; Zimmermann, Del Piccolo & Finset, 2007). In addition, empathic opportunities were marked on the transcripts as per the ECCS discussed in Chapter Five (Bylund & Makoul, 2005).

10.3.3 Identification of information-giving themes

Once transcripts had been coded for communication behaviours, they were marked to indicate episodes of information-giving exchanges that focused on patients' health behaviour. The length of these information-giving exchanges varied and depended on the topic being discussed. They consisted of sequential utterances from either the patient or the dietitian about the same topic and finished when the topic under discussion was changed (Kiuru et al, 2004).

These episodes of information-giving exchange were then reviewed to identify potential themes. The communication skills that had been coded onto the hard copy of the transcripts helped this process, by providing markers around the information-giving exchange. This guided the development of potential themes for these information-giving exchanges (Braun & Clarke, 2006). Themes were initially identified by exploring whether the patient or the dietitian led the information-giving exchanges. The patient-initiated episodes of information exchange were defined in terms of how the patient may have triggered the need for information, either directly, by requesting information, or indirectly through statements and cues and by offering information. The dietitian defined the dietitian-led episodes, and offered information that was either not requested by the patient, or occurred as a result of questioning. The initial coding map used to identify information-giving episodes is provided in Figure 10.12.



Figure 10.12: Initial coding map used to identify information-giving exchanges within the consultations

The process of initial coding with this scheme highlighted difficulties as it was found to be too simplistic. The themes used for the information-giving exchanges lacked context as prior questions or responses may have facilitated the subsequent flow of information (Anderson, 1981). For example the professional may be facilitating the discussion by exploring the patient's understanding, and consequently will be using questions to enable this information exchange to occur (Holli et al, 2003). Using open and exploratory questions encourages information giving from the patient and in this case the dietitian initiates the information exchange. This type of exchange differs from a professional-led, advice-giving episode, because the patient is supported and encouraged in the exchange of information; whereas professional-led exchanges are more likely to contain episodes of one-directional communication with limited patient input. Although the professional instigates both episodes of information exchange, the intent of the information provided is different. Patients have been reported to be reluctant to offer information or request help, unless prompted to do so (Barry et al, 2000). This supports the potentially facilitating nature of questions for information provision. The coding process is not simply a matter of recording who asks the questions, but must also consider the intention of the questions and how the information is handled once requested, or how the need for information is identified. Information-giving exchanges therefore need to be explored in terms of how the dietitian delivers the information and how the patient received it. This is further influenced by the potential reason for the information exchange, the level of involvement of the patient in the subsequent dialogue, as well as the dietitian's role within this information exchange.

To aid the development and identification of themes for the information-giving exchanges, four transcribed scripts were selected for provisional coding: two transcripts that had complete agreement on reported decisions made and two that had disagreement. These transcripts were reviewed to see how the provisional coding themes for information-giving exchanges worked in practice and to help identify additional codes or sub-themes that would allow a more thorough view of the allocated themes. This also enabled the flow of the conversation to be assessed, adding further meaning to the selected themes

(Silverman, 1985). The information-giving exchanges identified were collected together within a framework. This highlighted how information was provided, received, the impact on conversation flow and the intent of advice in order to help with identification and naming of themes (Bryman, 2008).

Once these coding themes had been established, the remaining 16 transcripts were reviewed and coded. In order to ensure that coding bias did not occur, data for these remaining transcripts was kept separate from data used to identify the agreement on reported decisions made.

As all 20 recordings were worked through, provisional themes were built up around the information-giving exchanges, final codes developed and concepts explored more fully. The identified information-giving exchanges were colourcoded on the electronic copies of the transcripts, according to the theme allocated. These helped to visualise the data and enabled quick identification and retrieval of coded themes for further analysis (Creswell, 1998).

10.3.4 Reviewing identified themes

Information-giving exchanges were collated on spreadsheets according to their assigned colour codes to allow further analysis and refinement of provisional coded themes. It was important to ensure that a little of the surrounding data was collected with the information-giving exchanges to help contextualise the identified themes (Braun & Clarke, 2006). This process of reviewing provisional identified themes helped to ensure consistency and clarity within the identified themes (Braun & Clarke, 2006).

Initially seven themes for the information-giving exchanges were identified within the consultations. These were given provisional names of confirming (permitting), persuading, supporting, teaching advice, teaching collaborative, teaching information general and provision of instructions. However, by exploring common features and areas of overlap, these seven themes were condensed down to four key themes. These four key themes are similar to the dietary advice giving styles identified in a study of Finnish nurses (Kiuru et al, 2004). Table 10.38 has been adapted from Kiuru et al (2004) to illustrate the key constructs used for the themes allocated to the information-giving exchanges.

	Persuasive	Supportive /	Confirming /	Recommending /
	Information	collaborative	permitting	teaching / instructions
How information is provided	Little or no understanding of patient's perspective could be judgemental	Patient's understanding and views sought before information given Supportive feedback provided	Response to patient's queries or statements checking information	Information, suggestions, recommendations provided by dietitian
How information is received	Met with resistance /Justification	Met with further collaborative exchange	Patient confirms information received	Minimal response from patient
Patient participation	Minimal input from the patient, or patient justifies actions, defends course of action taken	Generally patient talks more in these exchanges, active engagement Patient provides additional personal/ relevant information	Met with minimal input from patient, or patient confirms that they have understood the information received	Minimal input from the patient, clarifying questions occasionally used, passive response from patient
Dietitians role in information exchange (Intent of information)	Aimed at getting patient to follow advice persuasive	Aimed at facilitating patient understanding and ability to change, personalised advice provided	Aimed at confirming, permitting patient's thoughts, actions, general information.	Aimed at informing patient of correct course of action to take Specific advice or instruction given encourage patient to consider recommendations or suggestions offered.
Outcome	No clear decision or course of action made	Collaborative decision more likely	Confirmation of behaviour/ information	Clear instructions provided by dietitian

Table 10.38: Key features of the four identified themes for coding informationgiving exchanges adapted from Kiuru et al (2004) ¹²

¹² Reprinted from Kiuru et al (2004) 'Advice-giving Styles by Finish Nurses in Dietary Counseling Concerning Type 2 Diabetes Care' *Journal of Health Communication*, 9 pp. 337-354 with permission from Elsevier

10.3.5 Defining and naming themes

To improve the rigour in the qualitative analysis, a second coder checked the four provisional coding themes. Where discrepancies arose, the two coders went back to the original transcript to discuss the rationale for inclusion of the information-giving exchange within the stated theme. These discussions helped to clarify the distinctions between the themes identified for the information-giving exchanges and to ensure that rationale used for coding was consistently applied.

A coding manual was developed to define and illustrate codes used in developing the themes and to clarify the final definitions of themes used for the information-giving exchanges. The manual featured examples selected from the transcripts to serve as reference points for coding (Braun & Clarke, 2006). The coding manual can be found in Appendix 3.1. The final themes for the information-giving exchanges were named according to their dominant conversational characteristic (Kiuru et al, 2004). These were *persuasive*, *supportive/collaborative*, *recommending/teaching*, and *permission/confirming*.

Following discussions between the two coders, a number of revisions were subsequently made to the coding manual. All codes allocated to the identified information-giving exchanges in the 20 transcripts were then rechecked in order to ensure that the final criteria for themes defined in the coding manual were consistently applied to all the information-giving exchanges.

A total of 282 information-giving exchanges were identified from the transcripts. The four themes identified for the information-giving exchanges were seen to

alternate in the consultations. Others have noted this effect (Kiuru et al, 2004). All of the consultations analysed contained at least three of the four themes identified for the information-giving exchanges.

10.4 Comparison of data from consultations with agreement on reported decisions made and those with disagreement

The frequency of coded communication behaviours and information-exchange themes in consultations were summarised and data on agreement and disagreement on reported decisions made was applied to this data set. This will allow direct comparison of similarities and differences between those consultations with agreement on reported decisions made and those without.

Frequencies were chosen as a measure of the communicative behaviours and information exchanges in the consultations, this allows the differences in presence of skills and information-giving exchanges in the data sets to be compared. In addition, as there are two people in the consultation, the rapport established between the patient and the professional may determine the length of the consultation (Street et al, 1993). The measurement of the frequency of skills and information exchanges used thus provides more meaningful data on the skill set used in the consultation, as suggested by Street et al (1993).

10.5 Sequencing of information-giving exchanges and empathic opportunities in consultations

The colour-coded information-giving exchanges were entered onto an Excel spreadsheet for each consultation in order of occurrence within the consultation. This provided an overview of the flow of information-giving exchanges within the consultations. Data on the occurrence of empathic opportunities within the consultations was added to this sequential map (Bylund & Makoul, 2002; Bylund & Makoul, 2005). In addition, consultations with agreement on reported decisions made and those with disagreement are highlighted.

This thematic map provides a visual display of the sequence of informationgiving exchanges and occurrence of empathic statements within the consultations. It does not provide information on timing or length of the individual information exchanges. It allows the relationship between the different information-giving exchanges and empathic opportunities to be explored, as well as their impact on agreement on reported decisions made (Braun & Clarke, 2006).

10.4 Summary

This chapter has described the coding process used for the qualitative analysis. It explains how the embedded data for the qualitative analysis was extracted from the Phase I data and the process of coding of communication behaviours that occurred in the 20 selected transcripts. The identification of these communication behaviours provided additional markers to assist the development of themes around information-giving exchanges in the consultations. The process of theme identification and development has been described. Four key themes around information-giving exchanges were finally defined in relation to how information was provided and received, the level of patient participation and the overall intent of the advice that was provided. The presence of empathic statements made by patients was explored alongside the information-giving exchanges and communication behaviours. The findings from this analysis are presented in the next chapter, along with excerpts from

the consultations that serve to illustrate the identified themes for the information-giving exchanges that occurred within the consultations.

CHAPTER ELEVEN: RESULTS FROM PHASE II QUALITATIVE ANALYSIS

This chapter presents the data from the qualitative analysis of the 20 selected consultations, as described in Chapter Ten. It is usual to start to critique and interpret qualitative analysis as findings are presented (Creswell, 1998) and this can involve some discussion occurring in the results section. However, to ensure that this write-up conforms to the mixed method approach (Creswell, 2009) outlined in Chapter Six, the qualitative data will be explored and analysed alongside the quantitative data set. The analytical discussion of the qualitative and quantitative data sets will occur in Chapter Twelve.

The communication skills identified in the analysed consultations are not exhaustive, but have been chosen as key skills (highlighted from the literature reviewed in Chapter Three) that are important for patient-centred communication and for the development of the themes for information-giving exchanges. These communication skills have been quantified to illustrate the frequency of occurrence in consultations with agreement on decisions made and those with disagreement.

There follows a qualitative description of the identified themes for the information-giving exchanges. After this, there is a quantitative summary of the frequency of occurrence of identified themes in consultations, which differentiates between those with agreement on decisions made and those with disagreement.

Information on the sequencing of information-giving exchanges, alongside the presence of empathic statements, provides further unique insight into the

relationship between information-giving exchange and empathy in the consultation. This is the first study to undertake this type of analysis in order to explore dietetic consultations.

The exploration of this data set provides answers to the research questions outlined in the methods section in Chapter Ten.

11.0 Exploration of Key Communication Skills

The key communication behaviours identified have been quantified in terms of frequency of occurrence within the consultations. Full examples of the communication skills coded from the qualitative data are provided in the coding manual in Appendix 3.1. The coding of communication behaviours in this way allows comparison between those consultations with agreement on reported decisions made, and those with disagreement.

11.1 Is there a difference in the use of active listening skills in consultations with agreement on reported decisions made and those with disagreement?

Active-listening skills are essential for an empathic dialogue to occur (Egan,

2002), and Table 11.39 shows the active-listening skills that were identified and

coded from the consultations.

	Consultations with agreement on reported decisions made	Consultations with disagreement on reported decisions made
Reflections and clarifications	133	91
Summarising and paraphrasing	39	18
Supportive statements	35	29
Confirming statements	22	8
Total frequency	229	146

Table 11.39: Frequency of occurrence of active listening skills used in consultations

Reflections and clarifications occurred mainly in the information-gathering phase of the consultation. They appeared frequently and accounted for 224/375 (60%) of the total active-listening skills recorded. Summarising and paraphrasing were less evident, but were twice as likely to occur in those consultations where agreement on reported decisions occurred 57/375 (15%). Summarising and paraphrasing are used when reflecting large chunks of the dialogue back in order to check patient understanding. In consultations where this skill was absent, the patients generally talked very little, so the dietitian did not need to paraphrase chunks of the conversation. There were nearly three times as many confirming statements in consultations where there was agreement on reported decisions made.

Although there were differences seen in the frequency of active-listening skills used in consultations with agreement and those with disagreement, the Mann-Whitney test found these differences to be non-significant p > .05.

Overall, in consultations where there was complete agreement on reported decisions made, there was a greater use of active-listening skills, as indicated by the presence of paraphrasing, summarising, and confirming statements.

11.1.2 Is there a difference in the use of exploratory questions in consultations with agreement on reported decisions made and those with disagreement?

Patients with a chronic disease rely heavily on their own knowledge, skill base, and experience of living with the disease, which is a unique personal experience rather than a standardised experience (Thorne, Paterson & Russell, 2003). Consequently, when working with individuals with a chronic illness, it is important to gauge their perspective and understanding (Anderson & Funnell, 2005b; Epstein et al, 2005; Kinnersley et al, 1996; Lewin et al, 2001). This may give additional clues as to how patients use information to manage their condition on a daily basis (Lang, Floyd & Beine, 2000; Shillitoe, 1994). The dietitian would therefore be expected to gather information on the patient's perspectives, thoughts, and feelings in relation to issues raised during the consultation, in order to personalise and shape decisions made. Questions that specifically asked how patients thought or felt, or queried their current management practices were recorded from the consultations and these are summarised in Table 11.40

	Consultations	Consultations
	with agreement	with
	on reported	disagreement on
	decisions made	reported
		decisions made
Number of consultations where	10	4
these questions were asked		
Exploratory Questions		
Questions that explored		•
patient's thoughts, feelings,	29	8
ideas, opinions		
Questions that explored		
patient's self-care	11	1
management		
Total questions	40	9

Table 11.40: Frequency of patient-specific questions used in consultations

The Mann-Whitney test was conducted to test for significance. A significantly greater number of questions were asked about self-care management practices in consultations where there was agreement on reported decisions made. (U = 18.500, p = .007) However, the difference seen in the number of questions that explored the patient's thoughts, feelings ideas and opinions was non-significant p > .05.

11.1.3 Is there a difference in agenda setting in consultations with agreement on reported decisions made and those with disagreement?

Agenda setting helps to provide context and direction for the consultation. It is a facilitative process that suggests greater patient involvement, as patient views are sought at the beginning of the consultation (Berg-Smith et al, 1999; Feste & Anderson, 1995; Gable, 2007; Glasgow, Emont & Miller, 2006; Holli, Calabrese et al, 2003; Parkin, 2003; Rollnick et al, 1999). It may be an additional factor that influences patient and dietitian dialogue during the consultation. Table 11.41 illustrates the number of consultations where an agenda was set with patients during the opening stages of the consultation.

	Consultations with agreement on reported decisions made	Consultations with disagreement on reported decisions made
Number of consultations where an agenda was set.	8	1

Table 11.41: Number of consultations that set an agenda with the patient

In consultations where disagreement on reported decisions occurred, there was either a lack of a clear agenda at the start of the consultation or a lack of patient involvement in the setting of the agenda.

In summary, consultations with agreement on reported decisions made had a greater use of active listening skills, supportive and confirming statements and significantly more exploratory questions that explored patient self-care management practice. In addition, these consultations were more likely to set an agenda with input from the patient at the start of the consultation. Full examples of active-listening skills and questions coded can be found in the coding manual in the appendix (Appendix 3.1).

11.2 Exploration of the four identified information-giving themes in the consultations

The following section provides a detailed description of the four informationgiving themes identified from the 20 transcribed consultations. Excerpts from the coded transcripts are provided to illustrate key points.

11.2.1 Persuasive information-giving exchanges

Persuasive information-giving exchanges were clearly identified as advicegiving episodes led by the professionals, with the dietitian often offering information to the patient that was not requested. Before providing this information, the dietitian does not question the patient to explore how they think or feel about the situation under discussion or inquire as to their self-care management practices. There was little or no understanding therefore of the patient's point of view or perspective of the issue being discussed. The information provided by the dietitian was aimed at getting the patient to follow the advice, even though in some cases the patient had made it very clear that they did not wish to follow the advice provided. In an attempt to get the patient to take on board the advice, there were a few occasions where the *persuasive* information contained threats as to the long-term consequences of not changing. For example:

"It would help that, wouldn't it? Obviously you've had this by-pass so obviously you want to try and keep the blood vessels as clear as possible, don't you, so the walking would help that."

In this example, the dietitian is using health risks to try and coerce the patient to change rather than presenting the information in a more neutral manner (Rollnick, 1996).

Sometimes the dietitian's response was judgemental, in that it informed the patient of "what they should do". For example, the patient in the following instance has volunteered information about a dietary indiscretion and explained that this indiscretion did not impact on their blood glucose levels. Rather than exploring this situation with the patient to see what they have learnt from the
experience (Kolbe, 2002) in order to help the patient develop their problemsolving and self-management skills (King et al, 2002), the dietitian continues with a judgemental response:

"It's not going to be your best choice though because there will be quite a lot of sugar."

Provision of information in a persuasive manner was met by a number of responses. Either the patient responded by justifying the action they were taking; *"I only had one, not every day, just every now and then"* or they would present further resistance to the information provided, as shown by the following example:

Dietitian: "No but a couple of prunes would be good..."

Patient: "But they don't make no difference to my bowels."

In this case, the dietitian did not enquire as to how the patient currently manages the situation before providing this information. This may explain why the information is met with resistance by the patient, as the information did not fit with their understanding or previous experience of the situation.

Persuasive information-giving exchanges were more likely to follow on from recommending information-giving exchanges and previous *persuasive* information-giving exchanges. This suggests that when the patient's prior response to a recommendation from the dietitian is met with resistance, the dietitian may then attempt to persuade the patient that this recommendation is the right thing to do (Kiuru et al, 2004). In Figure 11.13 (Excerpt 1) the patient's partner is resisting attempts to change, and it is the patient that terminates this information exchange by focusing on something that the dietitian has said. By

questioning the advice provided, the patient successfully moves the topic away

from the area being covered and this information-giving exchange is terminated.

Excerpt 1:

Dietitian: "What about, have you tried any of the drinks like peppermint tea which is good for your digestion?"

Patient's partner: "He doesn't drink tea."

Dietitian: "No. It's not normal tea ... "

Patient's partner: "I still don't think he'd drink it."

Dietitian: "I just wondered if it might be worth a try, it's just that it's quite good for your digestion, so it might help with feeling – peppermint is very good."

Patient's partner: "Peppermint tea."

Dietitian: "It's just like herbal tea but just with peppermint leaves. You put a little bag in a cup and pour hot water in – it's just a hot drink with a peppermint flavour. You don't add milk, it's like a thin drink and it's quite good for your digestion. It's the sort of thing you might not have tried or ever thought about trying. Have you ever tried it?"

Patient's partner: "No, he doesn't like anything like that so we wouldn't have tried it."

Dietitian: "No, if I had a bag I could have given you one to try. It might be worth trying and it's only a suggestion, you don't have to try it, it's entirely up to you. I'm just thinking in terms of helping with the digestion as well and something that wouldn't be sugary and whether that would help at night time particularly. Coffee's fine, you can have as much as you like but I was trying to think if there was an alternative..."

Patient: "I thought too much coffee wasn't good for you?"

Excerpt 2:

Dietitian: "So, I think my only concern would be the amount of chocolate that you're eating, I know you're giving extra insulin for it but there's still quite a lot of fat and things in that."

Patient: "Yes I suppose so."

Dietitian: "Do you think one Snickers would be reasonable?"

Patient: "No, 2 I need because I need the physical food."

Dietitian: "Yes."

Patient: "Yes, they're OK I only normally have it in the morning or in the afternoon, sometimes..."

Dietitian: "Sometimes in the evening you might..."

Patient: "Yes but if you think about it I don't have a pudding do I, so?"

Dietitian: "No, you're not having a pudding are you."

Patient: "No and so it's a substitute as a pudding."

Dietitian: "Yes, OK, right."

Figure 11.13: Examples of *persuasive* information-giving exchanges

In Excerpt 2 (Figure 11.13) the dietitian is raising the issue of a reduction in chocolate consumption. This issue has already been raised earlier in the consultation where it was met with resistance by the patient. The patient justifies why it is ok to eat chocolate, and the exchange ends with the dietitian agreeing with the patient's point of view and moving on to another topic. Even though the dietitian agreed at that point, they brought the topic up again later in the consultation. This topic was raised on four separate occasions in the consultation and each time the patient clearly indicated no interest or willingness to change their consumption of chocolate. In fact, the patient was so adamant that this was not a change that they wish to make that they responded by writing on the questionnaire booklet *"I do not want to give up chocolate!!!!"*

These episodes of persuasion and resistance are terminated by either the dietitian or the patient changing the topic and are characterised by the lack of clear agreement at the end of the exchange (Kiuru et al, 2004). Lack of

understanding of the patient's perspective and current behaviour may be impacting on the dietitian's ability to engage effectively with the patient in these instances. This lack of insight may explain the reluctance and resistance with which the information is received (Rollnick et al,1999). In attempting to get the patient to change, the resulting lack of choices and the persuasive nature adopted in these exchanges suggests low provision of autonomy support from the dietitian.

The distinctive feature of *persuasive* information-giving exchange in the qualitative analysis was that information was offered with little or no understanding of the patient's perspective. This was characterised by lack of questioning by the dietitian to explore patient's perspective. Alternatively, the patient may have provided information that the dietitian failed to follow through as in the following example:

Patient: *"I cringe and I don't like needles"*. Dietitian: *"No, OK. If we just run through a typical day, what do you have for your breakfast?"*

In *persuasive* exchanges, the information received was generally met with resistance by the patient and justification offered for continuation of the behaviour. The dietitian would then terminate *persuasive* exchanges and move on to another topic of conversation, or minimal responses from the patient would lead to the patient changing the topic.

11.2.2 Supportive/collaborative information-giving exchanges

This type of information-giving exchange engaged the patient through facilitated discussion (Entwistle et al, 2008; Ong et al, 1995). These exchanges may have

occurred as a result of the patient asking a question, raising a query, or making a statement. They can, in the first instance, be followed by a simple reflection from the dietitian and would have been coded and picked up in the section looking at counselling skills. These simple reflections would be supportive and involve some form of praise, *"Well done, you've obviously worked hard on that...."* In a supportive way, the dietitian also handled relapses and difficulties encountered by patients. For example, a patient has explained a number of issues that have got in the way of their day-to-day management and may have impacted on blood glucose control. The dietitian summarised this by acknowledging what they have heard *"It sounds like you've had a lot to cope with..." "I noted 4 or 5 times you mentioned ..."* before going on to explore these issues and possible management solutions with the patient. Reflection and summarising were therefore key communication skills used in *supportive/collaborative* information-giving exchanges.

These exchanges may also have occurred as a result of a question that the dietitian had asked which seeks to explore the patient's understanding, thoughts, and feelings around a particular issue. For example; "Do you feel that you could eat more? " or "OK. So having told me about your eating habits, do you feel thinking about it in more detail that there's any particular things that might need changing?"

These open questions are facilitative in that they encourage patient participation and help to clarify information. The questions are used to support patient understanding and may result in provision of additional information to help guide choices made. Inevitably, these collaborative exchanges involved the patient talking more and providing information that was personally relevant. This wasn't necessarily demonstrated with long monologues from the patient, but could be illustrated through continuous exchange between the patient and the dietitian, thus illustrating a more fluid discourse. Information exchange was seen to be supportive as it followed the patient's agenda and supported their actions, ideas, and perception of control. This collaborative nature is illustrated in the guided reflection and questioning used by the dietitian. Supportive exchanges were distinct from other exchanges, in that they provided supportive feedback (either directly as reflection) or in the supportive nature of the response, which would be incorporated into the feedback to the patient (Kiuru et al, 2004).

In Excerpt 3 (Figure 11.14) the patient raises the issue of being hungry in the evening. The dietitian follows this through by exploring the patient's management of the issue of hunger in the evening. In this way, the dietitian ascertains how the patient is currently handling the situation, before providing further information. In this case, the dietitian ends the exchange with a summary of the patient's management, the supportive elements of which are wound into the feedback acknowledging and, at the same time, supporting the patient's actions.

Excerpt 3:

Patient: "I feel quite hungry at night."

Dietitian: "OK and that's sometimes when you have your biscuits?"

Patient: "Yes, I feel like I need a bit more but not ... "

Dietitian: "OK and do you find that you're OK when you say no or does it still niggle the hunger or does it go away?"

Patient: "Whenever I feel hungry at night when I'm at college I go for a walk or a run or something."

Dietitian: "And when you come back?"

Patient: "I'm too knackered so I go to bed!"

Dietitian: "OK and that's good because sometimes when we're really, really hungry that hunger doesn't go away and at that point you really need to eat something. If you're feeling like you might want to eat something and you go and do something else instead like a walk or the things you were talking about and you're not that bothered when you get back then that's a good sign that you probably weren't that hungry or in need of food in the first instance. So it's good that you use what we would call a distraction technique to sort of do something else instead to sort of test the water but if you really are feeling hungry then there's no reason why you shouldn't have something to eat."

Figure 11.14: Examples of *supportive/collaborative* information-giving exchanges

In Excerpt 4 (Figure 11.15), through a facilitative discussion, the dietitian is

trying to help the patient to clarify areas in their diet that require more attention.

By encouraging the patient to be specific about the identified problem, the

dietitian has helped the patient to focus and in doing so the patient has provided

additional information to clarify why certain behaviours occur.

Excerpt 4:

Dietitian: "OK, so thinking about your diet in a little bit more detail, if you were to summarise how you think your diet is generally..."

Patient: "Rubbish."

Dietitian: "OK and are there particular parts of that which are worse than others or particular problem areas that you think you have identified – there might not be..."

Patient: "Gosh, I'm not a great breakfast eater ... "

Dietitian: "So you don't always have breakfast ... "

Patient: "Of course that advert that says people who eat a proper breakfast maintain a proper kind of weight (laughing), I do believe it, I'm not saying it's not true..."

Dietitian: "So you don't always have breakfast or you don't ever have breakfast?"

Patient: "I probably, I'd say 4 to 5 mornings a week I would have breakfast ... "

Dietitian: "OK."

Patient: *"It varies from a bowl of cereal to probably a couple of biscuits dunked in my tea."*

Dietitian: "OK."

Patient: "It depends how I am in the morning because I, for some reason over the last couple of years, I produce a lot of acid or something. I get indigestion and in the morning I have got bile that's coming up, my heads in the sink nearly every morning and the last thing I want to do is eat something."

Figure 11.15: Examples of *supportive/collaborative* information-giving exchanges

Consultations that had agreement on reported decisions were more likely to

have longer exchanges attached to the supportive statements. These

statements tended to elaborate on issues raised by patients by providing

additional information that supported the patient's choices, opinion, or actions.

11.2.3 Confirming/permitting information-giving exchanges

Confirming/permitting information-giving exchanges were always patient-led and would arise as a result of a patient query or patient statement. For example:

Patient: "I'm not worried about that because sometimes I think it should be left to 7."

Dietitian: "Yes. It's not too late, some people leave it a lot later than that..."

As can be seen in the above example, the confirming response is a simple confirmation that does not go on to clarify or explore further the issue raised by the patient. In some ways, these exchanges can be considered as asking the dietitian for permission to continue a particular behaviour (Kiuru et al, 2004).

The majority of these exchanges resulted in minimal responses from the patients, as they were confirming thoughts, behaviours, or actions presented to the dietitian. Included in this theme are exchanges that give the patient permission to behave in a certain way.

For example, in Excerpt 5 (Figure 11.16) the patient has asked whether it is ok to eat cream cakes. Having clarified that it is fine to eat cream cakes occasionally, the dietitian proceeds to clarify the permission given with the statement, *"your scones and things are best..."* As can be seen from this example, these information-giving exchanges end with a clear confirming or permissive response as to what can or cannot be done.

In Excerpt 6 (Figure 11.16), the dietitian confirms what the patient has said but does not go on to explore why the patient only has two meals per day. The

permitting tone is seen in the response *"obviously it is healthy for all of us to ..."* So although at first glance this may be seen as supportive, the lack of exploration and clarification and the suggestion that *"it would be better for you"* clearly defines this as a permitting information exchange.

Excerpt 5:

Patient: "But nothing like cream cakes that I can have, you know some of its got cream in the middle, but I couldn't eat that, could I?"

Dietitian: "That would just be a real treat, it's alright to have something now and again as long as it's not forming the basis of your diet."

Patient: "Like for someone's birthday or something?"

Dietitian: "Exactly, exactly."

Patient: "But not every week, right, that's alright."

Dietitian: "I mean your scones and things are the best for regular things."

Patient: "Yes, yes."

Dietitian: "Cream cakes for treats."

Patient: "That's alright, as long as I know."

Excerpt 6:

Patient: "Yes, when I'm at University I tend to, well for the last year I've been dieting so I've only been on 2 meals a day. Should I be eating 3? I've been avoiding one or skipping one..."

Dietitian: "Erm, the benefit of the sort of insulin that you're on, the Humalog, that you have with meals is that if you choose not to have a meal then you don't take that. Obviously it is healthy for all of us to have 3 meals or 5 small, moderate meals per day rather than if we miss a meal we then tend to have a big evening meal or something, you know. So you do have that flexibility..."

Figure 11.16: Examples of *confirming/permitting* information-giving exchanges

Confirming/permitting information-giving exchanges generally result in little response from the patients, particularly when it follows a question that they have raised. However, when these responses are linked with additional information provision, resistant statements may arise. For example, the dietitian has summarised key areas of diet that the patient has changed, confirming that the changes are appropriate. At the end of the summary they add:

"as lean as possible, any meat really, cut off any visible fat – I think that's key really. Visible fat – get rid of it and don't even contemplate it." The patient's response to this is resistant *"You don't even see any fat in tongue."* This resistance may indicate the patient's perceived ability to manage this aspect of their diet (Kiuru et al, 2004).

This theme also included general information provision in response to a patient query or statement, for example:

Patient: "Can you use a sunflower butter or something?"

Dietitian: "You can get these olive oil type margarines which have got around 60% fat and you can still use them in baking..."

In summary, episodes of *confirmation/permitting* occur as a result of patient queries or statements. The information provided by the dietitian confirms the patient's observations, thoughts, behaviours, or actions or provides permission to act in a certain way. These episodes of information-giving are likely to arise as a result of a *supportive/collaborative* information-giving exchange, suggesting that the patient is feeling comfortable and able to ask for specific information from the dietitian. Alternatively, they may occur as a result of a

previous *recommending* information-giving exchange and may have been used as a way of changing the topic within that exchange.

11.2.4 Recommending/teaching/instruction information-giving exchanges

Recommending or teaching information-giving exchanges occurred when the dietitian provided a clear set of instructions as to what the patient should do in terms of action or behaviour. These sequences were characterised by minimal input from the patient. They could be dietitian-led advice-giving episodes, advice giving as a more direct response to a patient query or as a result of patients offering information that required elaboration.

Recommending exchanges can also include general advice giving that the dietitian believes is appropriate for the patient but without first checking what the patient is currently doing and therefore not knowing whether this information is needed. In Excerpt 7 (Figure 11.17) the patient is clarifying whether it's all right to eat cornflakes. The dietitian fails to use an open question to explore which breakfast cereals the patient usually eats before giving advice. Consequently, the dietitian examines a list of breakfast cereals that the patient may eat, until one of the cereals fits with the advice given. Thus, by a process of elimination, the general advice concerning increasing fibre from breakfast cereals has become more specific as the dietitian finally hits an acceptable choice of breakfast cereal.

Excerpt 7:

Patient: "And cornflakes."

Dietitian: "Yes, cornflakes but we try to encourage more fibre and roughage and there's not much in cornflakes, they're not sugar coated, so you can eat cornflakes but Weetabix and porridge are a better choice. Fruit and fibre?"

Patient: "I don't like that one."

Dietitian: "No, bran flakes?"

Patient: "No."

Dietitian: "Shreddies?"

Patient: "Yes, I'll eat Shreddies."

Excerpt 8:

Patient: "I've gone back to having porridge and I have it just as it is, with a bit of salt in it but no milk."

Dietitian: "Watch the old salt."

Patient: "Yes, I do put a tiny bit in."

- Dietitian: "Because it doesn't do the blood pressure much good. It would be better to put a bit of sweetener or fruit on it."
- Partner: "I've tried it and it tastes very salty to me. I think you're probably not quite aware of how much salt when you turn the salt pot and it puts quite a lot on."

Patient: "I'll watch it."

Dietitian: "Yes, I'd watch it or try not to add salt as an added thing. A little bit in cooking is OK because you lose a certain amount in the water when you cook but when we add it at the last minute, at the table, you get 100% of whatever you put on."

Partner: "Porridge you do of course because it's absorbed."

Dietitian: "Yes, so I would just watch that, bearing in mind that Dr [x] been treating your blood pressure because you don't want to exacerbate what is helping. So it's a balancing act really. Porridge is good, salt's not so good."

Figure 11.17: Examples of *recommending/teaching/instructing* informationgiving exchanges

Recommending episodes of advice are generally informative, and by their nature, the advice tells the patient what to do and provides some information as to why the recommendation is being made, as shown in Excerpt 8 (Figure 11.17).

Instructing information-giving exchanges were also included in this theme, as they require minimal response from patients and were generally not collaborative, principally serving to confirm patient's actions prior to the next visit. They usually covered issues such as completing food, insulin, and blood glucose diaries prior to the next clinic visit, and were used to help patients explore the relationship between insulin dose and carbohydrate intake and the subsequent effect on blood glucose levels. These diaries are a learning tool and, as such, patients have usually agreed to complete these forms, as the next stage of care is dependent on this information being available. *Instruction* exchanges therefore similar to *recommending* exchanges as patients are being told what to do in terms of collecting information and recording it appropriately prior to the next visit. For example:

Dietitian: "There are things on here that, there are household measures, but for most of them there's a weight by them as well to give you a guide. So you could start to build up your own reference library, I suppose, of what your typical portions are for boiled potatoes versus jacket potatoes versus rice and although they might vary a little bit from meal to the next in that if you had rice one day with a meal and rice the next day and it may not be the exact same amount another day but it's going to be fairly similar. So you've got an idea of what's typical for you in terms of the quantities that you eat and your portion sizes, so maybe weigh a few of those and get an accurate picture of that..."

Patient: "So just jot down for each meal..."

11.3 Is there a difference in the type of information-giving exchanges used in consultations with agreement on reported decisions made and those with disagreement?

The number of information-giving exchanges coded per theme and the

percentage use of the identified information-giving exchange themes within the

two data sets is recorded in Table 11.42.

	Agreemen reported dec	t on isions	Disagreeme reported dec	ent on cisions			
	Frequency of	%	Frequency of	%			
	occurrence		occurrence				
Persuasive	3	1.9	18	14.2			
Supportive/ collaborative	78	50.3	22	17.3			
Confirming/permitting	28	18.1	35	27.6			
Recommending/ teaching/instructions	46	29.7	52	40.9			
Total frequency of information-giving episodes	155		127				

Table 11.42: Summary of occurrence of information-giving exchange themes in consultations with agreement on reported decisions made and those with disagreement

A greater number of information-giving exchanges occurred in consultations with agreement on reported decisions made. *Supportive/collaborative* information-giving exchanges were three times more likely to occur in consultations with agreement on reported decisions made. The Mann-Whitney was conducted to test for significance and the difference in occurrence was found to be significant (U= 11, p = .003).

Persuasive information-giving exchanges were six times more likely to occur in consultations with disagreement. The Mann-Whitney test was conducted to test

for significance of this finding and the difference in occurrence was found to be significant. (U = 17, p = .007).

This data highlights significant differences in the way that information-giving exchange occurs in consultations. Those with agreement use significantly more *supportive/collaborative* information-giving exchanges, whereas those with disagreement use significantly more *persuasive* information-giving exchanges, and a greater number of *recommending* information-giving exchanges. This suggests that there are significant differences in the way that dietitians practise in this study.

11.4 What is the relationship between information-giving exchanges and empathy in the consultations reviewed?

Information-giving exchanges and empathic statements made by patients were placed on an Excel spreadsheet in order of occurrence for each consultation. The information-giving exchanges were colour-coded to provide a visual map of the flow of information-giving exchanges that occurred in consultations with agreement on reported decisions made and those with disagreement. The thematic map is shown in Figure 11.18. Figure 11.18: Thematic Map of information-giving exchanges and empathic statements for consultations

1-10 represents consultations with complete disagreement on reported decisions made

11-20 represents consultations with complete agreement on reported decisions made

Consultation	Consultation Time		Consultations with complete disagreement on reported decisions made																																
1	15.01	SP	SP S	Р	S	c sc		SC																											
2	15.53	SC				SP			SC																										
3	16.02																																		
4	16.31				S	2	SC																												
5	17.42	SE																																	
6	17.53	SP	SE S	E SE	S	>	SP	SP	SP	SP		SP	SP	SE																					
7	19.53	SP	SP S	P SC	;		SP	SP	SP		SC	SP						SC		SC															
8	23.13	SE		SP)	SP					SC																								
9	33.59	SC	SC S	c sc	;		SE					SC																							
10	39.00	SC	SP	SP	, si	sc sc	SE	SP			SE		SC				SP																		
A contract of the second se	•							-									-											_				_			
Consultation	Consultation Time					•							Со	nsult	atior	ns wit	h co	mple	ete ag	greer	nent	on rep	orte	d deo	cisio	ons n	nade	e				<u> </u>			
Consultation 11	Consultation Time 13.42	SP	sc s	E			SC						Co	nsult	atior	ns wit	h coi	mple	ete ag	greer	nent	on rep	orte	d deo	cisio	ons n	nade)							
Consultation 11 12	Consultation Time 13.42 16.41	SP	SC S	E P			SC SP						Co	nsuli	atior	ns wit	h coi	mple	ete aç	greer	nent	on rep	orte	d deo	cisio	ons n	nade	e				<u> </u>			
Consultation 11 12 13	Consultation Time 13.42 16.41 17.15	SP SE	SC S SP S SE	E P			SC SP		SE	SC			Co	SP	atior	ns wit	h coi	mple	ete aç	greer	nent	on rep	orte	d deo	cisio	ons n	nade	9				<u> </u>			
Consultation 11 12 13 14	Consultation Time 13.42 16.41 17.15 26.51	SP SE SC	SC S SP S SE SE	E P P	E SI	C SC	SC SP	SC	SE	SC SC	SC		Co	SP SC	atior SP	ns wit	h coi	mple	ete aç	greer	nent	on rep	orte	d deo	cisio	ons n	nade	e SE							
Consultation 11 12 13 14 15	Consultation Time 13.42 16.41 17.15 26.51 32.04	SP SE SC SC	SC S SP S SE SE S	E P SE		C SC	SC SP SE	SC	SE	SC SC	SC		Co	nsull SP SC	atior SP SC	ns wit		mple	ete aç	greer	ment		orte	d deo	cisio s	ons n	made	SE	SC						
Consultation 11 12 13 14 15 16	Consultation Time 13.42 16.41 17.15 26.51 32.04 33.25	SP SE SC SC SP	SC S SP S SE S SE S SE S	E P SE E SO	E S(C SC	SC SP SP SE	SC	SE	SC SC SC	SC		Co	SP SC SC	atior SP SC	ns wit			sp	greer	nent			d deo	cisio s	ons n	nade	e SE	SC						
Consultation 11 12 13 14 15 16 17	Consultation Time 13.42 16.41 17.15 26.51 32.04 33.25 35.53	SP SE SC SC SP SC	SC S SP S SE S SE S SE S SC S	E P SE E SC C SC		C SC SC	SC SP SP SE SE	SC	SE	SC SC SC SE SE	SC		Co	SP SC SC SC	atior SP SC	ns wit		mple	sp				orte			ons n	nade	SE	SC						
Consultation 11 12 13 14 15 16 17 18	Consultation Time 13.42 16.41 17.15 26.51 32.04 33.25 35.53 37.38	SP SE SC SC SP SC SP	SC S SP S SE S SE S SE S SC S SC S	E P SE E SC C SC C SC		C SC SC	SC SP SP SE SE	SC SC	SE	SC SC SC SE SP SC	SC		Co	SP SC SC SC	atior SP SC	ns wit		sc	sp	greer	ment				cisio s	ons n	nade	SE	SC						
Consultation 11 12 13 14 15 16 17 18 19	Consultation Time 13.42 16.41 17.15 26.51 32.04 33.25 35.53 37.38 52.50	SP SE SC SC SP SC SP	SC S SP S SE S SE S SC S SC S SC S SC S	E SC C SC C SC		C SC	SC SP SE SE SE SE	SC SC SC SP	SE	SC SC SE SP SC	SC SC	SC SP	Co Co SP	SP SC SC SC SC	atior SP SC	ns wit		sc	sp	greer SC						ens n	nade	SE	SC						

	Col	our Codes for Information-giving Themes	Codes for Empathic Statements
Agenda set with patient		Recommending / Teaching	SE = Statement of Emotion
at start of consultation		Confirming / Permitting	SP = Statement of Progress
		Persuasive	SC = Statement of Challenge
		Supportive /Collaborative	

Although the total length of the consultation was noted, the length of each, individual information-giving exchange was not recorded as discussed in Chapter Ten. Consultations that set agendas with patients were also highlighted. The thematic map therefore provides a visual guide as to the differences in the sequence of information-giving exchanges in consultations. It highlights consultations where agendas were set, and illustrates the relationship between empathic statements and information-giving exchanges within the consultations.

The thematic map allows the pattern of information-giving exchange to be explored between the two data sets. *Persuasive* information-giving exchanges are more likely to be followed by *recommending* information-giving exchanges and further episodes of *persuasive* information-giving exchange. In two of the consultations with disagreement on reported decisions made, *persuasive* information-giving exchanges were the last episodes to occur in the consultation.

11.4.1 Pattern of supportive/collaborative information-giving exchanges

The data showing the pattern of *supportive collaborative* information-giving exchanges found in consultations with agreement and those with disagreement is summarised in Table 11.43. This table illustrates how often the consultation starts and ends with *supportive/collaborative* information-giving exchanges and the frequency with which these exchanges occur before and after a patient's expression of an empathic opportunity.

	Consultations with agreement on reported decisions made	Consultations with disagreement on reported decisions made
% of consultations starting with supportive/collaborative exchanges	60%	20%
% of consultations ending with supportive/collaborative exchanges	60%	10%
% of information-giving exchanges that preceded empathic statements	50%	29%
% of information-giving exchanges that followed empathic statements	73%	24%

Table 11.43: Summary of the pattern of occurrence of supportive/collaborativeinformation-giving exchanges in consultations

The data in Table 11.43 appears to suggest that the *supportive/collaborative* information-giving exchanges may help to facilitate patients' expression of empathic opportunities and in turn facilitate more *supportive/collaborative* information-giving exchanges.

11.5 Exploring the communication skills that precede empathic statements in consultations

The presence of empathy has already been shown to be greater in those consultations where there is agreement on reported decisions made, as summarised in Chapter Eight. In light of the associations suggested between empathic statements and information-giving exchanges outlined in Table 11.43, further analysis was conducted to explore the potential verbal triggers that would encourage empathic statements from patients. Table 11.44 provides a summary of the frequency of occurrence of identified behaviours by the dietitian prior to an empathic statement from the patient.

	Err statem	pathic ents m	ade	Empathic statements made						
	in con with a on r dee	sultati greem eporte cisions	ons ent d	in consultations with disagreement on reported decisions						
Behaviours that precede	SP	SC	SE	SP	SC	SE				
empathic statements										
Asks about progress	2	6	3	4	6	1				
Supportive/positive statement	3	7	8	3	2	2				
Discuss, explores issue raised by patient	6	16	5	6	1	2				
Reflection/summarises	1	7	1	2	2	2				
Confirms patient statement	3	12	4	4	1	1				
Offers information	1	4	0	3	4	0				
Ignores patient question	0	0	0	1	0	0				
Total	20	49	23	27	21	10				

 Table 11.44: Frequency of behaviours preceding empathic statements of emotion (SE), progress (SP), and challenge (SC) in consultations

Consultations with agreement on reported decisions made had a greater use of active-listening skills, supportive and confirming statements, and exploratory questions. This appeared to encourage a greater number of empathic statements from patients. In the consultations that had complete agreement on reported decisions made, the greater use of active-listening skills resulted in the occurrence of twice as many statements of emotion and statements of challenge. Full examples of active listening skills and questions coded can be found in the coding manual in the appendix (Appendix 3.1).

11.6 Conclusion

Thematic analysis resulted in the identification of four distinct types of information-giving exchanges in dietetic consultations. These occurred in some but not all of the consultations analysed. *Supportive/collaborative* information-

giving exchanges occurred at a significantly greater frequency in consultations with agreement on decisions made, whereas *persuasive* information-giving exchanges were significantly more frequent in consultations with disagreement.

In consultations with agreement on reported decisions made there were more frequent empathic statements made by patients. Dietitians expressed greater empathy and used more active-listening skills and exploratory questions on self-care practices. A greater number of *supportive/collaborative* information-giving exchanges occurred and dietitians in these consultations were also more likely to set agendas with the patient at the start of the consultation.

In contrast, consultations with disagreement had a lower frequency of empathic statements made by patients. Dietitians expressed lower levels of empathy, and used fewer active-listening skills and few questions that explored self-care practices. A greater number of *persuasive* and *recommending* information-giving exchanges occurred and dietitians in these consultations did not generally set agendas with the patient at the start of the consultation.

The analysis has provided valuable insight into the process by which empathy, communication skills and information-giving exchanges interact with each other in the consultation; thus identifying a mechanism by which these behaviours and skills impact on the consultation process and result in agreement on reported decisions made.

In addition, agreement on reported decisions made appears to be a reliable marker of the presence of patient-centred communication skills and behaviours in the dietetic consultation.

Full interpretative analysis of these findings will occur alongside the interpretation of the quantitative findings from Chapter Eight and is presented in the next chapter.

CHAPTER TWELVE: DISCUSSION

"Good communication skills are crucial for optimising a care planning discussion and supporting individuals to self- care. It is vital that healthcare professionals have the right skills, approaches and behaviours to deliver high quality personalised services for individuals with long term conditions"

(Department of Health, 2010a).

This quote is taken from a pamphlet produced by the Department of Health aimed at professionals working with long-term conditions. The rhetoric is justified in that professionals need approaches and behaviours to improve patient experiences and outcomes, especially productive communication skills (Zandbelt et al, 2007). Street (1991) and Street et al (1993) in their studies exploring the communication styles of nurses and physicians demonstrated that the more patient-centred the professional, the more communicatively active the patient. It has been shown that increasing patient-participation behaviour also encourages professionals to be more patient-centred and collaborative (Moran, 2008; Street & Millay, 2001). A key factor in improving behaviours and self-care management is patient participation (Street et al, 1993).

When reviewing non-adherence to medication therapy, Epstein & Cluss (1982) found that physicians believed that it was others who had the poor consultations not themselves, despite the fact that 40% of the patients seeing these physicians did not follow the recommendations provided. This distorted perception means that professionals may be unaware of the deficit in their own skills. The present study confirmed that patient and dietitian perception of the consultations did not correlate, a finding supported by others exploring nurse, dietitian and physician perceptions of consultations (Brug et al, 2007; Parkin & Skinner, 2003; Pill et al, 1999).

A distorted perception demonstrated across different healthcare professional groups suggests that in order to engage professionals in reviewing and assessing the effectiveness of their practice, a simple but effective method for feedback needs to be implemented. For example, Parkin et al (2006) showed that the use of video feedback provided some insight, but these methods were cumbersome, required elaborate permissions and technology. A simple, effective and above all, low-cost, easily-administrated method needs to be developed and evaluated. This will allow professionals to reflect on skill use in their current practice (Entwistle & Watt, 2006) and identify potential training needs and areas for further skill development (Guimond et al. 2003; Parkin et al, 2006; Schön, 1983). However, awareness of the application of skills may not be enough to lead to the changes in the practitioner behaviour (Anderson & Funnell, 2005b; Brug et al, 2007; Pill et al, 1999; Skinner et al, 2008). Brug et al (2007) suggest that the level of competence and proficiency of the practitioner may impact on their ability to effectively apply new skills in the consultation. New skill acquisition may also be a problem (Rollnick, 1996). Frequency of use of new skills and the perceived importance and adequacy of training will all impact on ability to actually deliver the desired change (Delahanty & Halford, 1993). Poor acceptance may provide some explanation as to the difficulties seen in maintaining skills once trained (Doherty et al, 2000; Guimond et al, 2003; Latter et al, 2007; Lorenz et al, 1996; Pignone et al, 2003; Pill et al, 1998; Rollnick, 1996; Rollnick, Mason & Butler, 1999) and highlights the need for regular review to ensure that skills become embedded in habitual practice. There is a recognition that the skills and behaviours used while communicating with patients are just as important as the message being delivered (Bertakis, Roter & Putnam, 1991; Hall et al, 1994). There are problems with assessing

communication skills and behaviours because of the range of outcome measures employed to assess effectiveness in practice. Stewart's (1995) review highlights the need for clarity in terms of components of skills used and impact on specific measures, using a clear theory base to explain findings. When exploring whether communication skills are a primary source of improvement in outcome, a confounding factor is the multiple effects that medication and others therapies may have on the long-term markers for health outcomes, particularly medical markers (as outlined in Chapter One). If professionals are to assess the impact of their communication on outcomes, they need a specific marker that relates directly to their input. The measure of agreement on reported decisions made is potentially a clinically relevant marker for professional consultations.

This study chose to examine a method that can be easily adapted for day-today use in any health professional clinic or interview situation, yet provide valid and reliable feedback. Further, during the course of the work on demonstrating the tool and its fitness for purpose, this study also provided insight into the behaviours, skills, and patterns of consultations that promote interaction between the healthcare professional and the patient. It also shows how active *"care planning discussion and supporting individuals to self -care"* (Dept Health 2010) occurs within the consultation.

There are a few studies that have used this simple pen and paper technique of recording decisions reached by both the dietitian and the patient immediately after the consultation, in order to reflect on consultation outcomes. However, these studies have merely measured concepts such as patient satisfaction and

patient autonomy. Information on the specific communication skills and behaviours needed to achieve higher agreement rates on reported decisions made was not provided (Barnard et al, 2006; Goodchild et al, 2005; Parkin et al, 2006; Parkin & Skinner, 2003).

Low levels of patient/dietitian agreement on reported decisions could highlight the absence of patient-centred communication skills and behaviours used by the healthcare professional. Parkin & Skinner (2003) provided some evidence for this proposition, as they found that more controlling behaviours and low autonomy support were associated with disagreement on reported decisions made. Although the Parkin & Skinner study was conducted on small numbers of nurses and dietitians, it does suggest a role for agreement on recall as a marker of consultation behaviours. Parkin et al (2006) provide further evidence to support this view, as improvements in the level of agreement on decisions made were obtained following training in communication skills. Failure to measure how the training changed the communication skills used in the consultations meant that even though training resulted in improvement, the usefulness of these findings for identifying specific skills needed was lacking (Parkin et al, 2006). Nevertheless, it does suggest that recall may be a useful marker of communication skills in the consultation. The pen and paper method (recall) therefore has the potential to be a cheap easy and rapid tool to indicate effective use of communication skills in the consultation. By exploring the actual communication skills and behaviours used that resulted in higher levels of agreement, it would also be possible to identify how consultations can be made more effective in order to improve outcomes.

Satisfaction ratings immediately after a consultation are known to be affected. albeit in doctor/patient communication (Jackson et al, 2001; Jensen et al, 2010). Therefore, in this study, a patient-satisfaction questionnaire was tested as it has been shown to provide an independent method for examining communication skills. However, a lack of variation in response to the satisfaction questionnaire was seen in the current study, despite significant variations in agreement on reported decisions made between consultations. This is similar to findings from Trudeau & Dubé (1995) when exploring satisfaction with the counselling skills of dietitians. There is a suggestion by Latter et al (2002) and Latter et al (2007) when exploring nurse/patient concordance, that previous experience of consultations established certain expectations in the patient; a view also put forward by others (Epstein et al. 2005: Trudeau & Dubé, 1995). In this study, the satisfaction scores were consistently high. One possibility is that these high scores were because the cohort of patients chosen, were new to the dietetic service. (66% of the patients were registered as new visits to the dietitian. The classification of new for dietetic appointment systems was defined as those patients who had not been seen by the dietitian in the last year). However, this may not be the case, as there is another possibility. The mean duration of diabetes was 10.87 years with a range of 1-59 years; patients may therefore have had previous encounters with dietitians in the past and yet were still being recorded as new. Jensen et al, (2010) have shown that immediacy of data collection resulted in higher levels of satisfaction compared to data collected some 3-6 weeks later; a finding supported by others (Kinnersley et al, 1996). This may also have contributed to the higher satisfaction scores seen. Other possible explanations for these higher scores might include: the patient's previous experiences with dietitians (Jackson et al, 2001), alteration in patient

perception of the level of care to expect (Crow et al, 2002; Latter et al, 2007; Latter et al, 2002), patient modification of their reported satisfaction with the dietitian (Epstein et al, 2007; Hall et al, 2002; Latter et al, 2007; Sixma et al, 1998; Stiles et al, 1979; Zandbelt et al, 2004). Alternatively, patients may not have felt strongly enough about their encounter with the dietitian to make a judgement on their level of satisfaction with the consultation (Trudeau & Dubé, 1995).

Although the satisfaction questionnaire used in this study focused on the patient/dietitian relationship, high levels of reported satisfaction did not correlate with empathy or agreement on reported decisions made, as suggested in other studies (Crow et al, 2002; Goodchild et al, 2005; Kim, Kaplowitz & Johnston, 2004; Roter, 2000; Street et al, 2005). Kinmonth et al (1998) have identified problems with measuring satisfaction in GP/nurse consultations and found that greater satisfaction often masked deficiencies in other domains of practice, particularly negotiation skills. This resulted in patients reporting greater satisfaction, even though self-care management practices had not improved and health risks remained high (Kinmonth et al, 1998). Patients may therefore report being satisfied with behaviours that they find familiar, rather than behaviours that improve care (de Haes & Bensing, 2009; Epstein et al, 2005; Michie et al, 2003).

Patient-satisfaction questionnaires are affected by many factors beyond the actual consultation (Sixma et al, 1998). It can be concluded that a patient-satisfaction questionnaire does not provide reliable feedback to the healthcare

professional about the quality of their behaviour and the communication skills used in their consultations. An alternative method needs to be evaluated.

12.1 Level of agreement on reported decisions made in the consultation

Satisfaction questionnaires were seen to be affected by many factors and it may be that a number of factors will also influence agreement on decisions taken during the consultation, independent of the communication skills used. These may, for example, relate to the patient group being studied.

The chronic disease explored in this study was diabetes, and it has been suggested that poor disease control may affect recall ability in people with diabetes (Rost et al, 1990). Poor control may be indicated by the presence of complications. In this study, the presence of complications increased levels of agreement on reported decisions made, not decreased, as was expected. There are at least three possible reasons for this: Patients attending the dietetic clinic may be more motivated to deal with issues concerning food, in an attempt to cope with the complications that they are experiencing (Flocke & Stange, 2004). These food issues may be more pertinent to the individual as the presence of complications may assist recall (Bogardus et al, 2001; Kravitz et al, 1993). Patients who have had diabetes for longer may be more familiar with the self-care management messages, as they will have heard them before in previous consultations (Jansen et al, 2008b). These findings suggest that patients with complications appear to be more likely to be engaged in dietetic consultations. Sharing responsibility for decision-making has been shown to lead to agreement on recall in physicians' consultations (Heisler et al, 2003).

There are a limited number of studies that measure agreement on reported decisions made in the consultation. However, the few studies that have used this measure report similar agreement rates to those found in the present study (Flocke & Stange, 2004; Madan & Tichansky, 2005; Parkin & Skinner, 2003; Pesudovs, Luscombe & Coster, 2006; Roach et al, 1992). These findings (including the current study) all suggest that agreement on reported decisions made in the consultation between patients and nurses, dietitians and doctors, are fairly consistent, falling between 40-47.1% or 1-1.65 decisions per consultation. It would imply that between 53 and 60% of decisions made in the consultation cannot be recalled.

Furthermore, in the current study, the mean agreement rate for reported decisions was 1.65 decisions per consultation; better than the 1 decision per consultation reported by others exploring physician/nurse and dietitian consultations (Page et al, 1981; Skinner et al, 2007). These decisions account for just a small number of the total decisions made in the consultation. It raises the question as to how many decisions should be made in the consultation if patients are to recall and then implement them.

One study of GP consultations suggests that recall is greatest when 1-2 recommendations are set, with recall declining as the number of recommendations increases (Bravoa et al, 2010). However, in this present study, differences were seen for each dietitian in the number of decisions set and the number of decisions reported by patients. Dietitian 1 made few decisions per consultation (2.95) and had the lowest patient report of decisions

and agreement on decisions made (31%). Dietitian 3 made more decisions per consultation (4.43) and had the highest patient report of decisions and agreement on decisions made (51%). These figures suggest that in a dietetic consultation, the total number of decisions set is not the only factor that needs to be considered when reviewing agreement on reported decisions made. The process that occurs in the consultation, in terms of patient engagement, is likely to impact on recall as well as on the level of accuracy of decisions reported (Watson & McKinstry, 2009).

It has been shown that patients who report that they have followed specific dietary behaviours have been shown to have better HbA1c measurements than those who do not follow these behaviours (Delahanty & Halford, 1993). However, if patients cannot recall decisions made, or recall of decisions made is inaccurate, this is likely to impact on outcomes. Inaccurate recall may result in implementation of behaviours that do not lead to expected outcomes. One third of the decisions reported in this present study lacked clarity and this is similar to earlier findings in nurse/dietitian consultations (Parkin & Skinner, 2003). This highlights the need for good communication skills to ensure that decisions made are clear, specific, and relevant to the individual, thus increasing the chances of recall and implementation (Heisler et al, 2007; Miller et al, 2002; Skinner et al, 2007). Following professional training on agenda setting and goal setting with nurses and dietitians, Parkin et al (2006) were able to demonstrate a reduction from 17% to 11% on complete disagreement on decisions made. This highlights the potential of skills training to enhance clarity of communication and improve agreement on recall.

The point at which information is placed within the consultation also impacts on recall, in that retrieval of information improves when it is received at the very beginning or the very end of the consultation (Kravitz et al, 1993; Watson & McKinstry, 2009). Furthermore, the process of recapping (Bravoa et al, 2010; Poskiparta et al, 2001; Watson & McKinstry, 2009) and action planning (the *how* and *when* a decision will be applied; Bradley *et al*, 1999; Stadtlander & Coyne, 1990) enhances recall and increases the chances of implementation of the agreed decision/behaviour (Schnoll & Zimmerman, 2001; Schüz et al, 2006; Webb & Sheeran, 2008). Active planning therefore is a strategy that can increase recall and help patients develop autonomous motivation towards their dietary plan (Julien et al, 2009). Lack of action planning and exploration of strategies to address planned decision-implementation may therefore contribute to poor outcomes (Stetson *et al*, 1992). Action planning alongside decision recapping was found to be a significant contributor to the level of agreement on reported decisions made in this study.

The mixed method approach used in this study provides additional insight into the possible explanations for the varying levels of agreement on reported decisions made in the consultation. Themed analysis of the recordings of the consultations identified distinct differences in the type of information-giving exchanges used in consultations. *Supportive/collaborative* information exchanges accounted for 35.5% of the information exchanges used in dietetic consultations. Qualitative studies of Finnish nurse consultations suggest that 13.3% of the dietary information exchanges were *supportive* (Kiuru et al, 2004). This lower frequency is not surprising considering the cultural and professional differences between Finnish nurses and UK dietitians. In addition, Moran &

Latchford (2008) found that active involvement accounted for less than a quarter of GP consultations and is similar to findings from Goss et al (2004). Goss et al (2004) explored information-giving sequences in GP consultations and found a limited degree of patient involvement either before or after information exchange. Campbell et al (1990) suggested that patients were more expressive when interacting with nurses than with physicians; this may also be true for dietitians. Street et al's (2005) review of physician consultations highlighted that female physicians were more likely to use supportive talk than male physicians. The fact that all the dietitians used in this present study were female may be a further reason for the higher frequency of supportive information-giving exchanges seen. The findings from this present study appear to suggest that patients are more expressive when interacting with dietitians, and that dietitians enable that interaction to occur through supportive/collaborative information-giving exchanges. Although direct comparisons are hard to draw, due to the different methodologies used to assess collaborative interactions, these studies appear to suggest that dietitians may be more supportive/collaborative in their approach to working with patients than are physicians. However, further work is needed, using similar methods with larger groups of professionals, which may support this view.

The need to switch from a proscriptive to a more collaborative style of working with patients was one of the key recommendations from the Diabetes Control and Complications Trial (DCCT) (Lorenz et al, 1996). Collaborative working with patients is a method reported to achieve better control in diabetes (Aiken, Bingham & Piette, 2005; Flocke & Stange, 2004; Greenfield & Kaplan, 1988; *Spahn et al,* 2010; Stewart, 2001; Street, 1992b; Trudeau & Dubé, 1995). This

present study supports these findings, as the presence of *supportive/collaborative* information-giving exchanges was significantly greater in consultations with agreement on reported decisions made, and accounted for up to 50.3% of the total information-giving exchanges. This provides further evidence of the ability of the outcome measure of agreement on reported decisions made to act as a marker of effective patient-centred communication.

However, consultations with disagreement on reported decisions made were unsuccessful in achieving supportive/collaborative information-giving exchanges. Instead, these consultations favoured more professional-led information exchanges, which were identified as *persuasive and* recommending. The greater presence of directive behaviours in consultations has been related to poor metabolic control in other studies (Greenfield & Kaplan, 1988; Kaplan, Greenfield & Ware, 1989). Wolpert & Anderson (2001) suggest that these attempts at persuading and recommending the individual to change are well intentioned, as they are aimed at improving health outcomes. This has been referred to as the *theory of miscarried helping*, a process by which 'a helper's well intentioned efforts to help motivate a person paradoxically leads to interactions over time that are constraining and detrimental to the recipient' (ibid p 995). Although well intended, the persuasive and recommending information-giving exchanges fail to engage the patient perspective and discourage a sense of autonomy (Anderson, 1995; Carter et al, 1982; Parkin & Skinner, 2003; Street et al, 1993; Wolpert & Anderson, 2001). In this present study, this category of information-giving exchange resulted in a higher level or disagreement on reported decisions made.

Lack of engagement is likely to close down the relationship between the patient and the professional (Anderson et al, 1991; Gable, 2007; Suchman et al, 1997; Wolpert & Anderson, 2001). This was illustrated in this present study, where lack of exploratory questions prior to *recommending and persuasive* information-giving exchanges resulted in failure to engage patients, leading to disagreement on reported decisions made. Burley-Allen (1995) suggests that patients may be resistant to hearing advice that does not reflect their needs, and this lack of engagement therefore may discourage personal reflection (Marilynn et al, 1995). This may explain the disagreement on reported decisions made in consultations that used more *persuasive and recommending* information-giving exchanges.

Skilful questioning is therefore required to support active listening, as this helps the listener to develop an understanding of the patients' needs and engage them in the consultation process. It has been demonstrated as a useful means of establishing rapport with patients in nursing consultations (Poskiparta et al, 1998) and been found to encourage patients to think, reflect, and facilitate involvement in medical consultations (Guimond et al, 2003; Lang et al, 2000; Maguire et al, 1996). In this present study, a greater number of exploratory questions were used in consultations with agreement on reported decisions made, with significantly more questions being used that explored self-care management practices (p = .007). These types of questions will provide the dietitian with insight into patients' perceptions of their conditions (Bertakis et al, 1991) and its management (Miller & Rollnick, 2002; Rollnick et al, 1999). This helps in determining possible drivers for change as well as the individual preferences of the patient (Epstein et al, 2004; Stetson et al, 1992). This results
in greater agreement on reported decisions made, as decisions are more likely to reflect the patient's needs. The greater use of open questions in chronic disease consultations is related to positive reports of patients' perception of facilitation (Martin et al, 1999) and satisfaction with decisions (Guimond et al, 2003). Exploratory questions are therefore an important part of patient-centred communication, as they enable understanding and engagement with the individual to occur (Lang et al, 2000; Little et al, 2001b; Willems et al, 2005) and result in greater agreement on reported decisions made.

Although exploratory questions were used in some consultations that reported disagreement on decisions made, the resulting dialogue suggests that the professional needed to do more than ask the right type of question. How the question is asked (Paterson, 2001) and the ability to listen to, and respond appropriately to the answers provided by the patient is just as important (Egan, 2002; Guimond et al, 2003; Zachariae et al, 2003). The facilitation of patient participation in consultations, through skilful questioning is particularly important for influencing complex areas of behavioural change (Heisler et al, 2007; Pearson & Rapoport, 2007; Street, 1991). In particular, patient facilitation has been shown to improve dietary implementation (Trudeau & Dubé, 1995) and impact on blood sugar control and quality of life in people with diabetes (Greenfield & Kaplan, 1988). Improving communication skills to enable patient participation in consultations may help to reduce the discrepancy seen in agreement on reported decisions made (Jackson et al, 2005; Mead et al, 2008). In this present study, agreement on reported decisions made was more evident where patient participation was enhanced.

12.2 Level of patient engagement and consulting time

Beisecker & Beisecker (1990) suggest that situational variables such as length of consultation, diagnosis, and reason for visit may have more of an effect on the interaction in the consultation than the doctor-patient relationship. Some patients may not want to share in decision-making (McKinstry et al, 2004; O'Brien & Petrie, 1996). The resulting shorter duration of consultations may therefore reflect the patient's agenda and the lack of desire to be involved (Street et al, 1993). However, other studies suggest that longer consultation length is a result of concerns not being responded to adequately (Levinson et al, 2000). A review in 2004 stated that the evidence relating consultation length to level of patient involvement in the consultation was inconclusive (Griffin et al, 2004).

The use of a mixed-methods approach in this present study has provided some insights into this complex issue. Consultations with disagreement on reported decisions made were shorter, and contained younger patients with a shorter duration of diabetes. Duration of diabetes has been linked with poor dietary compliance (Shah et al, 2003), as has younger age of patients (Beisecker & Beisecker, 1990). Lack of agreement on reported decisions made may therefore be an indicator of lack of future compliance. Dietitians in this study used more controlling behaviours in consultations where disagreement occurred. It may be that dietitians were more controlling in an attempt to raise the importance of dietary change with patients. Who, because of their current age and health profile may not be interested or engaged in changing their diets. This contrasts with Street et al (2009), who found nurses were more controlling with less cooperative and adherent patients (ie, patients at higher risk of poor

control). The findings in this study appear to suggest that dietitians were more controlling in an attempt to raise awareness in those who are fit and well and presenting with no complications. This may account for the lower frequency of empathic opportunities recorded in consultations with patients who had a shorter duration of diabetes, reflecting the lack of engagement and subsequent disagreement on reported decisions made.

Conversely patients with a long duration of diabetes are more likely to have complications as a result of the diabetes and may be struggling with the uncertainties of the day-to-day management of the condition. It is likely that they will have more issues to discuss potentially creating longer consultations (Albright et al, 2001; Shah et al, 2003). When more time is provided in the consultation, patients with a long duration of diabetes may ask more questions (Beisecker & Beisecker, 1990) and receive more information (Street et al, 1993). Encouraging patients to take part in the conversation results in more interactive dialogue and is correlated with the length of the consultation (Roter et al, 2008). The present study supports this finding in that longer consultations were associated with greater use of patient-centred communication skills, empathy and active listening skills, and a greater number of *supportive* / *collaborative* information-giving exchanges.

The mixed-methods approach taken in this present study provides a unique and valuable insight into the sequential effect of information-giving exchanges and empathy in dietetic consultations (Pedersen, 2009). *Supportive/collaborative* information-giving exchanges are associated with greater empathy. In turn, more collaborative information-giving exchanges reflect greater patient

engagement, resulting in increased agreement on reported decisions made. The extent of patient engagement in the dietetic consultation resulted in longer consulting times, as suggested by Street et al (2009). Zimmermann et al (2007) suggest that the longer consultation is due to the greater number of empathic opportunities that patients present. In this present study, it appears that the extra time in the consultation arising from the *supportive/collaborative* information-giving exchanges and greater patient engagement results in the greater number of empathic opportunities presented by patients (Zimmermann et al, 2007).

Studies that have reviewed GP consultations have highlighted a minimum time of 19-20 minutes before patients were able to fully engage in a dialogue about their condition (Beisecker & Beisecker, 1990; Martin et al, 1999). This process is referred to as *patient enablement* whereby sufficient time is needed to provide opportunities to raise concerns and issues (Martin et al, 1999) and explore options for self-care decision-making (Beisecker & Beisecker, 1990). These studies also noted that short interactions could minimise: the level of question-asking by patients, the expression of patient attitudes (Beisecker & Beisecker, 1990; Moran et al, 2008) and expression of empathic opportunities.

These effects result in low rates of patient participation (O'Brien & Petrie, 1996). These studies reflect the low rate of empathic opportunities and low patient engagement that was seen in this present study, in the *supportive/collaborative* information-giving exchanges within those consultations that had disagreement on decisions made. In addition, in this present study, a minimum time of 18.27 and 23.33 minutes for follow-up and new patient appointments respectively was

found to be needed, before agreement on decisions made could occur. This finding suggests that even when there is an established relationship with the patient, time is still required in the consultation to enable the patient to actively participate (Marvel et al, 1999).

Using the measure of *number of agreements on decisions made* acts as a marker for the presence of patient-centred communication skills. The findings above also suggest that dietetic services wishing to improve their delivery of effective patient-centred consultations will need to consider the potential implications of longer consultation times on resource planning.

12.3 The focus of the consultation and its relevance to immediate need

The varying agendas (Heisler et al, 2003; Julien et al, 2009; Vijan et al, 2005) and priorities of both patients and professionals (Kravitz et al, 1993; Levy, 2009) may be reflected in the type of dietary decisions recalled and reported (Heisler et al, 2003; Jansen et al, 2008a; Michie et al, 1997; Skinner et al, 2007; Wolpert & Anderson, 2001).

Patients attending consultations with professionals are more likely to focus on immediate life concerns, such as integrating the demands of diabetes into their day-to-day lives (Anderson, 1995; Vijan et al, 2005; Watson & McKinstry, 2009; Wolpert & Anderson, 2001). This finding is reflected in this present study, with patients reporting a high number of decisions on snacking and eating out, thus illustrating the immediate personal and social context of their eating (Gentili et al, 2001).

Conversely, dietitians reported more decisions being made that related to increasing fruit and vegetable intake, reflecting their desire to improve the quality of the patient's diet in order to improve overall health (Diabetes and Nutrition Study Group of the European Association for The Study of Diabetes, 2000; England et al, 2009). The lower frequency of decisions relating to fruit and vegetable intake, reported by patients in this present study, may reflect the lower importance attached by them to this area of dietary change.

Agreeing an agenda at the start of the consultation is recognised as an important part of the consultation process for all professionals (Berg-Smith et al, 1999; Feste & Anderson, 1995; Gable, 2007; Glasgow et al, 2006; Holli et al, 2003; Parkin, 2003; Rollnick et al, 1999). Agenda agreement is a key feature in patient-centred counselling approaches such as empowerment (Anderson, 1995) and educational strategies that use motivational interviewing (Berg-Smith et al, 1999; Rollnick et al, 1999). Agenda agreement has been shown to be one of several strategies used in dietary motivational counselling that can lead to significant reductions in energy and fat intakes (Berg-Smith et al, 1999). In this present study, agendas were more likely to be set with patients in those consultations where there was agreement on reported decisions made and less likely in consultations where there was disagreement.

In order to establish and therefore address patients' needs, it is essential to focus on those needs through the process of agenda setting and the use of *supportive/collaborative* working by employing active listening skills and exploratory questions. The resultant information provided and the decisions made are likely to be more relevant to the individual, thereby assisting recall

and leading to greater agreement on reported decisions made (as seen in this present study).

12.4 Reporting of decisions that were not made in the consultation

Working collaboratively with patients may be one way of enhancing patient and dietitian agreement on reported decisions made and as a way to improve patient outcomes (Heisler et al, 2003). Lack of collaborative working would therefore be expected to result in disagreement on reported decisions made. Patients are more likely to recall information that supports their own opinions, attitudes and beliefs (Kiviniemi & Rothman, 2006) indicating the operation of a selective memory bias for recall (Skinner et al, 2007; Watson & McKinstry, 2009). Lack of engagement may be one way that this process is accentuated, contributing to low levels of agreement on reported decisions made and leading to the reporting of decisions that were not made in the consultation, as highlighted in this present study.

There is limited data on the reporting of unmade decisions in consultations; the few studies that have been conducted focus on physician/GP and nurse consultations (Makoul et al, 1995; Page et al, 1981; Skinner et al, 2007). They report unmade decisions figures of 24.3% in GP consultations and up to 40% in medical consultations. The latter higher reading is likely to result from the data being collected three months after the event, as recall is known to drop over time (Campos & Alonso-Quecuty, 2006; Pesudovs et al, 2006). In this present study a much lower figure of 11.5% was obtained for patient recall of decisions that were not made in the consultation. Skinner et al (2007) focused on nurse and dietitian consultations. However, data presented was for nurses and

dietitians, rather than for the individual professions involved, preventing comparison between nurses and dietitians to occur. The difference between professional groups is an important consideration, as patient expectations of care will differ in relation to the professionals involved in their care (Collins, 2005; Hampson & Mckay, 1996; Zandbelt et al, 2004). Consultations with dietitians have by definition a narrower focus; this narrower focus in the consultation may help to reduce the reported number of unmade decisions seen in this present study. In addition, these findings suggest that dietitians had better patient-centred communication skills than the nurses and physicians studied so far. This resulted in greater patient engagement, leading to few decisions being reported that did not occur in the consultation.

The discrepancies highlighted in this study between patients and dietitians in reported decisions made, underlines the importance of engaging patients through setting agreed agendas and using *supportive/collaborative* dialogue to assist both patient and professional understanding. This process will lead to discussions that are personally meaningful and relevant to the individual, resulting in decisions that support patient needs (Thorne et al, 2003). This relevance may assist recall, with greater agreement on reported decisions being suggested as a mechanism by which improvement in long-term health outcomes may occur (Heisler et al, 2003).

12.5 Presence of Empathy in the Consultation

A crucial part of the consultation process is how issues raised by patients during consultations are recognised, as these are likely to reflect matters of great importance. Empathy demonstrated during the consultation is required to elicit

patient engagement and will impact on the subsequent interaction between the patient and the dietitian. Indeed, in this study, empathy was correlated with patient and dietitian agreement on reported decisions made during the consultation ($\tau = .283$, p = .0005). This confirmed the hypothesis that greater empathy would result in greater agreement on reported decisions made. Recording agreement on decisions made provides a tangible marker of the presence of empathy in the consultation, which is quick and easy to assess. In addition, the mixed-methods approach provides a unique insight into how empathy interlinks with other communication skills and information giving in the consultation to influence the agreement level obtained.

Reviews by Stewart (2001) and Mead & Bower (2000) have contributed significantly to the understanding of concepts around patient-centred care and patient-centred communication. Additional studies support the growing evidence base for patient-centred care in order to communicate effectively with patients and improve health outcomes (Del Piccolo et al, 2008; Little et al, 2001a; Mead & Bower, 2000b; Stewart, 2001). An important construct in patient-centred care is the ability of the professional to express empathy (Neumann et al, 2009). Although empathy is highlighted as important, it is reported as occurring in low frequencies in medical consultations (Del Piccolo et al, 2008; Little et al, 2001a). Despite this low occurrence, when empathy is present in medical consultations, it has been linked to improvements in outcomes (Bertakis, 2009; Comstock et al, 1982; Larson & Yao, 2005; Levinson et al, 2000; Ong et al, 2000). These improvements include: better HbA1c measures in patients with diabetes (Hojat et al, 2011) and compliance to recommendations and improved patient satisfaction (Epstein et al, 2007;

Fuertes et al, 2008; Kim et al, 2004; Levinson & Roter, 1993; Stewart, 1984; Zachariae et al, 2003). A more recent study exploring GP counselling demonstrated that patients who perceived their GP as more empathic improved their weight-related attitudes and dietary behaviours (Cox et al, 2011). This present study was able to confirm these finding in as much as measured empathy in the dietetic consultation significantly improved agreement between patient and dietitians on reported decisions made in the consultation (p =.0005).

There were significant differences in the level of empathy demonstrated between dietitians participating in this study (p = .003) with the mean empathy level ranging from 2.49 to 3.07. Empathy correlated with significant differences on level of agreement on reported decisions made (p = .002) ranging from 1 to 2.09 decisions. This finding demonstrates the robustness with which the paperbased tool was able to reflect the presence of empathy within the consultation.

There are few studies comparing empathy in the different professions (Hojat et al, 2002), the difficulty being the range of tools and scoring systems that are used to assess empathy as outlined in Chapter Three. In addition, female physicians have been noted to have higher levels of empathy than male physicians (Bertakis, 2009; Bylund & Makoul, 2002; Nicolai & Demmel, 2007; Scmid Mast et al, 2007). This may further contribute to the differences seen in data largely drawn from studies using male physicians, which makes it harder to draw direct comparisons. The empathy score achieved in this present study was lower than that recorded in a previous study of female dietitians using the ECCS to measure empathy (Goodchild et al, 2005). The changes made to the

ECCS coding discussed in Chapter Four could have provided these lower than expected scores for empathy and may account for this difference. However, when compared to data on male consultants (Coughlan, 2005) and male / female physicians (Bylund & Makoul, 2002) using the unaltered ECCS, the level of empathy in this present study was comparable. The altered ECCS used in this present study provides lower scores for empathy than the original ECCS. Taking this factor into account, it seems that female dietitians in this study are demonstrating higher levels of empathy in their consultations than male/female physicians and male consultants.

Dietitians are key members of the diabetes team in their facilitation of patient self-care management and in the provision of good models to other members of the team. Parkin et al (2006) demonstrated that training the diabetes team in agenda setting and goal setting could improve agreement on reported decisions made. The data from this present study provides a clear indication of the significance of the level of empathy for improving agreement on reported decisions made. It also indicates that when the skills or behaviours that demonstrate empathy are absent (or occur infrequently) then patient and dietitians are likely to disagree on their reporting of decisions made in the consultation.

12.6 Relationship Between Patient-centred Communication Skills and Empathy

The varying levels of empathy demonstrated in this study may reflect the different skill sets employed by the dietitians. The mixed-methods approach to the analysis was critical in exploring this interplay between skill set and empathy, as well as the subsequent impact of empathy on the type and flow of

information-giving exchanges that occurred within the consultation (Pedersen, 2009).

When dietitians in this present study explored the empathic opportunities raised by patients, this led to more statements of challenge being raised by the patients in the consultation. In addition, when dietitians reflected, summarised, and confirmed issues raised by patients, further statements of challenge were presented. Active-listening skills included specific exploratory questions that explored how patients felt, their ideas, views and opinions as well as current self-care management practices. The use of active-listening skills by dietitians promoted a greater number of empathic opportunities from patients in the consultation and supported the dietitian's expressions of empathy. This finding is consistent with Eide et al (2011) recent finding in nurse consultations: that patient-centred communication is needed to support patient's expressions of concern. The greater levels of expressed empathy to statements of challenge in this present study resulted in greater patient recall (p = .011) and greater agreement on reported decisions made (p = .008).

Active-listening skills were also employed in the *supportive/collaborative* information-giving exchanges and these skills helped in engaging the patient in dialogue. These exchanges were commonly seen to precede empathic opportunities and were followed by further expressions of empathy and *supportive/collaborative* information-giving exchanges. The success of the collaborative approach is said to rely on the effect created by the communication (Finset & Mjaaland, 2009; Norfolk et al, 2007). Indeed in this present study, the effect of these patient-centred communication skills was most

pronounced in productive consultations (those that achieved agreement on reported decisions made) with half of the information-giving exchanges in these consultations being identified as *supportive/collaborative*. Active-listening skills were therefore seen to support expressions of empathic opportunities, allowing a more productive dialogue to occur, as patients were listened to and supported in a non-judgemental manner (Del Piccolo et al, 2007; Entwistle & Watt, 2006; Poskiparta et al, 2001; Zimmermann et al, 2007). This resulted in further empathic opportunities and supportive/collaborative information-giving exchanges that enhanced recall as they focused on issues with personal relevance to the patient. Indeed, in this study the greater use of active-listening skills and collaborative working resulted in a greater number of decisions being reported by patients and dietitians, and greater agreement on reported decisions made. This data is of interest, as previous studies exploring supportive dialogue and empathy have focused on nurse/physician consultations and only one of these studies matched these measures to an outcome, which was a broad general health questionnaire and a measure of emotional distress (Del Piccolo et al, 2007; Entwistle & Watt, 2006; Poskiparta et al, 2001; Zimmermann et al, 2007). Validation of patient experience and greater empathy may result in fewer negative emotions and more positive ones, which in turn may impact on outcomes (Fogarty et al, 1999; Ong et al, 2000). This present study therefore provides explicit data on dietetic consulting skills and (further to previous studies) links the presence of these skills with an outcome measure. It seems that as agreement reflects patient engagement and skilful communication on behalf of the professional, then this greater patient engagement may lead to significant improvement in patient behaviour change and improved health outcomes (Eide et al, 2011; Street et al, 2005).

This mixed-methods approach has provided valuable insight into the patientcentred behaviours and communication skills and patterns of occurrence in consultations that promote patient interaction and agreement on reported decisions made. In addition, this data highlights the usefulness of the measure of agreement on decisions made in detecting the presence of patient-centred communication skills and behaviours. It has also highlighted differences in the level of empathy provided by the different empathic statements made by patients.

Dietetic consultations focus predominately on food and this is reflected in the high percentage of food decisions reported by patients and dietitians. This is an area known to be difficult for patients to implement in the long term (Peyrot et al. 2005; Travis, 1997; Van Dulmen et al, 1997; Webb et al, 1984; Williamson et al, 2000). The high frequency of statements of challenge in dietetic consultations reflects the difficulties that patients experience in the daily management of their condition, particularly with regard to food. Dietitians responded to these statements of challenge with high levels of empathy, which suggests that they are attuned to the daily difficulties that patients may be facing in managing their diet. Responding empathically to statements of challenge resulted in more focused and collaborative information-giving exchanges that explored the patient's perspective (Heisler et al, 2003; Wolpert & Anderson, 2001). This collaborative working helps to facilitate problem solving (Lorenz et al, 1996; Spahn et al, 2010) and has been shown to lead to greater levels of engagement and improved dietary self-care and glycaemic control (Glasgow et al, 1992; Rubin et al, 1989). This may explain the high number of reported decisions, as well as the high level of agreement on reported decisions, made in

consultations where significant levels of supportive/collaborative informationgiving exchanges were found. Unlike others (Kiuru et al, 2004), this present study provides insight into the differences that occur in communication skills and information-giving exchanges in relation to a defined outcome: agreement on reported decisions made. It provides further evidence for the usefulness of these patient-centred communication skills in practice, and illustrates how these skills interact with each other. When patients feel listened to and understood they report greater levels of satisfaction (Golin et al, 1996; Jackson et al, 2001). Although patient satisfaction did not correlate with empathy in this present study, patient satisfaction was found to correlate significantly with high numbers of statements of challenge made by patients (p = .023). A supportive environment where patients feel respected and empowered (Stewart, 1995) is more likely to encourage patients to verbalise their concerns (Jansen et al, 2010; Ong et al, 2000). In addition, the statements of challenge raised require a more probing response from the professional to allow difficulties and challenges to be explored in more depth (Rollnick, 1996; Stetson et al, 1992). This level of exploration and focus on immediate difficulties that patients are addressing may explain the high reporting of satisfaction associated with statements of challenge.

However, it is interesting to note that statements of emotion did not appear to need a high empathic response. The act of being listened to is reported as being therapeutic in itself (Ong et al, 2000). An acknowledgement that validates feelings expressed appears to be an appropriate response to emotions raised in this present study (Lang et al, 2000; Suchman et al, 1997; Zachariae et al, 2003). Too much focus on emotions may accentuate the daily difficulties

associated with living with diabetes and thereby increase diabetes-related distress (Nakahara et al, 2006) and anxiety (Eide et al, 2004; Rost & Roter, 1987). It may also displace time for information giving and could impact negatively on patient recall (Neumann et al, 2007). This was reflected here when dietitian 2 demonstrated high levels of empathy to statements of emotion and achieved low agreement rate on reported decisions made. In contrast, dietitian 3 achieved the highest rate of agreement on reported decisions made and demonstrated low levels of empathy to statements of emotion. These findings are consistent with earlier studies, which found that the acknowledgment of feelings allowed patients to focus more clearly on their agenda (Lang et al, 2000; Morse et al, 2008; Neumann et al, 2007; Suchman et al, 1997; Zachariae et al, 2003).

Different empathic opportunities may therefore require different levels of empathic response, suggesting the need for flexibility and very careful activelistening skills in the way that dietitians communicate and respond to patients. The different number of empathic opportunities observed reflects both the level of expressed empathy and the level of patient involvement, as demonstrated through the different types of information-giving exchanges used in the consultation.

Where empathy was reported as being low, there were fewer empathic opportunities (Davenport et al, 1987; Goss et al, 2004; Street et al, 1993). In addition, the use of active-listening skills was much lower, suggesting low patient engagement, which may result in patients remembering less (Jansen et al, 2010). This effect is reflected in this present study, where the greater use of

controlling information exchanges was associated with fewer empathic opportunities, lower use of patient-centred communication skills and low agreement rates on reported decisions made.

12.7 Does patient autonomy or self-efficacy affect successful consultation outcomes?

The different tools employed to measure and assess empathy may also explain why some studies have been able to link empathy with patient autonomy and self-efficacy (Fuertes et al, 2008; Levinson & Roter, 1993) and why this present study was unable to make these associations. In two studies that both used indirect measures to assess empathy, the questionnaires that were used explored professional perception of empathy (Fuertes et al, 2008) and patient perception of empathy (Levinson & Roter, 1993). Bylund & Makoul (2002) have suggested that questionnaires that measure perception of empathy may be measuring a much broader concept of empathy compared to the direct measures as employed in this present study. Pedersen (2009) also raised this issue in his review of empirical research into empathy in medicine. The more conservative measures used for coding empathy directly from recordings using the ECCS, rather than using perception questionnaires that rely on recall, may therefore have resulted in smaller variations in identified practice. Brace et al (2006) cite small variations in practice as possible reasons for lack of significant correlation data. These small variations may be a possible reason for the lack of correlation seen in this study.

However, a trend towards significance was noted for patient autonomy and empathic opportunities and this picks up a trend noted in an earlier study using the ECCS to measure empathy (Goodchild et al, 2005). Even allowing for a larger sample size, this trend did not reach significance. In this present study, lack of significance may have resulted from the changes that were made to the ECCS coding level, leading to the lower measured empathy scores.

12.8 Does training influence the levels of empathy seen and the application of patient-centred communication skills needed to support this?

Additional postgraduate training in education, communication skills, and behaviour change skills has long been recommended for dietitians (King et al, 2002; Larme & Pugh, 1998; Rosal et al, 2001; Stetson et al, 1992). Recent NICE guidance has highlighted the need for advanced communication skills training in order to effectively implement a behavioural approach (NICE, 2007). In this study the uptake of postgraduate training for participating dietitians was variable. Data on dietetic postgraduate training in behaviour change and communication skills is not routinely collected, so it is difficult to estimate these training needs in the profession as a whole.

As issues that will impact on delivery, Doherty et al (2000) highlight the frequency of use of new skills and their perceived importance, as well as adequacy of training. In addition, skills needed for effective communication and behaviour change have been shown to be hard to utilise consistently in that they require ongoing reflection in order to ensure application (Doherty et al, 2000; Pill et al, 1999). This is further highlighted in this present study. Here, in comparison to the dietitian who had received the least training in this area, the dietitian with the fewest years of clinical experience, but the most frequent and recent training in behaviour change and communication skills, demonstrated

significantly greater levels of empathy in the consultation and the greatest agreement on reported decisions made compared.

These findings reflect the difficulties that professionals have in maintaining skills once trained and provide further compelling evidence for regular review and training to ensure that patient-centred communication skills are embedded in practice.

12.9 Limitations and future research directions

This study reviewed patients from a limited number of dietitians. It therefore cannot be considered to be a true representation of dietetic practice in the UK. The study was located in the South West and may illustrate cluster effects, rather than a pattern of practice across the UK profession as a whole. In addition, this study is an observational cross-sectional design and therefore conclusions around causality cannot be drawn from the correlations obtained (Field & Hole, 2003; Ong et al, 1995). The gualitative findings presented here on communication skills and information giving are consistent with other findings and support other research in this area (Goss et al, 2004; Neumann et al, 2009; Norfolk et al, 2007; Rubin et al, 2002). The findings here also support Neumann et al's (2009) model of empathic communication, which proposed a pathway between empathy, enhanced communication, and improvement in short-term outcomes. Longitudinal studies are now required to find whether agreement on reported decisions made does in fact translate into improved health outcomes and long-term dietary change as suggested (Kravitz et al, 1993).

The small sample size here may also have hampered the detection of correlations between patient autonomy and self-efficacy. Larger sample sizes of professionals and patients would allow the application of structured equational modelling to data analysis, and this would enable the causal pathways between empathy, patient autonomy and self-efficacy to be explored in more depth (Burke et al, 2003; Senécal et al, 2000). This present study highlights the complex nature of patient-centred communication and it is therefore important to consider a range of concepts when exploring communication effectiveness (Newman et al, 2004). These include the interactive nature of the identified skills, and the essential need to view these skills collectively in order to understand their impact more fully on communication effectiveness and outcomes. The use of larger populations could strengthen correlations between empathy and agreement. Techniques such as mixed methods are essential to improving understanding because outcomes are influenced by: low frequency of occurrence of empathy in consultations, multiple communication behaviours, patient and professional factors.

Although mixed methodology provides some challenges to the researcher, larger and more detailed studies of this nature are required. This will help to improve understanding of the interplay that the communication skills and behaviours may have with the models, theories, and concepts that are applied to the delivery of patient-centred care.

In addition, further studies will help to clarify the usefulness of measuring agreement on reported decisions made in the consultation, in highlighting lack

of patient-centred communication skills in other professional consultations. Further training in patient-centred communication skills may result in greater levels of agreement on reported decisions made in the consultation (Parkin et al, 2006). Further work is also required to test the validity of the changes made to the ECCS coding system and its application to consultations by other professionals.

Qualitative analysis is reported to have a research bias and care needs to be taken in interpretation of data to ensure that this does not occur (Creswell, 2009). Although the researcher was careful to limit bias by keeping outcome data separate from the qualitative data during analysis, it would be useful to explore the identified information-giving themes in a larger set of data with more researchers to increase the validity of the application of this process further. This is particularly important as the identified themes for information-giving exchanges provide clear direction on the communication behaviours within the consultations.

There is little evidence to show that recording consultations influences the behaviour of the patients or the professionals involved (Coleman, 2000). However, it could be argued that it introduces bias in those patients and professionals who agree to take part (Zandbelt et al, 2006). Despite this potential for bias, exceptional practice did not occur and significant variations in practice were observed including consultations where complete disagreement on reported decisions occurred. This appears to suggest that the audio recording of consultations had little impact on the usual care delivery of the

dietitians taking part. Themessl-Huber et al (2008) reached a similar conclusion.

12.10 Conclusions

There is a dearth of published data on the communication skills used or the effectiveness of dietetic consultations (Cant & Aroni, 2008; Whitehead et al, 2009). This study therefore provides a unique insight into the communication process that occurs in dietetic consultations and provides strong evidence to support the use of patient-centred communication. The mixed-methods approach taken provided valuable insight into the mechanisms by which empathy, active-listening skills and information giving interact with each other in the consultation. Highlighting the importance of these skills for effective patientcentred communication, as well as identifying the process by which these communication skills lead to agreement on reported decisions made in the consultation, add meaning to the data set observed (Epstein et al, 2007). The range of skills needed in the process of effective patient-centred consultation reflect its complex nature. In addition, specific behaviours such as agenda setting, decision recall towards the end of the consultation and action planning were all associated with greater agreement on reported decisions made. Although these findings are tentative, due to the small numbers used in the qualitative analysis, they provide support for the growing evidence base around patient-centred communication for effective consultations (Epstein et al, 2005). They also provide much needed evidence to support effective communication in dietetic consultations.

In addition the simple pen and paper measure of agreement on reported decisions made has been shown to be a reliable indicator of patient-centred communication. When key patient-centred communication skills of empathy, active listening, and *supportive/collaborative* information-giving exchanges were present, greater levels of agreement on reported decisions were made. Conversely, when these skills were lacking or absent, disagreement on reported decisions occurred.

It has been suggested that greater agreement on reported decisions made in the consultation may be a marker for success of the consultation leading to improved patient outcomes (Heisler et al, 2003; Kravitz et al, 1993). The level of agreement on reported decisions is therefore a viable method of engaging professionals in reflection on their patient-centred communication skills. These findings add support to the concept of the *therapeutic alliance* (Norfolk et al, 2007) and the *comprehensive model of empathic communication* (Neumann et al, 2009). Longitudinal studies are needed to substantiate the links seen between empathy, communication behaviours, and information-giving exchanges where there is agreement on reported decisions made. Such studies may confirm that agreement on reported decisions made translates into long-term improvement in health outcomes (Kravitz et al, 1993).

Professionals are increasingly expected to demonstrate their impact on patient care through improvements in health outcomes (BDA, 2009; BDA, 2011; Lacey & Pritchett, 2003; NHS, 2011). The outcome measure of agreement on reported decisions made is significant in that it provides a measure that relates directly to the dietitians interaction with the patient. Although guidance is

provided to dietitians on care delivery through the nutrition and dietetic care process (BDA, 2009; Lacey & Pritchett, 2003), this only provides information on each of the stages; assessment, nutritional diagnosis, intervention, evaluation and monitoring. This study is therefore significant as it examines the communication process that supports movement through the nutrition and dietetic care process, for chronic disease management, leading to improvement in the outcome measure of agreement on decisions made.

In addition, the data from this study suggests a need to consider post registration training in communication skills/behaviours for practicing dietitians. Although dietitians are required to engage in continued professional practice (CPD) this does not involve a compulsory update in communication skills or behaviours. Data from this study suggests that regular communication skills/behaviour training as part of a compulsory requirement for dietetic CPD is needed. Diabetes educators in Australia are required to register every 3 years, and in the United States every 5 years, with evidence to support continuing professional practice (ADEA, 2011; AADE, 2011). Based on the findings from this study it would not be unreasonable therefore, for dietitians working with chronic disease management to provide evidence of training in communication skills/behaviour every 3 years as part of their ongoing CPD and professional registration requirement.

In summary this study provides evidence for the specific skills/ behaviours needed for an effective patient-centred consultation for chronic disease management, and the rationale for regular review and development of

communication skills/behaviour training for dietitians to maintain competence

and improve patient outcomes.

CHAPTER THIRTEEN: IMPLICATIONS FOR PRACTICE AND FUTURE RESEARCH

The differences highlighted in the dietetic consultations reflect significant differences in the practice of the dietitians taking part in this study. This appears to be related to the amount of training received in communication and behaviour change, rather than years in practice or years working in the specialist area. The dietitian with the least experience in diabetes and the fewest number of years in practice had the highest level of empathy and agreement on reported decisions made. The same dietitian also had the most frequent and recent training in behaviour change and communication skills. This resulted in a greater demonstration of empathy in her consultations and higher levels of agreement on reported decisions made. This finding suggests that to remain competent in the delivery of patient-centred communication, professionals need to receive ongoing training and review to ensure application of the appropriate skill set.

Although this study involved a small number of dietitians, others have highlighted the need for regular training and assessment of skills in practice (Lorenz et al, 1996). Patient-centred communication skills and behaviours have been shown to be difficult to utilise consistently, requiring ongoing reflection to ensure application (Doherty et al, 2000; Pill et al, 1999) and embedding of these skills in practice (Hauenstein et al, 1987; Roach et al, 1992). The acquisition of new communication skills involves behaviour change in practitioners (Rollnick, Mason & Butler, 1999) and may present challenges if professionals are to shift away from their own well practised consulting styles (Moran et al, 2008;

Resnicow et al, 2001). Habit is therefore an issue (Rollnick, 1996) and this may prove resistant to change (Hulsman et al, 1999). The practitioner may also perceive these skills as too time consuming (Doherty et al, 2000; Lorenz et al, 1989) and therefore revert back to old methods before new skills have had a chance to become embedded (Hulsman et al, 1999). The following will impact on delivery (Delahanty & Halford, 1993): frequency of the use of new skills, perceived importance of new skills, monitoring of the use of the skills and adequacy of training.

Challenges over the perceptions of the professional's role and responsibilities (Francis et al, 2008) may also impact on effective skill utilisation (Pill et al, 1999; Resnicow et al, 2001), as will competing perceptions of real work and the importance attached to psychosocial aspects of care (Hauenstein et al, 1987; Jarret & Payne, 1995). The attitudes of professionals may therefore need to change in order to improve consultation effectiveness (Levinson & Roter, 1995). The extent to which dietitians feel supported in the workplace may also reflect their ability to take on board and use psychosocial skills in their practice (Francis et al, 2008).

Effectiveness in the delivery of patient-centred communication requires regular practice and review of skills used in order to ensure successful application to the complex area of dietary behaviour change (Forsetlund et al, 2009). This review of practice needs to focus on the reality of practice, not the perception, as the two may differ (Parkin & Skinner, 2003). Peer support will also assist in the embedding of skills in practice. The findings from this study are unique in that it is the first of its kind to use mixed methodology to explore the

communication skills used by dietitians in practice. In addition, the outcome measure of agreement on reported decisions made in the consultation was found to be consistently associated with the presence of empathy and patientcentred communication behaviours. Furthermore, when these skills were lacking, or absent, low levels of agreement on reported decisions made occurred.

The pen and paper method of checking patient and professional agreement on reported decisions made, through the use of simple written records, has the potential therefore to be a cheap, quick and easy marker of the presence or absence of patient-centred communication skills and behaviours in consultations. It requires no skill or special equipment and would be easily applied by any professional in a range of different settings. Agreement on reported decisions made is a more reliable marker than perception of practice and provides a quick method of assessment of the need to review skills used in the consultation. In addition, this study provides a clear indication of the patient-centred communication skills needed to support agreement on reported decisions made, providing direction for the necessary skills training to support professional's delivery of patient-centred care.

However, the frequency with which the communication process is reviewed is debatable. Roach et al (1992) noticed that overall improvement in teaching skills following training declined after three months, and whether this is true for consulting skills has yet to be established. Regular review and training may help to improve skills used (thereby reducing the level of discrepancy) by increasing agreement and reducing the reporting of unmade decisions in

consultations (Parkin et al, 2006). This would result in more accurate documentation in patient notes, providing a clear measure of the outcome effectiveness of the professional's consultation. In addition, improved recall may improve long-term health outcomes, thus increasing the clinical importance of the outcome measure of agreement on reported decisions made.

This study suggests that a minimum amount of time is needed in the consultation before patients become actively engaged. Dietetic services wishing to improve their delivery of effective patient-centred communication in order to improve health outcomes will, need to consider the potential implications of longer consultation times on resource planning.

Further research is needed, with dietitians and other professionals, to determine whether the relationship between agreement on reported decisions and patientcentred communication skills is consistent for larger populations of patients and for the variety of professionals involved. Additional training to improve patientcentred communication will result in more efficient use of patient-centred communication skills, with the potential for more skilled communicators delivering patient-centred communication in a shorter time frame. Studies that have explored GP consultations have looked at the impact of longer consultations times (Beisecker & Beisecker, 1990; Moran, et al, 2008) to see whether this reduces overall health cost and improves patient care as suggested (Wilson & Childs, 2006). Equally, there is no published work on ideal length of time for a dietetic consultation. Longitudinal studies are needed to ascertain whether agreement on reported decisions does in fact result in improved outcomes.

Successful patient-centred communication is achieved through the constant review of the skills and strategies needed to support this, alongside a structure that involves the patient as an active participant in this dialogue (Rollnick, 1996). The patient-centred communication skills employed in listening to patients will help to determine how active patients wish to be in this process and will result in a course of action that reflects their needs (de Haes & Bensing, 2009). The NICE (2007) guidelines on behaviour change at population and community levels state that advanced communication skills are needed for professionals to implement behavioural approaches effectively. These behavioural approaches will only be successful if the professionals employing them have the necessary communication skills to implement them appropriately. Lack of patient-centred communication skills may therefore explain the varied levels of success seen with the application of behaviour change theories and strategies in practice (Abraham & Michie, 2008).

The marker of agreement on reported decision made in the consultation provides a quick and simple tool that can highlight the need for communicationskill review. In addition, this study has identified key patient-centred communication skills that are needed to support agreement on reported decisions made. These are empathy, active-listening skills, exploratory questions, and *supportive/collaborative* information-giving exchanges. Patient self-care management can be improved through the regular review of the patient-centred communication skills employed by heath care professionals. This study provides evidence for a simple and quick method that will assist professionals in achieving this aim.

APPENDIX

1.1	RECRUITMENT, PARTICIPANTS AND METHOD FOR STUDY	319
	EXPLORING TOOLS SUITABLE FOR ANALYSING DIETETIC	
	CONSULTATIONS	
1.2	CODING FOR EMPATHY USING THE EMPATHIC	323
	COMMUNICATION CODING SYSTEM (ECCS)	
1.3	SUMMARY OF CODES OBTAINED USING ECCS	326
1.4	OVERVIEW OF CODING CATEGORIES FOR RIAS	327
1.5	SUMMARY OF CODES USING RIAS	328
1.6	SUMMARY OF CODES USING VERONA	329
1.7	SUMMARY OF CODES OBTAINED USING REFLECTIVE	330
	PRACTICE CODING SYSTEM	
2.1	PATIENT LETTER AND INFORMATION SHEET	331
2.2	PATIENT CONSENT FORM	333
2.3	DIETITIAN CONSENT FORM	334
2.4	DIETITIAN QUESTIONNAIRE BOOKLET	335
2.5	PATIENT QUESTIONNAIRE BOOKLET	338
2.6	INITIAL CODING FRAME FOR IDENTIFYING DECISIONS	344
2.7	ILLUSTRATION OF THE TEN SUBCATEGORIES FOR FOOD	346
2.8	AMENDED EMPATHY CODING LEVELS FOR ECCS	347
3.1	CODING MANUAL FOR QUALITATIVE ANALYSIS	349

APPENDIX 1.1

RECRUITMENT, PARTICIPANTS AND METHOD FOR STUDY EXPLORING TOOLS SUITABLE FOR ANALYSING DIETETIC CONSULTATIONS Exploring tools suitable for analysing dietetic consultations

In the recruitment of patients and dietitians to the exploratory study, the following outlines the process that was used to identify tools suitable for analysing patient-centred communication skills in dietetic consultations.

Recruitment

Two specialist diabetes dietitians working in a diabetes centre agreed to have their consultations used for exploratory analysis. Recruitment of participants took place over a two-week period. Portsmouth & Isle of Wight Research Ethics Committee provided ethical and clinical governance approval. Individuals attending for a scheduled out-patient appointment at a specialist diabetes centre were invited to participate in the study. Posters were displayed around the out-patient waiting area to inform patients that the study was taking place. On arrival, prospective participants were provided with an information sheet outlining the purpose of the research and how the data was to be used. Fifteen patients with diabetes who were attending the dietetic clinic agreed to participate. Participants were made aware that refusal to take part would not impact on their usual care delivery.

Participants

The following Inclusion and exclusion criteria were used in the selection of participants (patients and dietitians). Patients with diabetes who were attending scheduled out-patient appointments with the diabetes dietitians were eligible to take part. These routine out-patient clinics would have consisted of new and

follow-up patients with either type 1 or type 2 diabetes. Patients were excluded from taking part if they did not have fluency in English, were under the age of 16, had learning difficulties, mental health issues or a history of drug/alcohol abuse. Two female dietitians who provided out-patient diabetes care for between 5-15 years agreed to take part in this exploratory study. Each dietitian signed a consent form (see below).

Method

Posters displayed around the waiting area informed patients of the study and information sheets that were handed out on arrival at the clinic. I was on hand to answer any additional questions that the patients may have had about the study. If individuals agreed to take part in the study, they signed a consent form before they entered the clinic room and the whole consultation was audio-recorded (see below for consent form). The participants were reminded that the recording could be erased at any time during the consultation if they changed their mind and wished to withdraw from the study. The dietitian then undertook the consultation following usual procedures. At the end of the consultation, the audio recordings were collected and stored in a locked filing cabinet in accordance with the Data Protection Act 1998 on storage of personal information. Audio recordings were then transcribed and coded using the selected coding tools. Consent forms for dietitians and patients are below.

PATIENT CONSENT FORM TO ALLOW CONSULTATION TO BE AUDIO RECORDED FOR RESEARCH AND DEVELOPMENT PURPOSES

Date	Identification Number
Patients sexM/F	Diabetes type 1/ type 2
Age	Presence of complicationsyes/no
BMI	HbA1c

The dietitian who is seeing you today is hoping to have their communication skills reviewed. To do this we would like your help by giving your permission to audio-record today's session with the dietitian. The audio recordings will form part of a research project looking at how dietitians communicate with patients.

The recorder will be switched off at any time you wish.

Only the people directly involved in the research and the dietitian you are seeing today, will listen to the recording. It will only be used to evaluate the dietitian's communication skills. The recording will be securely stored and is subject to the same degree of confidentiality as your medical records.

The security and responsibility of the audio recording is the responsibility of the research team to whose care it is entrusted.

You do not have to agree to your session being recorded. If you do not want your session to be recorded, please tell the dietitian. This is not a problem and will not affect your session today with the dietitian. But if you do not mind your session being recorded, we are grateful to you. If you wish you may listen to the audio recording before confirming your consent.

If you consent to this session being recorded please sign below.

Thank you very much for your help.

If at the end of the consultation you wish to have the recording erased, please let us know.

TO BE COMPLETED BY PATIENT

I have read and understand the above information and give my permission for the session to be audio recorded.

Date Signature of patient

Date

Signature of researcher
Consent form for Dietitians taking part in the audio recording of consultations for research and development purposes.

Identification number.....

Study aim:-

This study is designed to look at the communication skills used by dietitians in consultations with people who have diabetes.

To do this I would like your help by giving your permission to audio record consultations with people who have diabetes. The audio recordings will form part of a research project looking at how dietitians communicate with people who have diabetes.

The audio recording can be listened to by you at your request and will only be listened to by the people directly involved in the research. The recording will be securely stored and is subject to the same degree of confidentiality as medical records.

The security and responsibility of the audio recording is the responsibility of the named researcher to whose care it is entrusted.

If you consent to your sessions being recorded, please sign below.

Thank you very much for your help.

TO BE COMPLETED BY DIETITIAN

I have read and understand the above information. I agree to take part in the above study and I am aware that the results of this study are for research purposes only. The outcomes of this study can therefore not be used by anyone in a senior position to criticise my performance or be used as part of my individual performance review. Agreement to take part or not to take part in this study is down to the individual, and will not and cannot be used as a marker of commitment to research in the Nutrition and Dietetic department.

Date	Signature of dietitian			
Date	Signature of researcher			

CODING FOR EMPATHY USING THE EMPATHIC COMMUNICATION CODING SYSTEM (ECCS)

Coding for empathic opportunities using the ECCS, examples in bold text have

been taken from the coded transcripts, italics are examples used in the original

ECCS coding manual (Bylund & Makoul, 2002).

Statement of Emotion: Emotion is defined as "an affective statement of consciousness in which joy, sorrow, fear or the like is experienced."

"I'm just scared because I never went through nothing – I've never had nothing wrong with me"

"Yeah I definitely eat, my emotions, to help. I tend to eat when I'm upset."

Statement of Progress: the patient states or describes a positive development in physical condition that has improved quality of life, a positive development in the psychosocial aspect of the patient's life, or a recent, very positive life changing event.

" I've been exercising more than the last time I saw you"

"Didn't go yesterday, no. I am missing it, it's obviously part of the routine that really works"

Statement of Challenge: the patient states or describes a negative effect a physical or psychosocial problem is having on the patients' quality of life, or a recent devastating, life-changing event.

"But sometimes it's just hard eating three pounces of meat you know what I mean?"

"At the weekends sometimes I have gluten-free porridge because I have time to make it, but during the week I pretty much end up eating Rice Crispies which isn't much good from the point of the view of my diabetes."

Table A. 45 Codes for empathic opportunities

Coding system for empathic responses made to patient empathic opportunities

(ECCS), examples in bold text have been taken from the coded transcripts,

italics are examples used in the original coding manual (Bylund & Makoul

2002).

Level 6: Statement of shared feeling or experience

Where the health professional makes an explicit statement that they either share the patients emotion or have had a similar experience.

"I understand how scary this must be for you. My husband recently had a biopsy and we were really scared"

There were no level 6 codes in the consultation recordings

Level 5: Confirmation

Where the health professional conveys to the patient that what they are feeling and describing is legitimate ie, congratulatory remarks, or acknowledging the difficulties the patient is having.

"You sound like you are very busy, I can see why it would be tough for you to find time to exercise"

"Yes, it sounds like you're trying very, very hard and you're very conscientious."

Level 4: Acknowledgement with pursuit

This is a response that explicitly acknowledges the central issues in the empathetic opportunity and conveys that the physician has "heard" the patient. Level 4 responses convey this by pursuing the topic, asking a question, elaborating on the point raised by the patient.

"You mention you've been feeling sad. Would you tell me more about that?"

"OK, what affect does it have on your diabetes? You mentioned twice that it's not good for your diabetes"

Level 3: Acknowledgement

This response also explicitly acknowledges the central issue in the empathic opportunity. Acknowledgement includes restatements of what patients have said. Information, advice, or offers help.

"Ok, so you've been cutting back on fats".

"So your key emotions throughout your eating, is boredom or tiredness?"

Acknowledgment should also be coded when the professional response to the empathic opportunity and provides verbal back-channeling cues such as "mm-hmm" "yes," whilst patient is talking about the empathic opportunity. It should be clear that the professional is actively listening to the patient.

"Yes, yes, ok".

Level 2: Implicit recognition of patient perspective

These responses concentrate on peripheral aspects of the empathic opportunity. These statements tend to be more content-base, not dealing directly with the patients concerns. These responses may also include questions or advice.

Patient: "this headache makes it difficult for me to work"

Professional: "yes, how is the insurance business lately?"

Patient: "Didn't go yesterday, no. I am missing it, it's obviously part of the routine that really works." Dietitian: "Yeah. So was that because your brother wasn't around?"

Level 1: Perfunctory recognition of patient perspective

A professional's automatic, script-type response (eg "uh-huh") to an empathic opportunity while the physician is doing something else, often with the health professional having their body orientated away from the patient.

There were no level 1 codes in the consultation recordings

Level 0: Denial of patient perspective

The professional either ignores the patient's empathic opportunity or makes a disconfirming statement. Such responses also include an immediate topic change, or not responding but focusing on a physician task.

Patient: "So I downed the morning one to 26, so I'm doing 26 in the morning and 26 at night and I'm fine at the moment, I haven't had that feeling."

Dietitian: "OK. OK. We've got about 40 minutes today to discuss whatever you want to discuss about your diabetes."

Table A.46 Coding levels for empathic responses ECCS

SUMMARY OF CODES OBTAINED USING ECCS

Recordings	Total empathic opportunities	Statement of challenge (SC)	Statement of progress (SP)	Statement of emotion (SE)	Mean response to SC	Mean response to SP	Mean response to SE	Mean empathic response
1	24	7	12	5	3.58	4.17	3.2	3.65
2	29	6	16	7	3.33	3.3	2.89	3.17
3	11	9	2	0	3.2	3	0	3.11
4	20	10	6	4	3.2	2.83	4.4	3.51
5	7	6	0	1	3	0	1	2
6	12	3	7	2	1	3	3	2.33
7	19	16	0	3	3.25	0	3.67	3.46
8	15	5	9	1	3.2	3.11	1	2.44
9	19	17	2	0	3.10	3.5	0	3.30
10	3	0	1	2	0	3	2.5	2.75
11	16	14	1	1	3.07	4	3	3.36
12	19	13	4	2	2.77	2.25	3.5	2.84

Table A.47 Summary of codes obtained using the ECCS

OVERVIEW OF CODING CATEGORIES FOR RIAS

Socio-emotional Exchange	Task-Focused Exchange
Personal remarks, social	Transition words
conversation	Gives orientation, instructions
Laughs, tells jokes	Paraphrase/Checks for understanding
Shows approval – direct	Bid for repetition
Gives compliment – general	Asks for understanding
Shows agreement or	Asks for opinion
understanding	Asks Questions (closed-ended) – medical condition
Back-channel responses	Asks Questions (closed-ended) – therapeutic
Empathy	regimen
Shows concern or worry	Asks Questions (closed-ended) – lifestyle
Reassures, encourages or	Asks Questions (closed-ended) – psychosocial-
shows optimism	feelings
Legitimises	Asks Questions (closed-ended) – other
Partnership	Asks questions (open- ended) – medical condition
Self-disclosure	Asks questions (open- ended) – therapeutic regimen
Shows disapproval – direct	Asks questions (open- ended) – lifestyle
Shows criticism – general	Asks questions (open- ended) – psychosocial-
Asks for reassurance	feelings
	Asks questions (open- ended) – other
	Gives information – medical condition
	Gives information - therapeutic regimen
	Gives information - lifestyle
	Gives information - psychosocial
	Gives information - other
	Counsels or directs behaviour – medical
	condition/therapeutic regimen
	Counsels or directs behaviour -
	lifestyle/Psychosocial
	Requests for services or medication

Table A.48 Overview of coding categories for RIAS (Roter, 2005)

SUMMARY OF CODES USING RIAS

Socio-emotional	Dietitian	Patient
exchanges		
Personal	35	0
Laughs	26	50
Approval	55	24
Agreement	110	613
Back-channeling	196	0
Empathy	9	0
Concerns	11	110
Reassurances	94	27
Legitimises	3	0
Partnership	2	0
Self-disclosure	2	0
Disapproval	10	5
Criticism	0	5
Asks	0	6
reassurance	U	Ö

Task-focused	Dietitian	Patient
exchanges		
Trans	262	333
Orient	23	0
Check	3	24
Bid	1	1
Asks	30	18
understanding	39	10
Asks opinion	52	2
Medical	25	0
Therapy	194	8
Lifestyle	99	1
Psychosocial	2	0
Other	6	0
Open med	20	0
Open therapy	68	8
Open lifestyle	94	17
Open	10	2
psychosocial	15	Z
Open other	25	2
Gives med	87	155
Gives	0	17/
psychosocial	0	174
Gives other	149	102
Direct med	132	0
therapy		
Direct lifestyle	13	0
Asks service	0	5

 Table A. 49 Summary of codes obtained using RIAS

SUMMARY OF CODES USING VERONA

	Code categories					
Recordings	Information gathering	Patient involving	Patient facilitating	Patient supporting	Patient education	Negative talk
1	58	23	48	11	25	0
2	11	14	28	10	21	3
3	59	17	43	12	26	0
4	29	4	49	16	15	1
5	54	11	39	13	66	13
6	18	6	37	8	11	2
7	16	14	34	13	84	4
8	16	7	31	15	30	2
9	24	9	34	7	42	0
10	19	14	40	12	117	5
11	26	20	28	10	88	0
12	40	12	59	4	68	0
Total	370	151	470	131	592	30

Table A.50 Summary of codes obtained using Verona

SUMMARY OF CODES OBTAINED USING THE REFLECTIVE PRACTICE CODING SYSTEM

Becordings	Total number of	Total +ve	Total -ve
Recordings	responses	scores	scores
1	169	47	3
2	96	22	7
3	154	18	23
4	121	17	13
5	197	13	18
6	113	17	2
7	177	12	7
8	117	5	1
9	107	16	9
10	219	5	2
11	191	16	8
12	183	24	4

Table A.51 Summary of codes obtained using reflective practice coding system

PATIENT LETTER AND INFORMATION SHEET



Faculty of Health and Social Work School of Health Professions University of Plymouth

Dear Sir/Madam

I am writing to notify you of some research that is being conducted in the dietetic clinic and to ask you whether you would be interested in taking part.

My name is Tracey Parkin and I work at the University of Plymouth, I am conducting research into how dietitians communicate with patients. The research is called

Improving Healthcare Professionals' Consultation skills to people with Diabetes

In order to do this I need to audio-record consultations between dietitians and patients so that I can listen to how the dietitians speak to patients.

I would also like to find out how patients feel about the consultation and so will be asking patients to fill in a short questionnaire at the end of the visit with the dietitian. This questionnaire should take no more than 10- 15 minutes to complete.

I am hoping that this information will help our understanding of how dietitians communicate to patients and provide information on ways to improve dietitians' communication skills.

Further details are provided on the patient information sheet (enclosed).

At your clinic visit you will be asked whether you would like to take part in this study.

If you agree a consent form asking for your permission to audio-record the consultation will be available at the clinic visit for you to sign.

I will also be available to answer any further queries that you may have.

Thank you very much for your help.

Yours Sincerely

Tracey Parkin BSc (Hons) RD, FETC, AdDP

Lecturer in Dietetic Practice

PATIENT INFORMATION SHEET



Faculty of Health and Social Work School of Health Professions University of Plymouth

Patient Information Sheet

Improving Healthcare Professionals Consultations

My name is Tracey Parkin and I work at the University of Plymouth I am conducting research into how dietitians communicate with patients. I am hoping to use this information long-term to try and improve the communication skills of healthcare professionals.'

In order to do this I need to audio-record consultations between dietitians and patients so that I can listen to how dietitians speak to patients.

As well as listening to how dietitians speak to patients I would like to ask patients about their experience of the consultation.

I will therefore be asking patients to complete a short questionnaire at the end of the visit with the dietitian. This questionnaire should take no more than 5 minutes to complete.

All information collected is assigned a code to ensure that audio recordings and questionnaires remain anonymous so that no one taking part in the study can be identified.

This information will only be used to evaluate the dietitian's communication skills.

The recording will be securely stored and is subject to the same degree of confidentiality as your medical records.

If you change your mind and wish to withdraw your consent during the consultation please inform the dietitian and the audio-recording will be erased immediately.

You do not have to agree to your session being recorded and you do not have to complete a questionnaire. Deciding not to take part in this research will not impact on your level of care or alter the session that you have with the dietitian.

If you decide to take part by allowing your consultation to be audio-recorded and by completing a short questionnaire at the end of the session we would be grateful.

PATIENT CONSENT FORM



Faculty of Health and Social Work School of Health Professions University of Plymouth

Date

Centre Number

Identification number

Improving healthcare Professionals' Consultation Skills to People with Diabetes

Patient Consent form to allow consultation to be audio-recorded	Please tick
for research	to confirm
I confirm that I have read and understand the information sheet dated	
January 2008 for the above study	
I have had the opportunity to consider the information, ask questions	
and had these answered satisfactorily	
I understand that my participation is voluntary and that I am free to	
withdraw at any time, without giving any reason.	
Agreement to take part or not to take part in this study is down to me,	
and will not and cannot be used as a marker of my commitment to	
research in the Nutrition and Dietetic department	
I understand that if I decide to withdraw from the study the audio	
recordings and questionnaires will be destroyed.	
I understand that only the people directly involved in the research will	
listen to the recording.	
I understand that the information collected will only be used to evaluate	
the communication skills of dietitians. The recording will be securely	
stored and is subject to the same degree of confidentiality as medical	
records.	
I am aware that the results of this study are for research purposes only.	
The outcomes of this study can therefore not be used by anyone in a	
senior position to criticise my performance or be used as part of my	
individual performance review.	
I agree to take part in this study	
I agree to be audio-recorded	

Name of dietitian	date	signature
Name of researcher	date	signature

DIETITIAN CONSENT FORM



Faculty of Health and Social Work School of Health Professions University of Plymouth

Date

Centre Number

Patient Identification number

Improving Healthcare Professionals' Consultation Skills to patients with diabetes

Patient Consent form to allow consultation to be audio-recorded for research

			Please tick
			to confirm
I confirm that I have read and	understand the info	rmation sheet dated	
January 2008 for the above st	udy		
I have had the opportunity to c and had these answered satis	consider the information factorily	tion, ask questions	
I understand that my participat	tion is voluntary and	I that I am free to	
or legal rights being affected.	giving any reason, v	vitnout my dietetic care	
I understand that if I decide to recording and guestionnaire w	withdraw from the sill be destroyed.	study the audio	
I understand that only the peo	ple directly involved	in the research will	
listen to the recording. It will c	only be used to eval	uate the dietitian's	
communication skills. The rec	ording will be secu	ely stored and is	
subject to the same degree of	confidentiality as m	y medical records.	
I agree to take part in the abov	ve research study		
I agree to be audio-recorded			
Name of patient	date	signature	

Name of researcher

date

signature

DIETITIAN QUESTIONNAIRE BOOKLET



Faculty of Health and Social Work School of Health Professions University of Plymouth

Date

Patient ID Number

Dietitian Questionnaire booklet

The questions in this booklet are designed to help identify which aspects of the consultation with the dietitian enable the patient to manage their diabetes better.

Please answer **all** the questions in the booklet and only make **one** choice per item.

Please answer the questions as honestly as you can.

Please circle the number that best matches your response. All responses to the questions are confidential and will not be shown to patients or other members of staff.

Additional Patient details

Male / female

D.O.B
Type 1 diabetes / Type 2 diabetes
How long has the patient had diabetes
Does the patient have any diabetic complications
New appointment / follow up appointment

Please could you write down any decisions that were made in the consultation today

 	 ••••••	

These questions are related to how you felt during the consultation. We would like to know more about how you felt about your encounter with your patient. Your responses are confidential. Please circle the number that best represents your answer.

1. I feel that I provided my patient with choices and options.

1	2	3	4	5	6	7
Strongly disagree		Ne	eutral			Strongly agree

2. I feel that I understood my patient.

1	2	3	4	5	6	7
Strongly		Ne	eutral			Strongly
disagree						agree

3. I felt that I was able to convey confidence in my patient's ability to make changes.

1	2	3	4	5	6	7
Strongly		Ne	eutral		:	Strongly
disagree						agree

4. I encouraged my patient to ask me questions.

1	2	3	4	5	6	7
Strongly		Ne	eutral		;	Strongly
disagree						agree

5. I listened to how my patient would like to do things.

1	2	3	4	5	6	7
Strongly		Ne	eutral		;	Strongly
disagree						agree

6. I try to understand how my patient sees things before suggesting new ways to do things.

1	2	3	4	5	6	7
Strongly		Ne	eutral		;	Strongly
disagree						agree

THANK YOU FOR COMPLETING THE QUESTIONNAIRE

APPENDIX 2.5 PATIENT QUESTIONNAIRE BOOKLET



Faculty of Health and Social Work School of Health Professions University of Plymouth

Date Patient ID Number

Questionnaire booklet

The questions in this booklet are designed to help identify which aspects of the consultation with the dietitian enable you to manage your diabetes better.

Please answer all the questions in the booklet and only make one choice per item.

Please answer the questions as honestly as you can.

Please circle the number that best matches your response. All responses to the questions are confidential and will not be shown to your dietitian. consultation today.

Please could you write down any decisions that were made in the

A) We would like to know how confident you are in doing certain activities. For each of the following questions, please choose the number that corresponds to your confidence in your ability to do the tasks regularly at the present time.

1. How on hours	confid every	lent do y day,	o you inclu	feel th ding b	nat yo preakfa	u can ast?	eat yo	our me	eals ev	very 4	to 5
Not at all confident	1	2	3	4	5	6	7	8	9	10	totally confident
2. How confident do you feel that you can follow your diet when you have to prepare or share food with other people who do not have diabetes?											
Not at all confident	1	2	3	4	5	6	7	8	9	10	Totally confident
3. How of to eat	confid when	lent de 1 you a	o you are hu	feel th Ingry (nat yo (for ex	u can cample	choo: e, sna	se the cks)?	appro	opriate	e foods
Not at all confident	1	2	3	4	5	6	7	8	9	10	Totally confident
4. How o 5 time	confid es a w	ent do eek?	o you	feel th	nat yo	u can	exerc	ise 15	5 to 30	minu	tes, 4 to
Not at all confident	1	2	3	4	5	6	7	8	9	10	Totally confident
5. How o blood	confid suga	lent de r leve	o you I from	feel th drop	nat yo ping v	u can vhen y	do so /ou e>	omethi (erciso	ing to e?	preve	nt your
Not at all confident	1	2	3	4	5	6	7	8	9	10	Totally confident
6. How o sugar	confid level	lent do goes	o you highe	feel th er or lo	nat yo ower t	u knov han it	w wha shou	at to d Id be?	o whe	n you	r blood
Not at all confident	1	2	3	4	5	6	7	8	9	10	Totally confident
7. How o your i	confid Ilness	lent do s meai	o you n you	feel th shoul	nat yo d visi	u can t the c	judge loctor	wher ?	n the c	hango	es in
Not at all confident	1	2	3	4	5	6	7	8	9	10	Totally confident
8. How confident do you feel that you can control your diabetes so that it does not interfere with the things you want to do?											
Not at all confident	1	2	3	4	5	6	7	8	9	10	Totally confident

B) Dietitians have different styles in dealing with patients. We would like to know more about how you felt about your encounter with your dietitian. Your responses are confidential. Please circle the number that best represents your answer.

1. I feel that my dietitian has provided me with choices and options.

2 1 3 4 5 6 7 Strongly Neutral Strongly disagree agree

2. I feel understood by my dietitian.

1 2 3 4 5 6 7 Strongly Neutral Strongly disagree agree

3. My dietitian conveys confidence in my ability to make changes

	1	2	3	4	5	6	7
Strongl	у			Neutra	al		Strongly
disagre	е						agree

My dietitian encourages me to ask questions. 4.

	1	2	3	4	5	6	7	
Strongly				Neutral		S	Strongly	/
disagree							agree	

5. My dietitian listens to how I would like to do things.

1	2	3	4	5	6	7
Strongly			Neutra	l	S	Strongly
disagree						agree

6. My dietitian tries to understand how I see things before suggesting a new way to do things.

1	2	3	4	5	6	7
Strongly		I	Neutral		Str	ongly
disagree					ag	ree

C) These statements are concerned with your view of the consultation today. Please answer every question only make one choice per item.

1. The dietitian gave me a chance to say what was really on my mind

	1	2	3	4	5	6	7
	Strongly disagree	9		Neut	ral		Strongly agree
2.	I really	felt und	lerstoo	od by th	e dieti	tian	
	1	2	3	4	5	6	7
	Strongly disagree	; ;		Neut	ral		Strongly agree
3.	After ta	lking to	o the di	etitian	l feel n	nuch k	better about my problems
	1	2	3	4	5	6	7
	Strongly disagree	; ;		Neut	ral		Strongly agree
4.	l felt fre	ee to tal	k to the	e dietiti	ian abo	out my	r thoughts
	1	2	3	4	5	6	7
	Strongly disagree)		Neut	ral		Strongly agree
5.	The die	etitian s	eemed	interes	sted in	me as	a person
	1	2	3	4	5	6	7
	Strongly disagree)		Neut	ral		Strongly agree
6.	The die	etitian s	eemed	to take	e my pr	oblen	ns seriously
	1	2	3	4	5	6	7
	Strongly			Neut	ral		Strongly

agree

disagree

7. The dietitian seemed warm and friendly to me

 1
 2
 3
 4
 5
 6
 7

 Strongly disagree
 Neutral
 Strongly agree

8. I felt that the dietitian did not treat me as an equal

1 2 3 4 5 6 7 Strongly Neutral Strongly disagree agree

9. The dietitian seemed to know what they were doing

1	2	3	4	5	6	7
Strongly			Neutral			Strongly
disagree						agree

THANK YOU FOR COMPLETING THE QUESTIONNAIRE.

INITIAL CODING FRAME FOR IDENTIFYING DECISIONS

	Initial topics coded
1.	Portion sizes/Smaller plate/Reduce meat portion/Increase
	vegetables to ½ plate
2.	Weight loss/Weight maintenance
3.	More exercise
4.	More fruit and vegetables/Include one fruit daily
5.	Continue current diet
6.	Cut out fat/Reduce high-fat food
7.	Change milk
8.	Change type of fat
9.	Eating out
10.	Reduce extras /Reduce snacks /Reduce evening high fat snacks
11.	Follow up visit
12.	Take a dietary supplement
13.	Food and drink monitoring diary
14.	Spread fruit out over the day
15.	DAFNE course/weight loss course/DESMOND
16.	Sugar free drinks
17.	Contact GP re depressed feelings
18.	Drink water with meals
19.	Moderate CHO/Decrease potatoes/CHO counting/Lower G.I.
20.	Include a bedtime snack
21.	Not ready to change
22.	Reduce alcohol
23.	DNS re travel
24.	Monitor BG levels
25.	Watch for hypo signs
26.	Sort out pills
27.	Insulin dose/Get injection sites checked/insulin pump
28.	Occasional treats
29.	No salt

30.	Food labels
31.	Calorie counting
32.	Hypoglycaemia treatment
33.	Comfort eating/Consider hunger when picking
34.	Meal variety
35.	Regular meals
36.	Social issues
37.	Readiness to change

Table A.52 Initial codes for decisions

ILLUSTRATION OF THE TEN SUB-CATEGORIES FOR FOOD

Food subcategory	Examples of coded themes for this subcategory
Portion sizes	Smaller plate (7" plate)/Reduce meat portion on plate/ Increase vegetable portion to half the plate
Fruit and vegetables	spread fruit over the day/ include 1 piece of fruit daily/ advice on increasing fruit and vegetable intake
Fat	cut out fat/reduce high fat food/change type of fat/ change type of milk, reduce intake of high fat foods,, eg pastry
Carbohydrate counting/Glycaemic index (G.I).	Carbohydrate counting/carbohydrate portions/ reducing carbohydrate or increasing carbohydrate intake/matching carbohydrate to insulin dose/ choices of lower glycaemic index foods discussed
Snacks	Reduce snacks eaten/Eat snacks/Change type of snack eaten
Eating out	Occasional treats/healthier food choices when eating out/meal variety
Sugar-free drinks	Options discussed
Food labels	How to read and interpret nutritional content
Salt	Advice on reducing intake at table/reducing salt in cooking/reducing high salt foods
Drink water with meals	Drink water with meals

Table A.53 Illustration of the ten subcategories for food

AMENDED EMPATHY CODING LEVELS FOR ECCS

Level 6: Statement of shared feeling or experience

Where the health professional makes an explicit statement that they either share the patients emotion or have had a similar experience.

"I understand how scary this must be for you. My husband recently had a biopsy and we were really scared"

There were no level 6 codes in the consultation recordings.

Level 5: Confirmation

Where the health professional conveys to the patient that what they are feeling and describing is legitimate, ie congratulatory remarks, or acknowledging the difficulties the patient is having.

"You sound like you are very busy, I can see why it would be tough for you to find time to exercise"

"Yes, it sounds like you're trying very, very hard and you're very conscientious."

Level 4: Acknowledgement with pursuit

This is a response that explicitly acknowledges the central issues in the empathetic opportunity and conveys that the physician has "heard "the patient. Level 4 responses convey this by pursuing the topic, asking a question, elaborating on the point raised by the patient.

"You mention you've been feeling sad. Would you tell me more about that?"

"OK, what affect does it have on your diabetes? You mentioned twice that it's not good for your diabetes"

Level 3: Acknowledgement

This response also explicitly acknowledges the central issue in the empathic opportunity. Acknowledgement includes restatements of what patients have said. Information, advice, or offers help.

"Ok, so you've been cutting back on fats".

"So your key emotions throughout your eating, is boredom or tiredness?"

Level 2: Minimal encouragers

These responses occur when the professional provides verbal minimal encouragers such as *"Um-hmmm..*", *"yes" .", "Really?"* while the patient is talking about the empathic opportunity.

Level 1: Implicit recognition of patient perspective

These responses concentrate on peripheral aspects of the empathic opportunity. These statements tend to be more content-base, not dealing directly with the patients concerns. These responses may also include questions or advice.

Patient: "this headache makes it difficult for me to work"

Professional: "yes, how is the insurance businesses lately?"

Patient: "Didn't go yesterday, no. I am missing it, it's obviously part of the routine that really works."

Dietitian: "Yeah. So was that because your brother wasn't around?"

Level 0: Denial of patient perspective

The professional either ignores the patient's empathic opportunity or makes a disconfirming statement. Such responses also include an immediate topic change, or not responding but focusing on a physician task.

Patient: "So I downed the morning one to 26, so I'm doing 26 in the morning and 26 at night and I'm fine at the moment, I haven't had that feeling."

Dietitian: "OK. OK. We've got about 40 minutes today to discuss whatever you want to discuss about your diabetes."

Table A.54 Amended coding levels for the ECCS

CODING MANUAL FOR QUALITATIVE ANALYSIS

All consultations to be coded require transcribing before analysis can begin. The coder needs to become familiar with the data through listening to the recordings and reading through the transcripts. This allows the coder to check transcripts for inaccuracies before coding commences. It also helps to familiarise the coder to the material on the recordings before coding starts.

Once transcribed, each transcript needs to be numbered sequentially for units of speech. A unit of speech is defined as an episode of speech either by the dietitian or by the patient; this may be one word long or contain several sentences. It begins when a person starts speaking and ends when they have finished speaking (Del Piccolo et al, 2005). The numbering of these speech units enables tracking and retrieval of identified features in the analysis. Recordings need to be listened to while coding to help clarify tonal qualities and interpretation of communication codes and themes as they are identified (Del Piccolo et al, 2005).

The coding occurs in two stages: Stage 1 is the coding of the communication behaviours needed to inform the identification of information-giving themes. Stage 2 is the identification and coding of themes for information-giving exchanges in the consultations.

Stage 1

Coding of communication behaviours

The following communication behaviours (counselling strategies that support active listening in the consultation) were coded on the transcripts: reflections, clarification, summarising and paraphrasing, confirming statements and supportive statements. Also coded were specific exploratory questions that asked patients about their thoughts, feelings, ideas, opinions and selfmanagement practices. The total frequency of the communication behaviours that occurred in each of the consultations was entered onto an Excel spreadsheet, whilst coding. Examples of the communication behaviours coded are listed below:-

Reflections and Clarifications

Reflections demonstrate that you are listening and provide an opportunity to check that you have correctly understood what the individual is trying to say. Reflections can be used to reflect content, meaning, or feeling (Gable, 2007). An example of a simple content reflection would be:-

Patient: *"I take 8 or 9 units".* Dietitian: *"8 or 9 units".*

In this example, the professional is simply reflecting back the insulin units that the patient reports using; this simple reflection acts as confirmation that the professional has heard this small piece of information correctly.

It is important to address feelings (Del Piccolo et al, 2008; Little et al, 2001); these feelings can also be reflected back to the individual, for example:

350

Professional: "How do you feel about your weight?"

Patient: "I'm not bothered"

Professional: "You're not bothered....."

Implied feelings from the words or behaviour described by patients need to be handled with care. These emotions should always be reflected back tentatively, as poor interpretation may result in the patient reacting defensively (Gable, 2007). The sensitive reflection of emotions demonstrates empathy and is a greater skill than simply reflecting content alone (Egan, 2002; Gable, 2007). It is also important to reflect feelings with the same intensity that they have been expressed (Gable, 2007; Reynolds & Scott, 1999) as illustrated in the example below:-

Patient: "I said you'll have to report it because my fingers couldn't cope with that as I've pins and needles in my fingers, I couldn't even dress myself, couldn't feed myself..."

Professional: "It must have been horrible for you, horrible."

Reflection with clarification:

Reflections can be used to clarify what has been heard by turning the simple

reflection in to a reflective question, for example:

Patient: "Hit and miss."

Dietitian: "Hit and miss, OK, what do you mean by that, can you explain a bit

more?"

Paraphrasing:

Paraphrasing is another counselling technique that helps to provide direction in

the conversation. Paraphrasing is used to sum up the essence of the

conversation and to check the accuracy of what has been heard. It should

always be applied tentatively, as this process demonstrates that you are

checking your understanding, rather than making definitive statements. It

requires the professional to reflect back what they have heard using their own

words (Gable, 2007); as illustrated below:-

Patient: "Yes and their menus are the same as my menus. There were no new ideas."

Dietitian: *"So you didn't learn anything from them?"* (paraphrase) Patient: *"No, but then that's fine because it says to me that my diet is OK"*

Another example of paraphrasing is illustrated below:

Patient: "No, no, by the time I get home I'll pick up the kids from the grandparents or great grandparents and get them home. We don't live in the same town so by the time I get home, my shift finishes around 5, 6 o'clock sometimes, every other week I'm on lates until 10 o'clock at night and by the time you've got the tea and relaxed and by the time my wife gets home work..."

Dietitian: "There's not much spare time..."

Summarising

Summarising is a method of recapping a large chunk of the conversation in

order to assist both parties in understanding where they have got to in the

dialogue. It is another method of checking information and understanding and

can be conducted at any stage during the consultation to recap and provide

direction. Summarising is usually used when the speaker has been talking for

sometime (Gable, 2007). In the example below, the professional is

summarising the previous patient dialogue around weight management, lack of

activity and comfort eating and trying to provide some sense of where to go next

with the information obtained.

Professional: "For me listening to you just the last thing you were talking about in relation to the comfort eating kind of thing, sounds like an area that might be associated with the weight gain as well as the activity and therefore it might be a useful place to start, because, perhaps lots of the other things may influence it and it may need a bit more unravelling."

The example below illustrates the use of summarising to get the patients

focused on the task in hand. (These are all issues that the patient has raised.)

- Patient:" I don't think about, you know, I do snack in between but for some reason I've got a mental block thinking that's what's causing a good part of my weight gain. I don't think if I eat this I'm going to put on weight, I just pick. I keep thinking it's my meals because before I moved in with my wife – to cut a long story short I moved back home after a bad relationship, so I could go to work and have a meal, do my job, go home and have a blooming great plate full of food but it would always be a mixture of vegetables and what have you. Now I still go home, have a meal but the vegetables have gone down and it's gone to processed but I still feel hungry or I still want to keep eating more.
- Professional: "Ok, so attacking the comfort eating or evening eating is perhaps a priority and then may be thinking about that other bit that has changed from what you were talking about which is the composition rather than the amount of the meal.."

Supportive Statements

These are simple reflections that support the patient's thoughts, feelings, or

actions. For example, in the following excerpt the patient has been reporting

on progress since the last visit and the dietitian has responded to this in a

supportive way.

Dietitian: "Well it certainly seems to be doing the trick, doesn't it?"

Confirming Statements

These are simple reflective statements that confirm the patient's thoughts,

feelings, or actions. In the following example, the dietitian has been exploring

with the patient potential areas of change by getting the patient to identify those

areas of the diet that could be improved on. The next example illustrates that the dietitian has confirmed that they have heard what the patient has said about breakfast before questioning about further possible areas of change:

Patient: "Not the best of breakfasts but something, yes."

Dietitian: "OK, so that's one area that you think possibly might need some attention. Any other..."

Questions

Certain types of questions are important to consider when communicating in a patient-centred way. These are the open questions that specifically explore patient's thoughts, feelings ideas and opinions. These were coded separately - some examples are illustrated below.

A question that explores patient's opinions:

"OK so having told me about your eating habits, do you feel thinking about it in more detail that there's any particular things that might need changing?"

A question that explores patient's thoughts and feelings:

"How do you feel about that?" and another example: "Sorry I interrupted you – you're feeling good because?"

Questions that specifically asked about patient's self-care management were

also coded. An example of this type of question would be:

"Yes, if you do wake up and you're feeling hypo, how would you treat it initially?"

and "What have you found works for you in that situation?"

Agenda Setting

Following coding of the identified active-listening skills, the recordings were listened to, to establish whether an agenda was set and whether the patient was involved in this process. This outcome was marked onto the Excel spreadsheet alongside the consultation. For example, in the following exchange, having confirmed where the referral has come from, the dietitian then proceeds to check the patient's point of view regarding the referral and therefore the purpose of the session.

- Dietitian: "Yes and you've got Type 1 diabetes and she said on her summary here, she said impaired hypo awareness and you're not as aware of your hypos happening as you used to be..."
- Patient: "No and I discussed that the last time I was here and I've got some results which I'll show as we move forward but that's part of the..."
- Dietitian: "She was saying you were trying to match your insulin to your food intake to a certain extent and I think that was the main thing she wanted us to kind of assess and see whether it had helped. I don't know if that's what you thought?"
- Patient: "Yes, I've been a diabetic 10 years and I think I may have had the offer of a Dietician early on and didn't take it. Whilst it's easy to say I've controlled it reasonably well in the last 10 years I have suffered highs and lows and I've tried to get on the Daphne Course as well, which I'm still waiting for..."

This example illustrates the collaborative approach, which establishes the

viewpoints of both parties to the consultation.

In contrast, the example below illustrates a professional-led agenda. The

dietitian has not clarified at the start of the consultation the purpose of the visit

and launches straight into gathering information from the patient

Dietitian: "OK so I don't know if you've seen a Dietitian before?"

Patient: "Not for a few years, no."

Dietitian: "So what can we help you with today, how are you getting on generally?"

Patient: "Emm, not too bad I don't think."

Dietitian: "Good, everything going alright with the diabetes?"

Patient: "Yes."

Dietitian: "OK, and food wise?"

Patient: "I'm quite good, I do have my lapses but overall I'm quite good, I do try to be."

Dietitian: "OK and the sort of insulin that you're on?"

This would therefore be coded as a professional led agenda.

Additional codes

The following codes were marked in the margins of the transcript to provide further guidance as to the intent of the information provision and to guide the theme development for information-giving exchanges. These are illustrated in tables A.55 and A.56 below:

Dietitian Codes

Dietitian questions the patient to gather additional information about an issue that the patient has raised

Dietitian requests new information

Dietitian offers new advice

Dietitian offers advice in response to patient request

Dietitian offers explanations

Dietitian offers advice but checks with patient first whether this is needed

Dietitian tells patient what to do

Dietitian asks patient opinion, ideas, preferences,

Dietitian checks information by reflecting back to patient

Dietitian confirms patient observation

Dietitian gives advice

Dietitian checks patient understanding

Table A.55 Additional codes used for dietitian talk

Patient Codes

Patient offers an explanation (makes a statement)

Patient offers new advice (not requested by the dietitian)

Patient offers information as was asked

Patient asks questions

Patient queries advice given

Patient concern

Patient confirms advice given

Table A.56 Additional codes used for patient talk
Stage 2

Following completion of coding for active-listening skills, the second stage of the analysis can begin. This is the exploration and development of themes for the identified information-giving exchanges that occurred in the consultations.

Coding of information-giving exchanges

The transcripts were marked for information-giving episodes that focused on patients' health behaviour. The length of these information exchanges varied and was dependent on the topic being discussed. It consisted of consecutive utterances from the patient and the dietitian about the same topic, and finished when the discussed topic changed (Kiuru et al, 2004).

Once identified, the information episodes can then be coded for how the information was provided and received, the level of patient participation and the overall intent of the information exchange. There are four identified themes for information-giving exchanges. Table A.55 is adapted from Kiuru et al (2004) and illustrates the key constructs of the identified themes for information-giving exchanges.

358

	Persuasive Information	Supportive/ collaborative	Confirming/ permitting	Recommending/ teaching/ instructions
How information is provided	Little or no understanding of patient's perspective could be judgemental	Patient's understanding and views sought before information given Supportive feedback provided	Response to patient's queries or statements checking information	Information, suggestions, recommendations provided by dietitian
How information is received	Met with resistance /Justification	Met with further collaborative exchange	Patient confirms information received	Minimal response from patient
Patients participation	Minimal input from the patient or patient justifies actions, defends course of action taken	Generally patient talks more in these exchanges, active engagement Patient provides additional personal/ relevant information	Met with minimal input from patient or patient confirms that they have understood the information received	Minimal input from the patient, clarifying questions occasionally used, passive response from patient
Dietitians role in information exchange (Intent of information)	Aimed at getting patient to follow advice persuasive	Aimed at facilitating patient's understanding and ability to change, personalised advice provided	Aimed at confirming, permitting patient's thoughts, actions, general information	Aimed at informing patient of correct course of action to take Specific advice or instruction given encourage patient to consider recommendations or suggestions offered.
Outcome	No clear decision or course of action made	Collaborative decision more likely	Confirmation of behaviour/ information	Clear instructions provided by dietitian

Table A.57 Key features of the four identified themes for information-giving exchanges adapted from (Kiuru *et al*, 2004)¹³

¹³ Reprinted from Kiuru et al (2004) 'Advice-giving Styles by Finish Nurses in Dietary Counseling Concerning Type 2 Diabetes Care' *Journal of Health Communication*, 9 pp. 337-354 with permission from Elsevier

Each episode of information-giving exchange on the transcripts was read through and coded according to the above framework and categorised into one of the four identified themes. Further details plus examples of each of the coding categories for the information-giving exchanges is provided below.

Persuasive information-giving exchange

Persuasive information-giving exchanges were clearly identified as advicegiving episodes led by the dietitian, where often the information offered to the patient was not requested. Here the dietitian did not question the patient to explore how they think or feel about the situation under discussion or inquire as to their self-care management practices before providing this information. There was little or no understanding of the issue being discussed, from the patient's viewpoint or perspective. The information provided by the dietitian was aimed at getting the patient to follow the advice, even though in some cases the patient had made it very clear that they did not wish to follow the advice provided.

These episodes of information-giving exchanges were characterised by lack of questioning by the dietitian to explore patient's perspective. Alternatively, the patient may have proved information that was missed or ignored by the dietitian. In *persuasive* exchanges the information received was generally met with resistance by the patient and justification offered for continuation of the behaviour. *Persuasive* exchanges would either be terminated by the dietitian (as they moved on to another topic of conversation) or would be met with minimal responses from the patient and then the patient would change the topic.

360

In the following excerpt the dietitian is trying to persuade the patient to eat dried fruit in order to manage their constipation. However, before providing advice on what to do, the dietitian failed to use open questions to explore how the patient has managed this in the past, or what they have tried. Consequently, the patient is a little resistant to the advice that is offered, especially as it is something that they are aware has little impact on their constipation. The patient terminates this line of questioning by laughing and saying that nothing makes a difference to their bowels; this response allows the conversation to move onto the next topic as illustrated in Figure A.19.

Excerpt 1

Dietitian: "OK, do you ever use dried fruit because that's quite good for fibre and roughage?"
Patient: "Yes but then I find it's very sweet and it crystallises and that and I think no."
Dietitian: "So you think it's too sweet?"
Patient: "For me it is because as I say I've never been a person to eat sweets..".
Dietitian: "No but a couple of prunes would be good..."
Patient: "But they don't make no difference to my bowels."

Figure A.19 Example of *persuasive* information-giving exchange

Confirming/permitting Information-giving exchange

Confirming/permitting information-giving exchanges occur as a result of the patient's queries or statements. The information provided by the dietitian confirms the patients' observation, thoughts, behaviour, or actions or provides permission to act in a certain way. These episodes of information giving are likely to arise as a result of either a *supportive/collaborative* information-giving

exchange, suggesting that the patient is feeling comfortable and able to ask for specific information from the dietitian. Alternatively these episodes may occur as a result of a previous recommending information-giving exchange and may have been used as a way of changing the topic within that exchange.

The majority of these exchanges resulted in minimal responses from the patients, as they were confirming, thoughts, behaviours or actions presented to the dietitian. In the following excerpt, the patient seeks confirmation from the dietitian about the suitability of using milk to treat hypoglycaemia. This is general advice and we are not clear from the patient whether this something that she currently does, as illustrated in Figure A.20.

Patient: "What would you suggest if it was low-ish, a biscuit or something like that? What do they feel about milk now?"

Dietitian: "Milk is fine but it's fairly slowly absorbed, it's fine to have as a carbohydrate source but it's no good for treating a hypo..."

Patient: "No, not like we used to."

Dietitian: "We used to put sugar in it."

Figure A.20 Example of confirming information-giving exchange

Another example of *confirming* information exchange is provided below, the patient has offered some information to the dietitian that they have read in the paper and the dietitian confirms what they have read (Figure A.21)

Patient: "In the paper I think I read, last week, week before, that one of the many experts said that when you take medicines you shouldn't have orange juice because it defeats the object of some of the medicines."

Dietitian: "The only one I know of that is very common isn't orange juice it's grapefruit juice. You can't have grapefruit with statins."

Patient: "One of the tablets that we take, it does say you mustn't take grapefruit."

Dietitian: ""That's a very common one that people have to avoid."

Figure A.21 Example of *confirming* information-giving exchange

Recommending/Teaching advice information-giving exchange

Recommending or *teaching* information-giving exchanges occurred when the dietitian provided a clear set of instructions as to what the patient should do in terms of action or behaviour. These sequences were characterised by minimal input from the patient. They could be professional-led advice-giving episodes, as a more direct response to a patient query or as a result of the patient offering information that required elaboration. *Recommending* episodes of advice are generally informative, and by their nature the advice tells the patient what to do and provides some information as to why the recommendation is being made. For example, in Figure A.22 the patient is concerned that the sugar that they put into the bread they make is having an impact on their blood glucose levels.

Patient: "I could put that stuff that I have, that sweetener, would that make any difference or not?"

Dietitian: "No because I think the yeast needs a little sugar."

Patient: "Oh right, but we're not ... "

Dietitian: "We're not talking sugar free diet, we're talking low sugar. A little bit of sugar spread throughout a loaf of bread and you're having a couple of slices and sandwich which is fine, the sugar is negligible so that's fine."

Patient: "Good".

Figure A.22 Example of *recommending* information-giving exchange

Supportive/collaborative information-giving exchange

This type of information exchange engaged the patient through facilitated discussion. These exchanges may have occurred as a result of the patient asking a question, raising a query, or making a statement. This type of *supportive* exchange may involve further clarification and provision of additional information to support the patient's actions and or views. Inevitably, these collaborative exchanges usually involved the patient talking more and providing information that was personally relevant. Information exchange was therefore supportive, as it followed the patient's agenda and supported their actions, ideas, and perception of control. *Supportive* information-giving exchanges were distinct from other exchanges in that they provided supportive feedback. This feedback was either in the form of direct reflection or in the supportive nature of the response incorporated within the feedback to the patient (Kiuru et al, 2004).

The excerpt in Figure A.23 illustrates the supportive and collaborative way that the dietitian explores the issue of reducing blood glucose levels to improve HbA1c measurements with the patient. The patient has highlighted their concerns with regards to increasing control, as this will result in a greater risk of hypoglycaemia. The supportive response of the dietitian is illustrated through their acknowledgement of the patient's concern, which is reflected back to the patient.

Patient: "Last time here they said to get to that it would be good but you can suffer a lot of lows in trying to achieve that."

Dietitian: "Yes and that's where to a certain extent the Daphne principles and the course as a whole may be helpful in helping to get closer to that point without too many risks of the hypos and the consequence because that quite often happens. Like you say your HbA1c might come down but if you suffer more hypos as a consequence that's not going to help your hypo awareness, so, you know it's getting that balance right really. So overall that's your main..."

Patient: "Yes to keep it on a level keel during the day, perhaps either Daphne or you can tell me if I'm testing at the right time..."

Dietitian: "So when are you testing?"

Patient: "Testing when I wake up in the morning before breakfast, lunch time, pm for tea and then evening meal."

Dietitian: "So you're doing before breakfast, before lunch?"

Patient: "Yes before lunch or slightly after, around 6 or 7 o'clock – 8 o'clock when we have tea and just before I go to bed."

Dietitian: "OK so before tea and before bed. If we're talking Daphne we would be suggesting that you test at least before a meal and before bed, which is basically what you're doing."

Patient: "So before a meal and before bed."

Dietitian: "Daphne is focused very much with trying to get you within target by the time you get to your next mealtime, there isn't any focus on what is going on between times unless you're driving or you're not sure whether you're having a hypo or not or you're exercising or something and you need to know exactly what it's doing. So we discourage people from testing between meals because we're not actually teaching them to do anything about it, so it gives them information that we're not going to do anything with."

- Patient: *"I think that's what I'm guilty of sometimes is testing between meals and finding myself at 8, 9 or 10 and perhaps putting a little bit of insulin in and then maybe doing some manual job around the house and then as you've seen quite frequently Jan will have spotted it before I know it in terms of just how I look."*
- Dietitian: "So before each meal and before bed is a good basis to work from if you're thinking ahead to what we would teach you on Daphne."

Figure A.23 Example of supportive/collaborative information-giving exchange

Once all the information-giving exchanges had been coded with the identified themes, and data entered onto an Excel spreadsheet, the data on level of agreement on reported decisions made was added to this information. This allowed comparison to be made on the presence of communication skills and information-giving exchanges in those consultations with agreement on decisions made and those with disagreement.

Empathic statements made by patients were identified using the ECCS (Bylund & Makoul, 2002) and marked onto the transcripts. The data on informationgiving exchange themes and empathic statements made by patients were then placed in order of occurrence for each consultation on an Excel spreadsheet. The information-giving exchanges were colour-coded to provide a visual map of the flow of information-giving exchanges that occurred in consultations with either agreement or disagreement on reported decisions made. Consultations that had an agenda set were also marked to identify whether this altered the flow of information-giving exchange within the consultation. The colour-coded map provided a visual illustration of the sequences of information-giving themes within the consultations. It highlighted consultations where agendas were set, and illustrates the relationship between empathic statements and information-giving exchanges within the consultations with relations where agendas were set, and illustrates the relationship between empathic statements and information-giving exchanges within the consultations.

367

GLOSSARY

Adherence the ability to adhere and carry out a set of recommendations

Agreement on reported decisions patient and professional agreement on independent reporting of decisions made in the consultation

Auditory recall is the ability to recall information that has been heard

Autonomy is the degree to which a person feels that they are in control

Autonomy support the degree to which the patient feels supported in making choices

Autonomous motivation relates to patients being willing to engage in a healthrelated behaviour because they have fully accepted its importance

Behaviour change relates to a change that a person makes to their behaviour in order to improve outcomes

Biomedical relates to medical aspects of care

Binge eating a pattern of disordered eating resulting in episodes of uncontrollable eating

Biopsychosocial an approach that considers emotions, thoughts and behaviours as well as social aspects of care

Blood glucose a measure of circulating dextrose (blood sugar) in the blood stream

Blood pressure tension of the blood within the arteries maintained by the contraction of the left ventricle of the heart usually noted as systolic pressure/ diastolic pressure

Body Mass Index (BMI) numerical indicator of health risk relates to height m²/weight kg

"**Burnout**" a concept that describes the unwillingness to change when there occurs previous lack of success, constant failure and general fatigue with trying to bring about change

Calorie modification alteration in calorie intake usually taken in an attempt to lose weight

Carbohydrate refers to simple sugars and complex starchy foods such as bread, potatoes, rice, and pasta

Cardiovascular disease relating to the heart and the circulation of blood through the blood vessels

Cholesterol a type of fat produced by the body, which is used as an emulsifying agent

Client-centred nutritional counselling relates to a process of counselling that puts the client/patient at the centre of the discussion to address their needs and concerns

Cognitive behavioural therapy relates to talking therapy that helps the individual to change their thoughts around behaviours in order to behave and respond differently to situations

Co-morbidities having more than one medical condition

Compliance the degree of adherence to a prescribed regimen

Concordance relates to the involvement of the patient in the decision-making process in order to increase compliance

Counselling skills refers to a set of skills used when using talk therapy includes; reflection, clarifying, paraphrasing and summarising

Chronic disease a long-term progressive condition for which there is no cure and which is managed through medication and/or lifestyle change

Chronic obstructive pulmonary disease relating to a long-term condition involving restriction of airflow to and from the lungs

Client-centred counselling relates to process of talk therapy where the patient's needs and values are of prime importance

Depression a sinking of spirits that constitutes a clinical diagnosed condition

Diabetes Mellitus is a chronic condition; a metabolic disorder resulting from the lack or insufficient production of a hormone called insulin, resulting in elevated circulating levels of glucose in the blood stream

Diastolic the lowest blood pressure reached during a ventricular cycle

Dietitian a registered professional who is trained to give dietary advice to manage, treat or prevent medical conditions

Dyspnoea relates to shortness of breath or difficulty in breathing

Eating disorders refers to a group of disordered behaviours around eating that may involve excessive intake of food or insufficient food intake, which impact on the individual's physical and mental health

Empathy the ability to demonstrate an understanding of another person's situation

Empathic clues these are clues presented by an individual in a dialogue that require an empathic response

Empathic opportunities these are clues that are presented by an individual in a dialogue that provide the listener with an opportunity to respond empathically

Empathic statements these are statements that an individual makes in a conversation that require an empathic response

Empirical refers to data produced by an experiment or through observation

Empowerment refers to the process by which people gain skills and knowledge to take control over the decisions and actions that impact on their health

Gangrene death of body tissue as a result of obstruction of the blood flow

Glucose a monosaccharide or simple sugar

Glycaemic control refers to the control of blood glucose levels

Glycosylated Haemoglobin (HbA1c) is a laboratory measure of the average amount of circulating glucose in the blood stream over the previous three months.

Health literacy refers to a patient's ability to access health information, either because they can't read it or because they cannot understand what they have read

Hypercholesterolemic elevated circulating levels of cholesterol in the blood

Hyperglycaemia elevated circulating levels of glucose in the blood

Hypertensive/Hypertension relates to a blood pressure measurement that exceeds the accepted level of normal

Hypoglycaemia low levels of circulating glucose in the blood

Insulin a hormone used in the uptake of glucose

Insulin resistance refers to the condition resulting from the body's lack of ability to respond to circulating insulin levels

Lipids refers to a range of fats in the blood stream

Meta-analysis combination of results from a number of studies that explore a similar research hypothesis

Motivation the driving force that helps us to achieve goals

Macrovascular disease damage occurring to the large blood vessels resulting in cardiovascular disease and stroke

Metabolic syndrome refers to a group of risk factors that occur together and increase the risk of type 2 diabetes, heart disease, and stroke

Microvascular disease damage occurring to the small blood vessels resulting in retinopathy, neuropathy, and nephropathy

Monotherapy single drug use

Morbidity disease state

Mortality death rate

Neuropathy a complication seen in diabetes resulting from small vessel damage resulting in damage to the nervous system

Nephropathy a complication seen in diabetes: kidney damage occurring as a result of small vessel damage

Non-verbal communication refers to the body language signals

Obese classification relating to an individual's height to weight ratio, indicating excessive amounts of body fat

Oral hypoglycaemics refers to a group of drugs used to lower blood glucose levels in people with diabetes

Orthotics manufacture and design of a device that supports or corrects the function of a limb

Patient autonomy refers to the patient's ability to make decisions as a result of information provision these choices are informed and not coerced by the professional

Patient enablement is a measure of the patient's perception of their ability to cope with life and their illness, to understand their illness, to keep healthy and to be confident in their ability to help themselves.

Peripheral vascular disease often referred to as arterial damage resulting from obstruction to the large arteries

Polydipsia excessive thirst, leading to drinking large volumes of water

Polyuria passing of large volumes of urea with increasing frequency

Psychosocial considers the psychological aspects in relation to social factors

Psychologist a specialist in psychology

Physiological refers to the mechanics, physical, and biochemical, of the human body

Recall to remember or to retrieve information

Respiratory disease refers to chronic diseases of the airways and other structures of the lungs

Retinopathy a complication seen in diabetes resulting from small vessel damage causing non-inflammatory degenerative disease of the retina

Self-care management refers to the daily activities that patients engage in to manage their condition

Self-efficacy is a person's belief in their ability to succeed in a particular situation

Self-esteem relates to an individual's favourable opinion of their character and abilities

Socio-economic relates to the social and economic position and work experience of individuals relative to others

Stroke loss of brain function as a result of lack of blood supply to the brain due to a blockage (blood clot or embolism) or a bleed

Sympathy participation in feeling/compassion

Type 1 diabetes a chronic condition where the body stops making the hormone insulin and blood glucose levels go very high

Type 2 diabetes a chronic condition where the body is either not making enough of the hormone insulin, or the body is failing to recognise and use the insulin that is being produced, resulting in very high blood glucose levels

Vascular team a group of artery and vein specialists

REFERENCES

Abraham, C. & Michie, S. (2008) 'A Taxonomy of Behaviour Change Techniques Used in Interventions', *Health Psychology*, 27 (3), pp. 379-387.

AbuSabha, R. & Ahterberg, C. (1997) 'Review of self-efficacy and locus of control for nutrition -and health-related behaviour', *J.Am Diet Assoc*, 97 (10), pp. 1122-1132.

Adams, A. S., Soumerai, S. B., Lomas, J. & Ross-Degnan, D. (1999) 'Evidence of self-report bias in assessing adherence to guidelines', *Int J Qual Health Care*, 11 (3), pp. 187-192.

ADVANCE Collaborative Group, Patel, A., MacMahon, S., Chalmers, J., Neal, B., Billot, L., Woodward, M., Marre, M., Cooper, M., Glasziou, P., Grobbee, D., Hamet, P., Harrap, S., Heller, S., Liu, L., Mancia, G., Mogensen, C. E., Pan, C., Poulter, N., Rodgers, A., Williams, B., Bompoint, S., de Galan, B. E., Joshi, R. & Travert, F. (2008) 'Intensive blood glucose control and vascular outcomes in patients with type 2 diabetes. Collaborators (1307)', *N Engl J Med*, 358 (24), pp. 2560-2572.

Ahlén, G. C., Mattsson, B. & Gunnarsson, R. K. (2007) 'Physician patient questionnaire to assess physician patient agreement at the consultation', *Family Practice*, 24 pp. 498-503.

Aiken, J. E., Bingham, R. & Piette, J. D. (2005) 'Patient-provider communication and self-care behaviour among type 2 diabetes patients', *Diabetes Educ*, 31 (5), pp. 681-690.

Albright, T. L., Parchman, M., Burge, S. K. & RRNeST Investigators (2001) 'Predictors of self care behavior in adults with type 2 diabetes: an RRNeST study', *Family Medicine*, 35 (5), pp. 354-360.

Ambady, R. & Chamukuttan, S. (2008) 'Early diagnosis and prevention of diabetes in developing countries', *Rev Endocr Metab Disord*, 9 (3), pp. 193-201.

American Diabetes Association Position Statement (2002) 'Evidence-based nutrition principles and recommendations for treatment and prevention of diabetes and related complications', *Diabetes Care*, 25 pp. S50-S60.

American Association of Diabetes Educators (AADE) <u>http://www.ncbde.org/practice_require.cfm (accessed 30/10/11)</u>

Ammerman, A. S., Lindquist, C. H., Lohr, K. N. & Hersey, J. (2002) 'The Efficacy of Behavioural Interventions to Modify Dietary Fat and Fruit and Vegetable Intake: A Review of the Evidence', *Preventive Medicine*, 35 (1), pp. 25-41.

Anderson, B. & Funnell, M. M. (2005a) *The Art of Empowerment: Stories and Strategies for Diabetes Educators.* Alexandria, Va: American Diabetes Assoictaion.

Anderson, D. C. (1981) 'Abstractive and observational methods of educational evaluation in a dietetic clinic', *Sociology of Health and Illness*, 3 (1), pp. 1-14.

Anderson, J. L., Dodman, S., Kopelman, M. & Fleming, A. (1979) 'Patient information recall in a rheumatology clinic.', *Rheumatol Rehabil* 18 pp. 18-22.

Anderson, L. & Zimmerman, M. A.-. (1993) 'Patient and physician perceptions of their relationship and patient satisfaction: A study of chronic Disease management.', *Patient Education and Counseling*, 20 (1), pp. 27-36.

Anderson, R. M. (1995) 'Patient Empowerment and the Traditional Medical Model', *Diabetes Care*, 18 (3), pp. 412-415.

Anderson, R. M., Frunnel, M. M., Butler, P. M., Arnold, M. S., Fitzgerald, J. T. & Feste, C. C. (1995) 'Patient empowerment. Results of a randomized controlled trial', *Diabet Care*, 18 pp. 943-949.

Anderson, R. M. & Funnell, M. M. (2000) 'Compliance and adherence are dysfunctional concepts in diabetes care', *Diabetes Educator*, 26 pp. 597-604.

Anderson, R. M. & Funnell, M. M. (2005b) 'Patient empowerment: reflections on the challenge of fostering the adoption of a new paradigm', *Patient Education & Counseling*, 57 pp. 153-157.

Anderson, R. M., Funnell, M. M., Barr, P. A., Dedrick, R. F. & Davis, W. K. (1991) 'Learning to empower patients. Results of professional education program for diabetes educators', *Diabetes Care*, 14 (7), pp. 584-590.

Anderson, R. M., Genther, R. W. & Alogna, M. (1982) 'Diabetes Patient Education from Philosophy to Delivery', *Diabetes Educ*, 8 pp. 33-36.

Anderson, R. T., Balkrishnan, R., Camacho, F., Bell, R. A., Duren-Winfield, V. & Goff, D. C. J. (2003) 'Patient-Centred Outocmes of Diabetes Self-Care: Associations with Satisfaction and General Health in a Community Clinic Setting', *NC Med J*, 64 (2), pp. 58-65.

Arborelius, E. (1996) 'Using doctor-patient communication to affect patients lifestyles theoretical and practical implications', *Psychological Health*, 11 pp. 845-855.

Arborelius, E. & Österberg, E. (1995) 'How do GP's discuss subjects other than illness?', *Patient Educ Couns*, 25 pp. 257-268.

Arundel, F., Cradock, S., Noeken, J. & Sikinner, T. C. (2003) 'Phase 1 evaluation of Starting Out with Type 2 Diabetes: A self -management education workshop for the newly diagnosed', *Diabetic Medicine*, 20 (Suppl 2), pp. 76.

Ary, D. V., Toobert, D. J., Wilson, W. & Glasgow, R. E. (1986) 'Patient perspective on factors contributing to nonadherence to diabetes regimen', *Diabetes Care*, 9 (2), pp. 1668-1672.

Asimakopoulou, K. G., Hampson, S. E. & Morrish, N. J. (2002) 'Neuropsychological functioning in older people with Type 2 diabetes: the effect of controlling for confounding factors.', *Diabet Med*, 19 pp. 311-316.

Astrup, A. & Finer, N. (2000) 'Redefining Type 2: diabesity or obesity dependent diabetes mellitus', *Obesity Review*, 1 (2), pp. 57-59.

August, K. J. & Sorkin, D. H. (2010) 'Marital status and gender differences in managing a chronic illness: the function of health-related social control', *Soc Sci Med*, 71 (10), pp. 1831-1838.

Australian Diabetes Educators Association (ADEA) <u>http://www.adea.com.au/main/diabeteseducators/credentialleddiabeteseducator</u> <u>s</u> (accessed 30/10/11)

Bagust, A., Hopkinson, P. K., Maslove, L. & Currie, C. J. (2002) 'The projected health care burden of Type 2 diabetes in the UK from 2000 to 2060', *Diabet Med*, 19 (Suppl 4), pp. 1-5.

Bakeman, R. & Gottman, J. M. (1997) *Observing Interaction An Introduction to Sequential Analysis.* 2nd edn. Cambridge: Cambridge University Press.

Bandura, A. (1977) 'Self-efficacy: toward a unifying theory of behavioural change', *Psychological Review*, 84 (2), pp. 191-215.

Baranowski, T., Cullen, K. W., Nicklas, T., Thompson, D. & Baranowski, J. (2003) 'Are Current Health Behaviour Change Models Helpful in Guiding Prevention of Weight Gain Efforts ', *obesity Research*, 11 pp. 23S - 43S.

Barlow, J., Wright, C., Sheasby, J., Turner, A. & Hainsworth, J. (2002) 'Selfmanagement approaches for people with chronic conditions: a review', *Patient Education & Counseling*, 48 pp. 177-187.

Barnard, K. D., Cradock, S., Parkin, T. & Skinner, T. C. (2006) 'Effectiveness of a computerised assessment tool to prompt individuals with diabetes to be more active in consultations', *Practical Diabetes International*, 24 (1), pp. 36-41.

Baron, R. M. & Kenny, D. A. (1986) 'The moderator-mediator variable distinction in social psychological research: Conceptual strategic, and statistical considerations. ', *Journal of Personality and Social Psychology*, 51 pp. 1173-1182.

Barr, S. I., Yarker, K. V., Levy-Milne, R. & Chapman, G. E. (2004) 'Canadian dietitians' views and practices regarding obesity and weight management ', *J Hum Nutr Diet*, 17 pp. 503-512.

Barry, C. A., Bradley, C. P., Britten, N., Stevenson, F. A. & Barber, N. (2000) 'Patients' unvoiced agendas in general practice consultations: qualitative study', *BMJ*, 320 pp. 1246-1250.

Bartlett, E. E., Grayson, M., Barker, R., Levine, D. M., Golden, A. & Libber, S. (1984) 'The effects of physician communication skills on patient satisfaction; recall, and adherence', *J Chronic Dis*, 37 (9-10), pp. 755-764.

BDA (2008) 'Guidance for Dietitiasn for Record & Record Keeping'. *British Dietetic Association*.

British Dieteitc Association (2009) 'Nutrition and Dietetic Care Process'. British Dietetic Associtation.

British Dietetic Associtation (2011) 'Model for Dietetic Outcomes'. *British Dietetic Association*.

Beck, R. S., Daughtridge, R. & Sloane, P. D. (2002) 'Physician-Patient Communication in the Primary Care Office: A Systematic Review', *J Am Board Fam Pract*, 15 (1), pp. 25-38.

Beisecker, A. E. & Beisecker, T. D. (1990) 'Patient Information -Seeking Behaviors When Communicating With Doctors', *Med Care*, 28 (1), pp. 19-28.

Bell, D. S. (1994) 'Stroke in the diabetic patient', *Diabetes Care*, 17 (3), pp. 213-219.

Bellet, P. S. & Maloney, M. J. (1991) 'The importance of empathy as an interviewing skill in medicine', *JAMA*, 266 pp. 1831-1832.

Benner, J. S., Glynn, R. J., Mogun, H., Neumann, P. J., Weinstein, M. C., Avorn, J. (2002) ' Long-term persistence in use of statin therapy in elderly patients. ', *Journal of the American Medical Association*, 288 (4), pp. 495-497.

Bensing, J. (2000) 'Bridging the gap. The separate worlds of evidence-based medicine and patient-centred medicine', *Patient Education & Counseling*, 39 pp. 17-25.

Berg-Smith, S. M., Stevens, V. J., Brown, K. M., Van, H. L., Gernhofer, N., Peters, E., Greeberg, R., Snetselaar, L., Ahrens, L. & Smith, K. (1999) 'A brief motivational intervention to improve dietary adherence in adolescents. The Dietary Intervention Study in Children (DISC) Research Group', *Health Education Research*, 14 (3), pp. 399-410.

Bertakis, K. D. (2009) 'The influence of gender on the doctor-patient interaction', *Patient Education and Counseling*, 76 pp. 356-360.

Bertakis, K. D., Roter, D. & Putnam, S. M. (1991) 'The relationship of physician medical interview style to patient satisfaction ', *J Fam Pract*, 32 pp. 175-181.

Bertoni, A. G., Case, D., Bell, R. A., Goff, D. C. J. & Narayan, K. M. (2007) 'Predicted risk of coronary heart disease among persons with type 2 diabetes', *Coron Artery Dis.*, 18 (8), pp. 595-600.

Beverly, E. A. & Wray, L. A. (2010) 'The role of collective efficacy in exercise adherence: a qualitative study of spousal support and type 2 diabetes management.', *Health Education Research*, 25 (2), pp. 211-223.

Bezie, Y., Molina, M., Hernandez, N., Batista, R., Niang, S. & Huet, D. (2006) 'Therapeutic compliance: a prospective analysis of various factors involved in the adherence rate in type 2 diabetes', *Diabetes Metab*, 32 (6), pp. 611-616.

Boeije, H. (2010) Analysis in Qualitative Research London: Sage.

Bogardus, S. T., Bradley, E. H., Wiliams, C. S., Maciejewski, P. K., Gallo, W. T. & Inouye, S. K. (2004) 'Achieving Goals in Geriatric Assessment: Role of Caregiver Agreement and Adherence to Recommendations', *J Am Geriatr Soc*, 52 pp. 99-105.

Bogardus, S. T., Bradley, E. H., Williams, C. S., Maciejewski, P. K., van Doorn, C. & Inouye, S. K. (2001) 'Goals for the Care of Frail Older Adults: Do Caregivers and Clinicians Agree?', *Am J Med*, 110 pp. 97-102.

Bonnet, C., Gagnayre, R. & d'Ivernois, J. F. (1998) 'Learning difficulties of diabetic patients. A survey of educators ', *Patient Educ Couns*, 35 pp. 139-147.

Bovbjerg, V. E., McCann, B. S., Brief, D. J., Follette, W. C., Retzlaff, B. M., Dody, A. A., Walden, C. E. & Knopp, R. H. (1995) 'Spouse support and long-term adherence to lipd-lowering diets', *Am J Epidemiology*, 141 pp. 451-460.

Boyatzis, R. E. (1998) 'Developing themes using the theory-driven and prior research driven method and then analysing the code'. in Boyatzis, R.E. (ed.) *Thematic analysis and code devleopment : Transforming Qualitative Information* Thousabd Oaks: CA: Sage.

Brace, N., Kemp, R. & Snelgar, R. (2006) SPSS for Psychologists: A Guide to Data Analysis using SPSS for Windows (versions 12 and 13). 3rd edn. Basingstoke: Palgrave Macmillan.

Bradley, E. H., Bogardus, S. T., Tinetti, M. E. & Inouye, S. K. (1999) 'Goalsetting in clinical medicine', *Social science & Medicine*, 49 pp. 267-278.

Braun, V. & Clarke, V. (2006) 'Using thematic analysis in psychology', *Qualitative Research in Psychology*, 3 pp. 77-101.

Bravoa, B. N., Postigoa, J. M. L., Seguraa, L. R., Selvaa, J. P. S., Trivesa, J. J. R., Córcolesa, M. J. A., Marta Nieto Lópeza, M. N. L. & Hidalgob, J. L.-T. (2010) 'Effect of the evaluation of recall on the rate of information recalled by patients in Primary Care', *Patient Education and Counseling*, 81 pp. 272-274.

Brekke, H. K., Sunesson, A., Axelsen, M. & Lenner, R. A. (2004) 'Attitudes and barriers to dietary advice aimed at reducing risk of type 2 diabetes in first-degree relatives of patients with type 2 diabetes', *J Hum Nutr Diet*, 17 (6), pp. 513-521.

Bridle, C., Riemsma, R. P., Pattenden, J., Sowden, A. J., Mather, L., Watt, I. S. & Walker, A. (2005) 'Systematic review of the effectiveness of health behavior interventions based on the transtheoretical model', *Psychology & Health*, 20 (3), pp. 283-301.

British Dietetic Association (2008a) 'Code of Professional Conduct', BDA

British Dietetic Association (2008b) 'Curriculum Framework for the Pre-Registration Education and Training of Dietitians'. [Online]. Available at: <u>http://members.bda.uk.com/education/curriculum_framework.html</u>.

Brown, S. A. (1990) 'Effects of educational interventions and outcomes in diabetes adults: a meta-analysis revisited', *Patient Education Counselling*, 16 pp. 189-215.

Brown, S. L., Pope, J. F., hunt, A. E. & Tolman, N. M. (1998) 'Motivational strategies used by dietitians to counsel individuals with diabetes', *Diabetes Educator*, 24 (3), pp. 313-318.

Brug, J., Spikmans, F., Aartsen, C., Breedveld, B., Bes, R. & Fereira, I. (2007) 'Training Dietitians in Basic Motivational Interviewing Skills Results in Changes in Their Counseling Style and in Lower Saturated Fat Intakes in Their Patients', *J Nutr Educ Behav*, 39 pp. 8-12. Brunero, S., Lamont, S. & Coates, M. (2010) 'A review of empathy education in nursing.', *Nurs Inq*, 17 (1), pp. 65-74.

Bryman, A. (2008) *Social Research Methods.* 3rd edn. Oxford: Oxford University Press.

Burke, L. E., Dunbar-Jacob, J., Sereika, S. M. & Ewart, C. K. (2003) 'Development and testing of the Cholesterol-Lowering Diet Self-Efficacy Scale', *Eur J Cardiovascular Nursing*, 2 (4), pp. 265-273.

Burley-Allen, M. (1995) *Listening the forgotten skill: A Self-Teaching guide.* 2nd edn. New York: John Wiley & Sons Inc.

Butow, P. N., Brown, R. F., Cogar, S., Tattersall, M. H. N. & Dunn, S. M. (2002) 'Oncologists reaction to cancer patients' verbal cues', *Psycho-Oncology*, 11 pp. 47-58.

Bylund, C. L. & Makoul, G. (2002) 'Empathic communication and gender in the physician-patient encounter', *Patient Education & Counseling*, 48 (3), pp. 207-216.

Bylund, C. L. & Makoul, G. (2005) 'Examining empathy in medical encounters: an observational study using the empathic communication coding system', *Health Communication*, 18 (2), pp. 123-140.

Byrne, P. S. & Long, B. E. L. (1976) Doctors talking to patients. HMSO London:

Calder, J. & Sapsford, R. (1996) 'Multivariate Analysis'. in Sapsford, R. and Jupp, V. (eds.) *Data Collection and Analysis.* London: Sage, 10 10 pp 268-274.

Campbell, J. D., Mauksch, H. O., Neikirk, H. J. & Hosokawa, M. C. (1990) 'Collaborative practice and provider styles of delivering health care ', *Soc Sci Med*, 30 pp. 1359-1365.

Campos, L. & Alonso-Quecuty, M. (2006) 'Remembering a criminal conversation: Beyond eyewitness testimony', *Memory*, 14 (1), pp. 27-36.

Cant, R. P. & Aroni, R. A. (2008) 'Exploring dietitians' verbal and nonverbal communication skills for effective dietitian-patient communication', *J Hum Nutr Diet*, 21 pp. 502-511.

Carter, W. B., Inui, T. S., Kukull, W. A. & Haigh, V. H. (1982) 'Outcome-based doctor-patient interaction analysis: II. Identifying effective provider and patient behavior.', *Med Care*, 20 pp. 550-566.

Cegala, D. J. (1997) 'A Study of Doctors' and Patients' Communication During a Primary Care Consultation: Implications for Communication Training', *Journal of Health Communication*, 2 (3), pp. 169-1942.

Chao, J., Nau, D. P., Aikens, J. E. & Taylor, S. D. (2005) 'The mediating role of health beliefs in the relationship between depressive symptoms and medication adherence in persons with diabetes', *Res Social Adm Pharm*, 1 (4), pp. 508-525.

Chapelle, A., Champion, P. & May, C. (1997) 'Clinical terminology: anxiety and confusion amongst families undergoing genetic counselling.', *Patient Educ Couns*, 32 pp. 81-91.

Chapman, G. E., Sellaeg, K., Levy-Milne, R., Ottem, A., Barr., S. I., Fierini, D., Marchessault, G., Nolan, D., Paquette, M., Saunders, N., Thiele, K. & "DO IT WITH FOCUS" RESEARCH GROUP (2005) 'Canadian Dietitians' Approaches to Counseling Adult Clients Seeking Weight-Management Advice ', *Journal of the American Dietetic Association*, 105 (8), pp. 1275-1279.

Choudhary, P. (2004) 'Review of dietary recommendations for diabetes mellitus', *Diabetes Res Clin Pract*, 65 (Suppl I), pp. 9 -15.

Ciechanowski, P. S., Katon, W. J. & Russo, J. E. (2000) 'Depression and diabetes: impact of depressive symptoms on adherence, function, and costs.', *Arch Intern Med*, 160 (21), pp. 3278-3285.

Clark, M. & Hampson, S. E. (2001) 'Implementing a psychological intervention to improve lifestyle self-management in patients with type 2 diabetes', *Patient Educ Couns*, 42 pp. 247-256.

Clark, M. & Hampson, S. E. (2003) 'Comparison of patients' and health care professionals' beliefs and attitudes towards Type 2 diabetes ', *Diabetic Medicine*, 20 (2), pp. 152-154.

Claxton, A. J., Cramer, J. A. & Pierce, C. (2001) 'A systematic review of the associations between dose regimens and medication compliance', *Clinical Theraputics*, 23 (8), pp. 1296-1310.

Clement, S. (1995) 'Diabetes self-management education', *Diabetes Care*, 18 pp. 1204–1214.

Coates, V. E. & Boore, J. R. P. (1996) 'Knowledge and diabetes selfmanagement', *Patient Education & Counseling*, 29 pp. 99-108.

Cohen, M. Z., Tripp-Reimer, T., Smith, C., Sorofman, B. & Lively, S. (1994) 'Explanatory models of diabetes: patient practitioner variation', *Social science & Medicine*, 38 (1), pp. 59 -66.

Coleman, T. (2000) 'Using video-recorded consultations for research in primary care: Advantages and limitations', *Family Practice*, 17 (5), pp. 422-427.

Coles, C. (1990) 'Diabetes education: letting the patient into the picture', *Practical Diabetes*, 15 pp. 1423-1432.

Collins, S. (2005) 'Explanations in consultations: the combined effectiveness of doctors' and nurses' communication with patients', *Med Educ*, 39 (8), pp. 785-796.

Comstock, L. M., Hooper, E. M., Goodwin, J. M. & Goodwin, J. S. (1982) 'Physician behaviors that correlate with patient satisfaction', *J Med Educ*, 57 pp. 105-112.

Connor, M., Fletcher, I. & Salmon, P. (2009) 'The analysis of verbal interaction sequences in dyadic clinical communication: A review of methods', *Patient Education and Counseling*, 75 pp. 169-177.

Contento, I., Balch, G. I. & Bronner, Y. I. (1995) 'The effectiveness of nutrition education and implications for nutrition education policy, programs and research: a review of research ', *Journal of Nutrition Education*, 27 pp. 277-418.

Corbin, J. & Strauss, A. (2008) *Basics of Qualitative Research.* Thousand Oaks CA: Sage.

Coughlan, P. (2005) 'Physician Empathy and Patient-Physician Concordance on Treatment Decisions'. *Unpublished MSc dissertation*. [MSc in Health Psychology] Southampton: Southampton University.

Coughlan, P., Skinner, T. C., Cranston, I. & Kar, P. (2006) 'Physician Empathy and Patient-Physician Concordance on Treatment Decisions', *Diabetic Medicine*, 23 (2 Suppl), pp. S114.

Cox, A., Jenkins, V., Catt, S., Langridge, C. & Fallowfield, L. (2006) 'Information needs and experiences: an audit of UK cancer patients.', *Eur J Oncol Nurs*, 10 (4), pp. 263-272.

Cox, M. E., Yancy, W. S. J., Coffman, C. J., Ostbye, T., Tulsky, J. A., Alexander, S. C., Brouwer, R. J., Dolor, R. J. & Pollak, K. I. (2011) 'Effects of counseling techniques on patients' weight-related attitudes and behaviors in a primary care clinic', *Patient Educ Couns*, pp. [Epub ahead of print].

Coyne, T., Olson, M., Bradham, K., Garcon, M., Gregory, P. & Scherch, L. (1995) 'Dietary satisfaction correlated with adherence in the Modification of Diet in Renal Disease Study', *Journal of the American Dietetic Association*, 95 (11), pp. 1301-1306.

Cramer, J. A. (2004) 'A Systematic Review of Adherence With Medications for Diabetes', *Diabetes Care*, 27 (5), pp. 1218-1224.

Creswell, J. W. (1998) *Qualitative Inquiry and Research Design: Choosing among five traditions.* London: Sage.

Creswell, J. W. (2009) *Research Design: Qualitative, Quantitative, and Mixed Methods Approaches.* 3rd edn. Thousand Oaks, California: Sage.

Creswell, J. W. & Plano Clark, V. L. (2007) *Describing and conducting mixed methods research.* Thousand Oaks. CA: Sage.

Crow, R., Gage, H., Hampson, S., Hart, J., Kimber, A., Storey, L. & Thomas, H. (2002) 'The measurement of satisfaction with healthcare: implications for practice from a systematic review of the literature', *Health Technology Assessment*, 6 (32), pp. 1-244.

Cullen, K. W., Baranowski, T. & Smith, S. P. (2001) 'Using goal setting as a strategy for dietary behavior change', *Journal of the American Dietetic Association*, 101 (5), pp. 562-566.

Curry, K. R. & Jaffe, A. (1998) *Nutrition Counseling & communication skills.* W.B.Saunders company.

D' Eramo-Melkus, G. A. & Demas, P. (1989) 'Patient Perceptions of Diabetes Treatment Goals', *The Diabetes Educator*, 15 (5), pp. 440-443.

DAFNE Study Group (2002) 'Training in flexible, intensive insulin management to enable dietary freedom in people with type 1 diabetes: dose adjustment for normal eating (DAFNE) randomised controlled trial.', *BMJ*, 325 (7367), pp. 746. Doi:1136/bmj.7367746.

Daneman, D., Siminerio, L. M., Transue, D., Betschart, J., Drash, A. & Becker, D. (1985) 'The role of self-monitoring of blood glucose in the routine management of children with insulin-dependent diabetes mellitus', *Diabetes Care*, 8 (1), pp. 1-4.

Davenport, S., Goldberg, D. & Millar, T. (1987) 'How psychiatric disorders are missed during medical consultations', *The Lancet.*, 22 pp. 439-441.

Davidson, C., Kowalska, A. Z., Nutman, P. N. & Pearson, G. C. (1987) 'Dietitian-patient communication: a critical appraisal and approach to training', *Human Nutrition - Applied Nutrition*, 41 (6), pp. 381-389.

Davies, M. J., Heller, S., Skinner, T. C., Campbell, M. J., Carey, M. E., Cradock, S., Dallosso, H. M., Daly, H., Doherty, Y., Eaton, S., Fox, C., Oliver, L., Rantell, K., Rayman, G. & Khunti, K., Diabetes Education and Self Management for Ongoing and Newly Diagnosed Collaborative, (2008) 'Effectiveness of the diabetes education and self management for ongoing and newly diagnosed (DESMOND) programme for people with newly diagnosed type 2 diabetes: cluster randomised controlled trial.', *BMJ*, 336 (7642), pp. 491-495.

Dawson, S. I., Willis, J., Florkowski, C. M. & Scott, R. S. (2008) 'Cause-specific mortality in insulin-treated diabetic patients: a 20-year follow-up', *Diabetes Res Clin Pract*, 80 (1), pp. 16-23.

Day, J. L. (2000) 'Diabetic patient education: determination of success', *Diabetes Metab Res Rev*, 16 (1 Suppl), pp. 570-574.

DCCT (1993) 'The Effect of intensive treatment of Diabetes on the development and progression of long-term complications in Insulin-dependant Diabetes Mellitus', *The New England Journal of Medicine*, 329 (14), pp. 977-986.

De Fine Olivarius, N., Beck-Nielsen, H., Andreasen, A. H., Hørder, M. & Pederson, P. A. (2001) 'Randomised controlled trial of structured personal care of type 2 diabetes mellitus', *BMJ*, 323 pp. 970-975.

de Haes, H. (2006) 'Dilemmas in patient -centredness and shared decision making: A case for vulnerability', *Patient Educ Couns*, 62 pp. 291-298.

de Haes, H. & Bensing, J. (2009) 'Endpoints in medical communication research, proposing a framework of functions and ouctomes', *Patient Educ Couns*, 74 pp. 287-294.

Deakin, T. A., Cade, J. E., Williams, R. & Greenwood, D. C. (2006) 'Structured patient education: the diabetes X-PERT Programme makes a difference', *Diabetic Medicine*, 23 (9), pp. 944-954.

Dean, A. J., Walters, J. & Hall, A. (2010) 'A systematic review of interventions to enhance medication adherence in children and adolescents with chronic illness', *Arch Dis Child*, 95 (9), pp. 717-723.

Deci, E. L. & Ryan, R. M. (1985) *Intrinsic motivation and self-determination in human behaviour.* New York: Plenum Press.

Del Piccolo, L., Mazzi, M. A., Dunn, G., Sandri, M. & Zimmermann, C. (2007) 'Sequence analysis in multilevel models. A study on different sources of patient cues in medical consultations', *Social science & Medicine*, 65 pp. 2357-2370.

Del Piccolo, L., Mazzi, M. A., Scardoni, S., Gobbi, M. & Zimmermann, C. (2008) 'A theory-based proposal to evaluate patient-centred communication in medical consultations', *Health Education*, 108 (5), pp. 355-372.

Del Piccolo, L., Mead, N., Gask, L., Mazzi, M. A., Goss, C., Rimondini, M. & Zimmermann, C. (2005) 'The English version of the Verona medical interview classification system (VR-MICS) An assessment of its reliability and a comparative cross-cultural test of its validity', *Patient Education & Counseling*, 58 pp. 252-264.

Delahanty, L. M., Grant, R. W., Wittenberg, E., Bosch, J. L., Wexler, D. J., Cagliero, E. & Meigs, J. B. (2007) 'Association of diabetes-related emotional distress with diabetes treatment in primary care patients with Type 2 diabetes', *Diabet Med*, 24 (1), pp. 48-54.

Delahanty, L. M. & Halford, B. N. (1993) 'The Role of Diet Behaviours in Achieving Improved Glycaemic Control in Intensively Treated patients in the Diabetes Control and Complications Trial', *Diabetes Care*, 16 (11), pp. 1453-1458.

Dent, E., Brown, R., Dowsett, S., Tattersall, M. & Butow, P. (2005) 'The Cancode interaction analysis system in the oncological setting: reliability and validity of video and audio tape coding', *Patient Education & Counseling*, 561 pp. 35-44.

Department of Health (2001) *The Expert Patient: A New Approach to Chronic Disease Management in the 21st Century.* London: UK Department of Health report. Available at: http://www.doh.gov.uk/cmo/ep-report.pdf.

Department of Health (2003) *The National Framework (NSF) for Diabetes: Standards.* London: Department of Health. Available at: http://www.dh.gov.uk/en/Publicationsandstatistics/Publications/PublicationsPolic yAndGuidance/DH_4003246.

Department of Health (2005) Creating a patient-led NHS: Delivering the NHS Improvement Plan. London HMSO. Available at:

http://www.dh.gov.uk/en/Publicationsandstatistics/Publications/PublicationsPolic yAndGuidance/DH_4106506.

Department of Health (2007) 'Working together for better diabetes care. Clinical case for change: Report by Sue Roberts, National Director for Diabetes'. [Online]. Available at:

http://www.dh.gov.uk/en/Publicationsandstatistics/Publications/PublicationsPolic yAndGuidance/DH_074702.

Department of Health (2008) 'Part 1:Why standardise the structure and content of medical records?'. *A Clinician's Guide to Record Standards* DH Publications. [Online]. Available at: http://www.rcplondon.ac.uk/resources/clinicalresources/standards-medical-record-keeping/structure-and-content-medicalnotes/de.

Department of Health (2010a) 'Improving care for people with long term conditions: 'at a glance' information sheets for healthcare professionals'. Department of Health. [Online]. Available at: http://www.dh.gov.uk/en/Publicationsandstatistics/Publications/PublicationsPolic yAndGuidance/DH_121603 (Accessed: 28/1/11).

Department of Health (2010b) 'Equity and excellence: Liberating the NHS'. Department of Health. [Online]. Available at: http://www.dh.gov.uk/en/Publicationsandstatistics/Publications/PublicationsPolic yAndGuidance/DH_117353 (Accessed: white paper).

Department of Health (2010c) 'Six years on: delivering the Diabetes National Service Framework'. Department of Health. [Online]. Available at: ttp://www.dh.gov.uk/en/Publicationsandstatistics/Publications/PublicationsPolicy AndGuidance/DH_112509 (Accessed: 13/2/11).

Department of Health (2011) 'Long Term Conditions'. DH Publications. [Online]. Available at: http://www.dh.gov.uk/en/Healthcare/Longtermconditions/index.htm (Accessed: 28/1/11).

Department of Health and Diabetes UK (2005) 'Structured Patient Education in Diabetes: Report from the Patient Education Working Group '. *269106*, DH Publications. [Online]. Available at: www.dh.gov.uk/publications.

Diabetes and Nutrition Study Group of the European Association for The Study of Diabetes (2000) 'Recommendations of the Nutritional Management of patients with diabetes', *Eur J Clin Nutr*, 54 pp. 353-355.

Diabetes UK (2009) *Imporving supported self-management for people with diabetes.* London. Available at: http://www.diabetes.org.uk/Professionals/Publications-reports-and-resources/Reports-statistics-and-case-studies/Reports/Supported-self-management/?print=1.

Dietrich, C. (1996) 'Factors influencing the attitudes held by women with type II diabetes: a qualitative study', *Patient Educ Couns*, 29 pp. 13-23.

DiMatteo, M. R., Taranta, A., Friedman, H. S. & Prince, L. M. (1980) 'Predicting satisfaction from physicians non-verbal communication skills', *Medical Care*, 18 pp. 376-387.

Doherty, Y., Hall, D., James, P. T., Roberts, S. H. & Simpson, J. (2000) 'Change counselling in diabetes: the development of a training programme for the diabetes team', *Patient Educ Couns*, 40 pp. 23-78.

Donnan, P. T., MacDonald, T. M. & Morris, A. D. (2002) 'Adherence to prescribed oral hypoglycaemic medication in a population of patients with type 2 diabetes: a retrospective cohort study', *Diabetic Medicine*, 19 pp. 279-284.

Donnelly, M. B. & Anderson, R. M. (1990) 'The role related attitude of physicians , nurses and dieticians in the treatment of diabetes', *Medical Care*, 28 (2), pp. 175-179.

Duke, S. A., Colagiuri, S. & Colagiuri, R. (2009) 'Individual patient education for people with type 2 diabetes mellitus', *Cochrane Database of Systematic Reviews*, Issue 1. Art. No.: CD005268. DOI: 10.1002/14651858.CD005268.pub2.

Dunn, S. M. (1988) 'Treatment-education. Bailieres', *Clin Endocrionolgy Metab*, 2 pp. 493-506.

Easter, D. W. & Beach, W. (2004) 'Competent patient care is dependent upon attending to empathic opportunities presented during interview sessions ', *Current Surgery*, 61 (3), pp. 313-318.

Eccles, M. P., Whitty, P. M., Speed, C., Steen, I. N., Vanoli, A., Hawthorne, G. C., Grimshaw, J. M., Wood, L. J. & McDowell, D. (2007) 'A pragmatic cluster randomised controlled trial of a Diabetes REcall And Management system: the DREAM trial.', *Implement Sci*, 2doi: 10.1186/1748-5908-2-6

Edwards, A., Elwyn, G., Hood, K., Robling, M., Atwell, C., Holmes-Rovner, M., Kinnersley, P., Houston, H. & Russell, I. (2003) 'The development of COMRADE-a patient based outcome measure to evaluate the effectiveness of risk communication and treatment decision making in consultations ', *Patient Education & Counseling*, 50 (311-22),

Egan, G. (2002) The Skilled Helper: A Problem -Management and Opportunity - Development Approach to Helping. 7th edn. Brooks/Cole.

Eide, H., Frankel, R., Haaversen, A. C., Vaupel, K. A., Graugaard, P. K. & Finest, A. (2004) 'Listening for feelings: identifying and coding empathic and potential empathic opportunities in medical dialogue', *Patient Education & Counseling*, 54 (3), pp. 291-297.

Eide, H., Sibbern, T. & Johannessen, T. (2011) 'Empathic accuracy of nurses' immediate responses to fibromyalgia patients' expressions of negative emotions: an evaluation using interaction analysis', *J Adv Nurs*, 10 pp. [Epub ahead of print].

Eldh, I. (2006) 'Conditions for Patient Participation and Non Participation in Health Care', *Nursing Ethics*, 13 (5), pp. 503-514.

Ellis, S.E, Speroff, T., Dittus, R., Brown, A., Pichert, J. W. & Elasy, T. (2004) 'Diabetes patient education: a meta-analysis and meta-regression', *Patient Education & Counseling*, 52 pp. 97-105.

England, C. Y., Coulman, K. D., gorton, J. G., Paxton, E. C., Andrews, R. C. & on behalf of the Early ACTID Study Group (2009) 'Does a diagnosis of type 2 diabetes lead to a change in diet?', *Diabetic Medicine, Abstracts for Diabetes UK 2009 annual conference*, 26 pp. 452.

Engum, A. (2007) 'The role of depression and anxiety in onset of diabetes in a large population- based study', *J Psychosom Res*, 62 (1), pp. 31-38.

Entwistle, V., Prior, M., Skea, Z. C. & Francis, J. J. (2008) 'Involvement in treatment decision-making: Its meaning to people with diabetes and implications for conceptualisation', *Social science & Medicine*, 66 pp. 362-375.

Entwistle, V. A. & Watt, I. S. (2006) 'Patient involvement in treatment decisionmaking: The case for a broader conceptual framework', *Patient Education and Counseling*, 63 pp. 268-278.

Epstein, R., M., Franks, P., Fiscella, K., Shields, C., G., Meldrum, S. M., Kravitz, R. L. & Duberstein, P. R. (2005) 'Measuring Patient-centred communication in patient –physician consultations: Theoretical and practical issues', *Social science & Medicine*, 61 pp. 1516-1528.

Epstein, R. M., Alper, B. S. & Quill, T. E. (2004) 'Communicating evidence for participatory decision making', *JAMA*, 291 (19), pp. 2359-2366.

Epstein, L. H. & Cluss, P. A. (1982) 'A behavioural medicine perspective on adherence to long term medical regimens', *J Consult Clin PSychol*, 50 pp. 950-971.

Epstein, R. M., Hadee, T., Carroll, J., Meldrum, S. C., Lardner, J. & Shields, C. G. (2007) "Could this Be Something Serious?" Reassurance, Uncertainty, and Empathy in Response to Patients' Expressions of Worry', *J Gen Intern Med*, 22 (12), pp. 1731-1739.

Eren, I., Erdi, O. & Sahin, M. (2008) 'The effect of depression on quality of life of patients with type II diabetes mellitus', *Depress Anxiety*, 25 (2), pp. 98-106.

Fagard, R. H. (2009) 'Smoking amplifies cardiovascular risk in patients with hypertension and diabetes', *Diabetes Care*, 32 (2), pp. S429-431.

Fallowfield, L. & Jenkins, V. (1999) 'Effective communication skills are the key to good cancer care', *European Journal of Cancer*, 35 (11), pp. 1592-1597.

Fallowfield, L., Jenkins, V., Farewell, V., Saul, J. & Eves, R. (2002) 'Efficacy of a communication skills training model for oncologists: a randomized controlled trial.', *Lancet*, 359 (9307), pp. 650-656.

Farmer, A. J., Toby Prevost, A. T., Hardeman, W., Craven, A., Sutton, S., Griffin, S. J., Kinmouth, A. L. & The Support and Advice for Medication Trial Group (2008) 'Protocol for SAMA (Support and Advice for Medication Study): A randomised controlled trial of an intervention to support patients with type 2 diabetes with adherence to medication '. *BMC Family Practice*, 9 (2), DOI: 10. 1186/1471-2296-9-20 (Accessed: 31/07/2008).

Feste, C. & Anderson, R. M. (1995) 'Empowerment: from philosophy to practice', *Patient Educ Couns*, 6 pp. 139-144.

Field, a. & Hole, G. (2003) *How to Design and Report Experiments.* London: Sage Publications. 2006

Fieldhouse, P. (1979) 'An interview model for use in dietetic training', *J Hum Nutr Diet*, 33 pp. 26-210.

Finset, A. & Mjaaland, T. A. (2009) 'The medical consultation viewed as a value chain: A neurobehavioral approach to emotion regulation in doctor–patient interaction', *Patient Education and Counseling*, 74 pp. 323-330.

Fleming, R. M. (1996) 'Assessing the independent effect of dietary counseling and hypolipidemic medications on serum lipids', *Angiology*, 47 (9), pp. 83-40.

Flocke, S. A. & Stange, K. C. (2004) 'Direct observation and patient recall of health behavior advice', *Preventative Medicine*, 38 pp. 343-349.

Fogarty, L. A., Curbow, B. A., Wingard, J. R., McDonnell, K. & Somerfield, M. R. (1999) 'Can 40 Seconds of Compassion Reduce Patient Anxiety?', *Journal of Clinical Oncology*, 17 (1), pp. 371.

Ford, S., Hall, A., Ratcliffe, D. & Fallowfield, L. (2000) 'The Medical Interaction Process System (MIPS): an instrument for analysing interviews of oncologists and patients with cancer', *Social science & Medicine*, 50 (4), pp. 553-566.

Foreyt, J. P. & Poston, W. S. n. (1999) 'The challenge of diet, exercise and lifestyle modification in the management of the obese diabetic patient ', *International Journal of Obesity and Related Metabolic Disorders*, 23 (Suppl 7), pp. S5-11.

Forsetlund, L., Bjørndal, A., Rashidian, A., Jamtvedt, G., O'Brien, M. A., Wolf, F., Davis, D., Odgaard-Jensen, J. & Oxman, A. D. (2009) 'Continuing education meetings and workshops: effects on professional practice and health care outcomes.', *Cochrane Database Syst Rev*, 15 (2), pp. CD003030.

Fossum, B. & Arborelius, E. (2004) 'Patient-centred communication: videotaped consultations', *Patient Educ Couns*, 54 (2), pp. 163-169.

Francis, J. J., Eccles, M. P., Johnston, M., Whitty, P., Grimshaw, J. M., Kaner, E. F. S., Smith, L. & Walker, A. (2008) 'Explaining the effects of an intervention designed to promote evidence-based diabetes care: a theory ', *Implementation Science*, 3 pp. 50. doi:10.1186/1748-5908-3-50.

Franz , M. J. (2004) 'Evidence-based medical nutrition therapy for diabetes', *Nutr Clin Pract*, 19 (2), pp. 137-144.

Freeman, J. & Loewe, R. (2000) 'Barriers to communication about diabetes mellitus. Patients' and physicians' different view of the disease', *Journal of Family Practice*, 49 (6), pp. 507-512.

Fuertes, J. N., Boylan, L. S. & Fontanella, J. A. (2008) 'Behavioral Indices in Medical Care Outcome: The Working Alliance, Adherence, and Related Factors', *J Gen Intern Med*, 24 (1), pp. 80-85.

Funnell, M. M. & Anderson, R. M. (2004) 'Empowerment and Self-Management of Diabetes', *Clinical Diabetes*, 22 pp. 123-127.

Funnell, M. M., Nwankwo, R., Gillard, M. L., Anderson, R. M. & Tang, T. S. (2005) 'Implementing an empowerment-based diabetes self-management education programme', *Diabetes Educator*, 31 (1), pp. 53-61.

Gable, J. (2007) Counselling Skills for Dietitians. 2nd edn. Blackwell Publishing.

Gaede, P., Lund-Andersen, H., Parving, H. H. & Pedersen, O. (2008) 'Effect of a multifactorial intervention on mortality in type 2 diabetes', *N Engl J Med*, 58 (6), pp. 580-589.

Garg, A. (1998) 'High-monounsaturated-fat diets for patients with diabetes mellitus: a meta-analysis', *The American Journal of Clinical Nutrition*, 67 (3 Suppl), pp. 577S.

Gebhardt, W. A. (2006) 'Contextualizing Health Behaviors: The Role of Personal Goals'. in De Ridder, D. and De Wit, J. (eds.) *Self-regulation in Health Behavior.* Chichester: John Wiley & Sons Ltd.

Gentili, P., Maldonato, A., Grieco, R. & Santini, A. (2001) 'Influence of patients' representations and beliefs about diabetes and its treatment on their adherence to therapy', *Diabetes, Nutrition & Metabolisim*, 14 (3), pp. 140-152.

Georgiadis, M. M., Biddle, S. J. H. & Stavrou, N. A. (2006) 'Motivation for weight-loss diets: A clustering, longitudinal field study using self-esteem and self-determination theory perspectives', *Health Education Journal*, 65 pp. 53-72.

Gillies, C. L., Abrams, K. R., Lambert, P. C., Cooper, N. J., Sutton, A. J., Hsu, R. T. & Khunti, K. (2007) 'Pharmacological and lifestyle interventions to prevent or delay type 2 diabetes in people with impaired glucose tolerance: systematic review and meta-analysis', *BMJ*, 334 (7588), pp. 299. doi:10.1136/bmj.39063.689375.55

Glasgow, R. E. & Anderson, R. M. (1999) 'In diabetes care, moving from compliance to adherence is not enough: something entirely different is needed', *Diabetes Care*, 22 pp. 2090-2092.

Glasgow, R. E., Emont, S. & Miller, D. C. (2006) 'Assessing delivery of the five 'As' for patient-centered counseling ', *Health Promotion International*, 21 (3), pp. 245-255.

Glasgow, R. E., Hampson, S. E., Strycker, L. A. & Ruggiero, L. (1997) 'Personal-model beliefs and social environmental barriers related to diabetes self management', *Diabet Care*, 20 pp. 556-561.

Glasgow, R. E., Marrero, D. & Fisher, E. B. (1999) 'Behavioural Science in Diabetes ', *Diabetes Care*, 22 (5), pp. 832-843.

Glasgow, R. E. & Osteen, V. L. (1992) 'Evaluating diabetes education: are we measuring the most important outcomes?', *Diabet Care*, 15 pp. 1423-1432.

Glasgow, R. E., Roland, G. H., Anderson, R. M., Freidman, N. M., Hayward, R. A., Marrero, D. G., Taylor, C. B. & Vinicor, F. (2001) 'Report of the health care delivery work group: Behavioral research related to the establishment of a chronic disease model for diabetes care', *Diabet Care*, 24 pp. 124-130.

Glasgow, R. E., Toobert, D., Hampson, S., Brown, J., Lewinsohn, P. & Donnelly, J. (1992) 'Improving self-care among older patients with type II diabetes: "the sixty something ..."study', *Patient Educ Couns*, 19 pp. 61-74.

Glasgow, R. E. & Toobert, D. J. (1988) 'Social Environment and Regimen Adherence Among Type II Diabetic Patients', *Diabetes Care*, 11 (5), pp. 377-386.

Glasgow, R. E., Toobert, D. J. & Gillette, C. D. (2001) 'Psychological barriers to diabetes self management and quality of life', *Diabetes Spectrum*, 14 pp. 33-41.

Goebel-Fabbri, A. E. (2009) 'Disturbed eating behaviors and eating disorders in type 1 diabetes: clinical significance and treatment recommodations', *Curr Diab Rep*, 9 (2), pp. 133-139.

Goldstein, M. G., O., W. E., DePue, J. & 61-79, A. J. P. M. S. (2004) 'Planning Committee of the Addressing of Multiple Behavioural Risk Factors in Primary Care project. Multiple behavioural factor interventions in primary care. Summary of research evidence', *Am J Prev Med*, 2 (Suppl), pp. 61-79.

Golin, C. E., DiMatteo, M. R. & Gelberg, L. (1996) 'The role of patient participation in the doctor visit. Implications for adherence to diabetes care', *Diabetes Care*, 19 (10), pp. 1153 - 1164.

Goodall, T. A. & Halford, W. K. (1991) 'Self- management of diabetes mellitus: a critical review', *Health Psychol*, 10 pp. 1-8.

Goodchild, C. E., Skinner, T. C. & Parkin, T. (2005) 'The value of empathy in dietetic consultations. A pilot study to investigate its effect on satisfaction, autonomy and agreement', *Journal of Human Nutrition & Dietetics*, 18 (3), pp. 181-185.

Goss, C., Mazzi, M. A., L., D. P., Rimondini, M. & Zimmermann, C. (2004) 'Information-giving sequences in general practice consultations', *Journal of Evaluation in Clinical Practice*, 11 (4), pp. 339-349.

Green, J. & Thorogood, N. (2009) *Qualitative methods for health research.* 2nd edn. Thousand Oaks, CA: Sage.

Greenfield, S. & Kaplan, S. H. (1988) 'Patients' Participation in Medical Care: Effects on Blood Sugar Control and Quality of Life in Diabetes', *J Gen Intern Med*, 3 pp. 448-457.

Griffin, S., Kinmouth, A. L., Skinner, C. & Kelly, J. (1999) 'Educational and Psychosocial Interventions for Adults with Diabetes', *British Diabetic Association Report*.

Griffin, S. J., Kinmonth, A. L., Veltman, M. W., Gillard, S., Grant, J. & Stewart, M. (2004) 'Effect on health-related outcomes of interventions to alter the interaction between patients and practitioners: a systematic review of trials', *Ann Fam Med*, 2 (6), pp. 595-608.

Guimond, P., Bunn, H., O'Conner, A. M., Jacobsen, M. J., Tait, V. K., Drake, E. R., Graham, I. D., D., S. & Elmslie, T. (2003) 'Validation of a tool to assess health practitioners' decision support and communication skills', *Patient Education & Counseling*, 50 pp. 235-245.

Hall, J. A. & Dornan, M. C. (1988) 'What patients like about their medical care and how often they are asked: a meta analysis of the satisfaction literature', *Soc Sci Med*, 27 pp. 935-939.

Hall, J. A., Horgan, T. G., Stein, T. S. & Roter, D. L. (2002) 'Liking in the physician--patient relationship.', *Patient Educ Couns*, 48 (1), pp. 69-77.

Hall, J. A., Irish, J. T., Roter, D. L., Ehrlich, C. M. & Miller, L. H. (1994) 'Satisfaction, gender, and communication in medical visits', *Med Care*, 32 pp. 1216-1231.

Hall, J. A., Stein, T. S., Roter, D. L. & Riser, N. (1999) 'Inaccuracies in physician' perceptions of their patients', *Medical Care*, 37 pp. 1164-1168.

Hambly, H., Robling, M., Crowne, E., Hood, K., Gregory, J. W. & DEPICTED Study Team (2009) 'Communication skills of healthcare professionals in paediatric diabetes services', *Diabet Med*, 26 (5), pp. 502-509.

Hampson, S. E., Glasgow, R. E. & Foster, L. S. (1995) 'Personal Models of Diabetes Among Older Adults: Relationship to Self-Management and Other Variables', *Diabetes Educator*, 21 (4), pp. 300-307.

Hampson, S. E. & Mckay, H. G. (1996) 'Patient-physician interactions in diabetes management: consistencies and variation in the structure and content of two consultations', *Patient Education and Counseling*, 29 (1), pp. 49-58.

Hancock, B., Windridge, K. & Ockleford, E. (2007) 'An Introduction to Qualitative Research.' *The NIHR RDS EM / YH*.

Harrington, J., Noble, L. M. & Newman, S. P. (2004) 'Improving patients' communication with doctors: A systematic review of intervention studies', *Patient Education and Counseling*, 5 (1), pp. 7-16.

Harsha, D. W. (1999) 'Dietary Approaches to Stop Hypertension: a summary of study results. DASH Collaborative Research Group', *Journal of the American Dietetic Association*, 99 (8 Suppl), pp. S35.

Hartz, A., Kent, S., James, P., Xu, Y., Kelly, M. & Daly, J. (2006) 'Factors that influence improvement for patients with poorly controlled type 2 diabetes', *Diabetes Res Clin Pract*, 74 (3), pp. 227-232.

Harvey, J. N. & Lawson, V. L. (2009) 'The importance of health belief models in determining self-care behaviour in diabetes', *Diabet Med*, 26 (1), pp. 5-13.

Hauenstein, D. J., Schiller, M. R. & Hurley, R. S. (1987) 'Motivational techniques of dietitians counselling individuals with type II diabetes', *JAMA*, 87 (1), pp. 37-42.

Haynes, R. B., Yao, X., Degani, A., Kripalani, S., Garg, A. & McDonald, H. P. (2005) 'Interventions for enhancing medication adherence', [Online]. Available at:www.mrw.interscience.wiley.com/cochrane/clsysrev/articles/cD000011/frame. htm.

Health Development Agency (2003) *The management of obesity and overweight: an analysis of reviews of diet, physical activity and behavioural approaches.* London: HDA.

Health Services Research Group (1992) 'A guide to direct measures of patient satisfaction in clinical practice', *Can Med Assoc J*, 146 (10), pp. 1727-1731.

Heisler, M., Bouknight, R. R., Hayward, R. A., Smith, D. M. & Kerr, E. A. (2002) 'The relative importance of physician communication, participatory decision making and patient understanding in diabetes self -management ', *J Gen Intern Med*, 17 (4), pp. 243-252.

Heisler, M., Cole, I., Weir, D., Kerr, E. A. & Hayward, R. A. (2007) 'Does physician communication influence older patients' diabetes self-management and glycaemic control? Results from the Health and Retirement Study (HRS)', *J Gerontol A Biol SCi Med Sci*, 62 (12), pp. 1435-1442.

Heisler, M., Vijan, S., Anderson, R. M., Ubel, P. A., Bernstein, S. J. & Hofer, T. P. (2003) 'When do patients and their physicians agree on diabetes treatment goals and strategies, and what difference does it make? ', *Journal of General Internal Medicine*, 18 (11), pp. 893-902.

Hemmerdinger, J. M., Stoddart, S. D. & Lilford, R. J. (2007) 'A systematic review of tests of empathy in medicine', *BMC Med Educ*, 25 (7), pp. 24.

Herpertz, S., Albus, C., Kielmann, R., hagemann-Patt, H., K., L., Köhle, K., Mann, K. & Sen, W. (2001) 'Comorbidity of diabetes mellitus and eating disorders: A follow-up study', *Psychosomatic Research*, 51 (5), pp. 673-678.

Hill-Briggs, F., Gary, T. L., Bone, L. R., Hill, M. N., Levine, D. M. & Brancati, F. L. (2005) 'Medication adherence and diabetes control in urban African Americans with type 2 diabetes', *Health Psychology*, 24 (4), pp. 349-357.

Hinton, P. S. & Olson, C. M. (2001) 'Postpartum exercise and food intake: the importance of behavior-specific self-efficacy.', *J Am Diet Assoc*, 101 (12), pp. 1430-1437.

Hiscock, J., Legard, R. & Snape, D. (2001) 'Listening to diabetes service users qualitative findings for the Diabetes National Service Framework'. [Online]. Available at

http://www.dh.gov.uk/en/Publicationsandstatistics/Publications/PublicationsPolic yAndGuidance/DH_4008645 (Accessed: 12/2/11).

Hojat, M., Gonnella, J. S., Nasca, T. J., Mangione, S., Vergare, M. & Magee, M. (2002) 'Physician empathy: definition, components, measurement, and relationship to gender and speciality', *Am J Psychiatry*, 159 (9), pp. 1563-1569.

Hojat, M., Louis, D. Z., Markham, F. W., Wender, R., Rabinowitz, C. & Gonnella, J. S. (2011) 'Physicians' Empathy and Clinical Outcomes for Diabetic Patients', *Acad Med*, 86 (3), pp. 359-364.

Holli, B., Calabrese, R. J. & O'Sullivan Maillet, J. (2003) *Communication and Education Skills for Dietetic Professionals.* 4th edn. Lippincott Williams & Wilkins

Holmström, I., Larsson, J., Lindberg, E. & Rosenqvist, U. (2003) 'Improving the diabetes-patient encounter by reflective tutoring for staff', *Patient Educ Couns*, 53 pp. 325-332.

Holmstrom, I. & Rosenqvist, U. (2005) 'Misunderstandings about illness and treatment among patients with type 2 diabetes', *J Adv Nurs*, 49 (2), pp. 146-154.

Hope-Stone, L. D. & Mills, B. J. (2001) 'Developing empathy to improve patient care: a pilot study of cancer nurses', *Int J Palliat Nurs*, 7 (3), pp. 1465-1450.

Horacek, T. M., Salomón, J. E. & Nelsen, E. K. (2007) 'Evaluation of dietetic students' and interns' application of a lifestyle-oriented nutrition -counseling model', *Patient Education & Counseling*, 68 pp. 113-120.

Hornsten, A., Lundman, B., Selstam, E. K. & Sandstrorm, H. (2005) 'Patient satisfaction with diabetes care', *Journal of Advanced Nursing*, 51 (6), pp. 609-617.

Hua, G. & Tuomilehtoa, J. (2007) 'Lifestyle and outcome among patients with type 2 diabetes ', *International Congress Series*, 1303 pp. 160-171.

Huang, E. S., Gorawara-Bhat, R. & Chin, M. H. (2005) 'Self-reported goals of older patients with type 2 diabetes mellitus', *J Am Geriatr Soc*, 53 (2), pp. 306-311.

Hulsman, R. L. (2009) 'Shifting goals in medical communication. Determinants of gaol detection and response formation', *Patient Educ Couns*, 74 pp. 302-308.

Hulsman, R. L., Ross, W. J., Winnubst, J. A. & Bensing, J. M. (1999) 'Teaching clinically experienced physicians communication skills. A review of evaluation studies', *Med Educ*, 33 pp. 655-668.

Hunt, P. (1995) 'Dietary counselling: theory into practice', *J Inst Health Educ*, 33 pp. 4-8.

Hurley, A. C. & Shea, C. A. (1992) 'Self-efficacy: strategy for enhancing diabetes self-care', *Diabetes Educ*, 18 (2), pp. 146-150.

Huyck, N. I. (1986) 'Teaching dietetic students in the clinical setting', *Journal of the American Dietetic Association*, 86 (2), pp. 234-236.

Ishikawa, H., Yano, E., Fujimori, S., Kinoshita, M., Yamanouchi, T., Yoshikawa, M., Yamazaki, Y. & Teramoto, T. (2009) 'Patient health literacy and patient– physician information exchange during a visit.', *Family Practice*, 26 pp. 517-523.

Isselmann, C., Deubner, L. S. & Hartman, M. (1993) 'A nutrition counseling workshop: Integrating counseling psychology into nutrition practice', *J Am Diet Assoc*, 93 (3), pp. 324-326.

Jackson, J. A., Kinn, S. & Dalgarno, P. (2005) 'Patient-centred outcomes in dietary research', *Journal of Human Nutrition & Dietetics*, 18 (2), pp. 83-92.

Jackson, J. L., Chamberlin, J. & Kroenke, K. (2001) 'Predictors of patient staisfaction', *Soc Sci Med*, 52 pp. 609-620.

Jansen, J., Butow, P. N., van Weert, J. C., van Dulmen, S., Devine, R. J., Heeren, T. J., Bensing, J. M. & Tattersall, M. H. (2008a) 'Does age really matter? Recall of information presented to newly referred patients with cancer', *Journal of Clinical Oncology*, 26 (33), pp. 5450-5457.
Jansen, J., van Weert, J., van der Meulen, N., van Dulmen, S., Heeren, T. & Bensing, J. (2008b) 'Recall in Older Cancer Patients: Measuring Memory for Medical Information', *The Gerontologist*, 48 pp. 149-157.

Jansen, J., van Weert, J. C. M., de Groot, J., van Dulmen, S., Heeren, T. J. & Bensing, J. M. (2010) 'Emotional and informational patient cues: The impact of nurses' responses on recall', *Patient Educ Couns*, 79 pp. 218-224.

Jarret, N. & Payne, S. (1995) 'A selective review of the literature on nurse patient communication; has the patient's contribution been neglected? ', *Journal of Advanced Nursing*, 22 (1), pp. 72-78.

Jenkins, D. J. A., Kendall, C. W. C., Marchie, A., Faulkner, D. A., Wong, J. M. W., de Souza, R., Emam, A., Parker, T. L., Vidgen, E., Trautwein, E. A., Lapsley, K. G., Josse, R. G., Leiter, L. A., Singer, W. & Connelly, P. W. (2005) 'Direct comparison of a dietary portfolio of cholesterol-lowering foods with a statin in hypercholesterolemic participants', *Am J Clin Nutr*, 81 (2), pp. 380-387.

Jensen, H. I., Ammentorp, J. & Kofoed, P. E. (2010) 'User satisfaction is influenced by the interval between a health care service and the assessment of the service.', *Soc Sci Med*, 70 (12), pp. 1882-1887.

Jeon, Y. H., Essue, B., Jan, S., Wells, R. & Whitworth, J. A. (2009) 'Economic hardship associated with managing chronic illness: a qualitative inquiry', *BMC Health Serv Res*, 9 (9), pp. 182.

Joffe, H. & Yardley, L. (2004) 'Content and thematic analysis'. in Marks, D.F. and Yardley, L. (eds.) *Research methods for Clinical and Health Psychology.* London: Sage.

Johnson, R. B. & Onwuegbuzie, A. J. (2004) "Mixed Methods Research: A Research Paradigm Whose Time Has Come', *Educational Researcher*, 33 (7), pp. 14-26.

Johnston-Brooks, C. H., Lewis, M. A. & Garg, S. (2002) 'Self-Efficacy Impacts Self -Care and HbA1c in Young Adults With Type 1 Diabetes.', *Psychosomatic Medicine*, 64 pp. 43-51.

Jones, H., Edwards, L., Vallis, T., Ruggiero, L., Rossi, S., Rossi, J., Greene, G., Prochaska, J. & Zinman, B. (2003) 'Changes in diabetes self-care behaviors make a difference in glycemic control: The Diabetes Stages of Change (DiSC) study', *Diabetes Care*, 26 pp. 732-737.

Julien, E., Senécal, C. & Guay, F. (2009) 'Longitudinal Relations among Perceived Autonomy Support from Health Care Practitioners, Motivation, Coping Strategies and Dietary Compliance in a Sample of Adults with Type 2 Diabetes', *Journal of Health Psychology*, 14 pp. 457-470.

Jupp, V. (2006) *The Sage Dictionary of Social Research Methods.* ed. Jupp, V., London: Sage.

Kaplan, S. H., Greenfield, S. & Ware, J. E. J. (1989) 'Assessing the effects of physician-patient interactions on the outcomes of chronic disease', *Med Care*, 27 pp. S110-127.

Kasila, K., Poskiparta, M., Karhila, P. & Kettunen, T. (2003) 'Patients' readiness for dietary change at the begining of counselling: a transtheoretical model-based assessment', *Jounal of Human Nutrition & Dietetics*, 16 (3), pp. 159-166.

Kasser, V. M. & Ryan, R. M. (1999) 'The relation of psychological needs for autonomy and relatedness to health, vitality, well-being and mortality in a nursing home', *Journal of Applied Social Psychology*, 29 pp. 935-954.

Kavookjian, J., Berger, B. A., Grimley, D. M., Villaume, W. A., Anderson, H. M. & Barker, K. N. (2005) 'Patient decision making: strategies for diabetes diet adherence intervention', *Research In Social & Administrative Pharmacy*, 1 (3), pp. 389-407.

Keating, N. L., Gandhi, T. K., Orav, E. J., Bates, D. W. & Ayanian, J. Z. (2004) 'Patient characteristics and experiences associated with trust in specialist physicians', *Arch Int Med*, 164 pp. 1015-1020.

Kelsey, K. S., Kirlkley, B. G., DeVellis, R. F., Earp, J. A., Ammerman, A. S., Keyserling, T. C., Shannon, J. & Simpson, R. J. J. (1996) 'Social support as a predictor of dietary change in low-income population', *Health Educ Res*, 11 pp. 383-395.

Kenny, D. A. (2008) 'Mediation'. [Online]. Available at: http://davidakenny.net/cm/mediate.htm (Accessed: 28/03/2008).

Kim, S. S., Kaplowitz, S. & Johnston, M. V. (2004) 'The effects of physician empathy on patient satisfaction and compliance', *Evaluation & the Health Professions*, 27 (3), pp. 237-251.

King, E. B., Schlundt, D. G., Pichert, J. W., Kinzer, C. K. & Backer, B. A. (2002) 'Improving the Skills of Health Professionals in Engaging Patients in Diabetes-Related Problem Solving', *The Journal of Continuing Education in the Health Professions*, 22 pp. 94-102.

Kingery, P. M. & Glasgow, R. E. (1989) 'Self-efficacy and outcome expectations in the self-regulation of non-insulin dependent diabetes mellitus', *Health Education*, 20 pp. 13-19.

Kinmonth, A. L., Woodcock, A., Griffin, S., Spiegel, N. & Campbell, M. J. (1998) 'Randomised controlled trial of patient centred care of diabetes in general practice:Impact on current wellbeing and future disease risk', *British Medical Journal*, 317 pp. 1202-1208.

Kinnersley, P., Stott, N., Peters, T., Harvey, I. & Hackett, P. (1996) "A comparison of methods for measuring patient satisfaction with consultations in primary care" ', *Family Practice*, 13 (1), pp. 41-51.

Kinnersley, P., Stott, N., Peters, T. J. & Harvey, I. (1999) 'The patient - centredness of consultations and outcome in primary care', *Br J Gen Pract*, 49 pp. 717-716.

Kirk, J. K., Bertoni, A. G., Case, D., Bell, R. A., Goff, D. C. J. & Narayan, K. M. (2007) 'Predicted risk of coronary heart disease among persons with type 2 diabetes', *Coron Artery Dis.*, 18 (8), pp. 595-600.

Kiuru, P., Poskiparta, M., Kettunen, T., Saltevo, J. & Limatainen, L. (2004) 'Advice-giving Styles by Finish Nurses in Dietary Counseling Concerning Type 2 Diabetes Care', *Journal of Health Communication*, 9 pp. 337-354.

Kiviniemi, M. & Rothman, A. (2006) 'Selective memory biases in individuals' memory for health-related information and behavior recommendations ', *Psychology and Health*, 21 (2), pp. 247-272.

Knight, K. M., Dornan, T. & Bundy, C. (2006) 'The diabetes educator: trying hard, but must concentrate more on behaviour', *Diabetic Medicine*, 23 pp. 485-501.

Kogut, S. J., Andrade, S. E., Willey, C. & Larrat, E. P. (2004) 'Nonadherence as a predictor of antidiabetic drug therapy intensification (augmentation)', *Pharmacoepidemiol Drug Saf.*, 13 (9), pp. 591-598.

Kolb, D. (1984) *Experiential Learning: Experience as the Source of Learning and Development* Englewood Cliffs, NJ: Prentice-Hall.

Kolbe, J. (2002) 'The Influence of Socioeconomic and Psychological Factors on Patient Adherence to Self- Management Strategies: Lessons Learned in Asthma', *Disease Management & Health Outcomes*, 10 (9), pp. 551-570.

Kravitz, R., Cope, D., Bhrany, V. & Leake, B. (1994) 'Internal medicine patients' expectations for care during office visits', *J Gen Intern Med*, 9 pp. 75-81.

Kravitz, R. L., Hays, R. D., Sherbourne, C. D., DiMatteo, R., Rogers, W. H., Ordway, L. & Greenfield, S. (1993) 'Recall of Recommendations and Adherence to Advice Among Patients with Chronic Medical Conditions', *Arch Int Med*, 153 pp. 1869-1878.

Krichbaum, K., Aarestad, V. & Buethe, M. (2003) 'Exploring the Connection between Self -Efficacy and Effective Diabetes Self-Management ', *The Diabetes Educator*, 29 (4), pp. 653-662.

Kristal, A. R., Andrilla, C. H., Koepsell, T. D., Diehr, P. H. & Cheadle, A. (1989) 'Dietary assessment instruments are susceptible to intervention associated response set bias ', *J Am Diet Assoc*, 98 pp. 40-43.

Kruger, J. & Dunning, D. (1999) 'Unskilled and unaware of it: how difficulties in recognizing one's own incompetence lead to inflated self-assessments', *J Pers Soc Psychol*, 77 pp. 1121-1134.

Krupt, E., Rosenkranz, S. L., Yeager, C. M., Barnard, K., Putnman, S. M. & Inui, T. S. (2000) 'The practice orientations of physicians and patients: the effect of doctor-patient congruence on satisfaction', *Patient Educ Couns*, 39 (1), pp. 49-59.

Krutz, S. M. S. (1990) 'Adherence to diabetes regimens: empirical status and clinical applications', *Diabetes Educator*, 16 pp. 50-59.

Kuijer, R. G. & De Ridder, D. T. D. (2003) 'Discrepancy in illness-realted goals and qualty of life in chronically ill patients: The role of self-efficacy', *Psychology and Health*, 18 (3), pp. 313-330.

Kunyk, D. & Olson, J. K. (2000) 'Clarification of conceptulizations of empathy', *J* Adv Nurs, 35 (3), pp. 317-325.

Lacey, K. & Pritchett, E. (2003) 'Nurtition care process and model; ADA adopts road map to quality care and outcomes management', *J Am Diet Assoc*, 103 pp.1061-1072.

Lally, P., Chipperfield, A. & Wardle, J. (2008) 'Healthy habits: efficacy of simple advice on weight control based on a habit-formation model', *Int J Obes (Lond)*, 32 (4), pp. 7000-7007.

Lang, F., Floyd, M. R. & Beine, K. L. (2000) 'Clues to Patients' Explanations and Concerns About Their Illnesses. A Call for Active Listening', *Arch Fam Med*, 9 pp. 222-227.

Larme, A. C. & Pugh, J. A. (1998) 'Attitudes of primary care providers toward diabetes: barriers to guide implementation', *Diabet Care*, 21 pp. 1391-1396.

Larson, E. B. & Yao, X. (2005) 'Clinical Empathy as Emotional Labor in the Patient-Physician Relationship', *JAMA*, 239 (9), pp. 1100-1106.

Latter, S., Maben, J., Myall, M. & Young, A. (2007) 'Perceptions and practice of concordance in nurses' prescribing consultations: Findings from a national questionnaire survey and case studies of practice in England', *International Journal of Nursing Studies*, 44 pp. 9-18.

Latter, S., Yerrell, P., Rycroft-Malone, J. & Shaw, D. (2002) 'Governance and health promotion: a case study of medication education', *Health Education Journal*, 59 pp. 253-266.

Lawrence, J. S. P. (2003) 'Communication and education skills for dietetic professionals ', *Journal of Human Nutrition & Dietetics*, 16 (6), pp. 469-470.

Lerman, I. (2005) 'Adherence to Treatment: The Key for Avoiding Long-Term Complications of Diabetes', *Archives of Medical Research*, 36 pp. 300-306.

Levinson, W., Gorawara-Bhat, R. & Lamb, J. (2000) 'A study of patient clues and physician responses in primary care and surgical settings', *Journal of the American Medical Association*, 284 (8), pp. 1021-1027.

Levinson, W. & Roter, D. (1993) 'The effects of two continuing medical education programs on communication skills of practising primary care physicians', *J Gen Intern Med*, 8 (6), pp. 318-324.

Levinson, W. & Roter, D. (1995) 'Physicians' psychosocial beliefs correlate with their patient communication skills', *J Gen Intern Med*, 10 pp. 375-379.

Levy, P. (2009) 'The current unmet need in type 2 diabetes mellitus: addressing glycaemia and cardiovascular disease', *Postgraduate Medicine*, 121 (3 Suppl 1), pp. 7-12.

Lewin, S. A., Skea, Z. C., Entwistle, V., Zwarenstein, M. & Dick, J. (2001) 'Interventions for providers to promote a patient -centred approach in clinical consultations', *The Cochrane Database of Systematic Reviews*, Issue 4. Art. No.: CD003267. DOI:10.1002/14651858.CD003267 Ley, P. (1982) 'Satisfaction, compliance and communication', *Br J Clin Psychol*, 21 (Pt 4), pp. 241-254.

Li, G., Zhang, P., Wang, J., Gregg, E. W., Yang, W., Gong, Q., Li, H., Li, H., Jiang, Y., An, Y., Shuai, Y., Zhang, B., Zhang, J., Thompson, T. J., Gerzoff, R. B., Roglic, G., Hu, Y. & Bennett, P. H. (2008) 'The long-term effect of lifestyle interventions to prevent diabetes in the China Da Qing Diabetes Prevention Study: a 20-year follow-up study', *Lancet*, 371 (9626), pp. 1783-1789.

Lichman, S. W., Pisarska, I., Berman, E. R., Pestone, M., Dowling, H., Offenbacher, E., Weisel, H., Heshka, S. E. & Heymsfield, S. B. (1992) 'Discrepancy between self-reported and actual caloric intake and exercise in obese subjects', *N Engl J Med*, 327 pp. 1893-1998.

Lin, E. H., Katon, W., Von Korff, M., Rutter, C., Simon, G. E., Oliver, M., Ciechanowski, P., Ludman, E. J., Bush, T. & Young, B. (2004) 'Relationship of depression and diabetes self-care, medication adherence, and preventive care', *Diabetes Care.*, 27 (9), pp. 2154-2160.

Lin, E. H., Rutter, C. M., Katon, W., Heckbert, S. R., Ciechanowski, P., Oliver, M. M., Ludman, E. J., Young, B. A., Williams, L. H., McCulloch, D. K. & Von Korff, M. (2010) 'Depression and advanced complications of diabetes: a prospective cohort study', *Diabetes Care*, 33 (2), pp. 264-269.

Little, P., Everitt, H., Willamson, I., Warner, G., Moore, M., Gould, C., Ferrier, K. & Payne, S. (2001a) 'Observational study of effect of patient centredness and positive approach on outcomes of general practice consultations', *BMJ*, 323 pp. 908-911.

Little, P., Everitt, H., Williamson, I., Warner, G., Moore, M., Gould, C., Ferrier, K. & Payne, S. (2001b) 'Preferences of patients for patient centred approach to consultation in primary care: observational study', *BMJ*, 322 (7284), pp. 468 - 472.

Loewe, R. & Freeman, J. (2000) ' Interpreting diabetes mellitus: differences between patient and provider models of disease and their implications for clinical practice', *Culture, Medicine & Psychiatry*, 24 (4), pp. 379-401.

Long, C. G., Simpson, C. M. & Allott, E. A. (1983) 'Psychological and dietetic counselling combined in the treatment of obesity: a compartive study in a hospital outpatient clinic', *Human Nutrition - Applied Nutrition*, 37 pp. 94-102.

Lorenz, R. A., Bubb, J., Davis, D., Jacobson, A. M., Jannasch, K., Kramer, J., Lipps, J. & Schlundt, D. (1996) 'Changing behaviour: Practical lessons from the Diabetes Control and Complications Trial', *Diabet Care*, 19 (6), pp. 648-652.

Lorenz, R. A., Pichert, J. W., Boswell, E. J. & Schlundt, D. G. (1989) 'Teaching skills training for health professionals: Effects on immediate recall by surrogate patients. ', *Teach Learn Med*, 1 pp. 26-30.

Lorig, K., Ritter, P. L., Villa, F. J. & Armas, J. (2009) 'Community-Based Peer-Led Diabetes Self-management: A Randomized Trial', *The Diabetes Educator*, 351 pp. 641-651. Lorig, K. R., Ritter, P., Stewart, A. L., Sobel, D. D., Drown, B., B.W, Jr., Bandura, A., Gonzalez, V. M., Laurent, D. D. & Holman, H. R. (2001) 'Chronic disease self management program: 2 year health status and health care utilization outcomes', *Medical Care*, 39 (11), pp. 1217-1223.

Lorig, K. R., Ritter, P. L. & Jacquez, A. (2005) 'Outcomes of border health Spanish/English chronic disease self-management programs', *Diabetes Educ*, 31 pp. 401-409.

Lu, A. H. & Dollahite, A. (2010) 'Assessment of dietitians' nutrition counselling self-efficacy and its positive relationship with reported skill usage', *J Hum Nutr Diet*, 23 pp. 144-153.

Lussier, M. & Richard, C. (2007) 'Refelcting back: Emapthic process', *Canadian Family Physician*, 53 pp. 827-829.

Lustman, P. J., Anderson, R. J., Freedland, K. E., de Groot, M., Carney, R. M. & Clouse, R. E. (2000) 'Depression and poor glycaemic control: a meta-analytic review of the literature', *Diabet Care*, 23 pp. 934-942.

Maclean, H. M. (1991) 'Patterns of diet related self-care in diabetes', *Soc Sci Med*, 32 (6), pp. 689-696.

MacLellan, D. & Berenbaum, S. (2003) 'Client Centred Nutrition Counselling: Do We Know What This Means?', *Canadian Journal of Dietetic Practice and Research*, 64 (1), pp. 12-15.

Maclellan, D. L. (2006) 'Dietitian's Opinions and Experiences of Client -Centred Nutrition Counselling', *Canadian Journal of Dietetic Practice and Research*, 67 pp. 1119-1124.

Madan, A. K. & Tichansky, D. S. (2005) 'Patients postoperatively forget aspects of preoperative patient education', *Obesity Surgery*, 15 (7), pp. 1066-1069.

Maddigan, S. L., Feeny, D. H. & Johnson, J. A. (2005) 'Health-related quality of life deficits associated with diabetes and comorbidities in a Canadian National Population Health Survey.', *Qual Life Res*, 14 (5), pp. 1311-1320.

Maddigan, S. L., Majumdar, S. R. & Johnson, J. A. (2005) 'Understanding the complex associations between patient-provider relationships, self-care behaviours, and health-related quality of life in type 2 diabetes: a structural equation modelling approach', *Qual Life Res*, 14 (6), pp. 1489-1500.

Maguire, P., Faulkner, A., Booth, K., Elliott, C. & Hillier, V. (1996) 'Helping cancer patients disclose their concerns', *Eur J Cancer*, 32A (1), pp. 78-81.

Maibach, E. & Murphy, D. A. (1995) 'Self-efficacy in health promotion research and practice: conceptulization and measurement', *Health Educ Res*, 101 pp. 37-50.

Makoul, G., Arntson, P. & Schofield, T. (1995) 'Health Promotion in Primary Care: Physician-Patient Communication and Decision Making about Prescription Medications ', *Social science & Medicine*, 41 (9), pp. 1241-1254. Maldonato, A., Bloise, D., Ceci, M., Fraticelli, E. & Fallucca, F. (1995) 'Diabetes mellitus: lessons form patient education', *Patient Education & Counseling*, 26 pp. 57-66.

Marilynn, S. A., Butler, P. M., Anderson, R. M., Funnell, M. M. & Feste, C. (1995) 'Guidelines for facilitating a client empowerment program', *The Diabetes Educator*, 21 pp. 308-312.

Martin, C. M., Banwell, C. L., Broom, D. H. & Nisa, M. (1999) 'Consultation length and chronic illness care in general practice: a qualitative study ', *Med J Aust*, 171 (2), pp. 77-81.

Marvel, M. K., Epstein, R. M., Flowers, K. & Beckman, H. B. (1999) 'Solicitating the patient's agenda: have we improved?', *J Am Med Assoc*, 281 pp. 283-287.

Matteucci, E. & Giampietro, O. (2003) 'Closing the gap between literature and practice: evaluation of a teaching programme (in the absence of a structured treatement) on both type 1 and type 2 diabetes', *Diabetes Nutr Metab*, 16 (5-6), pp. 298-305.

Mauffrey, C., Prempeh, E. M., John, J. & Vasario, G. (2008) 'The influence of written information during the consenting process on patients' recall of operative risks. A prospective randomised study', *International Orthapaedics*, 32 (4), pp. 425-429.

Mc Kinstry, B., Colthart, I. & Walker, J. (2006) 'Can doctors predict patients' satisfaction and enablement? A cross -sectional observational study', *Family Practice*, 23 pp. 240-245.

Mc Kinstry, B., Walker, J., Blaney, D., Heaney, D. & Begg, D. (2004) 'Do patients and expert doctors agree on the assessment of consultation skills?', *Family Practice*, 21 pp. 75-80.

Mc Nabb, W. L. (1997) 'Adherence in Diabetes: Can We Define It and Can We Measure It?', *Diabetes Care*, 20 (2), pp. 215-218.

McCann, S. & Weinman, J. (1996) 'Encouraging patient participation in general practice consultations: Effect on consultation length and content, patient satisfaction and health', *Psychology and Health*, 11 pp. 857-869.

McCaul, K. D., Glasgow, R. E. & Schafer, L. C. (1987) 'Diabetes regimen behaviours. Predicting adherence', *Med Care*, 25 (9), pp. 868-881.

McClusky, A. & Lovarini, M. (2005) ' Providing education on evidence-based practice improved knowledge but did not change behaviour: a before and after study.' *BMC Medical Education*, 5 (40), DOI: doi:10.1186/1472-6920-5-40 [Online]. Available at: http://www.biomedcentral.com/1472-6920/5/40 (Accessed: 10th August 2008).

McDonald-Miszczak, L., Neupert, S. D. & Gutman, G. (2005) 'Younger-old and older-old adults' recall of medication instructions', *Can J Aging*, 24 (4), pp. 409-417.

McKinstry, B. (2000) 'Do patients wish to be involved in decision making in the consultation? A cross sectional survey with video vignettes', *BMJ*, 321 pp. 867-871.

McWilliam, C. L., Brown, J. B., Carmichel, J. L. & Lehman, J. M. (1994) 'A new perspective on threatened autonomy in elderly persons: the disempowering process ', *Soc Sci Med*, 38 pp. 327-338.

Mead, N. & Bower, P. (2000a) 'Measuring patient-centredness: a comparison of three observation-based instruments', *Patient Education and Counseling*, 39 (10), pp. 71-80.

Mead, N. & Bower, P. (2000b) "Patient-centredness: a conceptual framework and review of the empirical literature", *Soc Sci Med*, 51 (7), pp. 78-81.

Mead, N., Bower, P. & Martin, R. (2008) 'Factors associated with enablement in general practice: cross-sectional study using routinely-collected data ', *British Journal of General Practice*, 58 (550), pp. 346-352.

Mercer, S. W., Neumann, M., Wirtz, M., Fitzpatrick, B. & Vojt, G. (2008) 'General practitioner empathy, patient enablement, and patient-reported outcomes in primary care in an area of high socio-economic deprivation in Scotland--a pilot prospective study using structural equation modeling', *Patient Educ Couns*, 73 (2), pp. 240-245.

Merkel, W. T. (1984) 'Physician perception of patient satisfaction. Do doctors know which patients are satisfied?', *Medical Care*, 22 pp. 453-459.

Michie, S., French, D., Allanson, A., Bobrow, M. & Marteau, T. M. (1997) 'Information recall in genetic counselling: a pilot study of its assessment', *Patient Education and Counseling*, 32 (1-2), pp. 93-100.

Michie, S., Miles, J. & Weinman, J. (2003) 'Patient -centredness in chronic illness: what is it and does it matter?', *Patient Education & Counseling*, 51 (3), pp. 197-206.

Miller, C. K., Edwards, L., Kissling, G. & Sanville, L. (2002) 'Evaluation of a theory-based nutrition intervention for older adults with diabetes mellitus', *Journal of the American Dietetic Association*, 102 (8), pp. 1069-1081.

Miller, W. R. & Rollnick, S. (2002) *Motivational interviewing: Preparing people to change addictive behaviour* New York: The Guildford Press.

Moran, J., Bekker, H. & Latchford, G. (2008) 'Everyday use of patient-centred, motivational techniques in routine consultations between doctors and patients with diabetes', *Patient Educ Couns*, 73 pp. 224-231.

Morris, A. D., Boyle, D. I., McMahon, A. D., Greene, S. A., MacDonald, T. M. & Newton, R. W. (1997) 'Adherence to insulin treatment, glycaemic control, and ketoacidosis in insulin-dependent diabetes mellitus. The DARTS/MEMO Collaboration. Diabetes Audit and Research in Tayside Scotland. Medicines Monitoring Unit.', *Lancet*, 350 (9090), pp. 1505-1510.

Morse, D. S., Edwardsen, E. A. & Gordon, H. S. (2008) 'Missed Opportunities for Interval Empathy in Lung Cancer Communication', *Arch Intern Med*, 168 (17), pp. 1853-1858.

Moser, A., van der Bruggen, H., Widdershoven, G. & Spreeuwenberg, C. (2008) 'Self-management of type 2 diabetes mellitus: a qualitative investigation from the perspective of participants in a nurse-led, shared-care programme in the Netherlands', *BMC Public Health*, 8 pp. 91.

Mulrow, C. D., Chiquette, E., Angel, L., Cornell, J., Summerbell, C., Anagnostelis, B., Grimm, R., Jr. & Brand, M. B. (2000) 'Dieting to reduce body weight for controlling hypertension in adults', *The Cochrane Database of Systematic Reviews*, Issue 2:CD000484

Murphy, E. & Kinmouth, A. L. (1995) 'No symptoms, no problems? Patients' understandings of non-insulin dependent diabetes', *Family Practice*, 12 pp. 184-192.

Naik, A. D., Michael A. Kallen, M. A., Walder, A. & Street, R. L. J. (2008) 'Improving Hypertension Control in Diabetes Mellitus. The Effects of Collaborative and Proactive Health Communication', *Circulation*, 117 pp. 1361-1368.

Najavits, L. M. & Weiss, R. D. (1994) 'Variations in therapist effectiveness in the treatment of patients with substance use disorders: an empirical review', *Addiction*, 89 (6), pp. 679-688.

Nakahara, R., Yoshuchi, K., Kumano, H., Hara, Y., Suematsu, H. & Kuboki, T. (2006) 'Prospective Study on Influence of Psychosocial Factors on Glycemic Control in Japanese Patients With Type 2 Diabetes', *Psychosomatics*, 47 (3), pp. 240-246.

Nathan, D. M., Cleary, P. A., Backlund, J. Y., Genuth, S. M., Lachin, J. M., Orchard, T. J., Raskin, P., Ziunman, B. & Diabetes Control and Complications Trial/Epidemiology of Diabetes Interventions and Complications (DCCT/EDIC) Study Research Group (2005) 'Intensive diabetes treatment and cardiovascular disease in patients with type 1 diabetes', *N Engl J Med*, 22 (353), pp. 2643-2653.

Nau, D. P., Aikens, J. E. & Pacholski, A. M. (2007) 'Effects of gender and depression on oral medication adherence in persons with type 2 diabetes mellitus', *Gend Med*, 4 (3), pp. 205-213.

Nestle, M., Wing, R., Birch, L., DiSogra, L., Drewnowski, A., Middleton, S., Sigman-Grant, M., Sobal, J., Winston, M. & Economos, C. (1998) 'Behavioral and social influences on food choice', *Nutr Rev*, 56 (5 Pt 2), pp. S50-64; discussion S64-74.

Neumann, M., Bensing, J., Mercer, S., Ernstmann, N., Ommena, O. & Pfaff, H. (2009) 'Analyzing the "nature" and "specific effectiveness" of clinical empathy: A theoretical overview and contribution towards a theory-based research agenda', *Patient Education and Counseling*, 74 pp. 339-346.

Neumann, M., Wirtz, M., Bollschweiler, E., Mercer, S. W., Warm, M., Wolf, J. & Pfaff, H. (2007) 'Determinants and patient-reported long-term outcomes of

physician empathy in oncology: A structural equation modelling approach', *Patient Education and Counseling*, 69 pp. 63-75.

Newman, S., Steed, L. & Mulligan, K. (2004) 'Self-management interventions for chronic illness', *Lancet*, 364 pp. 1523-1537.

NHS (2011) 'NHS Diabetes: Supporting, Improving, Caring'. [Online]. Available at: http://www.diabetes.nhs.uk/ (Accessed: 23/2/11).

NICE (2007) 'Public Health Programme Guidance No. 6, Behaviour change at population, community and individual levels.'. *NICE Guidelines,* NICE. [Online]. Available at: http://guidance.nice.org.uk/PH6/Guidance/pdf/English.

NICE (2008a) 'Type 2 Diabetes: National Clinical guidelines for management in primary and secondary care (update)'. *NICE Guidance; CG66*. [Online]. Available at: http://www.nice.org.uk/nicemedia/pdf/CG66NICEguidance.pdf.

NICE (2008b) 'NICE guidelines lipid modification and type 2 diabetes guidelines: CG67'. *NICE guidelines:* . [Online]. Available at: http://www.nice.org.uk/nicemedia/pdf/CG67fullguidelines.pdf

NICE (2009) 'Type 1 diabetes: diagnosis and management of type 1 diabetes in children, young people and adults '. *NICE Guidelines*. CG15. [Online]. Available at: http://guidance.nice.org.uk/CG15/NICEGuidance/pdf/English.

NICE (2003) 'Guidance on the use of patient education models for diabetes, Technology Appraisal 60'. NICE. [Online]. Available at: guidance.nice.org.uk/nicemedia/live/11496/32612/32612.pdf

Nicolai, J. & Demmel, R. (2007) 'The impact of gender stereotypes on the evaluation of general practitioners' communication skills: an experimental study using transcripts of physician-patient encounters.', *Patient Educ Couns*, 69 (1-3), pp. 200-205.

Norfolk, T., Birdi, K. & Walsh, D. (2007) 'The role of empathy in establishing rapport in the consultation: a new model', *Medical Education*, 41 pp. 690-697.

Norris, S. L., Engelgau, M. M. & Narayan, K. M. (2001) 'Effectiveness of selfmanagement training in type 2 diabetes: a systematic review of randomized controlled trials', *Diabetes Care*, 24 (3), pp. 561-587.

Norris, S. L., Lau, J., Smith, S. J., Schmid, C. H. & Engelgau, M. M. (2002) 'Selfmanagement education for adults with Type 2 diabetes: a meta-analysis of the effects on glycaemic control', *Diabetes Care*, 25 pp. 1159-1171.

Norris, S. L., Zhang, x., Avenell, A., Gregg, E., Schmid, C. H. & Lau, J. (2005) 'Long-term non-pharmacological weight loss interventions for adults with prediabetes', *The Cochrane Database of Systematic Reviews*, Issue 2. Art. No.:Cd005270.DOI:10.1002/14651858.CD005270

O'Brien, M. & Petrie, K. J. (1996) 'Examining patient participation in medical consultations: A combined quantitative and qualitative approach', *Psychology & Health*, 11 (6), pp. 871-890.

Ockene, J. K., Ockene, I. S., Quirk, M. E., Hebert, J. R., Saperia, G. M., Luippold, R. S., Merriam, P. A. & Ellis, S. (1995) 'Physician Training for PatientCentered Nutrition Counseling in a Lipid Intervention Trial', *Preventive Medicine*, 24 pp. 563-570.

Odegard, P. S. & Capoccia, K. (2007) 'Medication taking and diabetes: a systematic review of the literature', *Diabetes Educ*, 33 (6), pp. 1014-1029.

Odegard, P. S. & Gray, S. L. (2008) 'Barriers to medication adherence in poorly controlled diabetes mellitus.', *Diabetes Educ*, 34 (4), pp. 692-697.

Ogden, J., Ambrose, L., Khadra, A., Manthri, S., Symons, L., Vass, A. & Williams, M. (2002) 'A questionnaire study of GPs' and patients' beliefs about the different components of patient centredness', *Patient Educ Couns*, 47 pp. 223-227.

Ong, L. M., De Haes, J. C. J. M., Hoos, A. M. & Lammes, F. B. (1995) 'Doctorpatient communication: a review of the literature', *Soc Sci Med*, 3 pp. 157-174.

Ong, L. M., Visser, M. R., Lammes, F. B. & de Haes, J. C. (2000) ' Doctor– patient communication and cancer patients' quality of life and satisfaction', *Patient Educ Couns*, 41 pp. 145-156.

Orozco, L. J., Buchleitner, A. M., Gimenez-Perez, G., Rogué I Figuls, M., Richter, B. & Mauricio, D. (2008) 'Exercise or exercise and diet for preventing type 2 diabetes mellitus', *The Cochrane Database of Systematic Reviews*, Art. No.: CD003054. DOI: 10.1002/14651858.CD003054.pub3

Padgett, D., Mumford, E., Hynes, M. & Carter, R. (1988) 'Meta-analysis of the effects of educational and psychosocial interventions on management of diabetes mellitus', *Journal of Clinical Epidemiology*, 41 pp. 1007-10309.

Page, P., Verstraete, D. G., Robb, J. R. & Etzwiler, D. D. (1981) 'Patient recall of self-care recommendations in diabetes', *Diabetes Care*, 4 (1), pp. 96-98.

Pan, X. R, Li, G. W., Hu, Y.H., Wang, J. X., Yang, W. Y., An, Z. X., Hu, Z. X., Lin, J., Xiao, J. Z., Cao, H. B., Liu, P. A., Jiang, X. G., Jiang, Y. Y., Wang, J. P., Zheng, H., Zhang, H., Bennett, P. H., Howard, B. V. (1997) 'Effects of diet and exercise in preventing NIDDM in people with impaired glucose tolerance', *Diabetes Care*, 20 (4), pp. 537-544.

Parkin, T. (2003) 'Counselling in Diabetes'. in Frost, G., Dornhorst, A. and Moses, R. (eds.) *Nutritional Management of Diabetes Mellitus.* Chichester: John Wiley & Sons.

Parkin, T., Barnard, K., Cradock, S., Pettman, P. & Skinner, T. C. (2006) 'Does Professional Centred Training Improve Consultation Outcomes?', *Practical Diabetes International*, 23 (60), pp. 253-256.

Parkin, T. & Skinner, T. C. (2003) 'Discrepancies between patient and professionals recall of an outpatient consultation.', *Diabetic Medicine*, 20 (1), pp. 909-914.

Paterson, B. (2001) 'Myth of empowerment in chronic illness.', *J Adv Nurs*, 34 pp. 574-581.

Pearson, D. & Rapoport, L. (2007) 'Changing health behaviour '. in Briony Thomas (Editor), J.B.E. (ed.) *Manual of Dietetic Practice.* 4th edn. Chichester: Wiley.

Pedersen, R. (2009) 'Empirical research on empathy in medicine—A critical review', *Patient Education and Counseling*, 76 pp. 307-322.

Peel, E., Parry, O., Douglas, M. & Lawton, J. (2005) 'Taking the biscuit? A discursive approach to managing diet in type 2 diabetes', *J Health Psychol*, 10 (6), pp. 79-91.

Pesudovs, K., Luscombe, C. K. & Coster, D. J. (2006) 'Recall from informed consent counselling for cataract surgery', *Journal of Law and Medicine*, 13 (4), pp. 496-504.

Peyrot, M. & Rubin, R. R. (1994) 'Living with diabetes: the patient-centered perspective', *Diabetes Spectrum*,, 7 pp. 205-206.

Peyrot, M., Rubin, R. R., Lauritzen, T., Snoek, F. J., Matthews, D. R. & Skovlund, S. E. (2005) 'Psychosocial problems and barriers to improved diabetes management: results of the Cross-National Diabetes Attitudes, Wishes and Needs (DAWN) Study', *Diabet Med*, 22 (10), pp. 1379-1385.

Pignone, M. P., Ammerman, A., Fernandez, L., Orleans, C. T., Pender, N., Woolf, S., Lohr, K. N. & Sutton, S. (2003) 'Counseling to promote a healthy diet in adults: a summary of the evidence for the U.S. Preventive Services Task Force', *Am J Prev Med*, 24 (1), pp. 75-92.

Pill, R., Rees, M. E., Stott, N. C. & Rollnick, S. R. (1999) 'Can nurses learn to let go? issues arising from an intervention designed to improve patient's involvement in their own care', *Journal of Advanced Nursing*, 29 (6), pp. 1492-1499.

Pill, R., Stoot, N. C. H., Rollnick, S. R. & Rees, M. (1998) 'A Randomised controlled trial of an intervention designed to improve the care given in general practice to type II diabetic patients: patient outcomes and professional ability to change behaviour', *Family Practice*, 15 (3), pp. 229-235.

Polonsky, W. H. (1996) 'Understanding and treating patients with diabetes burnout'. *Practical applications for diabetes clinicians* Alexandria, Va American Diabetes Assocation, pp 183-192.

Polonsky, W. H., Fisher, L., Guzman, S., Sieber, W. J., Philis-Tsimikas, A. & Edelman, S. V. (2010) 'Are patients' initial experiences at the diagnosis of type 2 diabetes associated with attitudes and self-management over time?', *Diabetes Educ*, 36 (5), pp. 828-834.

Poskiparta, M., Kettunen, T. & Liimatainen, L. (1998) 'Reflective questions in health counselling', *Q Health Res*, 8 pp. 682-693.

Poskiparta, M., Kettunen, T. & Liimataninen, L. (2000) 'Questioning and advising in health counseling. Results from a study of Finish nurse counselors', *Health Edcuation Journal*, 95 pp. 47-67.

Poskiparta, M., Liimataninen, L., Kettunen, T. & Karhila, P. (2001) 'From nursecentred health counseling to empowermental health counseling', *Patient Educ Couns*, 95 pp. 69-79.

Povey, R., Conner, M., Sparks, P., James, R. & Shepherd, R. (1999) 'A critical examination of the application of the Transtheoretical Model's stages of change to dietary behaviours', *Health Education Research*, 14 (5), pp. 641-651.

Powell, C. K., Hill, E. G. & Clancy, D. E. (2007) 'The relationship between health literacy and diabetes knowledge and readiness to take health actions', *Diabetes Educ*, 33 (1), pp. 144-151.

Ram, P., Grol, R., Rethans, J. J., Schouten, B., van der Vleuten, C. & Kester, A. (1999) 'Assessment of general practitioners by video observation of communicative and medical performance in daily practice: issues of validity, reliability and feasibility', *Med Educ*, 33 (6), pp. 447-454.

Rapoport, L. (1998) 'Integrating cognitive behavioural therapy into dietetic practice: a challenge for dietitians', *Journal of Human Nutrition & Dietetics*, 11 (3), pp. 227-237.

Rapoport, L. & Perry, K. N. (2000) 'Do dietitians feel that they have had adequate training in behaviour change methods?', *Jounal of Human Nutrition & Dietetics*, 13 (4), pp. 287-298.

Reach, G. (2005) 'Role of habit in adherence to medical treatment', *Diabetic Medicine*, 22 pp. 415-420.

Realini, T., Kalet, A. & Sparling, J. (1995) 'Interruption in the medical interaction', *Arch Fam Med*, 4 pp. 1028-1033.

Renders, C. M., Valk, G. D., Griffin, S., Wagner, E. H., Eijk, J. T. v. & Assendelft, W. J. J. (2000) 'Interventions to improve the management of diabetes mellitus in primary care, outpatient and community settings.', *The Cochrane Database of Systematic Reviews*, (Issue 4), pp. Art. No.: CD001481. DOI: 001410.001002/14651858.CD14001481

Resnicow, K., Jackson, A., Wang, T., De, A. K., McCarty, F., Dudley, W. N. & Baranowski, T. (2001) 'A Motivational Interviewing Intervention to Increase Fruit and Vegetable Intake Through Black Churches: Results of the Eat for Life Trial', *Am J Public Health*, 91 (10), pp. 1686-1693.

Reynolds, W. J. & Scott, B. (1999) 'Empathy: a crucial component of the helping relationship', *J Psychiatr Ment Health Nurs*, 6 (5), pp. 363-370.

Roach, R. R., Pichert, J. W., Stetson, B. A., Lorenz, R. A., Boswell, E. J. & Schlundt, D. G. (1992) 'Improving dietitians' teaching skills', *J Am Diet Assoc*, 92 (12), pp. 1466-1470.

Rogers, C. R. (1957) 'The necessary and sufficient conditions of theraputic personality change ', *J Consult CLin PSychol*, 21 pp. 95-103.

Rogers, C. R. (1980) A Way of Being. Boston: Houghton Mittlin Company.

Rollnick, S. (1996) 'Behaviour change in practice: targeting individuals', *International Journal of Obesity*, 20 (Supp 1), pp. S22-S26.

Rollnick, S., Mason, P. & Butler, C. (1999) *Health Behaviour Change: A guide for practitioners.* . Edinburgh: Churchill Livingstone.

Roper, N. A., Bilous, R. W., Kelly, W. F., Unwin, N. C. & Connolly, V. M. (2001) 'Excess mortality in a population with diabetes and the impact of material deprivation: longitudinal, population based study', *BMJ*, 322 (7299), pp. 1389-1393.

Rosal, M. C., Ebbeling, C. B., Lofgren, I., Ockene, I. S. & Hebert, J. R. (2001) 'Facilitating dietary change: the patient-centred counseling model', *Journal of the American Dietetic Association*, 101 (3), pp. 332-341.

Rosenstock, I. M. (1985) 'Understanding and enhancing patient compliance with diabetic regimens', *Diabet Care*, 8 pp. 610-616.

Ross, F. M., Bower, P. J. & Ibbald, B. S. (1994) 'Practice nurses: characteristics, workload and training needs', *Br J Gen Pract*, 44 (37B), pp. 15-18.

Rost, K. (1989) 'The influence of patient participation on satisfaction and compliance', *Diabetes Educ*, 15 pp. 134-138.

Rost, K. & Roter, D. L. (1987) 'Predictors of Recall of medication regimens and recommendations for lifestyle change in elderly patients', *Gerontologist*, 27 (4), pp. 310-315.

Rost, K., Roter, D. L., Quill, T. & Bertakis, K. (1990) 'Capacity to remember prescription drug changes: deficits associated with diabetes. Collaborative Study Group of Task Force on Medical Interview', *Diabetes Research and Clinical Practice*, 10 (2), pp. 183-187.

Roter, D. (2000) 'The enduring and evolving nature of the patient-physician relationship', *Patient Educ Couns*, 39 pp. 5-15.

Roter, D. L. (2003) 'Observations on Methodological and measurement challenges in the assessment of communication during medical exchanges', *Patient Education & Counseling*, 50 pp. 17-21.

Roter, D. L. (2005) 'RIAS: Manual'. *Roter Interaction Analysis System,* Roter, D.L. [Online]. Available at: httpp://www.rias.org/manual (Accessed: 12th June 2005).

Roter, D. L. (2009) 'RIAS articles'. [Online]. Available at: http://www.rias.org/articles (Accessed: 6/5/06).

Roter, D. L., Hall, J. A. & Aoki, Y. (2002) 'Physician Gender Effects in Medical Communication: A Meta-analytic Review', *JAMA.*, 288 (6), pp. 756-764.

Roter, D. L., Hall, J. A. & Katz, N. R. (1987) 'Relations between physicians' behaviors and analogue patients' satisfaction, recall, and impressions.', *Med Care*, 25 (5), pp. 437-451.

Roter, D. L., Larson, S. M., Beach, M. C. & Cooper, L. A. (2008) 'Interactive and evaluative correlates of dialogue sequence: A simulation study applying the RIAS to turn taking structures', *Patient Education and Counseling*, 71 pp. 26-33.

Roter, D. L., Stewart, A. L., Putnam, S. M., Lipkin, M. J., Stiles, W. & Inui, T. S. (1997) 'Communication patterns of primary care physicians', *JAMA*, 277 (4), pp. 350-356.

Rowland-Morin, P. A. & Carroll, J. G. (1990) 'Verbal Communication Skills and Patient Satisfaction: A Study of Doctor-Patient Interviews', *Evaluation & the Health Professions*, 13 (2), pp. 168-185.

Rubak, S., Sandbaek, A., Lauritzen, T. & Christensen, B. (2005) 'Motivational interviewing: a systematic review and meta-analysis', *Br J Gen Pract*, 55 (513), pp. 305-312.

Rubin, R. R. (2005) 'Adherence to pharmalogical therapy in patients with type 2 diabetes mellitus', *Am J Med*, 118 (Suppl 5A), pp. 27S-34S.

Rubin, R. R., Anderson, R. M. & Funnell, M. M. (2002) 'Collaborative diabetes care', *Pract Diabetol*, 21 pp. 29-32.

Rubin, R. R., Peyrot, M. & Saudeck, C. D. (1993) 'The Effcet of a Diabetes Eductaion Program Incorporating Coping Skills and Emotional Well-Being and Diabetes Self-Efficacy', *The Diabetes Educator*, 19 (3`), pp. 210-214.

Rubin, R. R., Peyrot, M. & Saudek, C. D. (1989) 'The effect of a diabetes education program incorporating coping skills training on emotional well-being and diabetes self-eficacy', *Diabetes Educator*, 19 pp. 210-214.

Rubin, R. R., Peyrot, M. & Saudek, C. D. (1991) 'Differential effect of diabetes education on self-regulation and lifestyle beahviours', *Diabet Care*, 14 pp. 335-338.

Rustveld, L. O., Pavlik, V. N., Jibaja-Weiss, M. L., Kline, K. N., Gossey, J. T. & Volk, R. J. (2009) 'Adherence to diabetes self-care behaviors in English- and Spanish-speaking Hispanic men', *Patient Preference and Adherence*, 3 pp. 123-130.

Ryan, G. W. & Bernard, H. R. (2003) 'Techniques to identify themes ', *Field Methods*, 15 pp. 85-109.

Sacco, W. P., Wells, K. J., Friedman, A., Matthew, R., Perez, S. & Vaughan, C. A. (2007) 'Adherence, body mass index, and depression in adults with type 2 diabetes: the mediational role of diabetes symptoms and self-efficacy', *Health Psychology*, 26 (6), pp. 693-700.

Sacks, F. M., Svetkey, L. P., Vollmer, W. M. & et al (2001) 'Effects on blood pressure of reduced dietary sodium and the dietary approaches to stop hypertension (DASH) diet. DASH-sodium collaborative research group', *N Engl J Med*, 344 pp. 3-10.

Salmela, S., Poskiparta, M., Kasila, K., Vähäsarja, K. & Vanhala, M. (2009) 'Transtheoretical model-based dietary interventions in primary care: a review of the evidence in diabetes', *Health Education Research*, 24 (2), pp. 237-252. Salmon, P., Dowrick, C. F., Ring, A. & Humphris, G. M. (2004) 'Voiced but unheard agendas: qualitative analysis of the psychosocial cues that patients with unexplained symptoms present to general practitioners', *Br J Gen Pract.*, 54 (500), pp. 171-176.

Sandvik, M., Eide, H., Lind, M., Graugaard, P. K., Torper, J. & Finset, A. (2002) 'Analysing medical dialogues: strength and weakness of Roter's interaction analysis system (RIAS)', *Patient Education & Counseling*, 46 (4), pp. 235-241.

Satia, J. A., Kristal, A. R., Curry, S. J. & Trudeau, E. (2001) 'Motivations for healthful dietary change', *Public Health Nutrition*, 4 (5), pp. 953-959.

Schmitt, K. E., Edie, C. F., Laflam, P., Simbartl, L. A. & Thakar, C. V. (2010) 'Adherence to antihypertensive agents and blood pressure control in chronic kidney disease', *Am J Nephrol*, 32 (6), pp. 541-548.

Schnoll, R. & Zimmerman, B. J. (2001) 'Self-regulation training enhances dietary self-efficacy and dietary fiber consumption', *J Am Diet Assoc*, 101 pp. 1006-1011.

Schön, D. (1983) *The reflective Practitioner. How professionals think in action,.* London: Temple Smith.

Schrimer, J. M., Mauksch, L., Lang, F., Marvel, K., Zoppi, K., Epstein, R., M., Brock, D. & Pryzbylski, M. (2005) 'Assessing Communication Competence: A Review of Current Tools', *Family Medicine*, 37 (3), pp. 184-192.

Schüz, B., Sniehotta, F. F., Wiedemann, A. & Seemann, R. (2006) 'Adherence to a daily flossing regimen in university students: effects of planning when, where, how and what to do in the face of barriers ', *Journal of Clinical Periodontology*, 33 pp. 612-619.

Scmid Mast, M., Hall, J. A. & Roter, D. L. (2007) 'Disentangling physician sex and physician communciation style their effects on patient satisfaction in a virtual medical visit', *Patient Education & Counseling*, 68 (1), pp. 16-22.

Searle, A., Norman, P., Thompson, R. & Vedhara, K. (2007) 'Illness representations among patients with type 2 diabetes and their partners: relationships with self-management behaviors', *Journal of Psychosomatic Research*, 63 (2), pp. 175-184.

Selvais, P. L., Amoussou-Guenou, K. D. & Hermans, M. P. (2008) 'Belonging to a diabetes patients' association is predictive of better metabolic control', *Diabetes Metab*, 34 (3), pp. 279-282.

Senécal, C., Nouwen, A. & White, D. (2000) 'Motivation and dietary self-care in adults with diabetes: Are self-efficacy and autonomous self-regulation complementary or competing constructs?', *Health Psychology*, 19 (5), pp. 452-457.

Shah, J. H., Murata, G. H., Duckworth, W. C., Hoffman, R. M. & Wendel, C. S. (2003) 'Factors Affecting Compliance in Type 2 Diabetic Patients: Experience from the Diabetes Outcomes in Veterans Study (DOVES)', *Int J Diab Dev Ctries*, 23 pp. 75-82.

Shaw, B., Cheater, F., Baker, R., Gillies, C. L., hearnshaw, H., Flottorp, S. & Robertson, N. (2005) 'Tailored interventions to overcome identified barriers to change: effects on professional practice and health care outcomes', *The Cochrane Database of Systematic Reviews*, Issue 3. Art.No.: CD005470. DOI: 10.1002/14651858.CD005470

Sherman, A. M., Bowen, D. J., Vitolins, M., Perri, M. G., Rosal, M. C., Sevick, M. A. & Ockene, J. K. (2000) 'Dietary Adherence: Characterstics and Interventions', *Control Clin Trials*, 21 pp. 206S-211S.

Shigaki, C., Kruse, R. L., Mehr, D., Sheldon, K. M., Bin, G. e., Moore, C. & Lemaster, J. (2010) 'Motivation and diabetes self-management', *Chronic Illn*, 6 (3), pp. 202-214.

Shillitoe, R. (1994) *Counselling People with Diabetes.* ed. Davis, H., Communication and Counselling in Health Care Series. Leicester: British Psychological Society.

Shultz, J. A., Sprague, M. A., Branen, L. J. & Lambeth, S. (2001) 'A comparison of views of individuals with type 2 diabetes mellitus and diabetes educators about barriers to diet and exercise', *J Health Commun*, 6 (2), pp. 99-115.

Silverman, D. (1985) Qualitative methodology and Sociology. Aldershot: Gower.

Silverman, D. (2006) Interpreting Qualitative Data. 3rd edn. Sage publications.

Siminoff, L. A., Graham, G. C. & Gordon, N. M. (2006) 'Cancer communication patterns and influence of patient characteristics: disparities in information giving and affective behaviours', *Patient Education & Counseling*, 62 (3), pp. 355-360.

Simmons, D. (1992) 'Diabetes self help facilitated by local diabetes research: the Coventry Asian Diabetes Support Group', *Diabet Med*, 9 pp. 866-869.

Simmons, D., Lillis, S., Swan, J. & Haar, J. (2007) 'Discordance in perceptions of barriers to diabetes care between patients and primary care and secondary care', *Diabetes Care*, 30 (3), pp. 490-495.

Simpson, M., Buckman, R., Stewart, M., Maguire, P., Lipkin, M., Novack, D. & Till, J. (1991) 'Doctor-patient communication: the Toronto Consensus Statement ', *BMJ*, 303 pp. 1385-1387.

Sixma, H. J., Spreeuwenberg, P. M. & van der Pasch, M. A. (1998) 'Patient satisfaction with the general practitioner: a two-level analysis', *Med Care*, 36 (2), pp. 212-229.

Skinner, T. C. (2004) 'Psychological barriers', *European Journal of Endocrinology*, 151 pp. T13-T17.

Skinner, T. C., Barnard, K., Cradock, S. & Parkin, T. (2007) 'Patient and professional accuracy of recalled treatment decisions in out-patient consultations', *Diabetic Medicine*, 24 pp. 557-560.

Skinner, T. C., Carey, M. E., Cradock, S., Dallosso, H. M., Daley, H., Davies, M. J., Doherty, Y., Heller, S., Khunti, K. & Oliver, L., DESMOND Collaborative, (2008) "Educator talk' and patient change: some insights from the DESMOND

(Diabetes Education and Self Management for Ongoing and Newly Diagnosed) randomized controlled trial', *Diabetic Medicine*, 25 (9), pp. 1117-1120.

Skinner, T. C., Craddock, S., Arundel, F. & Graham, W. (2003a) 'Four Theories and a Philosophy: Self-Management Education for Individuals Newly Diagnosed With Type 2 Diabetes', *Diabetes Spectrum*, 16 (2), pp. 75-80.

Skinner, T. C. & Cradock, S. (2000) 'Empowerment: What About the Evidence?', *Practical Diabetes International*, 17 (3), pp. 91-95.

Skinner, T. C., Mehrshahi, R., Gregory, R. & Jackson, S. (2003b) 'Who remembers what happened in the consultation? Comparing patient and doctor recall with the video evidence', *Diabetologia*, 46 (Suppl 2), pp. 46.

Skinner, T. C., Tringham, J., Khunit, K., Jarvis, J., Farooqui, A. & Raymond, N. (2003c) 'Illness beliefs of individuals participating in the STAR Diabetes Screening study', *Diabetic Medicine*, 20 (Suppl 2), pp. 24.

Smith, P., Masterson, A. & Smith, S. L. (1999) 'Health promotion versus disease and care: failure to establish "blissful clarity" in British nurse education and practice', *Soc Sci Med*, 48 pp. 227-239.

Sousa, V. D., Zauszniewski, J. A., Musil, C. M., McDonald, P. E. & Milligan, S. E. (2004) 'Testing a conceptual framework for diabetes self-care management', *Res Theory Nurs Pract*, 18 (4), pp. 293-316.

Spahn, J. M., Reeves, R. S., Keim, K. S., Laquatra, I., Kellogg, M., Jortberg, B. & Clark, N. A. (2010) 'State of the evidence regarding behavior change theories and strategies in nutrition counseling to facilitate health and food behavior change', *J Am Diet Assoc*, 110 (6), pp. 879-891.

Spiro, H. (1992) 'What is empathy and can it be taught?', *Ann Intern Med.*, 116 (10), pp. 843-846.

Spraggins, E. F., Fox, E. A. & Carey, J. C. (1990) 'Empathy in clinical dietitians and dietetic interns', *J Am Diet Assoc.*, 90 (2), pp. 244-249.

Stadtlander, L. M. & Coyne, A. C. (1990) 'The effect of goal-setting and feedback on age differences in secondary memory', *Experimental Aging Research*, 16 pp. 91-94.

Stamler, J., Vaccaro, O., Neaton, J. D. & Wentworth, D. (1993) 'Diabetes, other risk factors and 12-yr cardiovascular mortality for men screened in the Multiple Risk Factor Intervention Trial', *Diabetes Care.*, 16 pp. 434-444.

Stanford Patient Education Research Centre (2009) 'Diabetes self-efficacy scale'. Patient Education Research Centre', Stanford University. [Online]. Available at: http://patienteducation.stanford.edu/research/sediabetes.html (Accessed: 9/11/09).

Steel, J. M., Young, R. J., Lloyd, G. G. & Macintyre, C. C. (1989) 'Abnormal eating attitudes in young insulin -dependent diabetics', *Br J Psychiatry*, 155 (51), pp. 515-521.

Stephens, M. A., Rook, K. S., Franks, M. M., Khan, C. & lida, M. (2010) 'Spouses use of social control to improve diabetic patients' dietary adherence', *Fam Syst Health*, 28 (3), pp. 199-208.

Stepien, K. A. & Baernstein, A. (2006) 'Educating for Empathy', *J Gen Intern Med*, 21 pp. 524-530.

Stetson, B. A., Pichert, J. W., Roach, R. R., Lorenz, R. A., Boswell, E. J. & Schlundt, D. G. (1992) 'Registered Dietitians' Teaching and Adherence Promotion Skills During Routine Patient Education', *Patient Education & Counseling*, 19 pp. 273-280.

Stewart, M. (2001) 'Towards a global definition of patient centred care', *BMJ*, 322 pp. 444 - 445.

Stewart, M., Brown, J. B., Donner, A., McWhinney, I. R., Oates, J. & Weston, W. W., et al. (2000) 'The impact of patient-centred care on outcomes', *Journal of Family Practice*, 49 pp. 796-804.

Stewart, M. A. (1984) 'What is a successful doctor patient interview? a study of interactions and outcomes', *Soc Sci Med*, 19 (2), pp. 167-175.

Stewart, M. A. (1995) 'Effective physician-patient communication and health outcomes; A review', *Canadian Medical Association Journal*, 152 (9), pp. 1423-1433.

Stewart, R. & Liolitsa, D. (1999) 'Type 2 diabetes mellitus, cognitive impairment and dementia', *Diabet Med*, 16 pp. 93-112.

Stiles, W. B., Putnam, S. M., Wolf, M. H. & James, S. A. (1979) 'Interaction exchange structure and patient satisfaction with medical interviews', *Med Care*, 17 (6), pp. 667-681.

Street, J., R. L. (1992a) 'Analyzing communication in medical consultations. Do behavioral measures correspond to patients' perceptions?', *Med Care*, 30 pp. 976-988.

Street, J., R. L., Gordon, H. S., Ward, M. M., Krupt, E. & Kravitz, R. L. (2005) 'Patient participation in medical consultations: why some patients are more involved than others', *Med Care*, 42 pp. 960-969.

Street, R. L. (1991) 'Information giving in medical consultations the influence of patients communicative styles and personal characteristics', *Soc Sci Med*, 32 pp. 421-548.

Street, R. L. (1992b) 'Communicative styles and adaptations in physician-parent consultations', *Soc Sci Med*, 34 pp. 1155-1163.

Street, R. L., Jr,, Makoul, G., Arora, N. K. & Epstein, R. M. (2009) 'How does communication heal? Pathways linking clinician–patient communication to health outcomes', *Patient Educ Couns*, 74 pp. 295-301.

Street, R. L., Jr,, Piziak, V. K., Carpentier, W. S., Herzog, J., Skinner, G. & McLellan, L. (1993) 'Provider-Patient Communication and Metabolic Control', *Diabetes Care*, 16 (5), pp. 714-721.

Street, R. L. J. & Millay, B. (2001) 'Analyzing patient participation in medical encounters', *Health Communication*, 13 pp. 61-73.

Suchman, A. L., Markakis, K., Beckman, H. B. & Frankel, R. (1997) 'A model of empathic communication in the medical interview', *JAMA*, 277 pp. 678-682.

Sullivan, E. D. & Joseph, D. H. (1998) 'Struggling with behaviour changes: A special case for clients with diabetes.', *Diabetes Educator*, 24 pp. 72-77.

Swift, B. (1996) 'Data Analysis: Preparing Numerical Data'. in Sapsford, R. and Jupp, V. (eds.) *Data Collection and Analysis.* London: Sage, pp 162-179.

Takii, M., Uchigata, Y., Tokunaga, S., Amemiya, N., Kinukawa, N., Nozaki, T., Iwamoto, Y. & Kubo, C. (2008) 'The duration of severe insulin omission is the factor most closely associated with the microvascular complications of Type 1 diabetic females with clinical eating disorders', *Int J Eat Disord*, 41 (3), pp. 259-264.

Tang, T. S., Brown, M. B., Funnell, M. M. & Anderson, R. M. (2008) 'Social Support, Quality of Life, and Self-Care behaviours Among African Americans With Type 2 Diabetes', *The Diabetes Educator*, 34 (2), pp. 266-276.

Tashakkori, A. & Teddlie, C. (1998) *Mixed methodology; Combining qualitative and quantitative approaches.* Thousand Oaks, CA: Sage.

Teddlie, C. & Tashakkori, A. (2003) 'Major issues and controversies in the use of mixed methods in the social and behavioural sciences'. in Teddlie C and Tashakkor A (ed.) *Handbook book of mixed methods in social & behavioral research.* Thousand Oaks, CA: Sage.

The DAWN (2002) '(Diabetes Attitudes, Wishes and Needs) Study', *Practical Diabetes International*, 19 pp. 22-24.

Themessl-Huber, M., Humphris, G., Dowell, J., Macgillivray, S., Rushmer, R. & Brian Williams, B. (2008) 'Audio-visual recording of patient–GP consultations for research purposes: A literature review on recruiting rates and strategies', *Patient Education and Counseling*, 71 pp. 157-168.

Theunissen, N. C., de Ridder, D. T., Bensing, J. M. & Rutten, G. E. (2003) 'Manipulation of patient -provider interaction: discussing illness representations or action plans concerning adherence', *Patient Educ Couns*, 51 (3), pp. 247-258.

Thomas, D. & Elliott, E. J. (2009) ' Low glycaemic index, or low glycaemic load, diets for diabetes mellitus.', *The Cochrane Database of Systematic Reviews*, Issue . Art. No.: DOI:10.1002/14651858

Thorn, D. H. (2000) 'Training physicians to increase patient trust ', *j Eval Clin Pract*, 6 pp. 245-253.

Thorne, S., Paterson, B. & Russell, C. (2003) 'The Structure of Everyday Selfcare Decision Making in Chronic Illness', *Qualitative Health Research*, 13 (10), pp. 1337-1352. Thorne, S. E., Harris, S. R., Mahoney, K., Con, A. & McGuinness, L. (2004) 'The context of health care communication in chronic illness', *Patient Educ Couns*, 54 (3), pp. 299-306.

Thorne, S. E. & Paterson, B. L. (2001) 'Health care professional support for selfcare management in chronic illness: insights from diabetes research', *Patient Educ Couns*, 42 pp. 81-90.

Toobert, D. J. & Glasgow, R. E. (1991) 'Problem solving and diabetes self-care', *J Behav Med*, 14 pp. 71-86.

Toobert, D. J., Hampson, S. E. & Glasgow, R. E. (2000) 'The summary of diabetes self-care activities measures: results from 7 studies and a revised scale', *Diabetes Care*, 23 pp. 943-950.

Travis, T. (1997) 'Patient Perceptions of Factors That Affect Adherence to Dietary Regimens for Diabetes Mellitus', *The Diabetes Educator*, 23 (2), pp. 152-156.

Trudeau, E. & Dubé, L. (1995) 'Moderators and determinants of satisfaction with diet cousneling for patients consuming a theraputic diet ', *J Am Diet Assoc*, 95 pp. 34-39.

Tuckett, A. G. (2005) 'Applying thematic analysis theory to practice: a researcher's experience', *Contemp Nurse*, 19 (1-2), pp. 75-87.

Tuomilehto, J., Lindstrom, J., Eriksson, J., Valle, T. T., Hamalainen, H., Ilanne-Parikka, P. & et al for the Finish Diabetes Prevention Study Group (2002) 'Prevention of type 2 diabetes mellitus by change in lifestyle among subjects with impaired glucose tolerance', *N Engl J Med*, 344 pp. 1343-1350.

UK Prospective Diabetes Study (UKPDS) Group (1998) 'Intensive blood glucose control with sulphonylureas or insulin compared with conventional treatment and risk of complications in patients with type 2 diabetes. UKPDS 33', *The Lancet.*, 352 pp. 837-853.

University of Rochester (2008) 'Self-Determination Theory: An Approach to Human Motivation and Personality', [Online]. Available at: http://www.psych.rochester.edu/SDT/measures/pashealth.php (Accessed: 8/12/09).

Uusitupa, M., Louheranta, A., Lindstrom, J., Valle, T., Sundvall, J., Eriksson, J. & Tuomilehto, J. (2000) 'The Finish Diabetes Prevention Study', *Br J Nutr*, 83 (Suppl 1), pp. S137-142.

van Beurden, E., James, R., Christian, J. & Church, D. (1991) 'Dietary selfefficacy in a community-based intervention: implications for effective dietary counseling', *Austr J Nutr Diet*, 48 (2), pp. 64-67.

Van Dulmen, A. M., Verhaak, P. F. M., Henk, J. G. & Bilo. (1997) 'Shifts in doctor-patient communication during a series of outpatient consultations in non-insulin dependent diabetes mellitus', *Patient Education and Counseling*, 30 (3), pp. 227-237.

Vickery, C. E. & Hodges, P. A. (1986) 'Counseling strategies for dietary management', *Journal of the American Dietetic Association*, 86 (7), pp. 924-928.

Vijan, S., Stuart, N. S., Fitzgerald, J. T., Ronis, D. L., Hayward, R. A., Slater, S. & Hofer, T. P. (2005) 'Barriers to following dietary recommendations in Type 2 diabetes', *Diabetic Medicine*, 22 (1), pp. 32-38.

Wansink, B. & Sobal, J. (2007) 'Mindless eating: the 200 daily food decisions we overlook', *Environ Behav*, 39 pp. 106-123.

Ward, J. & Sanson-Fisher, R. (1996) 'Accuracy of patient recall of opportunistic smoking cessation advice in general practice', *Tobacco Control*, 5 (2), pp. 110-113.

Watkins, K. W., Connell, C. M., Fitzgerald, J. T., Klem, L., Hickey, T. & Ingersoll-Dayton, B. (2000) 'Effetc of Adults' Self-Regulation of Diabetes on Quality-of-Life Outcomes', *Diabet Care*, 23 pp. 1511-1515.

Watson, P. W. B. & McKinstry, B. (2009) 'A systematic review of interventions to improve recall of medical advice in healthcare consultations', *J R Soc Med*, 102 pp. 235-243.

Webb, K. L., Dobson, A. J., O'Connell, D. L., Tupling, H. E., Harris, G. W., Moxon, J. A., Sulway, M. J. & Leeder, S. R. (1984) 'Dietary compliance among insulin-dependent diabetics', *J Chronic Dis*, 37 (8), pp. 633-643.

Webb, T. L. & Sheeran, P. (2008) 'Mechanisms of implementation intention effects: The role of goal intentions, self-efficacy, and accessibility of plan components', *Br J Soc Psychol*, 47 pp. 373-395.

Weston, W. W., Brown, J. B. & Stewart, M. A. (1989) 'Patient-Centred Interviewing Part I: Understanding Patient's Experiences', *Canadian Family Physician*, 35 pp. 147-151.

Wetzels, R., Harmsen, M., Van Weel, C., Grol, R. & Wensing, M. (2007) 'Interventions for improving older patients' involvement in primary care episodes', *Cochrane Database of Systematic Reviews*, Issue 1: Art No.:CD004273.DOI:10.1002/14651858.CD004273.pub2,

Whitehead, K., Langley-Evans, S. C., Tischler, V. & Swift, J. A. (2009) 'Communication skills for behaviour change in dietetic consultations', *J Hum Nutr Diet*, 22 pp. 493-500.

Whiteley, L., Padmanabhan, S., Hole, D. & Isles, C. (2005) 'Should diabetes be considered a coronary heart disease risk equivalent?: results from 25 years of follow-up in the Renfrew and Paisley survey', *Diabetes Care*, 28 (7), pp. 1588-1593.

Wikblad, K. (1991) 'Patients' perspective of diabetes care and education ', *J Adv Nurs*, 16 pp. 837-844.

Willems, S., De Maesschalck, S., Deveugele, M., Derese, A. & DE Maeseneer, J. (2005) 'Socio-economic status of the patient and doctor-patient communication: does it make a difference?', *Patient Educ Couns*, 56 pp. 139-146

Williams, G. & Pickup, J. C. (2004) *Handbook of Diabetes.* 3rd edn. Massachusetts: Blackwell Publishing.

Williams, G. C., Cox, E. M., Kouides, R. & Deci, E. L. (1999) 'Presenting the facts about smoking to adolescents: The effects of an autonomy supportive style ', *Archives of Pediatrics and Adolescent Medicine*, 153 pp. 959-964.

Williams, G. C., Frankel, R. C., ampbell, T. L. & Deci, E. L. (2000) 'Research on relationship-centered care and health-care_ outcomes from the Rochester biopsychosocial program: A self-determination theory integration', *Families, Systems & Health*, 18 pp. 79-90.

Williams, G. C., Freedman, Z. R. & Deci, E. L. (1998) 'Supporting autonomy to motivate patients with diabetes for glucose control', *Diabetes Care*, 21 (10), pp. 1644-1651.

Williams, G. C., Grow, V. M., Freedman, Z. R., Ryan, R. M. & Deci, E. L. (1996) 'Motivational predictors of weight loss and weight loss maintenance', *J. Pers Soc Psychol*, 70 pp. 115-126.

Williams, G. C., Lynch, M. & Glasgow, R. E. (2007) 'Computer-assisted intervention improves patient-centered diabetes care by increasing autonomy support', *Health Psychol*, 26 (6), pp. 728-734.

Williams, G. C., McGregor, H. A., King, D., Nelson, C. C. & Glasgow, R. E. (2005) 'Variation in perceived competence, glycaemic control, and patient satisfaction: relationship to autonomy support from physicians.', *Patient Education & Counseling*, 57 (1), pp. 39-45.

Williams, G. C., McGregor, H. A., Sharp, D., Levesque, C., Kouides, R. W., Ryan, R. M. & Deci, E. L. (2006) 'Testing a self-determination theory intervention for motivating tobacco cessation: supporting autonomy and competence in a clinical trial', *Health Psychol*, 25 (1), pp. 91-101.

Williams, G. C., McGregor, H. A., Zeldman, A., Freedman, Z. R. & Deci, E. L. (2004) 'Testing a self determination theory process model for promoting glycaemic control through diabetes self management', *Health Psychology*, 2 (5), pp. 58-66.

Williams, G. C., Rodin, G. C., Ryan, R. M., Grolnick, W. S. & Deci, E. L. (1998) 'Autonomous regulation and long-term medication adherence in adult outpatients', *Health Psychology*, 17 pp. 269-276.

Williams, J. (1995) 'Education for empowerment: implications for professional development and training in health promotion', *54*, (37-47),

Williamson, A. R., Hunt, A. E., Pope, J. F. & Tolman, N. M. (2000) 'Recommendations of dietitians for overcoming barriers to dietary adherence in individuals with diabetes', *Diabetes Educ*, 26 (2), pp. 272-279. Wilson, A. D. & Childs, S. (2006) 'Effects of interventions aimed at changing the length of primary care physicians' consultation ', *Cochrane Database of Systematic Reviews*, Issue 1. Art. No.: CD003540. DOI: 10.1002/14651858.CD003540.pub2

Wilson, P. M. (2002) 'The expert patient: issues and implications for community nurses', *Br J Commun Nurs*, 7 pp. 514-519.

Wing, R., Epstein, L., Norwalk, M. P., Scott, N. & Koeske, R. (1985) 'Compliance to self- monitoring of blood glucose: a marked-item technique compared with self report', *Diabet Care*, 8 pp. 456-460.

Wing, R. R., Phelan, S. & Tate, D. (2002) 'The role of adherence in mediating the relationship between depression and health outcomes', *Journal of Psychosomatic Research*, 53 pp. 877-881.

Wiseman, T. (1996) 'A concept analysis of empathy', *Adv Nurs*, 23 (6), pp. 1162-1167.

Wolpert, H. A. & Anderson, B. J. (2001) 'Management of diabetes: are doctors framing the benefits from the wrong perspective?', *BMJ*, 323 pp. 994-996.

World Health Organisation (2009) '2008-2013 Action Plan for the Global Strategy for the Prevention and Control of Noncommunicable Diseases'.[in World Health Assembly Document A61/8 Geneva: WHO. (Accessed:World Health Organisation 24/1/11).

Yamaoka, K. & Tango, T. (2005) 'Efficacy of lifestyle education to prevent type 2 diabetes: a meta-analysis of randomized controlled trials', *Diabet Care*, 28 (11), pp. 2780-2786.

Yannakoulia, M. (2006) 'Eating behavior among type 2 diabetic patients: a poorly recognized aspect in a poorly controlled disease', *Rev Diabet Stud*, 3 (1), pp. 11-16.

Yu, A. P., Yu, Y. F. & Nichol, M. B. (2010) 'Estimating the effect of medication adherence on health outcomes among patients with type 2 diabetes--an application of marginal structural models', *Value Health*, 13 (8), pp.1038-45.

Yu, J. & Kirk, M. (2008) 'Measurement of empathy in nursing research: systematic review', *J Adv Nurs.*, 64 (5), pp. 440-454.

Zachariae, R., Pedersen, C. G., Jensen, A. B., Ehrnrooth, E., Rossen, P. B. & von der, M. H. (2003) 'Association of perceived physician communication style with patient satisfaction, distress, cancer-related self-efficacy, and perceived control over the disease.', *British Journal of Cancer*, 88 (5), pp. 658-665.

Zandbelt, L., Smets, E. M. A., Oort, F. J., Godfried, M. H. & de Haes, H. C. J. M. (2004) 'Satisfaction with the Outpatient Encounter', *Journal of General Internal Medicine*, 19 (11), pp. 1088-1095.

Zandbelt, L. C., Smets, E. M., Oort, F. J. & de Haes, H. C. (2005) 'Coding patient-centred behaviour in the medical encounter', *Soc Sci Med*, 61 (3), pp. 661-671.

Zandbelt, L. C., Smets, E. M., Oort, F. J., Godfried, M. H. & de Haes, H. C. (2006) 'Determinants of physicians' patient-centred behaviour in the medical specialist encounter', *Soc Sci Med*, 63 (4), pp. 899-910

Zandbelt, L. C., Smets, E. M. A., Oort, F. J., Godfried, M. H. & de Haes, H. C. J. M. (2007) 'Patient participation in the medical specialist encounter: Does physicians' patient-centred communication matter?', *Patient Education and Counseling*, 65 pp. 396-406.

Zimmermann, C., Del Piccolo, L. & Finset, A. (2007) 'Cues and Concerns by Patients in Medical Consultations: A Literature Review', *Psychological Bulletin*, 133 (3), pp. 438-463.

BIBLIOGRAPHY

Anderson, B. & Funnell, M. M. (2005a) *The Art of Empowerment: Stories and Strategies for Diabetes Educators.* Alexandria, Va: American Diabetes Assoictaion.

Brace, N., Kemp, R. & Snelgar, R. (2006) SPSS for Psychologists: A Guide to Data Analysis using SPSS for Windows (versions 12 and 13). 3rd edn. Basingstoke: Palgrave Macmillan.

Corbin, J. & Strauss, A. (2008) *Basics of Qualitative Research.* Thousand Oaks CA: Sage.

Creswell, J. W. (2009) *Research Design: Qualitative, Quantitative, and Mixed Methods Approaches.* 3rd edn. Thousand Oaks, California:: Sage

Egan, G. (2002) The Skilled Helper: A Problem -Management and Opportunity - Development Approach to Helping. 7th edn. Brooks/Cole.

Field, a. & Hole, G. (2003) *How to Design and Report Experiments.* London: Sage Publications. 2006

Gable, J. (2007) Counselling Skills for Dietitians. 2nd edn. Blackwell Publishing.

Green, J. & Thorogood, N. (2009) *Qualitative methods for health research.* 2nd edn. Thousand Oaks, CA: Sage.

Joffe, H. & Yardley, L. (2004) 'Content and thematic analysis'. in Marks, D.F. and Yardley, L. (eds.) *Research methods for Clinical and Health Psychology.* London: Sage.

Polonsky, W. H. (1996) 'Understanding and treating patients with diabetes burnout'. *Practical applications for diabetes clinicians* Alexandria, Va American Diabetes Assocation, pp 183-192.

Rogers, C. R. (1980) A Way of Being. Boston: Houghton Mittlin Company.

Rollnick, S., Mason, P. & Butler, C. (1999) *Health Behaviour Change: A guide for practitioners.* Edinburgh: Churchill Livingstone.

Silverman, D. (2006) Interpreting Qualitative Data. 3rd edn. Sage publications.